Smaller reporting company □

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

POST-EFFECTIVE AMENDMENT NO. 1 TO

# FORM S-4

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

## LIBERTY MEDIA CORPORATION

(Exact name of Registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

Large accelerated filer □

4841

(Primary Standard Industrial Classification code number)

20-8988475

(I.R.S. Employer Identification No.)

12300 Liberty Boulevard, Englewood, Colorado 80112, (720) 875-5300

(Address, including zip code, and telephone number, including area code, of Registrant's principal executive offices)

Charles Y. Tanabe Liberty Media Corporation 12300 Liberty Boulevard Englewood, Colorado 80112 (720) 875-5300

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copy to:

Frederick H. McGrath Renee L. Wilm Baker Botts L.L.P. 30 Rockefeller Plaza New York, New York 10112 (212) 408-2500

Approximate date of commencement of proposed sale to the public: As soon as practicable after all conditions to the transaction registered hereby have been satisfied or waived, as applicable.

If the securities being registered on this form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box:
If this form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.
If this form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier registration statement for the same offering.
Registration Number: 333-171201
Indicate by check mark whether the Registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Non-accelerated filer

(Do not check if a smaller reporting company)

Accelerated filer □

## EXPLANATORY NOTE

This Post-Effective Amendment No. 1 to Form S-4 amends the Registration Statement on Form S-4 of Liberty Media Corporation (f/k/a Liberty CapStarz, Inc. and prior thereto Liberty Splitco, Inc.) (Registration No. 333-171201), as amended prior to the date hereto (the "Registration Statement"), which was declared effective by the Securities and Exchange Commission (the "Commission") on April 19, 2011 (the "Effective Date").

At 5:00 p.m. NYT, today, Liberty Interactive Corporation (f/k/a Liberty Media Corporation), the current parent company of the Registrant, will effect a redemption pursuant to which shares of its Liberty Capital tracking stock group and Liberty Starz tracking stock group will be redeemed in exchange for shares of the Registrant's Liberty Capital tracking stock group and Liberty Starz tracking stock group (the "Redemption"). Immediately following the Redemption, the Registrant will be separated from Liberty Interactive Corporation and become a separate publicly traded company.

This Post-Effective Amendment No. 1 is being filed for the sole purpose of replacing Exhibit 8.1: Form of Opinion of Baker Botts L.L.P. regarding certain tax matters, previously filed with the Registration Statement, with a final, executed version of Exhibit 8.1. The Registration Statement is hereby amended, as appropriate, to reflect the replacement of such exhibit. We are also including for completeness certain other exhibits which have been updated since the Effectiveness Date.

### Item 21. Exhibits And Financial Statement Schedules.

(a) Exhibits. The following is a complete list of Exhibits filed as part of this registration statement.

Exhibit No.	Document				
2.1	Reorganization Agreement by and between Liberty Media Corporation and Liberty CapStarz, Inc.				
3.1	Restated Certificate of Incorporation of the Registrant (to be in effect contemporaneously with the effective time of the Redemption)				
3.2	Certificate of Amendment to the Certificate of Incorporation of the Registrant, dated September 22, 2011				
3.3	Bylaws of the Registrant (to be in effect contemporaneously with the effective time of the Redemption)				
4.1	Specimen certificate for shares of the Registrant's Series A Capital common stock, par value \$.01 per share				
4.2	Specimen certificate for shares of the Registrant's Series B Capital common stock, par value \$.01 per share				
4.3	Specimen certificate for shares of the Registrant's Series A Starz common stock, par value \$.01 per share				
4.4	Specimen certificate for shares of the Registrant's Series B Starz common stock, par value \$.01 per share				
5.1	Opinion of Baker Botts L.L.P.*				
8.1	Opinion of Baker Botts L.L.P. regarding certain tax matters				
10.1	Liberty Media Corporation 2011 Incentive Plan				
10.2	Liberty Media Corporation 2011 Non-Employee Director Incentive Plan				
	II 1				

Exhibit No.	Document
10.3	Liberty Media Corporation Transitional Stock Adjustment Plan
10.4	Tax Sharing Agreement by and between Liberty Interactive Corporation and Liberty Media Corporation
10.5	Services Agreement by and between Liberty Interactive Corporation and Liberty Media Corporation
10.6	Facilities Agreement by and between Liberty Interactive Corporation and Liberty Media Corporation
10.7	Form of Indemnification Agreement by and between Liberty Media Corporation and its executive officers/directors*
10.8	Aircraft Time Sharing Agreements
10.9	Management and Allocation Policies of Liberty Media Corporation*
10.10	Tax Sharing Agreement dated as of March 9, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.1 to Liberty Media Corporation's Annual Report on Form 10-K for the year ended December 31, 2009 (File No. 001-33982) as filed on February 25, 2010 (the "Liberty Media 2009 10-K")).
10.11	First Amendment to Tax Sharing Agreement dated as of May 28, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.2 to the Liberty Media 2009 10-K).
10.12	Second Amendment to Tax Sharing Agreement dated as of September 24, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.3 to the Liberty Media 2009 10-K).
10.13	Third Amendment to Tax Sharing Agreement dated as of October 20, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.4 to the Liberty Media 2009 10-K).
10.14	Fourth Amendment to Tax Sharing Agreement dated as of October 28, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.5 to the Liberty Media

10.15 Fifth Amendment to Tax Sharing Agreement dated as of December 6, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.6 to the Liberty Media

2009 10-K).

2009 10-K).

Exhibit No. Document

- 10.16 Sixth Amendment to Tax Sharing Agreement dated as of December 10, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.7 to the Liberty Media 2009 10-K).
- 10.17 Seventh Amendment to Tax Sharing Agreement dated as of December 30, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.8 to the Liberty Media 2009 10-K).
- 10.18 Eighth Amendment to Tax Sharing Agreement dated as of July 25, 2000, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.9 to the Liberty Media 2009 10-K).
- 10.19 Instrument dated January 14, 2000, adding The Associated Group, Inc. as a party to the Tax Sharing Agreement dated as of March 9, 1999, as amended, among The Associated Group, Inc., AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.10 to the Liberty Media 2009 10-K).
- 10.20 Restated and Amended Employment Agreement dated November 1, 1992, between Tele-Communications, Inc. and John C. Malone (assumed by Liberty Media LLC as of March 9, 1999), and the amendment thereto dated June 30, 1999 and effective as of March 9, 1999, between Liberty Media LLC and John C. Malone (collectively, the "Malone Employment Agreement") (incorporated by reference to Exhibit 10.11 to the Liberty Media 2009 10-K).
- 10.21 Second Amendment to Malone Employment Agreement effective January 1, 2003 (incorporated by reference to Exhibit 10.12 to the Liberty Media 2009 10-K).
- 10.22 Third Amendment to Malone Employment Agreement effective January 1, 2007 (incorporated by reference to Exhibit 10.13 to Liberty Media Corporation's Annual Report on Form 10-K for the year ended December 31, 2008 (File No. 001-33982) as filed on February 27, 2009) (the "Liberty Media 2008 10-K")).
- 10.23 Fourth Amendment to Malone Employment Agreement effective January 1, 2009 (incorporated by reference to Exhibit 10.14 to the Liberty Media 2008 10-K).
- 10.24 Tax Sharing Agreement, dated as of November 19, 2009, by and between Liberty Media Corporation and Liberty Entertainment, Inc. ("LEI") (incorporated by reference to Exhibit 10.7 to Amendment No. 1 to the LEI Registration Statement on Form S-4 (File No. 333-158795) as filed on June 8, 2009).
- 10.25 Executive Employment Agreement, dated December 17, 2009, between Gregory B. Maffei and Liberty Media Corporation (incorporated by reference to Exhibit 10.1 to Liberty Media Corporation's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2010 (File No. 001-33982) as filed on August 9, 2010).

Exhibit No.	Document
10.26	Letter Agreement regarding personal use of Liberty's aircraft, dated as of February 22, 2008, between Gregory B. Maffei and Liberty Media Corporation (incorporated by reference to Exhibit 10.38 to Liberty Media Corporation's Annual Report on Form 10-K for the year ended December 31, 2007 (File No. 000-51990) as filed on February 29, 2008).
21.1	List of Subsidiaries of the Registrant*
23.1	Consent of KPMG LLP (for Liberty Interactive Corporation (f/k/a Liberty Media Corporation))*
23.2	Consent of KPMG LLP (for Liberty Media Corporation (f/k/a Liberty CapStarz, Inc. and Liberty Splitco, Inc.))*
23.3	Consent of Ernst & Young LLP*
23.4	Consent of Baker Botts L.L.P. (included in Exhibit 5.1)*
24.1	Power of Attorney*
99.1	Proxy Card for Liberty Capital and Liberty Starz stockholders*
99.2	Letter of Transmittal*

Previously filed.

<sup>(</sup>b) Financial Statement Schedules. Schedules not listed above have been omitted because the information set forth therein is not material, not applicable or is included in the financial statements or notes of the proxy statement/prospectus which forms a part of this registration statement.

## SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the Registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the city of Denver, state of Colorado, on September 23, 2011.

## LIBERTY MEDIA CORPORATION

By: /s/ CHARLES Y. TANABE

Name: Charles Y. Tanabe

Title: Executive Vice President and General Counsel

Pursuant to the requirements of the Securities Act of 1933, as amended, this registration statement has been signed by the following persons (which persons constitute a majority of the Board of Directors) in the capacities and on the dates indicated:

	Name	<u>Title</u>	Date	
	*	Chairman of the Board and Director		
	John C. Malone			
	*	Chief Executive Officer (Principal Executive		
	Gregory B. Maffei	Officer), President and Director		
	*	Senior Vice President and Treasurer (Principal		
David J.A. Flowers		Financial Officer)		
*		Senior Vice President and Controller (Principal		
	Christopher W. Shean	Accounting Officer)		
/s/ CHARLES Y. TANABE		Executive Vice President, General Counsel, and	September 23, 2011	
	Charles Y. Tanabe	Director		
*By:	/s/ CHARLES Y. TANABE		September 23, 2011	
	Charles Y. Tanabe Attorney-in-Fact			

# EXHIBIT INDEX

Exhibit No.	Document
2.1	Reorganization Agreement by and between Liberty Media Corporation and Liberty CapStarz, Inc.
3.1	Restated Certificate of Incorporation of the Registrant (to be in effect contemporaneously with the effective time of the Redemption)
3.2	Certificate of Amendment to the Certificate of Incorporation of the Registrant, dated September 22, 2011
3.3	Bylaws of the Registrant (to be in effect contemporaneously with the effective time of the Redemption)
4.1	Specimen certificate for shares of the Registrant's Series A Capital common stock, par value \$.01 per share
4.2	Specimen certificate for shares of the Registrant's Series B Capital common stock, par value \$.01 per share
4.3	Specimen certificate for shares of the Registrant's Series A Starz common stock, par value \$.01 per share
4.4	Specimen certificate for shares of the Registrant's Series B Starz common stock, par value \$.01 per share
5.1	Opinion of Baker Botts L.L.P.*
8.1	Opinion of Baker Botts L.L.P. regarding certain tax matters
10.1	Liberty Media Corporation 2011 Incentive Plan
10.2	Liberty Media Corporation 2011 Non-Employee Director Incentive Plan
10.3	Liberty Media Corporation Transitional Stock Adjustment Plan
10.4	Tax Sharing Agreement by and between Liberty Interactive Corporation and Liberty Media Corporation
10.5	Services Agreement by and between Liberty Interactive Corporation and Liberty Media Corporation
10.6	Facilities Agreement by and between Liberty Interactive Corporation and Liberty Media Corporation
10.7	Form of Indemnification Agreement by and between Liberty Media Corporation and its executive officers/directors*
10.8	Aircraft Time Sharing Agreements
10.9	Management and Allocation Policies of Liberty Media Corporation*
10.10	Tax Sharing Agreement dated as of March 9, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.1 to Liberty Media Corporation's Annual Report on Form 10-K for the year ended December 31, 2009 (File No. 001-33982) as filed on February 25, 2010 (the "Liberty Media 2009 10-K"))

Exhibit No. Document

- 10.11 First Amendment to Tax Sharing Agreement dated as of May 28, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.2 to the Liberty Media 2009 10-K).
- 10.12 Second Amendment to Tax Sharing Agreement dated as of September 24, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.3 to the Liberty Media 2009 10-K).
- 10.13 Third Amendment to Tax Sharing Agreement dated as of October 20, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.4 to the Liberty Media 2009 10-K).
- 10.14 Fourth Amendment to Tax Sharing Agreement dated as of October 28, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.5 to the Liberty Media 2009 10-K).
- 10.15 Fifth Amendment to Tax Sharing Agreement dated as of December 6, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.6 to the Liberty Media 2009 10-K).
- 10.16 Sixth Amendment to Tax Sharing Agreement dated as of December 10, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.7 to the Liberty Media 2009 10-K).
- 10.17 Seventh Amendment to Tax Sharing Agreement dated as of December 30, 1999, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.8 to the Liberty Media 2009 10-K).
- 10.18 Eighth Amendment to Tax Sharing Agreement dated as of July 25, 2000, by and among AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.9 to the Liberty Media 2009 10-K).
- 10.19 Instrument dated January 14, 2000, adding The Associated Group, Inc. as a party to the Tax Sharing Agreement dated as of March 9, 1999, as amended, among The Associated Group, Inc., AT&T Corp., Liberty Media LLC, Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc. and each Covered Entity listed on the signature pages thereof (incorporated by reference to Exhibit 10.10 to the Liberty Media 2009 10-K).

oit No.	Document
10.20	Restated and Amended Employment Agreement dated November 1, 1992, between Tele-Communications, Inc. and John C. Malone (assumed by Liberty Media LLC as of March 9, 1999), and the amendment thereto dated June 30, 1999 and effective as of March 9, 1999, between Liberty Media LLC and John C. Malone (collectively, the "Malone Employment Agreement") (incorporated by reference to Exhibit 10.11 to the Liberty Media 2009 10-K).
10.21	Second Amendment to Malone Employment Agreement effective January 1, 2003 (incorporated by reference to Exhibit 10.12 to the Liberty Media 2009 10-K).
10.22	Third Amendment to Malone Employment Agreement effective January 1, 2007 (incorporated by reference to Exhibit 10.13 the Liberty Media Corporation's Annual Report on Form 10-K for the year ended December 31, 2008 (File No. 001-33982) as filed on February 27, 2009) (the "Liberty Media 2008 10-K")).
10.23	Fourth Amendment to Malone Employment Agreement effective January 1, 2009 (incorporated by reference to Exhibit $10.14$ to the Liberty Media $2008\ 10$ -K).
10.24	Tax Sharing Agreement, dated as of November 19, 2009, by and between Liberty Media Corporation and Liberty Entertainment, Inc. ("LEI") (incorporated by reference to Exhibit 10.7 to Amendment No. 1 to the LEI Registration Statement on Form S-4 (File No. 333-158795) as filed on June 8, 2009).
10.25	Executive Employment Agreement, dated December 17, 2009, between Gregory B. Maffei and Liberty Media Corporation (incorporated by reference to Exhibit 10.1 to Liberty Media Corporation's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2010 (File No. 001-33982) as filed on August 9, 2010).
10.26	Letter Agreement regarding personal use of Liberty's aircraft, dated as of February 22, 2008, between Gregory B. Maffei and Liberty Media Corporation (incorporated by reference to Exhibit 10.38 to Liberty Media Corporation's Annual Report on Form 10-K for the year ended December 31, 2007 (File No. 000-51990) as filed on February 29, 2008).
21.1	List of Subsidiaries of the Registrant*
23.1	Consent of KPMG LLP (for Liberty Interactive Corporation (f/k/a Liberty Media Corporation))*
23.2	Consent of KPMG LLP (for Liberty Media Corporation (f/k/a Liberty CapStarz, Inc. and Liberty Splitco, Inc.))*
23.3	Consent of Ernst & Young LLP*
23.4	Consent of Baker Botts L.L.P. (included in Exhibit 5.1)*
24.1	Power of Attorney*
99.1	Proxy Card for Liberty Capital and Liberty Starz stockholders*
99.2	Letter of Transmittal*
	ously filed.

# QuickLinks

## EXPLANATORY NOTE

Item 21. Exhibits And Financial Statement Schedules.

SIGNATURES EXHIBIT INDEX

## REORGANIZATION AGREEMENT

#### between

## LIBERTY MEDIA CORPORATION

and

## LIBERTY CAPSTARZ, INC.

## Dated as of August 30, 2011

#### TABLE OF CONTENTS

		Page	
ARTICLE I TH	IE RESTRUCTURING	2	
1.1	Restructuring	2	
1.2	Transfer of CapStarz Assets and CapStarz Businesses; Assumption of CapStarz Liabilities	2 3	
1.3	Third Party Consents and Government Approvals	3	
1.4	Further Actions	3	
1.5 1.6	Restructuring Documents Qualification as Reorganization	3 3	
	HE REDEMPTIONS	3	
2.1	The Redemptions	3	
2.2 2.3	Conditions to the Redemptions Treatment of Outstanding Equity Awards	5 7	
		,	
	REPRESENTATIONS AND WARRANTIES	9	
3.1	Representations and Warranties of the Parties	9	
3.2	No Approvals or Notices Required; No Conflict with Instruments	10	
3.3	No Other Reliance	10	
ARTICLE IV C		10	
4.1	Cross-Indemnities	10	
4.2	Further Assurances	14	
4.3	Specific Performance	14	
4.4	Access to Information	14	
4.5	Confidentiality	15	
4.6	Notices Regarding Transferred Assets	16 16	
4.7	Treatment Of Payments	16	
ARTICLE V C	LOSING	16	
5.1	Closing	16	
5.2	Conditions to Closing	17	
5.3	Deliveries at Closing	17	
ARTICLE VI T	TERMINATION	18	
6.1	Termination	18	
6.2	Effect of Termination	18	
ARTICLE VII	MISCELLANEOUS	18	
7.1	Definitions	18	
7.2	No Third-Party Rights	24	
7.3	Notices	24	
7.4	Entire Agreement	25	
7.5	Binding Effect; Assignment	25	
7.6	Costs and Expenses	25	
	i		
7.7	Governing Law; Jurisdiction	25	
7.8	Waiver of Jury Trial	26	
7.9	Severability	26	
7.10	Amendments; Waivers	26	
7.11	No Strict Construction; Interpretation	27	
7.12	Conflicts with Tax Sharing Agreement	27	
7.13	Counterparts	27	
	Form of Aircraft Time Sharing Agreement		
EXHIBIT B —	Form of Facilities Sharing Agreement		
EXHIBIT C — Form of Services Agreement			

#### REORGANIZATION AGREEMENT

This **REORGANIZATION AGREEMENT** (together with all Schedules and Exhibits hereto, this "<u>Agreement</u>"), dated as of August 30, 2011, is entered into by and between **LIBERTY MEDIA CORPORATION**, a Delaware corporation ("<u>LMC</u>"), and **LIBERTY CAPSTARZ**, **INC**., a Delaware corporation ("<u>Splitco</u>"). Certain capitalized terms used herein have the meanings ascribed thereto in Section 7.1.

#### RECITALS:

WHEREAS, in accordance with and pursuant to the LMC Charter, the businesses, assets and liabilities of LMC are currently attributed among three tracking stock groups: the Capital Group; the Starz Group; and the Interactive Group;

WHEREAS, the LMC Board has determined that it is appropriate and in the best interests of LMC and its stockholders to separate LMC into two publicly traded companies (the "Split-Off"): (1) Splitco, which immediately after the Split-Off will conduct the Splitco Businesses, will own the Splitco Assets and will be responsible for the Splitco Liabilities; and (2) LMC, which immediately after the Split-Off will conduct the LMC Retained Businesses, will own the LMC Retained Assets and will be responsible for the LMC Retained Liabilities;

WHEREAS, in accordance with and pursuant to the Splitco Charter, the Splitco Businesses, the Splitco Assets and the Splitco Liabilities will be attributed between two tracking stock groups of Splitco: the Capital Group and the Starz Group;

WHEREAS, the LMC Board has determined that, in order to effect the Split-Off, it is appropriate and in the best interests of LMC and its stockholders: (i) for LMC and its Subsidiaries to effect the Restructuring, as a result of which Splitco will own, directly or indirectly, the Splitco Businesses, the Splitco Assets and the Splitco Liabilities and LMC will retain the LMC Retained Businesses, the LMC Retained Assets and the LMC Retained Liabilities; (ii) for LMC to redeem (the "LCAP Redemption") all of the issued and outstanding shares of LMC's Series A Liberty Capital common stock, par value \$.01 per share ("LCAPB" and, together with LCAPA, the "Liberty Capital Stock"), for shares of a corresponding series of Splitco Capital common stock, and (iii) for LMC to redeem (the "LSTZ Redemption." and together with the LCAP Redemption, the "Redemptions"), all of the issued and outstanding shares of LMC's Series A Liberty Starz common stock, par value \$.01 per share ("LSTZB" and, together with LSTZA, the "Liberty Starz Stock"), for shares of a corresponding series of Splitco Starz common stock, par value \$.01 per share ("LSTZB" and, together with LSTZA, the "Liberty Starz Stock"), for shares of a corresponding series of Splitco Starz common stock; in each case, on the terms and subject to the conditions set forth in the LMC Charter and this Agreement;

WHEREAS, the transactions contemplated by this Agreement, including the Restructuring and the Redemptions, have been approved by the LMC Board and, to the extent applicable, the Splitco Board and are motivated in whole or substantial part by certain substantial corporate business purposes of LMC and Splitco;

.

WHEREAS, this Agreement constitutes a "plan of reorganization" within the meaning of Section 368 of the Internal Revenue Code of 1986, as amended (the "Code"), and the Treasury Regulations promulgated thereunder; and

WHEREAS, the parties wish to set forth in this Agreement the terms on which, and the conditions subject to which, they intend to implement the measures referred to above and elsewhere herein.

**NOW, THEREFORE**, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements contained herein, the parties to this Agreement hereby agree as follows:

# ARTICLE I THE RESTRUCTURING

## 1.1 <u>Restructuring</u>.

- (a) The parties have taken or will take, and have caused or will cause their respective Subsidiaries to take, all actions that are necessary or appropriate to implement and accomplish the transactions contemplated by each of the steps set forth in the Restructuring Plan (collectively, the "Restructuring"); provided, that all of such steps shall be completed by no later than the Effective Time.
- (b) All the transactions contemplated by the Restructuring and the Redemptions are intended to be part of the same plan of reorganization, even though there may be delays between the completion of certain of the transactions.
- 1.2 <u>Transfer of Splitco Assets and Splitco Businesses; Assumption of Splitco Liabilities.</u>

On the terms and subject to the conditions of this Agreement, and in furtherance of the Restructuring and the Split-Off:

- (a) LMC, by no later than the Effective Time, shall cause all of its (or its Subsidiaries') rights, title and interest in and to all of the Splitco Assets and Splitco Businesses to be contributed, assigned, transferred, conveyed and delivered, directly or indirectly, to Splitco, and Splitco agrees to accept or cause to be accepted all such rights, title and interest in and to all the Splitco Assets and Splitco Businesses. All Splitco Assets are being transferred on an "as is, where is" basis, without any warranty whatsoever on the part of LMC.
- (b) LMC, by no later than the Effective Time, shall cause all of the Splitco Liabilities to be assigned, directly or indirectly, to Splitco, and Splitco agrees to accept, assume, perform, discharge and fulfill all of the Splitco Liabilities in accordance with their respective terms.
- (c) Upon completion of the transactions contemplated by Sections 1.2(a) and (b) above: (i) Splitco will own, directly or indirectly, the Splitco Businesses and the Splitco Assets and be subject to the Splitco Liabilities; and (ii) LMC will continue to own, directly or indirectly,

the LMC Retained Businesses and the LMC Retained Assets and continue to be subject to the LMC Retained Liabilities.

- 1.3 Third Party Consents and Government Approvals. To the extent that either Redemption or any step in the Restructuring Plan requires a consent of any third party or a Governmental Authorization, the parties will use commercially reasonable efforts to obtain each such consent and Governmental Authorization at or prior to the time such consent or Governmental Authorization is required in order to lawfully effect the Redemptions and each step in the Restructuring Plan.
- 1.4 <u>Further Actions.</u> From and after the Effective Time, upon the reasonable request of a party hereto, each other party hereto will promptly take, or cause its Subsidiaries to promptly take, all commercially reasonable actions necessary or appropriate to fully accomplish the Restructuring and to give effect to the transactions provided for in this Agreement, including each step in the Restructuring Plan, in accordance with the purposes hereof.
- 1.5 Restructuring Documents. All documents and instruments used to effect the Restructuring and otherwise to comply with this Agreement shall be in form satisfactory to LMC, Splitco and any additional signatories hereto.
- 1.6 Qualification as Reorganization. For U.S. federal income tax purposes, (1) each step of the Restructuring, other than Step 9 with respect to the distribution by LTWX V, Inc., is generally intended to be undertaken in a manner so that no gain or loss is recognized by LMC, Splitco or their respective Subsidiaries, and (2) the Contribution and the Redemptions are intended to qualify as a tax-free reorganization under Sections 368(a) and 355 of the Code.

# ARTICLE II THE REDEMPTIONS

#### 2.1 The Redemptions.

- (a) The LMC Board shall have the authority and right: (i) to effect the Redemptions, subject to the conditions set forth in Section 2.2, or terminate the Redemptions at any time prior to the Effective Time; (ii) to establish and change the date (the "Redemption Date") and time (the "Effective Time") at which the Redemptions will be effective; and (iii) prior to the Effective Time, to establish and change the procedures for effecting the Redemptions; subject, in all cases, to the applicable provisions of the DGCL and the LMC Charter.
- (b) On the Redemption Date, subject to the satisfaction or, where permissible, waiver of the conditions precedent to the Redemptions set forth in Section 2.2, LMC shall redeem:
  - (i) each outstanding share of LCAPA for one validly issued, fully paid and non-assessable share of Splitco's Series A Liberty Capital common stock, par value \$0.01 per share ("Splitco Series A Capital Stock");
  - (ii) each outstanding share of LCAPB for one validly issued, fully paid and non-assessable share of Splitco's Series B Liberty Capital common stock, par value \$0.01

3

per share ("Splitco Series B Capital Stock" and together with the Splitco Series A Capital Stock, the "Splitco Capital Stock");

- (iii) each outstanding share of LSTZA for one validly issued, fully paid and non-assessable share of Splitco's Series A Liberty Starz common stock, par value \$0.01 per share ("Splitco Series A Starz Stock"); and
- (iv) each outstanding share of LSTZB for one validly issued, fully paid and non-assessable share of Splitco's Series B Liberty Starz common stock, par value \$0.01 per share ("Splitco Series B Starz Stock") and together with the Splitco Series A Starz Stock, the "Splitco Starz Stock").
- (c) Immediately prior to the Effective Time and in accordance with the Restructuring Plan, Splitco shall cause the Splitco Charter to be filed with the Delaware Secretary of State, whereupon the issued and then outstanding shares of Splitco common stock, par value \$0.01 per share (all of which shall be owned by LMC), shall automatically be reclassified into: (i) a number of shares of Splitco Series A Capital Stock equal to the number of shares of LCAPA outstanding immediately prior to the Effective Time; (ii) a number of shares of Splitco Series A Starz Stock equal to the number of shares of LCAPA outstanding immediately prior to the Effective Time; (iii) a number of shares of Splitco Series A Starz Stock equal to the number of shares of LSTZA outstanding immediately prior to the Effective Time; and (iv) a number of shares of Splitco Series B Starz Stock equal to the number of shares of LSTZB outstanding immediately prior to the Effective Time.
- (d) LMC will provide notice of the Redemption Date and the Effective Time to holders of Liberty Capital Stock and Liberty Starz Stock in accordance with the requirements of Section A.2.(e)(iv) and Section A.2.(f)(iv), respectively, of the LMC Charter and the applicable provisions of the DGCL and the rules and regulations of the SEC.
- (e) LMC will take such action, if any, as may be necessary or appropriate under applicable state and foreign securities and "blue sky" laws to permit the Redemptions to be effected in compliance, in all material respects, with such laws.
- (f) Promptly following the Effective Time, LMC will deliver (or cause to be delivered) to the Redemption Agent all of the shares of Splitco Stock owned by LMC, and shall cause the Redemption Agent (i) to exchange (by means of debit and credit), on a one-for-one basis, the shares of each series of Liberty Capital Stock and Liberty Starz Stock held in book-entry form as of the Effective Time for the same series of Splitco Capital Stock and Splitco Starz Stock, respectively, and (ii) to mail to the holders of record of certificated shares of each series of Liberty Capital Stock and Liberty Starz Stock as of the Redemption Date a letter of transmittal (which will specify that delivery will be effected, and risk of loss and title to the redeemed shares will pass, only upon proper delivery of the certificates representing such shares to the Redemption Agent) with instructions for use in effecting the surrender, on a one-for-one basis, of the redeemed shares of Liberty Capital Stock and Liberty Starz Stock for the same series of Splitco Capital Stock and Splitco Starz Stock, respectively.

4

- (g) Shares of Splitco Capital Stock that are exchanged in the LCAP Redemption for shares of Liberty Capital Stock and shares of Splitco Starz Stock that are exchanged in the LSTZ Redemption for shares of Liberty Starz Stock will be deemed to have been distributed as of the Effective Time; provided, that until the surrender of any certificate representing redeemed shares of Liberty Capital Stock or Liberty Starz Stock for shares of Splitco Stock, Splitco may withhold and accumulate any dividends or distributions which become payable with respect to such shares of Splitco Stock pending the surrender of such certificate.
- (h) No fractional shares of Splitco Stock will be distributed, and no fractional shares of Liberty Capital Stock or Liberty Starz Stock will be redeemed, in connection with the Redemptions.

- 2.2 <u>Conditions to the Redemptions.</u> The completion of the Redemptions is subject to the satisfaction of the following conditions:
- (a) the approval of the LCAP Redemption by the holders of a majority in aggregate voting power of the shares of LCAPA and LCAPB that were present, in person or by proxy, and entitled to vote at the special meeting of holders of Liberty Capital Stock and Liberty Starz Stock called for the purpose of voting on the Redemptions (the "<u>Stockholder Meeting</u>"), voting together as a single class, shall remain valid and binding under the LMC Charter;
- (b) the approval of the LSTZ Redemption by the holders of a majority in aggregate voting power of the shares of LSTZA and LSTZB that were present, in person or by proxy, and entitled to vote at the Stockholders Meeting, voting together as a single class, shall remain valid and binding under the LMC Charter;
- (c) the private letter ruling received from the IRS on March 3, 2011 (the 'Ruling'), which provides to the effect that the Contribution and the Redemptions will qualify as a tax-free transaction under Sections 355 and 368(a)(1)(D) of the Code and that, for U.S. federal income tax purposes, (i) no gain or loss will be recognized by LMC upon the distribution of Splitco Stock pursuant to the Redemptions, and (ii) no gain or loss will be recognized by, and no amount will be included in the income of, holders of Liberty Capital Stock and Liberty Starz Stock for shares of Splitco Capital Stock and Splitco Starz Stock, respectively, pursuant to the Redemptions, shall not have been withdrawn, invalidated or modified in an adverse manner;
- (d) LMC shall have received a tax opinion from Baker Botts L.L.P., in form and substance reasonably acceptable to LMC, and which opinion will rely on the continued validity of the Ruling, to the effect that under applicable U.S. federal income tax law (i) the Contribution and the Redemptions will qualify as a tax-free transaction under Sections 355 and 368(a)(1)(D) of the Code, (ii) no gain or loss will be recognized by LMC upon the distribution of Splitco Capital Stock and Splitco Starz Stock pursuant to the Redemptions, (iii) no gain or loss will be recognized by, and no amount will be included in the income of, holders of Liberty Capital Stock and Liberty Starz Stock upon the exchange of their shares of Liberty Capital Stock and Liberty Starz Stock for shares of Splitco Capital Stock and Splitco Starz Stock, respectively, pursuant to the Redemptions, (iv) the Splitco Capital Stock and the Splitco Starz Stock distributed in the Redemptions will be treated as stock of Splitco for U.S. federal income tax purposes, and (v) the

5

Splitco Capital Stock and the Splitco Starz Stock distributed in the Redemptions will not constitute Section 306 stock within the meaning of Section 306(c) of the Code;

- (e) the Registration Statement shall be effective under the Securities Act and the registration of the Splitco Stock under Section 12(b) of the Exchange Act shall be effective, and no proceedings to suspend either such effectiveness shall have been initiated or threatened by the SEC;
- (f) the Splitco Stock shall have been approved for listing on The Nasdaq Stock Market (the 'Nasdaq'), subject to compliance with Nasdaq's listing requirements;
- (g) the necessary approvals and permits received from the Federal Communications Commission (<u>FCC</u>") with respect to the transfer of control of FCC licenses to be held by Splitco Subsidiaries or investees upon completion of the Restructuring shall not have been withdrawn, invalidated or modified in an adverse manner;
- (h) any other Governmental Authorizations that the LMC Board determines to obtain shall have been so obtained (or all relevant waiting periods with respect thereto shall have expired);
- (i) any approval, consent, ratification, waiver or other authorization of any Person that is not a party to this Agreement that the LMC Board determines to obtain shall have been so obtained;
- (j) with respect to the action captioned *Liberty Media Corporation and Liberty Media LLC vs. The Bank of New York Mellon Trust Company, as Trustee*(C.A. No. 5702-VCL), pending in the Delaware Court of Chancery (the "Delaware Action"), a final, non-appealable judgment that the Split-Off will not constitute a disposition of substantially all the assets and properties of Liberty Media LLC under the terms of Liberty Media LLC's Indenture, dated as of July 7, 1999 (as amended and supplemented, the "Indenture"), shall have been rendered; and
- (k) no Order or other legal restraint or prohibition preventing the consummation of the Split-Off or any of the other transactions contemplated by this Agreement, including the transactions to effect the Restructuring and the Redemptions, shall be pending or in effect.

The foregoing conditions are for the sole benefit of LMC and shall not in any way limit LMC's right to amend, modify or terminate this Agreement in accordance with Section 6.1. Any determination made by LMC prior to the Redemptions concerning the satisfaction of any condition set forth in this Section 2.2 shall be final and conclusive. The conditions to the completion of the Redemptions set forth in Sections 2.2(a), (b), (c), (d), (e), (f) and (k) may not be waived. The condition to the completion of the Redemptions set forth in Section 2.2(j) may be waived by the LMC Board but the Redemptions may not be completed until holders of Liberty Capital Stock and Liberty Starz Stock are notified of any such waiver and proxies are resolicited from such holders for the retaking of votes on the Redemptions. The conditions set forth in Sections 2.2(g), (h) and (i) may be waived by the LMC Board without any resolicitation of proxies from holders of Liberty Capital Stock and Liberty Starz Stock.

6

### 2.3 <u>Treatment of Outstanding Equity Awards.</u>

- (a) Certain current and former employees, non-employee directors and consultants of LMC, the Qualifying Subsidiaries and their respective Subsidiaries have been granted options, stock appreciation rights, and restricted shares in respect of Liberty Capital Stock and/or Liberty Starz Stock pursuant to various stock incentive plans of LMC administered by the LMC Board (collectively, "Awards"). LMC and Splitco shall use commercially reasonable efforts to take all actions necessary or appropriate so that Awards that are outstanding immediately prior to the Effective Time are adjusted as set forth in this Section 2.3.
- (b) Options. As of the Effective Time, and as determined by the LMC Board pursuant to its authority granted under the applicable stock incentive plan of LMC, the following shall occur:
  - (i) each outstanding LMC Option to purchase shares of Liberty Capital Stock, whether vested or unvested (each, an <u>Outstanding LMC Capital Option</u>"), will be converted, automatically, into a Splitco Option to purchase the same number and series of Splitco Capital Stock (a <u>Splitco Capital Option</u>") as the number and series of shares of Liberty Capital Stock subject to such Outstanding LMC Capital Option immediately prior to the Effective Time; and
  - (ii) each outstanding LMC Option to purchase shares of Liberty Starz Stock, whether vested or unvested (each, an <u>Outstanding LMC Starz Option</u>"), will be converted, automatically, into a Splitco Option to purchase the same number and series of shares of Splitco Starz Stock (a "<u>Splitco Starz Option</u>") as the number and series of shares of Liberty Starz Stock subject to such Outstanding LMC Starz Option immediately prior to the Effective Time.

In addition, the per share exercise price of each Splitco Capital Option will be equal to the per share exercise price of the corresponding Outstanding LMC Capital Option, and the per share exercise price of each Splitco Starz Option will be equal to the per share exercise price of the corresponding Outstanding LMC Starz Option;

provided, that the exercise price and number of shares subject to each Splitco Capital Option and Splitco Starz Option shall in any event be determined in a manner consistent with the requirements of Section 409A of the Code.

All other terms of the Splitco Capital Options and Splitco Starz Options (including the vesting terms thereof) will, in all material respects, be the same as those of the corresponding Outstanding LMC Capital Option and Outstanding LMC Starz Option, respectively, except that the Splitco Options will continue to vest so long as the holder provides service (whether as an employee, non-employee director or consultant, as the case may be) to any of LMC, a Qualifying Subsidiary or their respective Subsidiaries; provided, that the terms and conditions of exercise of the Splitco Options shall in any event be determined in a manner consistent with Section 409A of the Code.

(c) <u>SARs.</u> As of the Effective Time, and as determined by the LMC Board pursuant to its authority granted under the applicable stock incentive plan of LMC, the following shall occur:

7

- (i) each outstanding LMC SAR related to Liberty Capital Stock, whether vested or unvested (an 'Outstanding LMC Capital SAR'), will be converted, automatically, into a Splitco SAR related to the same number and series of Splitco Capital Stock (a "Splitco Capital SAR") as the number and series of shares of Liberty Capital Stock subject to such Outstanding LMC Capital SAR immediately prior to the Effective Time; and
- (ii) each outstanding LMC SAR related to Liberty Starz Stock, whether vested or unvested (an 'Outstanding LMC Starz SAR''), will be converted, automatically, into a Splitco SAR related to the same number and series of shares of Splitco Starz Stock (a "Splitco Starz SAR") as the number and series of shares of Liberty Starz Stock subject to such Outstanding LMC Starz SAR immediately prior to the Effective Time.

In addition, the per share base price of each Splitco Capital SAR will be equal to the per share base price of the corresponding Outstanding LMC Capital SAR, and the per share base price of each Splitco Starz SAR will be equal to the per share base price of the corresponding Outstanding LMC Starz SAR; *provided*, that the base price and the number of shares to which such Splitco SARs relate shall in any event be determined in a manner consistent with the requirements of Section 409A of the Code.

All other terms of the Splitco Capital SARs and Splitco Starz SARs (including the vesting terms thereof) will, in all material respects, be the same as those of the corresponding Outstanding LMC Capital SAR and Outstanding LMC Starz SAR, respectively, except that the Splitco SARs will continue to vest so long as the holder provides service (whether as an employee, consultant or nonemployee director, as the case may be) to any of LMC, a Qualifying Subsidiary or their respective Subsidiaries; provided, that the terms and conditions of exercise of the Splitco SARs shall in any event be determined in a manner consistent with Section 409A of the Code.

- (d) Restricted Stock. Shares of Liberty Capital Stock and Liberty Starz Stock that are subject to a restricted stock award granted under a stock incentive plan of LMC ("LMC Restricted Stock") will be treated in the Redemptions in the same manner as other outstanding shares of Liberty Capital Stock and Liberty Starz Stock, respectively. As of the Effective Time, and as determined by the LMC Board pursuant to its authority granted under the applicable stock incentive plan of LMC, shares of LMC Restricted Stock shall become Splitco Restricted Stock, and will otherwise be subject, in all material respects, to the same terms and conditions (including the vesting terms thereof) as those applicable to such shares of LMC Restricted Stock immediately prior to the Effective Time, except that the Splitco Restricted Stock will continue to vest so long as the holder provides service (whether as an employee, consultant or nonemployee director, as the case may be) to any of LMC, a Qualifying Subsidiary or their respective Subsidiaries.
- (e) From and after the Effective Time, Splitco Options, Splitco SARs and Splitco Restricted Stock, regardless of by whom held, shall be settled by Splitco pursuant to the terms of the Splitco Transitional Plan. The obligation to deliver (i) shares of Splitco Capital Stock or Splitco Starz Stock upon the exercise of Splitco Options, (ii) cash or shares of Splitco Capital Stock or Splitco Starz Stock in settlement of Splitco SARs or (iii) shares of Splitco Capital Stock

8

or Splitco Starz Stock upon vesting of shares of Splitco Restricted Stock shall be the sole obligation of Splitco, and LMC shall have no Liability in respect thereof.

- (f) It is intended that the Splitco Transitional Plan be considered, as to any Splitco Option, Splitco SAR or Splitco Restricted Stock that is issued as part of the adjustment provisions of this Section 2.3, to be a successor plan to the stock incentive plan of LMC pursuant to which the corresponding LMC Option, LMC SAR or LMC Restricted Stock was issued, and Splitco shall be deemed to have assumed the obligations under the applicable stock incentive plans of LMC to make the adjustments to the Awards set forth in this Section 2.3.
- (g) With respect to Awards adjusted pursuant to this Section 2.3, service after the Effective Time as an employee or non-employee director of, or consultant to, LMC, any Qualifying Subsidiary or their respective Subsidiaries shall be treated as service to Splitco and its Subsidiaries with respect to Splitco Options, Splitco SARs and Splitco Restricted Stock held after the Effective Time by such Persons.
- (h) Neither the Effective Time nor any other transaction contemplated by the Restructuring Plan or this Agreement shall be considered a termination of employment for any employee of Splitco or any of its Subsidiaries for purposes of any Award, and service as an employee or non-employee director of, or a consultant to, Splitco or any of its Subsidiaries shall be treated as service to LMC or its Subsidiaries with respect to LMC Options, LMC SARs and LMC Restricted Stock held after the Effective Time and any other awards granted under any stock incentive plan of LMC relating to shares of LMC Interactive common stock and held by employees or non-employee directors of, or consultants to, Splitco or any of its Subsidiaries.
- (i) Splitco agrees that, on and after the Effective Date, it shall use its reasonable efforts to cause to be effective under the Securities Act, on a continuous basis, a registration statement on Form S-8 with respect to shares of Splitco Capital Stock and Splitco Starz Stock issuable upon exercise of Splitco Options or settlement of Splitco SARs.

# ARTICLE III REPRESENTATIONS AND WARRANTIES

- 3.1 Representations and Warranties of the Parties. Each party hereto represents and warrants to the other as follows:
- (a) <u>Organization and Qualification</u>. Such party is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware, has all requisite corporate power and authority to own, use, lease or operate its properties and assets, and to conduct the business heretofore conducted by it, and is duly qualified to do business and is in good standing in each jurisdiction in which the properties owned, used, leased or operated by it or the nature of the business conducted by it requires such qualification, except in such jurisdictions where the failure to be so qualified and in good standing would not have a material adverse effect on its business, financial condition or results of operations or its ability to perform its obligations under this Agreement.
- (b) <u>Authorization and Validity of Agreement</u>. Such party has all requisite power and authority to execute, deliver and perform its obligations under this Agreement, the agreements

and instruments to which it is to be a party required to effect the Restructuring (the 'Restructuring Agreements') and the agreements to be delivered by it at the Closing pursuant to Section 5.3 (the "Other Agreements"). The execution, delivery and performance by such party of this Agreement, the Restructuring Agreements and the Other Agreements and the consummation by it of the transactions contemplated hereby and thereby have been duly and validly authorized by the board of directors, managing members or analogous governing body of such party and, to the extent required by law, its stockholders or members, and no other corporate or other action on its part is necessary to authorize the execution and delivery by such party of this Agreement, the Restructuring Agreements and the Other Agreements, the performance by it of its obligations hereunder and thereunder and the consummation by it of the transactions contemplated hereby and thereby. This Agreement has been, and each of the Restructuring Agreements and each of the Other Agreements, when executed and delivered, will be, duly executed and delivered by such party and each is, or will be, a valid and binding obligation of such party, enforceable in accordance with its terms.

- 3.2 No Approvals or Notices Required; No Conflict with Instruments. The execution, delivery and performance by such party of this Agreement, the Restructuring Agreements and the Other Agreements, and the consummation of the transactions contemplated hereby and thereby, do not and will not conflict with or result in a breach or violation of any of the terms or provisions of, constitute a default under, or result in the creation of any lien, charge or encumbrance upon any of its assets pursuant to the terms of, the charter or bylaws (or similar formation or governance instruments) of such party, any indenture, mortgage, deed of trust, loan agreement or other agreement or instrument to which it is a party or by which it or any of its assets are bound, or any law, rule, regulation, judgment, order or decree of any court or governmental authority having jurisdiction over it or its properties; provided, that no representation or warranty is given with respect to the Indenture.
- 3.3 No Other Reliance. In determining to enter into this Agreement, the Restructuring Agreements and the Other Agreements, and to consummate the transactions contemplated hereby and thereby, such party has not relied on any representation, warranty, promise or agreement other than those expressly contained herein or therein, and no other representation, warranty, promise or agreement has been made or will be implied. Except as otherwise expressly set forth herein or in the Restructuring Agreements or the Other Agreements, all Splitco Assets and Splitco Businesses are being transferred on an "as is, where is" basis, at the risk of the transferee, without any warranty whatsoever on the part of the transferor and from and after the Effective Time.

# ARTICLE IV COVENANTS

#### 4.1 <u>Cross-Indemnities</u>.

(a) Splitco hereby covenants and agrees, on the terms and subject to the limitations set forth in this Article IV, from and after the Closing, to indemnify and hold harmless LMC and its current and former directors, officers and employees, and each of the heirs, executors, trustees, administrators, successors and assigns of any of the foregoing (the "LMC Indemnified"

10

Parties") from and against any Losses incurred by the LMC Indemnified Parties (in their capacities as such) to the extent arising out of or resulting from any of the following:

- (i) the conduct of the Splitco Businesses (whether before or after the Closing);
- (ii) the Splitco Assets;
- (iii) the Splitco Liabilities (whether incurred before or after the Closing); or
- (iv) any breach of, or failure to perform or comply with, any covenant, undertaking or obligation of Splitco or any of its Subsidiaries under this Agreement, any Restructuring Agreement or any Other Agreement.
- (b) LMC hereby covenants and agrees, on the terms and subject to the limitations set forth in this Article IV, from and after the Closing, to indemnify and hold harmless Splitco and its current and former directors, officers and employees, and each of the heirs, executors, trustees, administrators, successors and assigns of any of the foregoing (the "Splitco Indemnified Parties") from and against any Losses incurred by the Splitco Indemnified Parties (in their capacities as such) to the extent arising out of or resulting from:
  - (i) the conduct of the LMC Retained Businesses (whether before or after the Closing);
  - (ii) the LMC Retained Assets;
  - (iii) the LMC Retained Liabilities (whether incurred before or after the Closing); or
  - (iv) any breach of, or failure to perform or comply with, any covenant, undertaking or obligation of LMC or any of its Subsidiaries (other than the Splitco Entities) under this Agreement, any Restructuring Agreement or any Other Agreement.
- (c) The indemnification provisions set forth in Sections 4.1(a) and (b) shall not apply to: (i) any Losses the responsibility for which is expressly covered by a Restructuring Agreement or an Other Agreement, including the Tax Sharing Agreement; (ii) any Losses incurred by any Splitco Entity pursuant to any contractual obligation (other than this Agreement, the Restructuring Agreements or the Other Agreements) existing on or after the Closing Date between (x) LMC or any of its Affiliates, on the one hand, and (y) Splitco or any of its Affiliates, on the Other Agreements) existing on or after the Closing Date between (x) LMC or any of its Affiliates, on the one hand, and (y) Splitco or any of its Affiliates, on the other hand.
- (d) (i) In connection with any indemnification provided for in this Section 4.1, the party seeking indemnification (the "<u>Indemnitee</u>") will give the party from which indemnification is sought (the "<u>Indemnitor</u>") prompt notice whenever it comes to the attention of the Indemnitee that the Indemnitee has suffered or incurred, or may suffer or incur, any Losses

11

for which it is entitled to indemnification under this Section 4.1, and, if and when known, the facts constituting the basis for such claim and the projected amount of such Losses (which shall not be conclusive as to the amount of such Losses), in each case in reasonable detail. Without limiting the generality of the foregoing, in the case of any Action commenced by a third party for which indemnification is being sought (a "Third-Party Claim"), such notice will be given no later than ten business days following receipt by the Indemnitee of written notice of such Third-Party Claim. Failure by any Indemnitee to so notify the Indemnitor will not affect the rights of such Indemnitee hereunder except to the extent that such failure has a material prejudicial effect on the defenses or other rights available to the Indemnitor with respect to such Third Party Claim. The Indemnitee will deliver to the Indemnitor as promptly as practicable, and in any event within five business days after Indemnitee's receipt, copies of all notices,

court papers and other documents received by the Indemnitee relating to any Third-Party Claim.

After receipt of a notice pursuant to Section 4.1(d)(i) with respect to any Third-Party Claim, the Indemnitor will be entitled, if it so elects, to take control of the defense and investigation with respect to such Third-Party Claim and to employ and engage attorneys reasonably satisfactory to the Indemnitee to handle and defend such claim, at the Indemnitor's cost, risk and expense, upon written notice to the Indemnitee of such election, which notice acknowledges the Indemnitor's obligation to provide indemnification under this Agreement with respect to any Losses arising out of or relating to such Third-Party Claim. The Indemnitor will not settle any Third-Party Claim that is the subject of indemnification without the written consent of the Indemnitee, which consent will not be unreasonably withheld, conditioned or delayed; provided, however, that, after reasonable notice, the Indemnitor may settle a claim without the Indemnitee's consent if such settlement (A) makes no admission or acknowledgment of Liability or culpability with respect to the Indemnitee, (B) includes a complete release of the Indemnitee and (C) does not seek any relief against the Indemnitee other than the payment of money damages to be borne by the Indemnitor. The Indemnitee will cooperate in all reasonable respects with the Indemnitor and its attorneys in the investigation, trial and defense of any lawsuit or action with respect to such Claim and any appeal arising therefrom (including the filing in the Indemnitee's name of appropriate crossclaims and counterclaims). The Indemnitee may, at its own cost, participate in any investigation, trial and defense of any Third-Party Claim controlled by the Indemnitor and any appeal arising therefrom, including participating in the process with respect to the potential settlement or compromise thereof. If the Indemnitee has been advised by its counsel that there may be one or more legal defenses available to the Indemnitee that conflict with those available to, or that are not available to, the Indemnitor ("Separate Legal Defenses"), or that there may be actual or potential differing or conflicting interests between the Indemnitor and the Indemnitee in the conduct of the defense of such Third-Party Claim, the Indemnitee will have the right, at the expense of the Indemnitor, to engage separate counsel reasonably acceptable to the Indemnitor to handle and defend such Third-Party Claim, provided, that, if such Third-Party Claim can be reasonably separated between those portion(s) for which Separate Legal Defenses are available ("Separable Claims") and those for which no Separate Legal Defenses are available, the Indemnitee will instead have the right, at the expense of the Indemnitor, to engage separate counsel reasonably acceptable to the Indemnitor to handle and defend the Separable Claims, and the Indemnitor will not have the right to control the defense or investigation of such Third-Party Claim or such Separable Claims, as the case may be (and, in which latter case, the Indemnitor will have the right to control the defense or investigation of the remaining portion(s) of such Third-Party Claim).

12

- (iii) If, after receipt of a notice pursuant to Section 4.1(d)(i) with respect to any Third-Party Claim as to which indemnification is available hereunder, the Indemnitor does not undertake to defend the Indemnitee against such Third-Party Claim, whether by not giving the Indemnitee timely notice of its election to so defend or otherwise, the Indemnitee may, but will have no obligation to, assume its own defense, at the expense of the Indemnitor (including attorneys fees and costs), it being understood that the Indemnitee's right to indemnification for such Third Party Claim shall not be adversely affected by its assuming the defense of such Third Party Claim. The Indemnitor will be bound by the result obtained with respect thereto by the Indemnitee; provided, that the Indemnitee may not settle any lawsuit or action with respect to which the Indemnitee is entitled to indemnification hereunder without the consent of the Indemnitor, which consent will not be unreasonably withheld, conditioned or delayed; provided further, that such consent shall not be required if (i) the Indemnitor had the right under this Section 4.1 to undertake control of the defense of such Third-Party Claim and, after notice, failed to do so within thirty days of receipt of such notice (or such lesser period as may be required by court proceedings in the event of a litigated matter), or (ii) (x) the Indemnitor does not have the right to control the defense of the entirety of such Third-Party Claim pursuant to Section 4.1(d)(ii) or (y) the Indemnitor does not have the right to control the defense of any Separable Claim pursuant to Section 4.1(d)(ii) (in which case such settlement may only apply to such Separable Claims), the Indemnitor provides reasonable notice to Indemnitor of the settlement, and such settlement (A) makes no admission or acknowledgment of Liability or culpability with respect to the Indemnitor, (B) does not seek any relief against the Indemnitor is responsible other than the payment of money damages.
- (e) In no event will the Indemnitor be liable to any Indemnitee for any special, consequential, indirect, collateral, incidental or punitive damages, however caused and on any theory of liability arising in any way out of this Agreement, whether or not such Indemnitor was advised of the possibility of any such damages; provided, that the foregoing limitations shall not limit a party's indemnification obligations for any Losses incurred by an Indemnitee as a result of the assertion of a Third Party Claim.
- (f) The Indemnitor and the Indemnitee shall use commercially reasonable efforts to avoid production of confidential information, and to cause all communications among employees, counsel and others representing any party with respect to a Third Party Claim to be made so as to preserve any applicable attorney-client or work-product privilege.
- (g) The Indemnitor shall pay all amounts payable pursuant to this Section 4.1 by wire transfer of immediately available funds, promptly following receipt from an Indemnitee of a bill, together with all accompanying reasonably detailed backup documentation, for any Losses that are the subject of indemnification hereunder, unless the Indemnitor in good faith disputes the amount of such Losses or whether such Losses are covered by the Indemnitor's indemnification obligation in which event the Indemnitor shall promptly so notify the Indemnitee. In any event, the Indemnitor shall pay to the Indemnitee, by wire transfer of immediately available funds, the amount of any Losses for which it is liable hereunder no later than three (3) days following any final determination of the amount of such Losses and the Indemnitor's liability therefor. A "final determination" shall exist when (a) the parties to the dispute have reached an agreement in

13

writing or (b) a court of competent jurisdiction shall have entered a final and non-appealable order or judgment.

- (h) If the indemnification provided for in this Section 4.1 shall, for any reason, be unavailable or insufficient to hold harmless an Indemnitee in respect of any Losses for which it is entitled to indemnification hereunder, then the Indemnitor shall contribute to the amount paid or payable by such Indemnitee as a result of such Losses, in such proportion as shall be appropriate to reflect the relative benefits received by and the relative fault of the Indemnitor on the one hand and the Indemnitee on the other hand with respect to the matter giving rise to such Losses.
- (i) The remedies provided in this Section 4.1 shall be cumulative and shall not preclude assertion by any Indemnitee of any other rights or the seeking of any and all other remedies against an Indemnitor, subject to Section 4.1(e).
  - (j) The rights and obligations of the LMC Indemnified Persons and the Splitco Indemnified Persons under this Section 4.1 shall survive the Split-Off.
- (k) For the avoidance of doubt, the provisions of this Section 4.1 are not intended to, and shall not, apply to any Loss, claim or Liability to which the provisions of the Tax Sharing Agreement are applicable.
- (l) To the fullest extent permitted by applicable law, the Indemnitor will indemnify the Indemnitee against any and all reasonable fees, costs and expenses (including attorneys' fees), incurred in connection with the enforcement of his, her or its rights under this Section 4.1.
- 4.2 <u>Further Assurances</u>. At any time before or after the Closing, each party hereto covenants and agrees to make, execute, acknowledge and deliver such instruments, agreements, consents, assurances and other documents, and to take all such other commercially reasonable actions, as any other party may reasonably request and as may reasonably be required in order to carry out the purposes and intent of this Agreement and to implement the terms hereof.
- 4.3 <u>Specific Performance</u>. Each party hereby acknowledges that the benefits to the other party of the performance by such party of its obligations under this Agreement are unique and that the other party is willing to enter into this Agreement only in reliance that such party will perform such obligations, and agrees that monetary damages may not afford an adequate remedy for any failure by such party to perform any of such obligations. Accordingly, each party hereby agrees that the other party will

have the right to enforce the specific performance of such party's obligations hereunder and irrevocably waives any requirement for securing or posting of any bond or other undertaking in connection with the obtaining by the other party of any injunctive or other equitable relief to enforce their rights hereunder.

#### 4.4 Access to Information.

(a) Each party will provide to the other party, at any time before or after the Redemption Date, upon written request and promptly after the request therefor (subject in all cases, to any bona fide concerns of attorney-client or work-product privilege that any party may reasonably have and any restrictions contained in any agreements or contracts to which any party or its Subsidiaries is a party (it being understood that each of LMC and Splitco will use its

14

reasonable best efforts to provide any such information in a manner that does not result in a violation of a privilege), any information in its possession or under its control that the requesting party reasonably needs (i) to comply with reporting, filing or other requirements imposed on the requesting party by a foreign or U.S. federal, state or local judicial, regulatory or administrative authority having jurisdiction over the requesting party or its Subsidiaries, (ii) to enable the requesting party to institute or defend against any action, suit or proceeding in any foreign or U.S. federal, state or local court or (iii) to enable the requesting party to implement the transactions contemplated hereby, including but not limited to performing its obligations under this Agreement, the Restructuring Agreements and the Other Agreements.

- (b) Any information belonging to a party that is provided to another party pursuant to Section 4.4(a) will remain the property of the providing party. The parties agree to cooperate in good faith to take all reasonable efforts to maintain any legal privilege that may attach to any information delivered pursuant to this Section 4.4 or which otherwise comes into the receiving party's possession and control pursuant to this Agreement. Nothing contained in this Agreement will be construed as granting or conferring license or other rights in any such information.
- (c) The party requesting any information under this Section 4.4 will reimburse the providing party for the reasonable out of pocket costs, if any, of creating, gathering and copying such information, to the extent that such costs are incurred for the benefit of the requesting party. No party will have any Liability to any other party if any information exchanged or provided pursuant to this Agreement that is an estimate or forecast, or is based on an estimate or forecast, is found to be inaccurate, absent willful misconduct or fraud by the party providing such information.
- (d) For the avoidance of doubt, the provisions of this Section 4.4 are not intended to, and shall not, apply to any information relating to matters governed by the Tax Sharing Agreement, which shall be subject to the provisions thereof in lieu of this Section 4.4.
- 4.5 <u>Confidentiality.</u> Each party will keep confidential for five years following the Closing Date (or for three years following disclosure to such party, whichever is longer), and will use reasonable efforts to cause its officers, directors, members, employees, Affiliates and agents to keep confidential during such period, all Proprietary Information of the other party, in each case to the extent permitted by applicable law.
- (a) "Proprietary Information" means any proprietary ideas, plans and information, including information of a technological or business nature, of a party (in this context, the "Disclosing Party") (including all trade secrets, intellectual property, data, summaries, reports or mailing lists, in whatever form or medium whatsoever, including oral communications, and however produced or reproduced), that is marked proprietary or confidential, or that bears a marking of like import, or that the Disclosing Party states is to be considered proprietary or confidential, or that a reasonable and prudent person would consider proprietary or confidential under the circumstances of its disclosure. Without limiting the foregoing, all information of the types referred to in the immediately preceding sentence to the extent used by Splitco or the Splitco Businesses or which constitute Splitco Assets on or prior to the Closing Date will

15

constitute Proprietary Information of Splitco for purposes of this Section 4.5 (it being understood that LMC or its Subsidiaries may retain copies of such information).

- (b) Anything contained herein to the contrary notwithstanding, information of a Disclosing Party will not constitute Proprietary Information (and the other party (in this context, the "Receiving Party") will have no obligation of confidentiality with respect thereto), to the extent such information: (i) is in the public domain other than as a result of disclosure made in breach of this Agreement or breach of any other agreement relating to confidentiality between the Disclosing Party and the Receiving Party; (ii) was lawfully acquired by the Disclosing Party from a third party not bound by a confidentiality obligation; (iii) is approved for release by prior written authorization of the Disclosing Party, or (iv) is disclosed in order to comply with a judicial order issued by a court of competent jurisdiction, or to comply with the laws or regulations of any governmental authority having jurisdiction over the Receiving Party, in which event the Receiving Party will give prior written notice to the Disclosing Party of such disclosure as soon as or to the extent practicable and will cooperate with the Disclosing Party in using reasonable efforts to disclose the least amount of such information required and to obtain an appropriate protective order or equivalent, and provided that the information will continue to be Proprietary Information to the extent it is covered by a protective order or equivalent or is not so disclosed.
- 4.6 Notices Regarding Transferred Assets. Any transferor of an Asset or Liability in the Restructuring that receives a notice or other communication from any third party, or that otherwise becomes aware of any fact or circumstance, after the Restructuring, relating to such Asset or Liability, will use commercially reasonable efforts to promptly forward the notice or other communication to the transferee thereof or give notice to such transferee of such fact or circumstance of which it has become aware. The parties will cause their respective Subsidiaries to comply with this Section 4.6.
- 4.7 <u>Treatment Of Payments.</u> The parties agree to treat all payments made pursuant to this Agreement in accordance with Section 4.6 of the Tax Sharing Agreement and to increase or reduce any amount paid hereunder if such payment would have been required to be increased or reduced under such section if it was a payment made pursuant to the Tax Sharing Agreement.

#### ARTICLE V CLOSING

5.1 Closing. Unless this Agreement is terminated and the transactions contemplated by this Agreement abandoned pursuant to the provisions of Article VI, and subject to the satisfaction of all conditions set forth in Sections 2.2 and 5.2 (or the waiver of such conditions, to the extent such conditions may be waived), the closing of the Redemptions (the "Closing") will take place at the offices of LMC, at 12300 Liberty Boulevard, Englewood, Colorado, at a mutually acceptable time and date to be determined by LMC (the "Closing Date").

16

## 5.2 <u>Conditions to Closing.</u>

(a) The obligations of the parties to complete the transactions provided for herein are conditioned upon the satisfaction or, if applicable, waiver of the

- (b) The performance by each party of its obligations hereunder is further conditioned upon:
- (i) the performance in all material respects by the other party of its covenants and agreements contained herein to the extent such are required to be performed at or prior to the Closing; and
- (ii) the representations and warranties of the other party being true and complete in all material respects as of the Closing Date with the same force and effect as if made at and as of the Closing Date.
- 5.3 <u>Deliveries at Closing</u>
- (a) <u>LMC</u>. At the Closing, LMC will deliver or cause to be delivered to Splitco:
  - the Tax Sharing Agreement duly executed by an authorized officer of LMC and Liberty Media LLC;
  - (ii) the Services Agreement duly executed by an authorized officer of LMC;
  - (iii) the Facilities Sharing Agreement duly executed by an authorized officer of LMC;
  - (iv) each Aircraft Time Sharing Agreement duly executed by an authorized officer of Liberty Media LLC (a Subsidiary of LMC);
- (v) a secretary's certificate certifying that the LMC Board has authorized the execution, delivery and performance by LMC of this Agreement, the Restructuring Agreements and the Other Agreements, which authorization will be in full force and effect at and as of the Closing; and
  - (vi) such other documents and instruments as Splitco may reasonably request.
- (b) <u>Splitco</u>. At the Closing, Splitco will deliver or cause to be delivered to LMC:
  - (i) the Tax Sharing Agreement duly executed by an authorized officer of Splitco;
  - (ii) the Services Agreement duly executed by an authorized officer of Splitco;
  - (iii) the Facilities Sharing Agreement duly executed by an authorized officer of Liberty Property Holdings, Inc. (a Subsidiary of Splitco);
  - (iv) each Aircraft Time Sharing Agreement duly executed by an authorized officer of Splitco;

17

- (v) a secretary's certificate certifying that the Splitco Board has authorized the execution, delivery and performance by Splitco of this Agreement, the Restructuring Agreements and the Other Agreements, which authorizations will be in full force and effect at and as of the Closing; and
  - (vi) such other documents and instruments as LMC may reasonably request.

### ARTICLE VI TERMINATION

- 6.1 <u>Termination</u>. This Agreement may be terminated and the transactions contemplated hereby may be amended, modified, supplemented or abandoned at any time prior to the Effective Time by and in the sole and absolute discretion of LMC without the approval of Splitco or the holders of Liberty Capital Stock or Liberty Starz Stock. For the avoidance of doubt, from and after the Effective Time, this Agreement may not be terminated (or any provision hereof modified, amended or waived) without the written agreement of all the parties.
- 6.2 <u>Effect of Termination</u>. In the event of any termination of this Agreement in accordance with Section 6.1, this Agreement will immediately become void and the parties will have no Liability whatsoever to each other with respect to the transactions contemplated hereby.

#### ARTICLE VII MISCELLANEOUS

- 7.1 <u>Definitions</u>.
- (a) For purposes of this Agreement, the following terms have the corresponding meanings:

"Action" means any demand, action, claim, suit, countersuit, litigation, arbitration, prosecution, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation whether or not commenced, brought, conducted or heard by or before, or otherwise involving, any court, grand jury or other governmental authority or any arbitrator or arbitration panel.

"Affiliates" means with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such first Person; provided, that none of LMC and its Subsidiaries shall be deemed to be Affiliates of Splitco or any of its Subsidiaries, and none of Splitco and its Subsidiaries shall be deemed to be Affiliates of LMC or any of its Subsidiaries, for any purpose hereunder, in each case both before and after the Effective Time; and provided further that none of Expedia, Inc. or any of its present or former Affiliates will be deemed to be Affiliates of LMC, Splitco or any of their respective Subsidiaries for any purpose hereunder during any period in which the voting stock of such corporation owned by LMC, Splitco or any of their respective Subsidiaries is subject to a proxy in favor of Mr. Barry Diller to vote those shares during such period.

18

"Aircraft Time Sharing Agreement" means the Aircraft Time Sharing Agreements to be entered into between Liberty Media, LLC and Splitco, one for each of the two aircraft owned by Splitco, substantially in the form attached hereto as <a href="Exhibit A.">Exhibit A.</a>.

"Assets" means assets, properties, interests and rights (including goodwill), wherever located, whether real, personal or mixed, tangible or intangible, movable or immovable, in each case whether or not required by GAAP to be reflected in financial statements or disclosed in the notes thereto.

"Contribution" has the meaning given to such term in the Restructuring Plan.

"Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of securities or partnership, membership, limited liability company, or other ownership interests, by contract or otherwise and the terms "Controlling" and "Controlled" have meanings correlative to the foregoing.

"DGCL" means the Delaware General Corporation Law.

"Exchange Act" means the Securities Exchange Act of 1934, together with the rules and regulations promulgated thereunder.

"Excluded Assets" means all Assets attributed to LMC's Capital Group or Starz Group which, pursuant to the Restructuring Plan, have been, or will be, reattributed to LMC's Interactive Group by no later than immediately prior to the Effective Time.

"Excluded Liabilities" means all Liabilities attributed to LMC's Capital Group or Starz Group which, pursuant to the Restructuring Plan, have been or will be reattributed to LMC's Interactive Group by no later than immediately prior to the Effective Time.

"Facilities Sharing Agreement" means the Facilities Sharing Agreement to be entered into between Liberty Property Holdings, Inc. and LMC, substantially in the form attached hereto as Exhibit B.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

"Governmental Authorization" means any authorization, approval, consent, license, certificate or permit issued, granted, or otherwise made available under the authority of any court, governmental or regulatory authority, agency, stock exchange, commission or body.

"IRS" means the Internal Revenue Service.

"Liabilities" means any and all debts, liabilities, commitments and obligations, whether or not fixed, contingent or absolute, matured or unmatured, direct or indirect, liquidated or unliquidated, accrued or unaccrued, known or unknown, and whether or not

19

required by GAAP to be reflected in financial statements or disclosed in the notes thereto (other than taxes).

"LMC Board" means the Board of Directors of LMC or a duly authorized committee thereof.

"LMC Charter" means the Restated Certificate of Incorporation of LMC, as in effect immediately prior to the Redemption Date.

"LMC Entity" or "LMC Entities" means and includes each of LMC and its Subsidiaries (other than the Splitco Entities), after giving effect to the Restructuring.

"LMC Option" means an option to purchase shares of Liberty Capital Stock or Liberty Starz Stock pursuant to a stock incentive plan of LMC.

"LMC Retained Assets" means all Assets which are attributed at the Effective Time to LMC's Interactive Group, including the Excluded Assets.

"LMC Retained Businesses" means all businesses which are attributed at the Effective Time to LMC's Interactive Group, including QVC, Inc., Backcountry.com, Bodybuilding.com., Celebrate Interactive, LLC and Provide Commerce, Inc.

"LMC Retained Liabilities" means all Liabilities which are attributed at the Effective Time to LMC's Interactive Group, including the Excluded Liabilities.

"LMC SARs" means stock appreciation rights with respect to shares of Liberty Capital Stock or Liberty Starz Stock granted under a stock incentive plan of LMC.

"Losses" means any and all damages, losses, deficiencies, Liabilities, penalties, judgments, settlements, claims, payments, fines, interest, costs and expenses (including the fees and expenses of any and all actions and demands, assessments, judgments, settlements and compromises relating thereto and the costs and expenses of attorneys', accountants', consultants' and other professionals' fees and expenses incurred in the investigation or defense thereof or the enforcement of rights hereunder), whether in connection with a Third-Party Claim or otherwise.

"Order" means any order, injunction, judgment, decree or ruling of any court, governmental or regulatory authority, agency, commission or body.

"Person" means any individual, corporation, company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Proxy Statement/Prospectus" means the proxy statement of LMC relating to the Stockholder Meeting which includes the prospectus of Splitco and forms a part of the Registration Statement.

"Qualifying Subsidiary" means a former direct or indirect Subsidiary of LMC (or its predecessor Liberty Media LLC), any successor of any such former Subsidiary, and

20

the parent company (directly or indirectly) of any such former Subsidiary or successor, including Splitco, Ascent Capital Group, Inc., DIRECTV, Discovery Communications, Inc. and Liberty Global, Inc.

"Redemption" means the LCAP Redemption or the LSTZ Redemption.

"Redemption Agent" means Computershare Trust Company, N.A., P.O. Box 430023, Providence, Rhode Island 02940-3023.

"Registration Statement" means the registration statement of Splitco on Form S-4 relating to the Redemptions.

"Restructuring Plan" means the Plan of Restructuring attached hereto as Schedule 1.1.

"SEC" means the Securities and Exchange Commission.

"SEC Filings" means the Registration Statement, the Proxy Statement/Prospectus, any amendments or supplements thereto, including any preliminary filings of the same, and any other registration statements or reports filed under the Securities Act or Exchange Act, in connection with the Redemptions.

"Securities Act" means the Securities Act of 1933, together with all rules and regulations promulgated thereunder.

"Services Agreement" means the Services Agreement to be entered into between LMC and Splitco, substantially in the form attached hereto a Exhibit C.

"Shares" has the meaning given to such term in the Restructuring Plan.

"Splitco Assets" means all Assets which are attributed immediately prior to the Effective Time to either LMC's Capital Group or Starz Group. The Splitco Assets do not include the Excluded Assets.

"Splitco Board" means the Board of Directors of Splitco or a duly authorized committee thereof.

"Splitco Businesses" means all businesses which are attributed immediately prior to the Effective Time to either LMC's Capital Group or Starz Group, including Atlanta National League Baseball Club, TruePosition, Inc., Starz Entertainment, LLC, Starz Media, LLC and Liberty Sports Interactive, Inc.

"Splitco Charter" means the Restated Certificate of Incorporation of Splitco to be filed with the Delaware Secretary of State immediately prior to the Effective Time, substantially in the form attached hereto as Exhibit D.

"Splitco Entity" or "Splitco Entities" means and includes each of Splitco and its Subsidiaries, after giving effect to the Restructuring.

21

"Splitco Liabilities" means all Liabilities which are attributed immediately prior to the Effective Time to either LMC's Capital Group or Starz Group. The Splitco Liabilities do not include the Excluded Liabilities.

"Splitco Option" means any option to purchase shares of Splitco Capital Stock or Splitco Starz Stock issued pursuant to the Splitco Transitional Plan.

"Splitco Restricted Stock" means any shares of Splitco Capital Stock or Splitco Starz Stock subject to restricted stock awards issued pursuant to the Splitco Transitional Plan.

"Splitco SARs" means stock appreciation rights with respect to Splitco Capital Stock or Splitco Starz Stock issued pursuant to the Splitco Transitional Plan.

"Splitco Stock" means the Splitco Capital Stock and the Splitco Starz Stock.

"Splitco Transitional Plan" means the 2011 Splitco Transitional Stock Adjustment Plan.

"Subsidiary" when used with respect to any Person, means (i)(A) a corporation a majority in voting power of whose share capital or capital stock with voting power, under ordinary circumstances, to elect directors is at the time, directly or indirectly, owned by such Person, by one or more Subsidiaries of such Person, or by such Person and one or more Subsidiaries of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, (B) a partnership or limited liability company in which such Person or a Subsidiary of such Person is, at the date of determination, (1) in the case of a partnership, a general partner of such partnership with the power affirmatively to direct the policies and management of such partnership or (2) in the case of a limited liability company, the managing member or, in the absence of a managing member, a member with the power affirmatively to direct the policies and management of such limited liability company, or (C) any other Person (other than a corporation) in which such Person, one or more Subsidiaries of such Person or such Person and one or more Subsidiaries of such Person, directly or indirectly, at the date of determination thereof, has or have (1) the power to elect or direct the election of a majority of the members of the governing body of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, or (2) in the absence of such a governing body, at least a majority ownership interest or (ii) any other Person of which an aggregate of 50% or more of the equity interests are, at the time, directly or indirectly, owned by such Person and/or one or more Subsidiaries of such Person. For purposes of this Agreement, both prior to and after the Effective Time, none of Splitco and its Subsidiaries shall be deemed to be Subsidiaries of LMC or any of its Subsidiaries, and neither IAC/InteractiveCorp nor Expedia, Inc. (nor any of their respective present or former Subsidiaries) will be treated as Subsidiaries of LMC or Spl

22

"Tax Sharing Agreement" means the Tax Sharing Agreement to be entered into among LMC, Liberty Media LLC and Splitco, substantially in the form attached hereto as Exhibit E.

"Treasury Regulations" means the Treasury regulations promulgated under the Code.

(b) As used herein, the following terms will have the meanings set forth in the applicable section of this Agreement set forth below:

Agreement	Preamble
Awards	Section 2.3(a)
Closing	Section 5.1
Closing Date	Section 5.1
Code	Recitals
Delaware Action	Section 2.2(j)
Disclosing Party	Section 4.5(a)
Effective Time	Section 2.1(a)
FCC	Section 2.2(g)
Indemnitee	Section 4.1(d)(i)
Indemnitor	Section 4.1(d)(i)
Indenture	Section 2.2(j)
LCAPA	Recitals
LCAPB	Recitals
LCAP Redemption	Recitals
Liberty Capital Stock	Recitals

Liberty Starz Stock	Recitals
LMC	Preamble
LMC Indemnified Parties	Section 4.1(a)
LMC Restricted Stock	Section 2.3(d)
LSTZA	Recitals
LSTZB	Recitals
LSTZ Redemption	Recitals
Nasdaq	Section 2.2(f)
Outstanding LMC Capital Option	Section 2.3(b)(i)
Outstanding LMC Capital SAR	Section 2.3(c)(i)
Outstanding LMC Starz Option	Section 2.3(b)(ii)
Outstanding LMC Starz SAR	Section 2.3(c)(ii)
Other Agreements	Section 3.1(b)
Proprietary Information	Section 4.5(a)
Receiving Party	Section 4.5(b)
Redemptions	Recitals
Redemption Date	Section 2.1(a)
Restructuring	Section 1.1(a)
Restructuring Agreements	Section 3.1(b)
Ruling	Section 2.2(c)
Separable Claims	Section 4.1(d)(ii)

23

Separate Legal Defenses	Section 4.1(d)(ii)
Splitco	Preamble
Splitco Capital Option	Section 2.3(b)(i)
Splitco Capital SAR	Section 2.3(c)(i)
Splitco Capital Stock	Section 2.1(b)
Splitco Indemnified Parties	Section 4.1(b)
Splitco Series A Capital Stock	Section 2.1(b)
Splitco Series B Capital Stock	Section 2.1(b)
Splitco Series A Starz Stock	Section 2.1(b)
Splitco Series B Starz Stock	Section 2.1(b)
Splitco Starz Option	Section 2.3(b)(ii)
Splitco Starz SAR	Section 2.3(c)(ii)
Splitco Starz Stock	Section 2.1(b)
Split-Off	Recitals
Stockholders Meeting	Section 2.2(a)
Third-Party Claim	Section 4.1(d)(i)

- 7.2 No Third-Party Rights. Except for the indemnification rights of the LMC Indemnified Persons and the Splitco Indemnified Persons pursuant to Section 4.1, nothing expressed or referred to in this Agreement is intended or will be construed to give any Person other than the parties hereto and their respective successors and assigns any legal or equitable right, remedy or claim under or with respect to this Agreement, or any provision hereof, it being the intention of the parties hereto that this Agreement and all of its provisions and conditions are for the sole and exclusive benefit of the parties to this Agreement and their respective successors and assigns.
- 7.3 Notices. All notices and other communications hereunder shall be in writing and shall be delivered in person, by facsimile (with confirming copy sent by one of the other delivery methods specified herein), by overnight courier or sent by certified, registered or express air mail, postage prepaid, and shall be deemed given when so delivered in person, or when so received by facsimile or courier, or, if mailed, three (3) calendar days after the date of mailing, as follows:

if to any LMC Entity: Liberty Media Corporation

12300 Liberty Boulevard Englewood, Colorado 80112 Facsimile (720) 875-5401 Attention: General Counsel

if to any Splitco Entity: Liberty CapStarz, Inc.

12300 Liberty Boulevard Englewood, Colorado 80112 Facsimile (720) 875-5401 Attention: General Counsel

24

or to such other address as the party to whom notice is given may have previously furnished to the other party in writing in the manner set forth above.

- 7.4 Entire Agreement. This Agreement (including the Exhibits and Schedules attached hereto) together with the Restructuring Agreements and the Other Agreements (including the Tax Sharing Agreement and the Services Agreement) embodies the entire understanding among the parties relating to the subject matter hereof and thereof and supersedes and terminates any prior agreements and understandings among the parties with respect to such subject matter, and no party to this Agreement shall have any right, responsibility or Liability under any such prior agreement or understanding. Any and all prior correspondence, conversations and memoranda are merged herein and shall be without effect hereon. No promises, covenants or representations of any kind, other than those expressly stated herein, have been made to induce either party to enter into this Agreement.
- 7.5 <u>Binding Effect; Assignment.</u> This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Except with respect to a merger of a party, neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other parties; *provided*, *however*, that LMC and Splitco may assign their respective rights, interests, duties, liabilities and obligations under this Agreement to any of their respective wholly-owned Subsidiaries, but such assignment shall not relieve LMC or Splitco, as the assignor, of its obligations hereunder.

- 7.6 Costs and Expenses. LMC and Splitco will each pay their respective costs and expenses incurred in connection with the Restructuring and the Redemptions.
- 7.7 Governing Law; Jurisdiction. This Agreement and the legal relations among the parties hereto will be governed in all respects, including validity, interpretation and effect, by the laws of the State of Delaware applicable to contracts made and performed wholly therein, without giving effect to any choice or conflict of laws provisions or rules that would cause the application of the laws of any other jurisdiction. Each of the parties hereto irrevocably agrees that any legal action or proceeding with respect to this Agreement, and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware). Each of the parties hereto hereby irrevocably submits with regard to any such action or proceeding for itself and in respect of its property, generally and unconditionally, to the personal jurisdiction of the aforesaid courts and agrees that it will not bring any action relating to this Agreement or the transactions contemplated hereby in any court other than the aforesaid courts. Each of the parties hereto hereby irrevocably waives, and agrees not to assert as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement (a) any claim that it is not personally subject to the jurisdiction of the above named courts for any reason other than the failure to serve in accordance with Section 7.3 and this Section 7.7, (b) any claim that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of

25

notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) to the fullest extent permitted by applicable law, any claim that (i) the suit, action or proceeding in such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper or (iii) this Agreement or the subject matter hereof may not be enforced in or by such courts. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 7.3 shall be deemed effective service of process on such party.

- 7.8 Waiver of Jury Trial. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF SUCH ACTION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 7.8.
- 7.9 Severability. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof. Any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Upon a determination that any provision of this Agreement is prohibited or unenforceable in any jurisdiction, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the provisions contemplated hereby are consummated as originally contemplated to the fullest extent possible.
- 7.10 Amendments; Waivers. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed, in the case of an amendment, by each party to this Agreement, or in the case of a waiver, by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise provided herein, the rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by applicable law. Any consent provided under this Agreement must be in writing, signed by the party against whom enforcement of such consent is sought.

26

#### 7.11 No Strict Construction; Interpretation.

- (a) LMC and Splitco each acknowledge that this Agreement has been prepared jointly by the parties hereto and shall not be strictly construed against any party hereto.
- (b) When a reference is made in this Agreement to an Article, Section, Exhibit or Schedule, such reference shall be to an Article of, a Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. All terms defined in this Agreement shall have the defined meanings when used in any certificate or other document made or delivered pursuant hereto unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein. References to a Person are also to its permitted successors and assigns and references to a party means a party to this Agreement.
- 7.12 <u>Conflicts with Tax Sharing Agreement.</u> In the event of a conflict between this Agreement and the Tax Sharing Agreement, the provisions of the Tax Sharing Agreement shall prevail.
- 7.13 <u>Counterparts.</u> This Agreement may be executed in two or more identical counterparts, each of which shall be deemed to be an original, and all of which together shall constitute one and the same agreement. The Agreement may be delivered by facsimile transmission of a signed copy thereof.

27

By: /s/ Craig Troyer

Name: Craig Troyer Title: Vice President

LIBERTY CAPSTARZ, INC.

By: /s/ Pamela Coe

Name: Pamela Coe Title: Vice President

[SIGNATURE PAGE TO REORGANIZATION AGREEMENT]

## Exhibit A

Form of Aircraft Time Sharing Agreement

#### Exhibit B

Form of Facilities Sharing Agreement

#### Exhibit C

Form of Services Agreement

### Exhibit D

Form of Splitco Charter

## Exhibit E

Form of Tax Sharing Agreement

### Schedule 1.1

Restructuring Plan

## List of Omitted Exhibits and Schedules

The following exhibits and schedules to the Reorganization Agreement, dated as of August 30, 2011, by and between Liberty Media Corporation and Liberty CapStarz, Inc., have not been provided herein:

Exhibit A: Form of Aircraft Time Sharing Agreement

Exhibit B: Form of Facilities Sharing Agreement

Exhibit C: Form of Services Agreement

Exhibit D: Form of Splitco Charter

Exhibit E: Form of Tax Sharing Agreement

Schedule 1.1—Restructuring Plan

The undersigned registrant hereby undertakes to furnish supplementally a copy of any omitted exhibit or schedule to the Securities and Exchange Commission upon request.

# RESTATED CERTIFICATE OF INCORPORATION OF LIBERTY MEDIA CORPORATION

LIBERTY MEDIA CORPORATION, a corporation organized and existing under the laws of the State of Delaware, hereby certifies as follows:

- (1) The name of the Corporation is Liberty Media Corporation. The original Certificate of Incorporation of the Corporation was filed on May 7, 2007. The name under which the Corporation was originally incorporated is ANLBC/LA Holdings Inc.
  - (2) This Restated Certificate of Incorporation restates and amends the Certificate of Incorporation of the Corporation.
- (3) This Restated Certificate of Incorporation has been duly adopted in accordance with Sections 228, 242 and 245 of the General Corporation Law of the State of Delaware.
  - (4) This Restated Certificate of Incorporation will become effective on September 23, 2011 at 5:00 p.m. EST.
- (5) Pursuant to Sections 242 and 245 of the General Corporation Law of the State of Delaware, the text of the Certificate of Incorporation is hereby restated to read in its entirety as follows:

#### ARTICLE I

#### NAME

The name of the corporation is Liberty Media Corporation (the 'Corporation').

#### ARTICLE II

#### REGISTERED OFFICE

The address of the registered office of the Corporation in the State of Delaware is 2711 Centerville Road, Suite 400, in the City of Wilmington, County of New Castle, 19808. The name of its registered agent at such address is the Corporation Service Company.

1

#### ARTICLE III

#### **PURPOSE**

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of Delaware (as the same may be amended from time to time, "DGCL").

### ARTICLE IV

## AUTHORIZED STOCK

The total number of shares of capital stock which the Corporation will have authority to issue is twelve billion two hundred seventy-five million (12,275,000,000) shares, which will be divided into the following classes:

- (a) twelve billion two hundred twenty-five million (12,225,000,000) shares will be of a class designated Common Stock, par value \$0.01 per share ("Common Stock"), such class to be divided in series as provided in Section A of this Article IV; and
- (b) fifty million (50,000,000) shares will be of a class designated Preferred Stock, par value \$0.01 per share (**Preferred Stock**"), such class to be issuable in series as provided in Section B of this Article IV.

Upon this Certificate becoming effective pursuant to the DGCL (the 'Effective Time'), all shares of Common Stock, par value \$0.01 per share, issued and outstanding immediately prior to the Effective Time shall automatically be reclassified as (i) W (as defined below) number of shares of the Corporation's Series A Liberty Starz Common Stock, par value \$0.01 per share, (ii) X (as defined below) number of shares of the Corporation's Series B Liberty Starz Common Stock, par value \$0.01 per share, (iii) Y (as defined below) number of shares of the Corporation's Series B Liberty Starz Common Stock, par value \$0.01 per share, and (iv) Z (as defined below) number of shares of the Corporation's Series B Liberty Capital Common Stock, par value \$0.01 per share, in each case without any action by the holder thereof. As used in this paragraph, "W" means the number of outstanding shares of Liberty Interactive Corporation's (formerly known as Liberty Media Corporation) Series A Liberty Starz Common Stock, par value \$.01 per share, "X" means the number of outstanding shares of Liberty Interactive Corporation's Series B Liberty Starz Common Stock, par value \$.01 per share, "Y" means the number of outstanding shares of Liberty Interactive Corporation's Series B Liberty Starz Common Stock, par value \$.01 per share, in each case, as of the Effective Time.

The description of the Common Stock and the Preferred Stock of the Corporation, and the relative rights, preferences and limitations thereof, or the method of fixing and establishing the same, are as hereinafter in this Article IV set forth:

2

## SECTION A

## COMMON STOCK

### 1. General.

Capital Common Stock"), seventy five million (75,000,000) shares of Common Stock will be of a series designated Series B Liberty Capital Common Stock (the 'Series B Liberty Capital Common Stock"), and two billion (2,000,000,000) shares of Common Stock will be of a series designated as Series C Liberty Capital Common Stock (the "Series C Liberty Capital Common Stock") and together with the Series A Liberty Capital Common Stock and the Series B Liberty Capital Common Stock, the 'Liberty Capital Common Stock"). Four billion (4,000,000,000,000) shares of Common Stock will be of a series designated Series A Liberty Starz Common Stock (the 'Series A Liberty Starz Common Stock)"), one hundred fifty million (150,000,000) shares of Common Stock will be of a series designated Series B Liberty Starz Common Stock (the 'Series B Liberty Starz Common Stock'), and four billion (4,000,000,000) shares of Common Stock will be of a series designated as Series C Liberty Starz Common Stock (the 'Series C Liberty Starz Common Stock will be of a series designated as Series C Liberty Starz Common Stock (the 'Series C Liberty Starz Common Stock) and together with the Series A Liberty Starz Common Stock and the Series B Liberty Starz Common Stock, the 'Liberty Starz Common Stock').

#### 2. <u>Liberty Capital Common Stock and Liberty Starz Common Stock.</u>

Each share of Series A Liberty Capital Common Stock, Series B Liberty Capital Common Stock and Series C Liberty Capital Common Stock will, except as otherwise provided in this Section A.2., be identical in all respects and will have equal rights, powers and privileges.

Each share of Series A Liberty Starz Common Stock, Series B Liberty Starz Common Stock and Series C Liberty Starz Common Stock will, except as otherwise provided in this Section A.2., be identical in all respects and will have equal rights, powers and privileges.

#### (a) <u>Voting Powers</u>.

(i) Series A Liberty Capital Common Stock, Series B Liberty Capital Common Stock, Series A Liberty Starz Common Stock and Series B Liberty Capital Common Stock will be entitled to one vote for each share of such stock held of record, holders of Series A Liberty Starz Common Stock will be entitled to one vote for each share of such stock held of record and holders of Series B Liberty Starz Common Stock will be entitled to ten votes for each share of such stock held of record, upon all matters that may be submitted to a vote of stockholders of the Corporation (regardless of whether such holders are voting together with the holders of all Voting Securities, or as a separate class with the holders of one or more series of Common Stock, or as a separate series of Common Stock, or otherwise).

3

- (ii) Series C Liberty Capital Common Stock and Series C Liberty Starz Common Stock. Holders of Series C Liberty Capital Common Stock and holders of Series C Liberty Starz Common Stock will not be entitled to any voting powers, except as (and then only to the extent) required by the laws of the State of Delaware. If a vote of the holders of Series C Liberty Capital Common Stock or Series C Liberty Starz Common Stock should at any time be required by the laws of the State of Delaware on any matter, the holders of Series C Liberty Capital Common Stock or Series C Liberty Starz Common Stock, as applicable, will be entitled to 1/100th of a vote on such matter for each share held of record.
- Delaware or (C) as may otherwise be provided in any Preferred Stock Designation, the holders of shares of Series A Liberty Capital Common Stock, the holders of shares of Series B Liberty Capital Common Stock, the holders of shares of Series B Liberty Capital Common Stock, the holders of shares of Series B Liberty Capital Common Stock, the holders of shares of Series B Liberty Starz Common Stock and the holders of shares of Preferred Stock that is designated as a Voting Security and is entitled to vote thereon in accordance with the terms of the applicable Preferred Stock Designation will vote as one class with respect to the election of directors and with respect to all other matters to be voted on by stockholders of the Corporation (including, without limitation and irrespective of the provisions of Section 242(b)(2) of the DGCL, any proposed amendment to this Certificate that (i) would increase (x) the number of authorized shares of Common Stock or any series thereof, (y) the number of authorized shares of Preferred Stock or any series thereof or (z) the number of authorized shares of any other class or series of capital stock of the Corporation hereafter established, or (ii) decrease (x) the number of authorized shares of any other class or series of capital stock of the Corporation hereafter established (but, in each case, not below the number of shares of such class or series of capital stock (as the case may be) then outstanding)), and no separate class or series vote of the holders of shares of any class or series of capital stock of the Corporation will be required for the approval of any such matter. In the event the holders of the Series C Liberty Capital Common Stock and/or the holders will vote as one class with all other stockholders of the Corporation entitled to vote on any matter that may be submitted to a vote of stockholders of the Corporation, such holders will vote as one class with all other stockholders of the Corporation entitled to vote on such matter, unless otherwise required by this C
- (iv) Special Voting Rights in Connection with Dispositions (A) If the Board of Directors, at its election, determines to seek the approval of the holders of Liberty Capital Voting Securities entitled to vote thereon to classify a proposed Capital Group Disposition as an Exempt Capital Group Disposition, then such proposed Capital Group Disposition will constitute an Exempt Capital Group Disposition if approved by the holders of record, as of the record date for the meeting at which such vote is taken, of Liberty Capital Voting Securities representing a majority of the aggregate voting power of Liberty Capital Voting Securities that are present in person or by proxy at such meeting, voting together as a separate class.

4

- (B) If the Board of Directors, at its election, determines to seek the approval of the holders of Liberty Starz Voting Securities entitled to vote thereon to classify a proposed Starz Group Disposition as an Exempt Starz Group Disposition, then such proposed Starz Group Disposition will constitute an Exempt Starz Group Disposition if approved by the holders of record, as of the record date for the meeting at which such vote is taken, of Liberty Starz Voting Securities representing a majority of the aggregate voting power of Liberty Starz Voting Securities that are present in person or by proxy at such meeting, voting together as a separate class.
- (C) Any vote taken pursuant to clause (A) or (B) of this paragraph (a)(iv) will be in addition to, and not in lieu of, any vote of the stockholders of the Corporation required pursuant to Article IX of this Certificate or the DGCL to be taken with respect to the applicable Disposition.
- (v) Special Voting Rights in Connection with Certain Redemptions (A) If the Corporation proposes to redeem outstanding shares of Liberty Capital Common Stock for securities of a Subsidiary pursuant to paragraph (e)(i) of this Section A.2., such redemption will be subject to, and will not be undertaken unless, the Corporation has received the approval of the holders of record, as of the record date for the meeting at which such vote is taken, of Liberty Capital Voting Securities representing a majority of the aggregate voting power of Liberty Capital Voting Securities that are present in person or by proxy at such meeting, voting together as a separate class (a "Capital Group Redemption Stockholder Approval").
- (B) If the Corporation proposes to redeem outstanding shares of Liberty Starz Common Stock for securities of a Subsidiary pursuant to paragraph (f)(i) of this Section A.2., such redemption will be subject to, and will not be undertaken unless, the Corporation has received the approval of the holders of record, as of the record date for the meeting at which such vote is taken, of Liberty Starz Voting Securities representing a majority of the aggregate voting power of Liberty Starz Voting Securities that are present in person or by proxy at such meeting, voting together as a separate class (a "Starz Group Redemption Stockholder Approval").
- (C) Any vote taken pursuant to clause (A) or (B) of this paragraph (a)(v) will be in addition to, and not in lieu of, any vote of the stockholders of the Corporation required by the DGCL to be taken with respect to the applicable redemption.

(i) (A) Conversion of Series B Liberty Capital Common Stock into Series A Liberty Capital Common Stock; Other Each share of Series B Liberty Capital Common Stock will be convertible at any time, at the option of the holder thereof, into one fully paid and non-assessable share of Series A Liberty Capital Common Stock. Any such conversion may be effected by any holder of Series B Liberty Capital Common Stock by surrendering such holder's certificate or certificates representing the Series B Liberty Capital Common Stock to be converted, duly endorsed, at the principal office of the Corporation or any transfer agent for the

- 5

Series B Liberty Capital Common Stock, together with a written notice to the Corporation at such office that such holder elects to convert all or a specified whole number of shares of Series B Liberty Capital Common Stock represented by such certificate or certificates and stating the name or names in which such holder desires the certificate or certificates representing shares of Series A Liberty Capital Common Stock to be issued and, if less than all of the shares of Series B Liberty Capital Common Stock represented by one certificate are to be converted, the name or names in which such holder desires the certificate or certificates representing the unconverted shares of Series B Liberty Capital Common Stock to be issued. Any certificate representing shares surrendered for conversion in accordance with this paragraph will, if so required by the Corporation or its transfer agent, be accompanied by instruments of transfer, in form satisfactory to the Corporation or its transfer agent, duly executed by the holder of such shares or the duly authorized representative of such holder, and will, if required by the next succeeding paragraph, be accompanied by payment, or evidence of payment, of applicable issue or transfer taxes. Promptly thereafter, the Corporation will issue and deliver to such holder or such holder's nominee or nominees, a certificate or certificates representing the number of shares of Series A Liberty Capital Common Stock to which such holder will be entitled as herein provided. If less than all of the shares of Series B Liberty Capital Common Stock represented by any one certificate are to be converted, the Corporation will issue and deliver to such holder or such holder's nominee or nominees a new certificate representing the shares of Series B Liberty Capital Common Stock not converted. Such conversion will be deemed to have been made at the close of business on the date of receipt by the Corporation or any such transfer agent of the certificate or certificates, notice and, if required, instruments of transfer and payment or evidence of payment of taxes referred to above, and the Person or Persons entitled to receive the Series A Liberty Capital Common Stock issuable on such conversion will be treated for all purposes as the record holder or holders of such Series A Liberty Capital Common Stock on that date. A number of shares of Series A Liberty Capital Common Stock equal to the number of shares of Series B Liberty Capital Common Stock outstanding from time to time will be set aside and reserved for issuance upon conversion of shares of Series B Liberty Capital Common Stock as provided herein. Shares of Series A Liberty Capital Common Stock and shares of Series C Liberty Capital Common Stock will not be convertible at the option of the holder into shares of any other series of Common Stock.

The Corporation will pay any and all documentary, stamp or similar issue or transfer taxes that may be payable in respect of the issue or delivery of a certificate or certificates representing shares of Liberty Capital Common Stock on conversion of shares of Series B Liberty Capital Common Stock pursuant to this paragraph (b)(i)(A). The Corporation will not, however, be required to pay any tax that may be payable in respect of any issue or delivery of a certificate or certificates representing any shares of Liberty Capital Common Stock in a name other than that in which the shares of Series B Liberty Capital Common Stock so converted were registered and no such issue or delivery will be made unless and until the person requesting the same has paid to the Corporation or its transfer agent the amount of any such tax or has established to the satisfaction of the Corporation or its transfer agent that such tax has been paid.

Liberty Capital Common Stock will be convertible at the option of the Corporation, in whole or in part, in accordance with the other provisions of this Section A.2.

6

Conversion of Series B Liberty Starz Common Stock into Series A Liberty Starz Common Stock; Other Each share of Series B Liberty Starz Common Stock will be convertible at any time, at the option of the holder thereof, into one fully paid and non-assessable share of Series A Liberty Starz Common Stock. Any such conversion may be effected by any holder of Series B Liberty Starz Common Stock by surrendering such holder's certificate or certificates representing the Series B Liberty Starz Common Stock to be converted, duly endorsed, at the principal office of the Corporation or any transfer agent for the Series B Liberty Starz Common Stock, together with a written notice to the Corporation at such office that such holder elects to convert all or a specified whole number of shares of Series B Liberty Starz Common Stock represented by such certificate or certificates and stating the name or names in which such holder desires the certificate or certificates representing shares of Series A Liberty Starz Common Stock to be issued and, if less than all of the shares of Series B Liberty Starz Common Stock represented by one certificate are to be converted, the name or names in which such holder desires the certificate or certificates representing the unconverted shares of Series B Liberty Starz Common Stock to be issued. Any certificate representing shares surrendered for conversion in accordance with this paragraph will, if so required by the Corporation or its transfer agent, be accompanied by instruments of transfer, in form satisfactory to the Corporation or its transfer agent, duly executed by the holder of such shares or the duly authorized representative of such holder, and will, if required by the next succeeding paragraph, be accompanied by payment, or evidence of payment, of applicable issue or transfer taxes. Promptly thereafter, the Corporation will issue and deliver to such holder or such holder's nominee or nominees, a certificate or certificates representing the number of shares of Series A Liberty Starz Common Stock to which such holder will be entitled as herein provided. If less than all of the shares of Series B Liberty Starz Common Stock represented by any one certificate are to be converted, the Corporation will issue and deliver to such holder or such holder's nominee or nominees a new certificate representing the shares of Series B Liberty Starz Common Stock not converted. Such conversion will be deemed to have been made at the close of business on the date of receipt by the Corporation or any such transfer agent of the certificate or certificates, notice and, if required, instruments of transfer and payment or evidence of payment of taxes referred to above, and the Person or Persons entitled to receive the Series A Liberty Starz Common Stock issuable on such conversion will be treated for all purposes as the record holder or holders of such Series A Liberty Starz Common Stock on that date. A number of shares of Series A Liberty Starz Common Stock equal to the number of shares of Series B Liberty Starz Common Stock outstanding from time to time will be set aside and reserved for issuance upon conversion of shares of Series B Liberty Starz Common Stock as provided herein. Shares of Series A Liberty Starz Common Stock and shares of Series C Liberty Starz Common Stock will not be convertible at the option of the holder into shares of any other series of Common Stock.

The Corporation will pay any and all documentary, stamp or similar issue or transfer taxes that may be payable in respect of the issue or delivery of a certificate or certificates representing shares of Liberty Starz Common Stock on conversion of shares of Series B Liberty Starz Common Stock pursuant to this paragraph (b) (i)(B). The Corporation will not, however, be required to pay any tax that may be payable in respect of any issue or delivery of a certificate or certificates representing any shares of Liberty Starz Common Stock in a name other than that

7

in which the shares of Series B Liberty Starz Common Stock so converted were registered and no such issue or delivery will be made unless and until the person requesting the same has paid to the Corporation or its transfer agent that such tax has been paid.

Liberty Starz Common Stock will be convertible at the option of the Corporation, in whole or in part, in accordance with the other provisions of this

#### (ii) <u>Conversion of Liberty Starz Common Stock into Liberty Capital Common Stock at the Option of the Corporation</u>

- (A) At the option of the Corporation, exercisable at any time by resolution of its Board of Directors: (I) each share of Series A Liberty Starz Common Stock will be converted into a number (or fraction) of fully paid and non-assessable shares of Series A Liberty Capital Common Stock equal to the Starz/Capital Group Optional Conversion Ratio, (II) each share of Series B Liberty Starz Common Stock will be converted into a number (or fraction) of fully paid and non-assessable shares of Series B Liberty Capital Common Stock equal to the Starz/Capital Group Optional Conversion Ratio, and (III) each share of Series C Liberty Starz Common Stock will be converted into a number (or fraction) of fully paid and non-assessable shares of Series C Liberty Capital Common Stock equal to the Starz/Capital Group Optional Conversion Ratio.
- (B) For purposes of this paragraph (b)(ii), the "Starz/Capital Group Optional Conversion Ratio" means the amount (calculated to the nearest five decimal places) obtained by dividing (I) the Average Market Value of the Liberty Starz Reference Share over the 20-Trading Day period ending on the Trading Day preceding the Determination Date, by (II) the Average Market Value of the Liberty Capital Reference Share over the 20-Trading Day period ending on the Trading Day preceding the Determination Date.
- (C) If the Corporation determines to convert shares of Liberty Starz Common Stock into Liberty Capital Common Stock pursuant to this paragraph (b)(ii), such conversion will occur on a Starz Group Conversion Date on or prior to the 45th day following the Determination Date and will otherwise be effected in accordance with the provisions of paragraph (f)(iv) of this Section A.2. If the Corporation determines not to undertake such conversion following the determination of the Starz/Capital Group Optional Conversion Ratio, the Corporation may at any time thereafter establish a new Determination Date, in which event the Starz/Capital Group Optional Conversion Ratio will be recalculated as of such new Determination Date and, if the Corporation determines to convert shares of Liberty Starz Common Stock into shares of Liberty Capital Common Stock, a new Starz Group Conversion Date will be established, in each case, in accordance with this paragraph (b)(ii).
- (D) The Corporation will not convert shares of a series of Liberty Starz Common Stock into shares of Liberty Capital Common Stock pursuant to this paragraph (b)(ii) without converting all outstanding shares of each series of Liberty Starz Common Stock into shares of Liberty Capital Common Stock, in each case, in accordance with this paragraph (b)(ii).

9

#### (iii) Conversion of Liberty Capital Common Stock into Liberty Starz Common Stock at the Option of the Corporation

- (A) At the option of the Corporation, exercisable at any time by resolution of its Board of Directors: (I) each share of Series A Liberty Capital Common Stock will be converted into a number (or fraction) of fully paid and non-assessable shares of Series A Liberty Starz Common Stock equal to the Capital/Starz Group Optional Conversion Ratio, (II) each share of Series B Liberty Capital Common Stock will be converted into a number (or fraction) of fully paid and non-assessable shares of Series B Liberty Starz Common Stock equal to the Capital/Starz Group Optional Conversion Ratio, and (III) each share of Series C Liberty Capital Common Stock will be converted into a number (or fraction) of fully paid and non-assessable shares of Series C Liberty Starz Common Stock equal to the Capital/Starz Group Optional Conversion Ratio.
- (B) For purposes of this paragraph (b)(iii), the "Capital/Starz Group Optional Conversion Ratio" means the amount (calculated to the nearest five decimal places) obtained by dividing (I) the Average Market Value of the Liberty Capital Reference Share over the 20-Trading Day period ending on the Trading Day preceding the Determination Date, by (II) the Average Market Value of the Liberty Starz Reference Share over the 20-Trading Day period ending on the Trading Day preceding the Determination Date.
- (C) If the Corporation determines to convert shares of Liberty Capital Common Stock into Liberty Starz Common Stock pursuant to this paragraph (b)(iii), such conversion will occur on a Capital Group Conversion Date on or prior to the 45th day following the Determination Date and will otherwise be effected in accordance with the provisions of paragraph (e)(iv) of this Section A.2. If the Corporation determines not to undertake such conversion following the determination of the Capital/Starz Group Optional Conversion Ratio, the Corporation may at any time thereafter establish a new Determination Date, in which event the Capital/Starz Group Optional Conversion Ratio will be recalculated as of such new Determination Date and, if the Corporation determines to convert shares of Liberty Capital Common Stock into shares of Liberty Starz Common Stock, a new Capital Group Conversion Date will be established, in each case, in accordance with this paragraph (b)(iii).
- (D) The Corporation will not convert shares of a series of Liberty Capital Common Stock into shares of Liberty Starz Common Stock pursuant to this paragraph (b)(iii) without converting all outstanding shares of each series of Liberty Capital Common Stock into shares of Liberty Starz Common Stock, in each case, in accordance with this paragraph (b)(iii).

## (c) <u>Dividends Generally</u>.

(i) <u>Dividends on Liberty Capital Common Stock.</u> Subject to the applicable terms of any Preferred Stock Designation, dividends on the Liberty Capital Common Stock may be declared and paid only out of the lesser of (A) assets of the Corporation legally available therefor and (B) the Capital Group Available Dividend Amount. Whenever a dividend, other

10

than a dividend that consists of a Share Distribution, is paid to the holders of one or more series of Liberty Capital Common Stock, the Corporation will also pay to the holders of each other series of Liberty Capital Common Stock a dividend per share equal to the dividend per share paid to the holders of such first one or more series of Liberty Capital Common Stock, such that the dividend paid on each share of Liberty Capital Common Stock, regardless of series, is the same. Whenever a dividend that consists of a Share Distribution is paid to the holders of one or more series of Liberty Capital Common Stock, the Corporation will also pay a dividend that consists of a Share Distribution to the holders of each other series of Liberty Capital Common Stock as provided in paragraph (d)(i) of this Section A.2.

If the Capital Group Outstanding Interest Fraction is less than one (1) on the record date for any dividend, including a dividend that consists of a Share Distribution, with respect to the Liberty Capital Common Stock, then concurrently with the payment of any dividend on the outstanding shares of Liberty Capital Common Stock:

(A) if such dividend consists of cash, securities (other than shares of Liberty Capital Common Stock or Liberty Starz Common Stock) or other assets, at the election of the Board of Directors, the Corporation will (I) attribute (a "Capital Group Inter-Group Dividend") to the Starz Group an aggregate amount of cash, securities or other assets, or a combination thereof (the "Capital Group Inter-Group Dividend Amount"), with a Fair Value equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest as of the record date for such dividend, by (y) the per share Fair Value of such dividend payable to the holders of outstanding shares of Liberty Capital Common Stock, as determined by the Board of Directors or (II) increase the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by dividing (x) the Capital Group Inter-Group Dividend Amount, by (y) the Fair Value of the Liberty Capital Reference Share as of the "ex" date or any similar date for such dividend;

(B) if such dividend consists of shares of Liberty Capital Common Stock, the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest will be increased by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest as of the record date for such dividend, by (y) the Capital Group Share Distribution Ratio applicable to such dividend; or

(C) if such dividend consists of shares of Liberty Starz Common Stock, subject to paragraph (d)(i)(B), the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest will be decreased by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by adding (I) the number of shares of Liberty Starz Common Stock distributed to holders of Liberty Capital Common Stock, plus (II) the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Starz Group with Respect to the Capital

П

Group Inter-Group Interest as of the record date for such dividend, by (y) the Starz Group Share Distribution Ratio applicable to such dividend.

In the case of a dividend paid pursuant to clause (D) of paragraph (e)(ii) of this Section A.2. in connection with a Capital Group Disposition, the Capital Group Inter-Group Dividend Amount may be increased, at the election of the Board of Directors, by the aggregate amount of the dividend that would have been payable with respect to the shares of Liberty Capital Common Stock converted into Liberty Starz Common Stock in connection with such Capital Group Disposition if such shares were not so converted and received the same dividend per share as the other shares of Liberty Capital Common Stock received in connection with such Capital Group Disposition.

A Capital Group Inter-Group Dividend may, at the discretion of the Board of Directors, be reflected by an allocation or by a direct transfer of cash, securities or other assets, or a combination thereof, and may be payable in kind or otherwise.

(ii) <u>Dividends on Liberty Starz Common Stock.</u> Subject to the applicable terms of any Preferred Stock Designation, dividends on the Liberty Starz Common Stock may be declared and paid only out of the lesser of (A) assets of the Corporation legally available therefor and (B) the Starz Group Available Dividend Amount. Whenever a dividend, other than a dividend that consists of a Share Distribution, is paid to the holders of one or more series of Liberty Starz Common Stock, the Corporation will also pay to the holders of each other series of Liberty Starz Common Stock a dividend per share equal to the dividend per share paid to the holders of such first one or more series of Liberty Starz Common Stock, such that the dividend paid on each share of Liberty Starz Common Stock, regardless of series, is the same. Whenever a dividend that consists of a Share Distribution is paid to the holders of one or more series of Liberty Starz Common Stock, the Corporation will also pay a dividend that consists of a Share Distribution to the holders of each other series of Liberty Starz Common Stock as provided in paragraph (d)(ii) of this Section A.2.

If the Starz Group Outstanding Interest Fraction is less than one (1) on the record date for any dividend, including a dividend that consists of a Share Distribution, with respect to the Liberty Starz Common Stock, then concurrently with the payment of any dividend on the outstanding shares of Liberty Starz Common Stock:

(A) if such dividend consists of cash, securities (other than shares of Liberty Starz Common Stock or Liberty Capital Common Stock) or other assets, at the election of the Board of Directors, the Corporation will (I) attribute (a "Starz Group Inter-Group Dividend") to the Capital Group an aggregate amount of cash, securities or other assets, or a combination thereof (the "Starz Group Inter-Group Dividend Amount"), with a Fair Value equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest as of the record date for such dividend, by (y) the per share Fair Value of such dividend payable to the holders of outstanding shares of Liberty Starz Common Stock, as determined by the Board of Directors or (II) increase the Number of Shares Issuable to

12

the Capital Group with Respect to the Starz Group Inter-Group Interest by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by dividing (x) the Starz Group Inter-Group Dividend Amount, by (y) the Fair Value of the Liberty Starz Reference Share as of the "ex" date or any similar date for such dividend;

(B) if such dividend consists of shares of Liberty Starz Common Stock, the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Inter-Group Interest will be increased by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest as of the record date for such dividend, by (y) the Starz Group Share Distribution Ratio applicable to such dividend; or

(C) if such dividend consists of shares of Liberty Capital Common Stock, subject to paragraph (d)(ii)(B), the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest will be decreased by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by adding (I) the number of shares of Liberty Capital Common Stock distributed to holders of Liberty Starz Common Stock, plus (II) the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest as of the record date for such dividend, by (y) the Capital Group Share Distribution Ratio applicable to such dividend.

In the case of a dividend paid pursuant to clause (D) of paragraph (f)(ii) of this Section A.2. in connection with a Starz Group Disposition, the Starz Group Inter-Group Dividend Amount may be increased, at the election of the Board of Directors, by the aggregate amount of the dividend that would have been payable with respect to the shares of Liberty Starz Common Stock converted into Liberty Capital Common Stock in connection with such Starz Group Disposition if such shares were not so converted and received the same dividend per share as the other shares of Liberty Starz Common Stock received in connection with such Starz Group Disposition.

A Starz Group Inter-Group Dividend may, at the discretion of the Board of Directors, be reflected by an allocation or by a direct transfer of cash, securities or other assets, or a combination thereof, and may be payable in kind or otherwise.

(iii) <u>Discrimination Between or Among Series of Common Stock.</u> Subject to the provisions of paragraphs (c) and (d) of this Section A.2., the Board of Directors will have the authority and discretion to declare and pay (or to refrain from declaring and paying) dividends, including, without limitation, dividends consisting of Share Distributions, on outstanding shares of Liberty Capital Common Stock or Liberty Starz Common Stock, or all such series, and in equal or unequal amounts, or only on the Liberty Capital Common Stock or the Liberty Starz Common Stock (subject to applicable law), notwithstanding the relationship between or among the Capital Group Available Dividend Amount, or the respective amounts of prior dividends declared on, or the liquidation rights of, the Liberty Capital Common Stock or the Liberty Starz Common Stock, or any other factor.

13

### (d) Share Distributions.

If at any time a Share Distribution is to be made with respect to the Series A Liberty Capital Common Stock, Series B Liberty Capital Common Stock or Series C Liberty Capital Common Stock, then, in addition to the applicable requirements of paragraph (c)(i) of this Section A.2., such Share Distribution may be declared and paid only as follows:

- (A) a Share Distribution consisting, at the election of the Board of Directors, of: (I) shares of Series A Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Capital Common Stock, Series B Liberty Capital Common Stock, Series B Liberty Capital Common Stock, Series B Liberty Capital Common Stock, on an equal per share basis; or (II) shares of Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Capital Common Stock) may be declared and paid to holders of Series A Liberty Capital Common Stock, or an equal per share basis; or (III) shares of Series A Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Capital Common Stock) may be declared and paid to holders of Series A Liberty Capital Common Stock, shares of Series B Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Capital Common Stock and shares of Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Capital Common Stock and shares of Series C Liberty Capital Common Stock, in each case, on an equal per share basis;
- (B) a Share Distribution consisting, at the election of the Board of Directors, of: (I) shares of Series A Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Starz Common Stock) may be declared and paid to holders of Series A Liberty Capital Common Stock, Series B Liberty Capital Common Stock and Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Starz Common Stock) may be declared and paid to holders of Series A Liberty Capital Common Stock, Series B Liberty Capital Common Stock and Series C Liberty Capital Common Stock, on an equal per share basis; or (III) shares of Series A Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Starz Common Stock) may be declared and paid to holders of Series B Liberty Capital Common Stock, where Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock (or Convertible Securities Convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Starz Common Stock) may be declared and paid to holders of Series C Liberty Starz Common Stock) may be declared and paid to holders of Series C Liberty Starz Common Stock) may be declared and paid to holders of Series C Liberty Starz Common Stock) may be declared and paid to holders of Series C Liberty Starz Common Stock) may be dec

14

Liberty Capital Common Stock, in each case, on an equal per share basis; provided, however, that no such Share Distribution will be declared and paid if the amount obtained by adding (x) the aggregate number of shares of Liberty Starz Common Stock to be so distributed pursuant to this paragraph (d)(i)(B) (including the number of such shares that would be issuable upon conversion, exercise or exchange of any Convertible Securities to be so distributed pursuant to such Share Distribution), plus (y) the number of shares of Liberty Starz Common Stock that are subject to issuance upon conversion, exercise or exchange of any Convertible Securities then outstanding that are attributed to the Capital Group Outstanding Interest Fraction is less than one (1) on the record date for the Share Distribution, the number of shares of Liberty Starz Common Stock equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (I) the Number of Shares Issuable to the Starz Group Interest as of the record date for such Share Distribution, by (II) the Starz Group Share Distribution Ratio, is greater than the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest; or

a Share Distribution consisting of any class or series of securities of the Corporation or any other Person, other than Liberty Capital Common Stock or Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Capital Common Stock or Liberty Starz Common Stock), may be declared and paid, at the election of the Board of Directors, either on the basis of a distribution of (x) identical securities, on an equal per share basis, to holders of each series of Liberty Capital Common Stock, (y) separate classes or series of securities, on an equal per share basis, to the holders of each series of Liberty Capital Common Stock or (z) a separate class or series of securities to the holders of one or more series of Liberty Capital Common Stock and, on an equal per share basis, a different class or series of securities to the holders of all other series of Liberty Capital Common Stock; provided, that in the case of clauses (y) and (z), (1) such separate classes or series of securities (and, if the distribution consists of Convertible Securities, the securities into which such Convertible Securities are convertible or for which they are exercisable or exchangeable) do not differ in any respect other than their relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), with holders of shares of Series B Liberty Capital Common Stock receiving the class or series of securities having (or convertible into or exercisable or exchangeable for securities having) the highest relative voting rights and the holders of shares of each other series of Liberty Capital Common Stock receiving securities of a class or series having (or convertible into or exercisable or exchangeable for securities having) lesser relative voting rights, in each case, without regard to whether such rights differ to a greater or lesser extent than the corresponding differences in voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) among the Series A Liberty Capital Common Stock, the Series B Liberty Capital Common Stock and the Series C Liberty Capital Common Stock, and (2) in the event the securities to be received by the holders of shares of Liberty Capital Common Stock other than the Series B Liberty Capital Common Stock consist of different classes or series of securities, with each such class or series of securities (or the securities into which such class or series is convertible or for which such class or series is exercisable or exchangeable) differing only with respect to the relative voting rights of such class or series (and any related differences in

15

designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), then such classes or series of securities will be distributed to the holders of each series of Liberty Capital Common Stock (other than the Series B Liberty Capital Common Stock) (i) as the Board of Directors determines or (ii) such that the relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) of the class or series of securities (or the securities into which such class or series is exercisable or exchangeable) to be received by the holders of each series of Liberty Capital Common Stock (other than the Series B Liberty Capital Common Stock) corresponds to the extent practicable to the relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) of such series of Liberty Capital Common Stock, as compared to the other series of Liberty Capital Common Stock (other than the Series B Liberty Capital Common Stock).

- (ii) <u>Distributions on Series A Liberty Starz Common Stock, Series B Liberty Starz Common Stock and Series C Liberty Starz Common Stock If at any time a Share Distribution is to be made with respect to the Series A Liberty Starz Common Stock, Series B Liberty Starz Common Stock or Series C Liberty Starz Common Stock, then, in addition to the applicable requirements of paragraph (c)(ii) of this Section A.2., such Share Distribution may be declared and paid only as follows:</u>
- (A) a Share Distribution consisting, at the election of the Board of Directors, of: (I) shares of Series A Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Starz Common Stock) may be declared and paid to holders of Series A Liberty Starz Common Stock, Series B Liberty Starz Common Stock and Series C Liberty Starz Common Stock, on an equal per share basis; or (II) shares of Series C Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Starz Common Stock) may be declared and paid to holders of Series A Liberty Starz Common Stock, shares of Series A Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Starz Common Stock) may be declared and paid to holders of Series A Liberty Starz Common Stock, shares of Series B Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock and shares of Series C Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock) may be declared and paid to holders of Series C Liberty Starz Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Starz Common Stock, in each case, on an equal per share basis;

(B) a Share Distribution consisting, at the election of the Board of Directors, of: (I) shares of Series A Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Capital Common Stock) may be declared and paid to holders of Series A Liberty Starz Common Stock, Series B Liberty Starz Common Stock and Series C Liberty Starz Common Stock, on an equal per share

16

basis; or (II) shares of Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Capital Common Stock) may be declared and paid to holders of Series A Liberty Starz Common Stock, on an equal per share basis; or (III) shares of Series A Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series A Liberty Capital Common Stock) may be declared and paid to holders of Series A Liberty Capital Common Stock) may be declared and paid to holders of Series B Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock and shares of Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series B Liberty Starz Common Stock and shares of Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Series C Liberty Capital Common Stock) may be declared and paid to holders of Series C Liberty Starz Common Stock, in each case, on an equal per share basis; provided, however, that no such Share Distribution will be declared and paid if the amount obtained by adding (x) the aggregate number of shares of Liberty Capital Common Stock to be so distributed pursuant to this paragraph (d)(ii)(B) (including the number of such shares that would be issuable upon conversion, exercise or exchange of any Convertible Securities then outstanding that are attributed to the Starz Group, plus (z) if the Starz Group Outstanding Interest Fraction is less than one (1) on the record date for the Share Distribution, the number of shares of Liberty Capital Common Stock equal to the amount (rounded, if necessary, to the nearest whole number) obtained

(C) a Share Distribution consisting of any class or series of securities of the Corporation or any other Person, other than Liberty Starz Common Stock or Liberty Capital Common Stock (or Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Starz Common Stock or Liberty Capital Common Stock), may be declared and paid, at the election of the Board of Directors, either on the basis of a distribution of (x) identical securities, on an equal per share basis, to holders of each series of Liberty Starz Common Stock, (y) separate classes or series of securities, on an equal per share basis, to the holders of each series of Liberty Starz Common Stock or (z) a separate class or series of securities to the holders of one or more series of Liberty Starz Common Stock and, on an equal per share basis, a different class or series of securities to the holders of all other series of Liberty Starz Common Stock; provided, that in the case of clauses (y) and (z), (1) such separate classes or series of securities (and, if the distribution consists of Convertible Securities, the securities into which such Convertible Securities are convertible or or which they are exercisable or exchangeable) do not differ in any respect other than their relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), with holders of shares of Series B Liberty Starz Common Stock receiving the class or series of securities having (or convertible into or

17

exercisable or exchangeable for securities having) the highest relative voting rights and the holders of shares of each other series of Liberty Starz Common Stock receiving securities of a class or series having (or convertible into or exercisable or exchangeable for securities having) lesser relative voting rights, in each case, without regard to whether such rights differ to a greater or lesser extent than the corresponding differences in voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) among the Series A Liberty Starz Common Stock, the Series B Liberty Starz Common Stock, and (2) in the event the securities to be received by the holders of shares of Liberty Starz Common Stock other than the Series B Liberty Starz Common Stock consist of different classes or series of securities, with each such class or series of securities (or the securities into which such class or series is convertible or for which such class or series is exercisable or exchangeable) differing only with respect to the relative voting rights of such class or series (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), then such classes or series of securities will be distributed to the holders of each series of Liberty Starz Common Stock (other than the Series B Liberty Starz Common Stock) (i) as the Board of Directors determines or (ii) such that the relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) of the class or series of securities into which such class or series is convertible or for which such class or series is exercisable or exchangeable) to be received by the holders of each series of Liberty Starz Common Stock (other than the Series B Liberty Starz Common Stock). Corresponds to the extent practicable to the

#### (e) Redemption and Other Provisions Relating to the Liberty Capital Common Stock.

(i) Redemption for Securities of one or more Capital Group Subsidiaries. At any time at which a Subsidiary of the Corporation holds, directly or indirectly, assets and liabilities attributed to the Capital Group, the Corporation may, at its option and subject to assets of the Corporation being legally available therefor, but subject (in addition to any other approval of the Corporation's stockholders (or any series thereof) required under the DGCL in respect of such redemption, if any) to the Corporation having received the Capital Group Redemption Stockholder Approval (and, to the extent applicable, the Starz Group Redemption Stockholder Approval), redeem outstanding shares of Liberty Capital Common Stock to be redeemed, the "Capital Group Redemption Shares") for securities of such Subsidiary (a "Distributed Capital Group Subsidiary"), as provided herein. The number of Capital Group Redemption Shares will be determined, by the Board of Directors, by multiplying (A) the number of outstanding shares of Liberty Capital Common Stock as of the Capital Group Redemption Selection Date, by (B) the percentage of the Fair Value of the Capital Group that is represented by the Fair Value of the Corporation's equity interest in the Distributed Capital Group Subsidiary which is attributable to the Capital Group, in each case, as determined by the Board of Directors as of a date selected by the Board of Directors, as such percentage may

18

be adjusted by the Board of Directors in its discretion to take into account such things as it deems relevant. The aggregate number of securities of the Distributed Capital Group Subsidiary to be delivered (the "Capital Group Distribution Subsidiary Securities") in redemption of the Capital Group Redemption Shares will be equal to: (A) if the Board of Directors makes a Capital Group Inter-Group Redemption Election as described below, the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the product of (I) the number of securities of the Distributed Capital Group Subsidiary owned by the Corporation and (II) the percentage of the Fair Value of the Corporation's equity interest in the Distributed Capital Group Subsidiary that is represented by the Fair Value of the Corporation's equity interest in the Distributed Capital Group Subsidiary which is attributable to the Capital Group (such product, the "Distributable Capital Group Subsidiary Securities"), by (y) the Capital Group Outstanding Interest Fraction, in each case, as of the Capital Group Redemption Selection Date, or (B) if the Board of Directors does not make a Capital Group Inter-Group Redemption Election, all of the Distributable Capital Group Subsidiary Securities, in each case, subject to adjustment as provided below. The number of securities of the Distributed Capital Group Subsidiary to be delivered in redemption of each Capital Group Redemption Share will be equal to the amount (rounded, if necessary, to the nearest five decimal places) obtained by dividing (x) the number of Capital Group Distribution Subsidiary Securities, by (y) the number of Capital Group Redemption Shares.

distribution of the Capital Group Distribution Subsidiary Securities in redemption of Capital Group Redemption Shares, the Corporation will attribute to the Starz Group an aggregate number of Distributable Capital Group Subsidiary Securities (the "Capital Group Inter-Group Interest Subsidiary Securities") equal to the difference between the total number of Distributable Capital Group Subsidiary Securities and the number of Capital Group Distribution Subsidiary Securities, subject to adjustment as provided below. If a Capital Group Inter-Group Redemption Election is made, then: (I) the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest will be decreased as described in subparagraph (ii)(D) of the definition of "Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest" in paragraph (i) of this Section A.2.; (II) the attribution of Capital Group Inter-Group Interest Subsidiary Securities to be made to the Starz Group may, at the discretion of the Board of Directors, be reflected by an allocation or by a direct transfer of Capital Group Inter-Group Interest Subsidiary Securities to the Starz Group; and (III) the Board of Directors may determine that the Capital Group Inter-Group Interest Subsidiary Securities so allocated or transferred to the Starz Group will be distributed to holders of shares of Liberty Starz Common Stock as a Share Distribution pursuant to paragraph (d)(ii)(C) of this Section A.2.

If at the time of a redemption of Liberty Capital Common Stock pursuant to this paragraph (e)(i), there are outstanding any Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Capital Common Stock that would become convertible into or exercisable or exchangeable for Distributable Capital Group Subsidiary Securities as a

19

result of such redemption, and the obligation to deliver securities of such Distributed Capital Group Subsidiary upon exercise, exchange or conversion of such Convertible Securities is not assumed or otherwise provided for by the Distributed Capital Group Subsidiary, then the Board of Directors may make such adjustments as it determines to be appropriate to the number of Capital Group Redemption Shares, the number of Capital Group Distribution Subsidiary Securities and the number of Capital Group Inter-Group Interest Subsidiary Securities (and any related adjustment to the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest) to take into account the securities of the Distributed Capital Group Subsidiary into which such Convertible Securities are convertible or for which such Convertible Securities are exercisable or exchangeable.

In the event that not all outstanding shares of Liberty Capital Common Stock are to be redeemed in accordance with this paragraph (e)(i) for Capital Group Distribution Subsidiary Securities, then (1) the number of shares of each series of Liberty Capital Common Stock to be redeemed in accordance with this paragraph (e)(i) will be determined by multiplying the aggregate number of Capital Group Redemption Shares by a fraction, the numerator of which is the aggregate number of shares of such series and the denominator of which is the aggregate number of shares of all series of Liberty Capital Common Stock, in each case, outstanding as of the Capital Group Redemption Selection Date, and (2) the outstanding shares of each series of Liberty Capital Common Stock to be redeemed in accordance with this paragraph (e)(i) will be redeemed by the Corporation pro rata among the holders of each series of Liberty Capital Common Stock or by such other method as may be determined by the Board of Directors to be equitable.

To the extent that a Distributed Capital Group Subsidiary to be distributed pursuant to this paragraph (e)(i) also holds, directly or indirectly, assets and liabilities attributed to the Starz Group, then (x) such Distributed Capital Group Subsidiary will also be deemed a Distributed Starz Group Subsidiary for purposes of paragraph (f)(i) and (y) in connection with the redemption of Capital Group Redemption Shares pursuant to this paragraph (e)(i) the Corporation will also redeem shares of Liberty Starz Common Stock pursuant to the provisions of paragraph (f)(i), subject to the Corporation obtaining the Capital Group Redemption Stockholder Approval and the Starz Group Redemption Stockholder Approval. In connection with any such redemption of Liberty Capital Common Stock and Liberty Starz Common Stock, the Board of Directors will effect such redemption in accordance with the terms of paragraphs (e)(i) and (f)(i), as determined by the Board of Directors in good faith, with such changes and adjustments as the Board of Directors determines are reasonably necessary in order to effect such redemption in exchange for securities of a single Subsidiary holding the assets and liabilities of more than one Group. In effecting such redemption, the Board of Directors may determine to redeem the Capital Group Redemption Shares and the Starz Group Redemption Shares in exchange for one or more classes or series of securities of such Subsidiary, including, without limitation, for separate classes or series of securities of such Subsidiary, including, without limitation, for separate classes or series of securities of such Subsidiary Securities intended to track the performance of the former assets and liabilities attributed to the Capital Group held by such Subsidiary and (II) with holders of Starz Group Redemption Shares to receive Starz Group Distribution Subsidiary Securities intended to track the performance of the

20

former assets and liabilities attributed to the Starz Group held by such Subsidiary, subject, in each case, to the applicable limitations on the class and series of securities of the Distributed Capital Group Subsidiary set forth in the last paragraph of paragraphs (e)(i) and (f)(i).

Any redemption pursuant to this paragraph (e)(i) will occur on a Capital Group Redemption Date set forth in a notice to holders of Liberty Capital Common Stock (and Convertible Securities convertible into or exercisable or exchangeable for shares of any series of Liberty Capital Common Stock (unless provision for notice is otherwise made pursuant to the terms of such Convertible Securities)) pursuant to paragraph (e)(iv)(C).

In effecting a redemption of Liberty Capital Common Stock pursuant to this paragraph (e)(i), the Board of Directors may determine either to (x) redeem shares of each series of Liberty Capital Common Stock in exchange for a single class or series of securities of the Distributed Capital Group Subsidiary without distinction among series of Liberty Capital Common Stock, on an equal per share basis, (y) redeem shares of each series of Liberty Capital Common Stock in exchange for separate classes or series of securities of the Distributed Capital Group Subsidiary, on an equal per share basis, or (z) redeem shares of one or more series of Liberty Capital Common Stock in exchange for a separate class or series of securities of the Distributed Capital Group Subsidiary and, on an equal per share basis, redeem shares of all other series of Liberty Capital Common Stock in exchange for a different class or series of securities of the Distributed Capital Group Subsidiary; provided, that, in the case of clauses (y) and (z), (1) such separate classes or series do not differ in any respect other than their relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), with holders of shares of Series B Liberty Capital Common Stock receiving securities of a class or series having the highest relative voting rights and the holders of shares of each other series of Liberty Capital Common Stock receiving securities of a class or series having lesser relative voting rights, in each case, without regard to whether such rights differ to a greater or lesser extent than the corresponding differences in voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) among the Series A Liberty Capital Common Stock, the Series B Liberty Capital Common Stock and the Series C Liberty Capital Common Stock, and (2) in the event the securities to be received by the holders of shares of Liberty Capital Common Stock other than the Series B Liberty Capital Common Stock in such redemption consist of different classes or series of securities, with each such class or series differing only with respect to the relative voting rights of such class or series (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), then such classes or series of securities will be distributed to the holders of each series of Liberty Capital Common Stock (other than the Series B Liberty Capital Common Stock) (i) as the Board of Directors determines or (ii) such that the relative voting rights of the class or series of securities to be received by the holders of each series of Liberty Capital Common Stock corresponds to the extent practicable to the relative voting rights (as compared to the other series of Liberty Capital Common Stock, other than the Series B Liberty Capital Common Stock) of such series of Liberty Capital Common Stock. If the Board of Directors has made a Capital Group Inter-Group Redemption Election, then the determination as to the classes or series of securities of the Distributed Capital Group Subsidiary comprising

- (ii) Mandatory Dividend, Redemption or Conversion in Case of Capital Group Disposition. In the event of a Capital Group Disposition (other than an Exempt Capital Group Disposition), the Corporation will, on or prior to the 120th Trading Day following the consummation of such Capital Group Disposition and in accordance with the applicable provisions of this Section A.2., take the actions referred to in one of clauses (A), (B), (C) or (D) below, as elected by the Board of Directors:
- (A) Subject to the first sentence of paragraph (c)(i) of this Section A.2. the Corporation may declare and pay a dividend payable in cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, to the holders of outstanding shares of Liberty Capital Common Stock, with an aggregate Fair Value (subject to adjustment as provided below) equal to the Capital Group Allocable Net Proceeds of such Capital Group Disposition as of the record date for determining the holders entitled to receive such dividend, as the same may be determined by the Board of Directors, with such dividend to be paid in accordance with the applicable provisions of paragraphs (c)(i) and (d)(i) of this Section A.2.; or
- (B) Provided that there are assets of the Corporation legally available therefor and the Capital Group Available Dividend Amount would have been sufficient to pay a dividend pursuant to clause (A) of this paragraph (e)(ii) in lieu of effecting the redemption provided for in this clause (B), then:
- (I) if such Capital Group Disposition involves all (not merely substantially all) of the assets of the Capital Group, the Corporation may redeem all outstanding shares of each series of Liberty Capital Common Stock for cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, with an aggregate Fair Value (subject to adjustment as provided below) equal to the Capital Group Allocable Net Proceeds of such Capital Group Disposition as of the Capital Group Redemption Date, as determined by the Board of Directors, such aggregate amount to be allocated among the shares of all series of Liberty Capital Common Stock outstanding as of the Capital Group Redemption Date on an equal per share basis (subject to the provisions of this paragraph (e)(ii)); or
- (II) if such Capital Group Disposition involves substantially all (but not all) of the assets of the Capital Group, the Corporation may apply an aggregate amount (subject to adjustment as provided below) of cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, with a Fair Value equal to the Capital Group Allocable Net Proceeds of such Capital Group Disposition as of the Capital Group Redemption Selection Date (the "Capital Group Redemption Amount") to the redemption of outstanding shares of each series of Liberty Capital Common Stock, such Capital Group Redemption Amount to be allocated (subject to the provisions of this paragraph (e)(ii)) to the redemption of shares of each series of Liberty Capital Common Stock in the ratio of (x) the number of shares of such series outstanding as of the Capital Group Redemption Selection Date to (y) the aggregate number of shares of all series of Liberty Capital Common Stock outstanding as of such date, and

22

the number of shares of each such series to be redeemed will equal the lesser of (1) the number of shares of such series outstanding as of the Capital Group Redemption Selection Date and (2) the whole number nearest the number obtained by dividing the aggregate amount so allocated to the redemption of such series by the Average Market Value of the Liberty Capital Reference Share over the period of 10 consecutive Trading Days beginning on the 2nd Trading Day following the consummation of such Capital Group Disposition; provided that, if following the foregoing allocation there remains any amount of the Capital Group Redemption Amount which is not being applied to the redemption of shares of a series of Liberty Capital Common Stock, then such excess amount will be allocated to the redemption of shares of each series of Liberty Capital Common Stock that, following the initial allocation referred to above, would have shares outstanding and not redeemed, with the number of outstanding and not redeemed shares to be redeemed from each such series to be calculated in accordance with clauses (1) and (2) of the immediately preceding sentence based upon such excess amount of the Capital Group Redemption Amount. The outstanding shares of a series of Liberty Capital Common Stock to be redeemed will be selected on a pro rata basis among the holders of such series or by such other method as the Board of Directors may determine to be equitable; or

- (C) The Corporation may convert each outstanding share of Series A Liberty Capital Common Stock into a number (or fraction) of fully paid and non-assessable shares of Series A Liberty Starz Common Stock, each outstanding share of Series B Liberty Capital Common Stock into a number (or fraction) of fully paid and non-assessable shares of Series B Liberty Starz Common Stock, and each outstanding share of Series C Liberty Capital Common Stock into a number (or fraction) of fully paid and non-assessable shares of Series C Liberty Starz Common Stock, in each case, equal to 110% of the average daily ratio (calculated to the nearest five decimal places) of (I) the Average Market Value of the Liberty Capital Reference Share over the period of 10 consecutive Trading Days beginning on the 2nd Trading Day following the consummation of such Capital Group Disposition, to (II) the Average Market Value of the Liberty Starz Reference Share over the same 10-Trading Day period; or
- (D) The Corporation may combine the conversion of a portion of the outstanding shares of Liberty Capital Common Stock into Liberty Starz Common Stock as contemplated by clause (C) of this paragraph (e)(ii) with the payment of a dividend on or the redemption of shares of Liberty Capital Common Stock as described below, subject to the limitations specified in clause (A) (in the case of a dividend) or clause (B) (in the case of a redemption) of this paragraph (e)(ii) (including the limitations specified in other paragraphs of this Certificate referred to therein). In the event the Board of Directors elects the option described in this clause (D), the portion of the outstanding shares of Liberty Capital Common Stock to be converted into fully paid and non-assessable shares of Liberty Starz Common Stock will be determined by the Board of Directors and will be so converted at the conversion rate determined in accordance with clause (C) above and the Corporation will either (x) pay a dividend to the holders of record of all of the remaining shares of Liberty Capital Common Stock outstanding, with such dividend to be paid in accordance with the applicable provisions of paragraphs (e)(i) and (d)(i) of this Section A.2., or (y) redeem all or a portion of such remaining shares of Liberty Capital Common Stock. The aggregate amount of such dividend, in the case of a dividend, or the portion of the Capital Group Allocable Net Proceeds to be applied to such

23

redemption, in the case of a redemption, will be equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (I) an amount equal to the Capital Group Allocable Net Proceeds of such Capital Group Disposition as of, in the case of a dividend, the record date for determining the holders of Liberty Capital Common Stock entitled to receive such dividend and, in the case of a redemption, the Capital Group Redemption Selection Date (in the case of a partial redemption) or the Capital Group Redemption Date (in the case of a full redemption), in each case, before giving effect to the conversion of shares of Liberty Capital Common Stock in connection with such Capital Group Disposition in accordance with this clause (D) and any related adjustment to the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest, by (II) one minus a fraction, the numerator of which will be the number of shares of Liberty Capital Common Stock to be converted into shares of Liberty Starz Common Stock in accordance with this clause (D) and the denominator of which will be the aggregate number of shares of Liberty Capital Common Stock outstanding as of the record date, Capital Group Redemption Selection Date or Capital Group Redemption Date used for purposes of clause (I) of this sentence. In the event of a redemption concurrently with or following any such partial conversion of shares of Liberty Capital Common Stock, if the Capital Group Disposition was of all (not merely substantially all) of the assets of the Capital Group, then all remaining outstanding shares of Liberty Capital Common Stock will be redeemed for cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, with an aggregate Fair Value equal to the portion of the Capital Group Allocable Net Proceeds to be applied to such redemption determined in accordance with this clause (D), such aggregate amount to be allocated among all such shares to be redeemed on an equal per share basis (subject to the provisions of this paragraph (e)(ii)). In the event of a redemption concurrently with or following any such partial conversion of shares of Liberty Capital Common Stock, if the Capital Group Disposition was of substantially all (but not all) of the assets of the Capital Group, then the number of shares of each series of Liberty Capital Common Stock to be redeemed will be determined in accordance with clause (B)(II) of this paragraph (e)(ii), substituting for the Capital Group Redemption Amount referred to therein the portion of the Capital Group Allocable Net Proceeds to be applied to such redemption as determined in accordance with this clause (D), and such shares will be redeemed for cash, securities (other than Common Stock) or other assets, or any combination thereof, with an aggregate Fair Value equal to such portion of the Capital Group Allocable Net Proceeds and allocated among all such shares to be redeemed on an equal per share basis (subject to the provisions of this paragraph (e)(ii)). The aggregate number of shares of Liberty Capital Common Stock to be converted in any partial conversion in accordance with this clause (D) will be allocated among the series of Liberty Capital Common Stock in the ratio of the number of shares of each such series outstanding to the aggregate number of shares of all series of Liberty Capital Common Stock outstanding as of the Capital Group Conversion Selection Date, and the shares of each such series to be converted will be selected on a pro rata basis or by such other method as the Board of Directors may determine to be equitable. In the case of a redemption, the

allocation of the cash, securities (other than shares of Common Stock) and/or other assets to be paid in redemption and, in the case of a partial redemption, the selection of shares to be redeemed will be made in the manner contemplated by clause (B) of this paragraph (e)(ii).

For purposes of this paragraph (e)(ii):

24

- (1) as of any date, "**substantially all of the assets of the Capital Group**" means a portion of such assets that represents at least 80% of the then-Fair Value of the assets of the Capital Group as of such date;
- (2) in the case of a Capital Group Disposition of assets in a series of related transactions, such Capital Group Disposition will not be deemed to have been consummated until the consummation of the last of such transactions:
- (3) if the Board of Directors seeks the approval of the holders of Liberty Capital Voting Securities entitled to vote thereon to qualify a Capital Group Disposition as an Exempt Capital Group Disposition and such approval is not obtained, the date on which such approval fails to be obtained will be treated as the date on which such Capital Group Disposition was consummated for purposes of making the determinations and taking the actions prescribed by this paragraph (e)(ii) and paragraph (e)(iv), and no subsequent vote may be taken to qualify such Capital Group Disposition as an Exempt Capital Group Disposition;
- (4) in the event of a redemption of a portion of the outstanding shares of Liberty Capital Common Stock pursuant to clause (B)(II) or (D) of this paragraph (e)(ii) at a time when the Capital Group Outstanding Interest Fraction is less than one, if the Board of Directors so elects (a "Capital Group Inter-Group Partial Redemption Election"), in its discretion, the Corporation will attribute to the Starz Group concurrently with such redemption an aggregate amount (the "Capital Group Inter-Group Redemption Amount") of cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, subject to adjustment as described below, with an aggregate Fair Value equal to the difference between (x) the Capital Group Net Proceeds and (y) the portion of the Capital Group Allocable Net Proceeds applied to such redemption as determined in accordance with clause (B)(II) or clause (D) of this paragraph (e)(ii). If the Board of Directors makes such election, the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest" in paragraph (i) of this Section A.2. The Capital Group Inter-Group Redemption Amount may, at the discretion of the Board of Directors, be reflected by an allocation to the Starz Group or by a direct transfer to the Starz Group of cash, securities and/or other assets;
- (5) if at the time of a Capital Group Disposition subject to this paragraph (e)(ii), there are outstanding any Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Capital Common Stock that would give the holders thereof the right to receive any consideration related to such Capital Group Disposition upon conversion, exercise or exchange or otherwise, or would adjust to give the holders equivalent economic rights, as a result of any dividend, redemption or other action taken by the Corporation with respect to the Liberty Capital Common Stock pursuant to this paragraph (e)(ii), then the Board of Directors may make such adjustments to (x) the amount of consideration to be issued or delivered as contemplated by this paragraph (e)(ii) as a dividend on or in redemption or conversion of shares of Liberty Capital Common Stock and/or, if applicable, (y) the Capital Group Inter-Group Redemption Amount and the Number of Shares Issuable to the Starz Group

25

with Respect to the Capital Group Inter-Group Interest as it deems appropriate to take into account the Liberty Capital Common Stock into which such Convertible Securities are convertible or for which such Convertible Securities are exercisable or exchangeable;

- the Corporation may pay the dividend or redemption price referred to in clause (A), (B) or (D) of this paragraph (e)(ii) payable to the holders of Liberty Capital Common Stock in cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, that the Board of Directors determines and which has an aggregate Fair Value of not less than the amount allocated to such dividend or redemption pursuant to the applicable of clauses (A), (B) or (D) of this paragraph (e)(ii), regardless of the form or nature of the proceeds received by the Corporation from the Capital Group Disposition; and
- if all or any portion of the redemption price referred to in clause (B) or (D) of this paragraph (e)(ii) payable to the holders of Liberty Capital Common Stock is paid in the form of securities of an issuer other than the Corporation, the Board of Directors may determine to pay the redemption price, so payable in securities, in the form of (x) identical securities, on an equal per share basis, to holders of each series of Liberty Capital Common Stock, (y) separate classes or series of securities, on an equal per share basis, to the holders of each series of Liberty Capital Common Stock or (z) a separate class or series of securities to the holders of one or more series of Liberty Capital Common Stock and, on an equal per share basis, a different class or series of securities to the holders of all other series of Liberty Capital Common Stock; provided, that, in the case of clauses (y) and (z), (1) such separate classes or series do not differ in any respect other than their relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), with holders of shares of Series B Liberty Capital Common Stock receiving securities of a class or series having the highest relative voting rights and the holders of shares of each other series of Liberty Capital Common Stock receiving securities of a class or series having lesser relative voting rights, in each case, without regard to whether such rights differ to a greater or lesser extent than the corresponding differences in voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) among the Series A Liberty Capital Common Stock, the Series B Liberty Capital Common Stock and the Series C Liberty Capital Common Stock and (2) in the event the securities to be received by the holders of shares of Liberty Capital Common Stock other than the Series B Liberty Capital Common Stock consist of different classes or series of securities, with each such class or series differing only with respect to the relative voting rights of such class or series (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), then such classes or series of securities will be distributed to the holders of each series of Liberty Capital Common Stock (other than the Series B Liberty Capital Common Stock) (i) as the Board of Directors determines or (ii) such that the relative voting rights of the class or series of securities to be received by the holders of each series of Liberty Capital Common Stock corresponds to the extent practicable to the relative voting rights (as compared to the other series of Liberty Capital Common Stock, other than the Series B Liberty Capital Common Stock) of such series of Liberty Capital Common Stock.

26

(iii) Certain Provisions Respecting Convertible Securities. Unless the provisions of any Convertible Securities that are or become convertible into or exercisable or exchangeable for shares of any series of Liberty Capital Common Stock provide specifically to the contrary, or the instrument, plan or agreement evidencing such Convertible Securities or pursuant to which the same were issued grants to the Board of Directors the discretion to approve or authorize any adjustment or adjustments to the conversion, exercise or exchange provisions of such Convertible Securities so as to obtain a result different from that which would otherwise occur pursuant to this paragraph (e)(iii), and the Board of Directors so approves or authorizes such adjustment or adjustments, after any Capital Group Conversion Date or Capital Group Redemption Date on which all outstanding shares of Liberty Capital Common Stock were converted or redeemed, any share of Liberty Capital Common Stock that is issued on conversion, exercise or exchange of any such Convertible Security will, immediately upon issuance and without any notice or any other action on the part of the Corporation or its Board of Directors or the holder of such share of Liberty Capital Common Stock, be redeemed in exchange for, to the extent assets of the Corporation are legally available therefor, the amount of \$.01 per share in cash.

#### (iv) <u>General</u>.

- (A) Not later than the 10th Trading Day following the consummation of a Capital Group Disposition referred to in paragraph (e)(ii) of this Section A.2., the Corporation will announce publicly by press release (x) the Capital Group Net Proceeds of such Capital Group Disposition, (y) whether the Capital Group Disposition qualifies as an Exempt Capital Group Disposition, and (z) if it does not so qualify at the time of such announcement (including in the event the Board of Directors had not sought stockholder approval to qualify such Capital Group Disposition as an Exempt Capital Group Disposition in connection with any required stockholder approval obtained by the Corporation, if applicable), whether the Board of Directors will seek the approval of the holders of Liberty Capital Voting Securities entitled to vote thereon to qualify such Capital Group Disposition as an Exempt Capital Group Disposition. Not later than the 30<sup>th</sup> Trading Day (and in the event a 10 Trading Day valuation period is required in connection with the action selected by the Board of Directors pursuant to clause (I) of this paragraph (e)(iv)(A), not earlier than the 11<sup>th</sup> Trading Day) following the later of (x) the consummation of such Capital Group Disposition and (y), if applicable, the date of the stockholder meeting at which a vote is taken to qualify such Capital Group Disposition as an Exempt Capital Group Disposition, the Corporation will announce publicly by press release (to the extent applicable):
- (I) which of the actions specified in clauses (A), (B), (C) or (D) of paragraph (e)(ii) of this Section A.2. the Corporation has irrevocably determined to take;
- (II) as applicable, the record date for determining holders entitled to receive any dividend to be paid pursuant to clause (A) or (D) of paragraph (e)(ii), the Capital Group Redemption Selection Date for the redemption of shares of Liberty Capital Common Stock pursuant to clause (B)(II) or (D) of paragraph (e)(ii) or the Capital Group Conversion Selection Date for the partial conversion of shares of Liberty Capital Common Stock

27

pursuant to clause (D) of paragraph (e)(ii), which record date, Capital Group Redemption Selection Date or Capital Group Conversion Selection Date will not be earlier than the 10th day following the date of such public announcement;

- (III) the anticipated dividend payment date, Capital Group Redemption Date and/or Capital Group Conversion Date, which in each case, will not be more than 85 Trading Days following such Capital Group Disposition; and
- (IV) unless the Board of Directors otherwise determines, that the Corporation will not be required to register a transfer of any shares of Liberty Capital Common Stock for a period of 10 Trading Days (or such shorter period as such announcement may specify) next preceding the specified Capital Group Redemption Selection Date or Capital Group Conversion Selection Date.

If the Corporation determines to undertake a redemption of shares of Liberty Capital Common Stock, in whole or in part, pursuant to clause (B) or (D) of paragraph (e)(ii) of this Section A.2., or a conversion of shares of Liberty Capital Common Stock, in whole or in part, pursuant to clause (C) or (D) of paragraph (e)(ii), the Corporation will announce such redemption or conversion (which, for the avoidance of doubt, may remain subject to the satisfaction or waiver of any applicable condition precedent at the time of such announcement) publicly by press release, not less than 10 days prior to the Capital Group Redemption Date or Capital Group Conversion Date, as applicable:

- (1) the Capital Group Redemption Date or Capital Group Conversion Date;
- (2) the number of shares of Liberty Capital Common Stock to be redeemed or converted or, if applicable, stating that all outstanding shares of Liberty Capital Common Stock will be redeemed or converted and the series of Liberty Starz Common Stock issuable to the holders of each series of Liberty Capital Common Stock upon any such conversion;
- (3) in the case of a redemption or a conversion, in each case, in whole or in part, of outstanding shares of Liberty Capital Common Stock, the kind and amount of per share consideration to be received with respect to each share of Liberty Capital Common Stock to be redeemed or converted and the Capital Group Outstanding Interest Fraction as of the date of such notice;
- (4) with respect to a partial redemption under clause (B)(II) or (D) of paragraph (e)(ii), if the Board of Directors has made a Capital Group Inter-Group Partial Redemption Election, the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest as of the Capital Group Redemption Selection Date;
- (5) with respect to a dividend under clause (A) or (D) of paragraph (e)(ii), the Number of Shares Issuable to the Starz Group with Respect to the Capital Group

28

Inter-Group Interest as of the record date for the dividend and the Capital Group Inter-Group Dividend Amount attributable to the Starz Group; and

- (6) instructions as to how shares of Liberty Capital Common Stock may be surrendered for redemption or conversion.
- (B) In the event of any conversion of shares of Liberty Capital Common Stock pursuant to paragraph (b)(iii) of this Section A.2., not less than 10 days prior to the Capital Group Conversion Date, the Corporation will announce publicly by press release:
- (1) that all outstanding shares of Liberty Capital Common Stock will be converted pursuant to paragraph (b)(iii) of this Section A.2. on the Capital Group Conversion Date;
  - (2) the Capital Group Conversion Date, which will not be more than 45 days following the Determination Date;
  - (3) a statement that all outstanding shares of Liberty Capital Common Stock will be converted;
- (4) the per share number and series of shares of Liberty Starz Common Stock to be received with respect to each share of each series of Liberty Capital Common Stock; and
  - (5) instructions as to how shares of Liberty Capital Common Stock may be surrendered for conversion.
- (C) If the Corporation determines to obtain the Capital Group Redemption Stockholder Approval and, subject to the receipt of such approval, to redeem shares of Liberty Capital Common Stock pursuant to paragraph (e)(i), the Corporation will announce publicly by press release:
  - (I) that the Corporation intends to redeem shares of Liberty Capital Common Stock for securities of a Distributed Capital Group

Subsidiary pursuant to paragraph (e)(i) of this Section A.2., subject to any applicable conditions, including the receipt of the Capital Group Redemption Stockholder Approval if such approval has not been obtained at the time of the press release;

(II) the number of shares of Liberty Capital Common Stock to be redeemed or, if applicable, stating that all outstanding shares of Liberty Capital Common Stock will be redeemed;

(III) the class or series of securities of the Distributed Capital Group Subsidiary to be received with respect to each share of each series of Liberty Capital Common Stock to be redeemed and the Capital Group Outstanding Interest Fraction as of the date of such notice, if any;

29

(IV) if applicable, the Capital Group Redemption Selection Date, which will not be earlier than the 10th day following the date of

the press release;

(V) the Capital Group Redemption Date, which will not be earlier than the 10th day following the date of the press release and will not be later than the 120th Trading Day following the date of the press release;

(VI) if the Board of Directors has made a Capital Group Inter-Group Redemption Election, the number of Capital Group Inter-Group Interest Subsidiary Securities attributable to the Starz Group, and the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest used in determining such number and attribution of Capital Group Interest Subsidiary Securities;

(VII) instructions as to how shares of Liberty Capital Common Stock may be surrendered for redemption; and

(VIII) if the Board of Directors so determines, that the Corporation will not be required to register a transfer of any shares of Liberty Capital Common Stock for a period of 10 Trading Days (or such shorter period as such announcement may specify) next preceding the specified Capital Group Redemption Selection Date.

If, at the time of the issuance of the press release required by this paragraph (C), the Capital Group Redemption Stockholder Approval has not yet been obtained, such press release shall include as much of the information set forth in subparagraphs (I) to (VIII) as is then available, and the Corporation will issue a second press release once the Capital Group Redemption Stockholder Approval is obtained setting forth any such required information not included in the first press release.

- (D) The Corporation will give such notice to holders of Convertible Securities convertible into or exercisable or exchangeable for Liberty Capital Common Stock as may be required by the terms of such Convertible Securities or as the Board of Directors may otherwise deem appropriate in connection with a dividend, redemption or conversion of shares of Liberty Capital Common Stock pursuant to this Section A.2., as applicable.
- (E) All public announcements (including any proxy materials to the extent approval of the stockholders of the Corporation is sought or required) made pursuant to clauses (A), (B) or (C) of this paragraph (e)(iv) will include such further statements, and the Corporation reserves the right to make such further public announcements, as may be required by law or the rules of the principal national securities exchange on which the Liberty Capital Common Stock is listed or as the Board of Directors may, in its discretion, deem appropriate.
- (F) No adjustments in respect of dividends will be made upon the conversion or redemption of any shares of Liberty Capital Common Stock; provided, however, that, except as otherwise contemplated by paragraph (e)(ii)(D), if the Capital Group Conversion Date or the Capital Group Redemption Date with respect to any shares of Liberty Capital Common Stock will be subsequent to the record date for the payment of a dividend or other

30

distribution thereon or with respect thereto, but prior to the payment of such dividend or distribution, the holders of record of such shares of Liberty Capital Common Stock at the close of business on such record date will be entitled to receive the dividend or other distribution payable on or with respect to such shares on the date set for payment of such dividend or other distribution, notwithstanding the prior conversion or redemption of such shares.

- (G) Before any holder of shares of Liberty Capital Common Stock will be entitled to receive a certificate or certificates representing shares of any kind of capital stock or cash, securities or other assets to be received by such holder with respect to shares of Liberty Capital Common Stock pursuant to paragraph (b) of this Section A.2. or this paragraph (e), such holder will surrender at such place as the Corporation will specify certificates representing such shares of Liberty Capital Common Stock, properly endorsed or assigned for transfer (unless the Corporation will waive such requirement). The Corporation will as soon as practicable after such surrender of a certificate or certificates representing shares of Liberty Capital Common Stock, deliver, or cause to be delivered, at the office of the transfer agent for the shares or other securities to be delivered, to the holder for whose account shares of Liberty Capital Common Stock were so surrendered, or to the nominee or nominees of such holder, a certificate or certificates representing the number of whole shares of the kind of capital stock or cash, securities or other assets to which such Person will be entitled as aforesaid, together with any payment for fractional securities contemplated by paragraph (e)(iv)(I). If less than all of the shares of Liberty Capital Common Stock represented by any one certificate are to be redeemed or converted, the Corporation will issue and deliver a new certificate for the shares of Liberty Capital Common Stock not redeemed or converted. Shares selected for redemption may not thereafter be converted pursuant to paragraph (b)(i)(A) of this Section A.2.
- Liberty Capital Common Stock that were converted or redeemed on such Capital Group Conversion Date or Capital Group Redemption Date, as applicable, will cease except for the right, upon surrender of a certificate or certificates representing such shares of Liberty Capital Common Stock, to receive a certificate or certificates representing shares of the kind and amount of capital stock or cash, securities (other than capital stock) or other assets for which such shares were converted or redeemed, as applicable, together with any payment for fractional securities contemplated by paragraph (e)(iv)(I) of this Section A.2. and such holder will have no other or further rights in respect of the shares of Liberty Capital Common Stock so converted or redeemed, including, but not limited to, any rights with respect to any cash, securities or other assets which are reserved or otherwise designated by the Corporation as being held for the satisfaction of the Corporation's obligations to pay or deliver any cash, securities or other assets upon the conversion, exercise or exchange of any Convertible Securities outstanding as of the date of such conversion or redemption. No holder of a certificate which immediately prior to the applicable Capital Group Conversion Date or Capital Group Redemption Date represented shares of Liberty Capital Common Stock will be entitled to receive any dividend or other distribution with respect to shares of any kind of capital stock into or in exchange for which the Liberty Capital Common Stock was converted or redeemed until surrender of such holder's certificate or certificates representing shares of such kind of capital stock. Upon such surrender, there will be paid to the holder the

represented by the certificate or certificates issued upon such surrender. From and after a Capital Group Conversion Date or Capital Group Redemption Date, as the case may be, the Corporation will, however, be entitled to treat certificates representing shares of Liberty Capital Common Stock that have not yet been surrendered for conversion or redemption in accordance with clause (G) above as evidencing the ownership of the number of whole shares of the kind or kinds of capital stock for which the shares of Liberty Capital Common Stock represented by such certificates will have been converted or redeemed in accordance with paragraph (b) of this Section A.2 or this paragraph (e), notwithstanding the failure of the holder thereof to surrender such certificates.

- (I) The Corporation will not be required to issue or deliver fractional shares of any class or series of capital stock or any other securities in a smaller than authorized denomination to any holder of Liberty Capital Common Stock upon any conversion, redemption, dividend or other distribution pursuant to paragraph (b) of this Section A.2 or this paragraph (e). In connection with the determination of the number of shares of any class or series of capital stock that will be issuable or the amount of other securities that will be deliverable to any holder of record of Liberty Capital Common Stock upon any such conversion, redemption, dividend or other distribution (including any fractions of shares or securities), the Corporation may aggregate the shares of Liberty Capital Common Stock held at the relevant time by such holder of record. If the aggregate number of shares of capital stock or other securities to be issued or delivered to any holder of Liberty Capital Common Stock includes a fraction, the Corporation will pay, or will cause to be paid, a cash adjustment in lieu of such fraction in an amount equal to the "value" of such fraction, as the Board of Directors shall in good faith determine to be appropriate (without interest).
- (J) Any deadline for effecting a dividend, redemption or conversion prescribed by this paragraph (e) may be extended if deemed necessary or appropriate, in the discretion of the Board of Directors, to enable the Corporation to comply with the U.S. federal securities laws, including the rules and regulations promulgated thereunder.
  - (f) Redemption and Other Provisions Relating to the Liberty Starz Common Stock.
- (i) Redemption for Securities of one or more Starz Group Subsidiaries. At any time at which a Subsidiary of the Corporation holds, directly or indirectly, assets and liabilities attributed to the Starz Group, the Corporation may, at its option and subject to assets of the Corporation being legally available therefor but subject (in addition to any other approval of the Corporation's stockholders (or any series thereof) required under the DGCL in respect of such redemption, if any) to the Corporation having received the Starz Group Redemption Stockholder Approval (and, to the extent applicable, the Capital Group Redemption Stockholder Approval), redeem outstanding shares of Liberty Starz Common Stock (such shares of Liberty Starz Common Stock to be redeemed, the "Starz Group Redemption Shares") for securities of such Subsidiary (a "Distributed Starz Group Subsidiary"), as provided herein. The number of

32

Starz Group Redemption Shares will be determined, by the Board of Directors, by multiplying (A) the number of outstanding shares of Liberty Starz Common Stock as of the Starz Group Redemption Selection Date, by (B) the percentage of the Fair Value of the Starz Group that is represented by the Fair Value of the Corporation's equity interest in the Distributed Starz Group Subsidiary which is attributable to the Starz Group, in each case, as determined by the Board of Directors as of a date selected by the Board of Directors, as such percentage may be adjusted by the Board of Directors in its discretion to take into account such things as it deems relevant. The aggregate number of securities of the Distributed Starz Group Subsidiary to be delivered (the "Starz Group Distribution Subsidiary Securities") in redemption of the Starz Group Redemption Shares will be equal to: (A) if the Board of Directors makes a Starz Group Inter-Group Redemption Election as described below, the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the product of (I) the number of securities of the Distributed Starz Group Subsidiary owned by the Corporation's equity interest in the Distributed Starz Group Subsidiary that is represented by the Fair Value of the Corporation's equity interest in the Distributed Starz Group Subsidiary that is represented by the Fair Value of the Corporation's equity interest in the Distributed Starz Group Subsidiary Securities'), by (y) the Starz Group Subsidiary Securities' (B) if the Board of Directors does not make a Starz Group Inter-Group Redemption Election, all of the Distributable Starz Group Subsidiary Securities, in each case, subject to adjustment as provided below. The number of securities of the Distributed Starz Group Subsidiary to be delivered in redemption of each Starz Group Redemption Share will be equal to the amount (rounded, if necessary, to the nearest five decimal places) obtained by dividing (x) the number of Starz Group Distribution Subsidiary Secu

If the Starz Group Outstanding Interest Fraction is less than one (1) on the Starz Group Redemption Selection Date for any redemption pursuant to this paragraph (f)(i) and if (but only if) the Board of Directors so determines in its discretion (a "Starz Group Inter-Group Redemption Election"), then concurrently with the distribution of the Starz Group Distribution Subsidiary Securities in redemption of Starz Group Redemption Shares, the Corporation will attribute to the Capital Group an aggregate number of Distributable Starz Group Subsidiary Securities (the "Starz Group Inter-Group Interest Subsidiary Securities") equal to the difference between the total number of Distributable Starz Group Subsidiary Securities and the number of Starz Group Distribution Subsidiary Securities, subject to adjustment as provided below. If a Starz Group Inter-Group Redemption Election is made, then: (I) the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest will be decreased as described in subparagraph (ii)(D) of the definition of "Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest" in paragraph (i) of this Section A.2.; (II) the attribution of Starz Group Inter-Group Interest Subsidiary Securities to be made to the Capital Group may, at the discretion of the Board of Directors, be reflected by an allocation or by a direct transfer of Starz Group Inter-Group Interest Subsidiary Securities to such Group; and (III) the Board of Directors may determine that the Starz Group Inter-Group Interest Subsidiary Securities so allocated or transferred to the Capital Group will be distributed to holders of shares of Liberty Capital Common Stock as a Share Distribution pursuant to paragraph (d)(i)(C) of this Section A.2.

33

If at the time of a redemption of Liberty Starz Common Stock pursuant to this paragraph (f)(i), there are outstanding any Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Starz Common Stock that would become convertible into or exercisable or exchangeable for Distributable Starz Group Subsidiary Securities as a result of such redemption, and the obligation to deliver securities of such Distributed Starz Group Subsidiary upon exercise, exchange or conversion of such Convertible Securities is not assumed or otherwise provided for by the Distributed Starz Group Subsidiary, then the Board of Directors may make such adjustments as it determines to be appropriate to the number of Starz Group Redemption Shares, the number of Starz Group Distribution Subsidiary Securities and the number of Starz Group Inter-Group Interest Subsidiary Securities (and any related adjustment to the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest), to take into account the securities of the Distributed Starz Group Subsidiary into which such Convertible Securities are convertible or for which such Convertible Securities are exercisable or exchangeable.

In the event that not all outstanding shares of Liberty Starz Common Stock are to be redeemed in accordance with this paragraph (f)(i) for Starz Group Distribution Subsidiary Securities, then (1) the number of shares of each series of Liberty Starz Common Stock to be redeemed in accordance with this paragraph (f)(i) will be determined by multiplying the aggregate number of Starz Group Redemption Shares by a fraction, the numerator of which is the aggregate number of shares of such series and the denominator of which is the aggregate number of shares of all series of Liberty Starz Common Stock, in each case, outstanding as of the Starz Group Redemption Selection Date, and (2) the outstanding shares of each series of Liberty Starz Common Stock to be redeemed in accordance with this paragraph (f)(i) will be redeemed by the Corporation pro rata among the holders of each series of Liberty Starz Common Stock or by such other method as may be determined by the Board of Directors to be equitable.

To the extent that a Distributed Starz Group Subsidiary to be distributed pursuant to this paragraph (f)(i) also holds, directly or indirectly, assets and liabilities attributed to the Capital Group, then (x) such Distributed Starz Group Subsidiary will also be deemed a Distributed Capital Group Subsidiary for purposes of paragraph (e)(i) and (y) in connection with the redemption of Starz Group Redemption Shares pursuant to this paragraph (f)(i) the Corporation will also redeem shares of Liberty Capital Common Stock pursuant to the provisions of paragraph (e)(i), subject to the Corporation obtaining the Starz Group Redemption Stockholder Approval and the Capital Group Redemption Stockholder Approval. In connection with any such redemption of Liberty Capital Common Stock and Liberty Starz Common Stock, the Board of

Directors will effect such redemption in accordance with the terms of paragraphs (e)(i) and (f)(i), as determined by the Board of Directors in good faith, with such changes and adjustments as the Board of Directors determines are reasonably necessary in order to effect such redemption in exchange for securities of a single Subsidiary holding the assets and liabilities of more than one Group. In effecting such redemption, the Board of Directors may determine to redeem the Capital Group Redemption Shares and the Starz Group Redemption Shares, in exchange for one or more classes or series of securities of such Subsidiary, including, without limitation, for separate classes or series of securities of such Subsidiary, (I) with the holders of Capital Group Redemption Shares to receive Capital Group Distribution Subsidiary Securities

34

intended to track the performance of the former assets and liabilities attributed to the Capital Group held by such Subsidiary and (II) with holders of Starz Group Redemption Shares to receive Starz Group Distribution Subsidiary Securities intended to track the performance of the former assets and liabilities attributed to the Starz Group held by such Subsidiary, subject, in each case, to the applicable limitations on the class and series of securities of the Distributed Starz Group Subsidiary set forth in the last paragraph of paragraphs (e)(i) and (f)(i).

Any redemption pursuant to this paragraph (f)(i) will occur on a Starz Group Redemption Date set forth in a notice to holders of Liberty Starz Common Stock (and Convertible Securities convertible into or exercisable or exchangeable for shares of any series of Liberty Starz Common Stock (unless provision for notice is otherwise made pursuant to the terms of such Convertible Securities)) pursuant to paragraph (f)(iv)(C).

In effecting a redemption of Liberty Starz Common Stock pursuant to this paragraph (f)(i), the Board of Directors may determine either to (x) redeem shares of each series of Liberty Starz Common Stock in exchange for a single class or series of securities of the Distributed Starz Group Subsidiary without distinction among series of Liberty Starz Common Stock, on an equal per share basis, (y) redeem shares of each series of Liberty Starz Common Stock in exchange for separate classes or series of securities of the Distributed Starz Group Subsidiary, on an equal per share basis, or (z) redeem shares of one or more series of Liberty Starz Common Stock in exchange for a separate class or series of securities of the Distributed Starz Group Subsidiary and, on an equal per share basis, redeem shares of all other series of Liberty Starz Common Stock in exchange for a different class or series of securities of the Distributed Starz Group Subsidiary; provided, that, in the case of clauses (y) and (z), (1) such separate classes or series do not differ in any respect other than their relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), with holders of shares of Series B Liberty Starz Common Stock receiving securities of a class or series having the highest relative voting rights and the holders of shares of each other series of Liberty Starz Common Stock receiving securities of a class or series having lesser relative voting rights, in each case, without regard to whether such rights differ to a greater or lesser extent than the corresponding differences in voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) among the Series A Liberty Starz Common Stock, the Series B Liberty Starz Common Stock and the Series C Liberty Starz Common Stock, and (2) in the event the securities to be received by the holders of shares of Liberty Starz Common Stock other than the Series B Liberty Starz Common Stock in such redemption consist of different classes or series of securities, with each such class or series differing only with respect to the relative voting rights of such class or series (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), then such classes or series of securities will be distributed to the holders of each series of Liberty Starz Common Stock (other than the Series B Liberty Starz Common Stock) (i) as the Board of Directors determines or (ii) such that the relative voting rights of the class or series of securities to be received by the holders of each series of Liberty Starz Common Stock corresponds to the extent practicable to the relative voting rights (as compared to the other series of Liberty Starz Common Stock, other than the Series B Liberty

35

Starz Common Stock) of such series of Liberty Starz Common Stock. If the Board of Directors has made a Starz Group Inter-Group Redemption Election, then the determination as to the classes or series of securities of the Distributed Starz Group Subsidiary comprising the Starz Group Inter-Group Interest Subsidiary Securities to be so transferred or allocated to the Capital Group will be made by the Board of Directors in its discretion.

- (ii) Mandatory Dividend, Redemption or Conversion in Case of Starz Group Disposition. In the event of a Starz Group Disposition (other than an Exempt Starz Group Disposition), the Corporation will, on or prior to the 120th Trading Day following the consummation of such Starz Group Disposition and in accordance with the applicable provisions of this Section A.2., take the actions referred to in one of clauses (A), (B), (C) or (D) below, as elected by the Board of Directors:
- (A) Subject to the first sentence of paragraph (c)(ii) of this Section A.2. the Corporation may declare and pay a dividend payable in cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, to the holders of outstanding shares of Liberty Starz Common Stock, with an aggregate Fair Value (subject to adjustment as provided below) equal to the Starz Group Allocable Net Proceeds of such Starz Group Disposition as of the record date for determining the holders entitled to receive such dividend, as the same may be determined by the Board of Directors, with such dividend to be paid in accordance with the applicable provisions of paragraphs (c)(ii) and (d)(ii) of this Section A.2.; or
- (B) Provided that there are assets of the Corporation legally available therefor and the Starz Group Available Dividend Amount would have been sufficient to pay a dividend pursuant to clause (A) of this paragraph (f)(ii) in lieu of effecting the redemption provided for in this clause (B), then:
- (I) if such Starz Group Disposition involves all (not merely substantially all) of the assets of the Starz Group, the Corporation may redeem all outstanding shares of each series of Liberty Starz Common Stock for cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, with an aggregate Fair Value (subject to adjustment as provided below) equal to the Starz Group Allocable Net Proceeds of such Starz Group Disposition as of the Starz Group Redemption Date, as determined by the Board of Directors, such aggregate amount to be allocated among the shares of all series of Liberty Starz Common Stock outstanding as of the Starz Group Redemption Date on an equal per share basis (subject to the provisions of this paragraph (f)(ii)); or
- (II) if such Starz Group Disposition involves substantially all (but not all) of the assets of the Starz Group, the Corporation may apply an aggregate amount (subject to adjustment as provided below) of cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, with a Fair Value equal to the Starz Group Allocable Net Proceeds of such Starz Group Disposition as of the Starz Group Redemption Selection Date (the "Starz Group Redemption Amount") to the redemption of outstanding shares of each series of Liberty Starz Common Stock, such Starz Group Redemption Amount to be allocated (subject to the provisions of this paragraph (f)(ii)) to the redemption of shares of

36

each series of Liberty Starz Common Stock in the ratio of (x) the number of shares of such series outstanding as of the Starz Group Redemption Selection Date to (y) the aggregate number of shares of all series of Liberty Starz Common Stock outstanding as of such date, and the number of shares of each such series to be redeemed will equal the lesser of (1) the number of shares of such series outstanding as of the Starz Group Redemption Selection Date and (2) the whole number nearest the number obtained by dividing the aggregate amount so allocated to the redemption of such series by the Average Market Value of the Liberty Starz Reference Share over the period of 10 consecutive Trading Days beginning on the 2nd Trading Day following the consummation of such Starz Group Disposition; provided that, if following the foregoing allocation there remains any amount of the Starz Group Redemption Amount which is not being applied to the redemption of shares of a series of Liberty Starz Common Stock, then such excess amount will be allocated to the redemption of shares of each series of Liberty Starz Common Stock that, following the initial allocation referred to above, would have shares outstanding and not redeemed, with the number of outstanding and not redeemed shares to be redeemed from each such series to be calculated in

accordance with clauses (1) and (2) of the immediately preceding sentence based upon such excess amount of the Starz Group Redemption Amount. The outstanding shares of a series of Liberty Starz Common Stock to be redeemed will be selected on a pro rata basis among the holders of such series or by such other method as the Board of Directors may determine to be equitable; or

- (C) The Corporation may convert each outstanding share of Series A Liberty Starz Common Stock into a number (or fraction) of fully paid and non-assessable shares of Series A Liberty Capital Common Stock, each outstanding share of Series B Liberty Starz Common Stock into a number (or fraction) of fully paid and non-assessable shares of Series B Liberty Capital Common Stock, and each outstanding share of Series C Liberty Starz Common Stock into a number (or fraction) of fully paid and non-assessable shares of Series C Liberty Capital Common Stock, in each case, equal to 110% of the average daily ratio (calculated to the nearest five decimal places) of (I) the Average Market Value of the Liberty Starz Reference Share over the period of 10 consecutive Trading Days beginning on the 2nd Trading Day following the consummation of such Starz Group Disposition, to (II) the Average Market Value of the Liberty Capital Reference Share over the same 10-Trading Day period; or
- (D) The Corporation may combine the conversion of a portion of the outstanding shares of Liberty Starz Common Stock into Liberty Capital Common Stock as contemplated by clause (C) of this paragraph (f)(ii) with the payment of a dividend on or the redemption of shares of Liberty Starz Common Stock as described below, subject to the limitations specified in clause (A) (in the case of a dividend) or clause (B) (in the case of a redemption) of this paragraph (f)(ii) (including the limitations specified in other paragraphs of this Certificate referred to therein). In the event the Board of Directors elects the option described in this clause (D), the portion of the outstanding shares of Liberty Starz Common Stock to be converted into fully paid and non-assessable shares of Liberty Capital Common Stock will be determined by the Board of Directors and will be so converted at the conversion rate determined in accordance with clause (C) above and the Corporation will either (x) pay a dividend to the holders of record of all of the remaining shares of Liberty Starz Common Stock outstanding, with such dividend to be paid in accordance with the applicable provisions of

37

paragraphs (c)(ii) and (d)(ii) of this Section A.2., or (y) redeem all or a portion of such remaining shares of Liberty Starz Common Stock. The aggregate amount of such dividend, in the case of a dividend, or the portion of the Starz Group Allocable Net Proceeds to be applied to such redemption, in the case of a redemption, will be equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (I) an amount equal to the Starz Group Allocable Net Proceeds of such Starz Group Disposition as of, in the case of a dividend, the record date for determining the holders of Liberty Starz Common Stock entitled to receive such dividend and, in the case of a redemption, the Starz Group Redemption Selection Date (in the case of a partial redemption) or the Starz Group Redemption Date (in the case of a full redemption), in each case, before giving effect to the conversion of shares of Liberty Starz Common Stock in connection with such Starz Group Disposition in accordance with this clause (D) and any related adjustment to the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest, by (II) one minus a fraction, the numerator of which will be the number of shares of Liberty Starz Common Stock to be converted into shares of Liberty Capital Common Stock in accordance with this clause (D) and the denominator of which will be the aggregate number of shares of Liberty Starz Common Stock outstanding as of the record date, Starz Group Redemption Selection Date or Starz Group Redemption Date used for purposes of clause (I) of this sentence. In the event of a redemption concurrently with or following any such partial conversion of shares of Liberty Starz Common Stock, if the Starz Group Disposition was of all (not merely substantially all) of the assets of the Starz Group, then all remaining outstanding shares of Liberty Starz Common Stock will be redeemed for cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, with an aggregate Fair Value equal to the portion of the Starz Group Allocable Net Proceeds to be applied to such redemption determined in accordance with this clause (D), such aggregate amount to be allocated among all such shares to be redeemed on an equal per share basis (subject to the provisions of this paragraph (f)(ii)). In the event of a redemption concurrently with or following any such partial conversion of shares of Liberty Starz Common Stock, if the Starz Group Disposition was of substantially all (but not all) of the assets of the Starz Group, then the number of shares of each series of Liberty Starz Common Stock to be redeemed will be determined in accordance with clause (B)(II) of this paragraph (f)(ii), substituting for the Starz Group Redemption Amount referred to therein the portion of the Starz Group Allocable Net Proceeds to be applied to such redemption as determined in accordance with this clause (D), and such shares will be redeemed for cash, securities (other than Common Stock) or other assets, or any combination thereof, with an aggregate Fair Value equal to such portion of the Starz Group Allocable Net Proceeds and allocated among all such shares to be redeemed on an equal per share basis (subject to the provisions of this paragraph (f)(ii)). The aggregate number of shares of Liberty Starz Common Stock to be converted in any partial conversion in accordance with this clause (D) will be allocated among the series of Liberty Starz Common Stock in the ratio of the number of shares of each such series outstanding to the aggregate number of shares of all series of Liberty Starz Common Stock outstanding as of the Starz Group Conversion Selection Date, and the shares of each such series to be converted will be selected on a pro rata basis or by such other method as the Board of Directors may determine to be equitable. In the case of a redemption, the allocation of the cash, securities (other than shares of Common Stock) and/or other assets to be paid in redemption and, in the case of a partial redemption, the selection of

38

shares to be redeemed will be made in the manner contemplated by clause (B) of this paragraph (f)(ii).

For purposes of this paragraph (f)(ii):

- (1) as of any date, "substantially all of the assets of the Starz Group" means a portion of such assets that represents at least 80% of the then-Fair Value of the assets of the Starz Group as of such date;
- (2) in the case of a Starz Group Disposition of assets in a series of related transactions, such Starz Group Disposition will not be deemed to have been consummated until the consummation of the last of such transactions;
- (3) if the Board of Directors seeks the approval of the holders of Liberty Starz Voting Securities entitled to vote thereon to qualify a Starz Group Disposition as an Exempt Starz Group Disposition and such approval is not obtained, the date on which such approval fails to be obtained will be treated as the date on which such Starz Group Disposition was consummated for purposes of making the determinations and taking the actions prescribed by this paragraph (f)(ii) and paragraph (f)(iv), and no subsequent vote may be taken to qualify such Starz Group Disposition as an Exempt Starz Group Disposition;
- (4) in the event of a redemption of a portion of the outstanding shares of Liberty Starz Common Stock pursuant to clause (B)(II) or (D) of this paragraph (f)(ii) at a time when the Starz Group Outstanding Interest Fraction is less than one, if the Board of Directors so elects (a "Starz Group Inter-Group Partial Redemption Election"), in its discretion, the Corporation will attribute to the Capital Group concurrently with such redemption an aggregate amount (the Starz Group Inter-Group Redemption Amount") of cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, subject to adjustment as described below, with an aggregate Fair Value equal to the difference between (x) the Starz Group Net Proceeds and (y) the portion of the Starz Group Allocable Net Proceeds applied to such redemption as determined in accordance with clause (B)(II) or clause (D) of this paragraph (f)(ii). If the Board of Directors makes such election, the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest will be decreased in the manner described in subparagraph (ii)(E) of the definition of "Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest" in paragraph (i) of this Section A.2. The Starz Group Inter-Group Redemption Amount may, at the discretion of the Board of Directors, be reflected by an allocation to the Capital Group or by a direct transfer to the Capital Group of cash, securities and/or other assets:
- (5) if at the time of a Starz Group Disposition subject to this paragraph (f)(ii), there are outstanding any Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Starz Common Stock that would give the holders thereof the right to receive any consideration related to such Starz Group Disposition upon conversion, exercise or exchange or otherwise, or would adjust to give the holders equivalent economic rights, as a result of any dividend, redemption or other action taken by the Corporation with

respect to the Liberty Starz Common Stock pursuant to this paragraph (f)(ii), then the Board of Directors may make such adjustments to (x) the amount of consideration to be issued or delivered as contemplated by this paragraph (f)(ii) as a dividend on or in redemption or conversion of shares of Liberty Starz Common Stock and/or, if applicable, (y) the Starz Group Inter-Group Redemption Amount and the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest as it deems appropriate to take into account the Liberty Starz Common Stock into which such Convertible Securities are exercisable or exchangeable;

the Corporation may pay the dividend or redemption price referred to in clause (A), (B) or (D) of this paragraph (f)(ii) payable to the holders of Liberty Starz Common Stock in cash, securities (other than shares of Common Stock) or other assets, or any combination thereof, that the Board of Directors determines and which has an aggregate Fair Value of not less than the amount allocated to such dividend or redemption pursuant to the applicable of clauses (A), (B) or (D) of this paragraph (f)(ii), regardless of the form or nature of the proceeds received by the Corporation from the Starz Group Disposition; and

if all or any portion of the redemption price referred to in clause (B) or (D) of this paragraph (f)(ii) payable to the holders of Liberty Starz Common Stock is paid in the form of securities of an issuer other than the Corporation, the Board of Directors may determine to pay the redemption price, so payable in securities, in the form of (x) identical securities, on an equal per share basis, to holders of each series of Liberty Starz Common Stock, (y) separate classes or series of securities, on an equal per share basis, to the holders of each series of Liberty Starz Common Stock or (z) a separate class or series of securities to the holders of one or more series of Liberty Starz Common Stock and, on an equal per share basis, a different class or series of securities to the holders of all other series of Liberty Starz Common Stock; provided, that, in the case of clauses (y) and (z), (1) such separate classes or series do not differ in any respect other than their relative voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), with holders of shares of Series B Liberty Starz Common Stock receiving securities of a class or series having the highest relative voting rights and the holders of shares of each other series of Liberty Starz Common Stock receiving securities of a class or series having lesser relative voting rights, in each case, without regard to whether such rights differ to a greater or lesser extent than the corresponding differences in voting rights (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.) among the Series A Liberty Starz Common Stock, the Series B Liberty Starz Common Stock and the Series C Liberty Starz Common Stock and (2) in the event the securities to be received by the holders of shares of Liberty Starz Common Stock other than the Series B Liberty Starz Common Stock consist of different classes or series of securities, with each such class or series differing only with respect to the relative voting rights of such class or series (and any related differences in designation, conversion, redemption and share distribution provisions such as those set forth in this Section A.2.), then such classes or series of securities will be distributed to the holders of each series of Liberty Starz Common Stock (other than the Series B Liberty Starz Common Stock) (i) as the Board of Directors determines or (ii) such that the relative voting rights of the class or series of securities to be received by the holders of each

40

series of Liberty Starz Common Stock corresponds to the extent practicable to the relative voting rights (as compared to the other series of Liberty Starz Common Stock, other than the Series B Liberty Starz Common Stock) of such series of Liberty Starz Common Stock.

(iii) Certain Provisions Respecting Convertible Securities. Unless the provisions of any Convertible Securities that are or become convertible into or exercisable or exchangeable for shares of any series of Liberty Starz Common Stock provide specifically to the contrary, or the instrument, plan or agreement evidencing such Convertible Securities or pursuant to which the same were issued grants to the Board of Directors the discretion to approve or authorize any adjustment or adjustments to the conversion, exercise or exchange provisions of such Convertible Securities so as to obtain a result different from that which would otherwise occur pursuant to this paragraph (f)(iii), and the Board of Directors so approves or authorizes such adjustment or adjustments, after any Starz Group Conversion Date or Starz Group Redemption Date on which all outstanding shares of Liberty Starz Common Stock were converted or redeemed, any share of Liberty Starz Common Stock that is issued on conversion, exercise or exchange of any such Convertible Security will, immediately upon issuance and without any notice or any other action on the part of the Corporation or its Board of Directors or the holder of such share of Liberty Starz Common Stock, be redeemed in exchange for, to the extent assets of the Corporation are legally available therefor, the amount of \$.01 per share in cash.

### (iv) General.

(A) Not later than the 10th Trading Day following the consummation of a Starz Group Disposition referred to in paragraph (f)(ii) of this Section A.2., the Corporation will announce publicly by press release (x) the Starz Group Net Proceeds of such Starz Group Disposition, (y) whether the Starz Group Disposition qualifies as an Exempt Starz Group Disposition, and (z) if it does not so qualify at the time of such announcement (including in the event the Board of Directors had not sought stockholder approval to qualify such Starz Group Disposition as an Exempt Starz Group Disposition in connection with any required stockholder approval obtained by the Corporation, if applicable), whether the Board of Directors will seek the approval of the holders of Liberty Starz Voting Securities entitled to vote thereon to qualify such Starz Group Disposition as an Exempt Starz Group Disposition. Not later than the 30<sup>th</sup> Trading Day (and in the event a 10 Trading Day valuation period is required in connection with the action selected by the Board of Directors pursuant to clause (I) of this paragraph (f)(iv)(A), not earlier than the 11<sup>th</sup> Trading Day) following the later of (x) the consummation of such Starz Group Disposition and (y), if applicable, the date of the stockholder meeting at which a vote is taken to qualify such Starz Group Disposition, the Corporation will announce publicly by press release (to the extent applicable):

(I) which of the actions specified in clauses (A), (B), (C) or (D) of paragraph (f)(ii) of this Section A.2. the Corporation has

irrevocably determined to take;

(II) as applicable, the record date for determining holders entitled to receive any dividend to be paid pursuant to clause (A) or (D) of

paragraph (f)(ii), the

41

Starz Group Redemption Selection Date for the redemption of shares of Liberty Starz Common Stock pursuant to clause (B)(II) or (D) of paragraph (f)(ii) or the Starz Group Conversion Selection Date for the partial conversion of shares of Liberty Starz Common Stock pursuant to clause (D) of paragraph (f)(ii), which record date, Starz Group Redemption Selection Date or Starz Group Conversion Selection Date will not be earlier than the 10th day following the date of such public announcement;

(III) the anticipated dividend payment date, Starz Group Redemption Date and/or Starz Group Conversion Date, which in each case, will not be more than 85 Trading Days following such Starz Group Disposition; and

(IV) unless the Board of Directors otherwise determines, that the Corporation will not be required to register a transfer of any shares of Liberty Starz Common Stock for a period of 10 Trading Days (or such shorter period as such announcement may specify) next preceding the specified Starz Group Redemption Selection Date or Starz Group Conversion Selection Date.

If the Corporation determines to undertake a redemption of shares of Liberty Starz Common Stock, in whole or in part, pursuant to clause (B) or (D) of paragraph (f)(ii) of this Section A.2., or a conversion of shares of Liberty Starz Common Stock, in whole or in part, pursuant to clause (C) or (D) of paragraph (f)(ii), the

Corporation will announce such redemption or conversion (which, for the avoidance of doubt, may remain subject to the satisfaction or waiver of any applicable condition precedent at the time of such announcement) publicly by press release, not less than 10 days prior to the Starz Group Redemption Date or Starz Group Conversion Date, as applicable:

(1) the Starz Group Redemption Date or Starz Group Conversion Date;

(2) the number of shares of Liberty Starz Common Stock to be redeemed or converted or, if applicable, stating that all outstanding shares of Liberty Starz Common Stock will be redeemed or converted and the series of Liberty Capital Common Stock issuable to the holders of each series of Liberty Starz Common Stock upon any such conversion;

(3) in the case of a redemption or a conversion, in each case, in whole or in part, of outstanding shares of Liberty Starz Common Stock, the kind and amount of per share consideration to be received with respect to each share of Liberty Starz Common Stock to be redeemed or converted and the Starz Group Outstanding Interest Fraction as of the date of such notice;

(4) with respect to a partial redemption under clause (B)(II) or (D) of paragraph (f)(ii), if the Board of Directors has made a Starz Group Inter-Group Partial Redemption Election, the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest as of the Starz Group Redemption Date;

42

- (5) with respect to a dividend under clause (A) or (D) of paragraph (f)(ii), the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Inter-Group Interest as of the record date for the dividend and the Starz Group Inter-Group Dividend Amount attributable to the Capital Group; and
  - (6) instructions as to how shares of Liberty Starz Common Stock may be surrendered for redemption or conversion.
- (B) In the event of any conversion of shares of Liberty Starz Common Stock pursuant to paragraph (b)(ii) of this Section A.2., not less than 10 days prior to the Starz Group Conversion Date, the Corporation will announce publicly by press release:
- (1) that all outstanding shares of Liberty Starz Common Stock will be converted pursuant to paragraph (b)(ii) of this Section A.2. on the Starz Group Conversion Date;
  - (2) the Starz Group Conversion Date, which will not be more than 45 days following the Determination Date;
  - (3) a statement that all outstanding shares of Liberty Starz Common Stock will be converted;
- (4) the per share number and series of shares of Liberty Capital Common Stock to be received with respect to each share of each series of Liberty Starz Common Stock; and
  - (5) instructions as to how shares of Liberty Starz Common Stock may be surrendered for conversion.
- (C) If the Corporation determines to obtain the Starz Group Redemption Stockholder Approval and, subject to the receipt of such approval, to redeem shares of Liberty Starz Common Stock pursuant to paragraph (f)(i), the Corporation will announce publicly by press release:
- (I) that the Corporation intends to redeem shares of Liberty Starz Common Stock for securities of a Distributed Starz Group Subsidiary pursuant to paragraph (f)(i) of this Section A.2, subject to any applicable conditions, including the receipt of the Starz Group Redemption Stockholder Approval if such approval has not been obtained at the time of the press release;
- (II) the number of shares of Liberty Starz Common Stock to be redeemed or, if applicable, stating that all outstanding shares of Liberty Starz Common Stock will be redeemed;
- (III) the class or series of securities of the Distributed Starz Group Subsidiary to be received with respect to each share of each series of Liberty Starz

43

Common Stock to be redeemed and the Starz Group Outstanding Interest Fraction as of the date of such notice, if any;

- (IV) if applicable, the Starz Group Redemption Selection Date, which will not be earlier than the 10th day following the date of the press release;
- (V) the Starz Group Redemption Date, which will not be earlier then the 10th day following the date of the press release and will not be later than the 120th Trading Day following the date of the press release;
- (VI) if the Board of Directors has made a Starz Group Inter-Group Redemption Election, the number of Starz Group Inter-Group Interest Subsidiary Securities attributable to the Capital Group, and the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest used in determining such number and attribution of Starz Group Inter-Group Interest Subsidiary Securities; and
  - (VII) instructions as to how shares of Liberty Starz Common Stock may be surrendered for redemption; and
- (VIII) if the Board of Directors so determines, that the Corporation will not be required to register a transfer of any shares of Liberty Starz Common Stock for a period of 10 Trading Days (or such shorter period as such announcement may specify) next preceding the specified Starz Group Redemption Selection Date.

If, at the time of the issuance of the press release required by this paragraph (C), the Starz Group Redemption Stockholder Approval has not yet been obtained, such press release shall include as much of the information set forth in subparagraphs (I) to (VIII) as is then available, and the Corporation will issue a second press release once the Starz Group Redemption Stockholder Approval is obtained setting forth any such required information not included in the first press release.

(D) The Corporation will give such notice to holders of Convertible Securities convertible into or exercisable or exchangeable for Liberty Starz Common Stock as may be required by the terms of such Convertible Securities or as the Board of Directors may otherwise deem appropriate in connection with a dividend, redemption or conversion of shares of Liberty Starz Common Stock pursuant to this Section A.2., as applicable.

(E) All public announcements (including any proxy materials to the extent approval of the stockholders of the Corporation is sought or required) made pursuant to clauses (A), (B) or (C) of this paragraph (f)(iv) will include such further statements, and the Corporation reserves the right to make such further
public announcements, as may be required by law or the rules of the principal national securities exchange on which the Liberty Starz Common Stock is listed or as the Board of Directors may, in its discretion, deem appropriate.
(F) No adjustments in respect of dividends will be made upon the conversion or redemption of any shares of Liberty Starz Common Stock; provided, however,
44
that, except as otherwise contemplated by paragraph (f)(ii)(D), if the Starz Group Conversion Date or the Starz Group Redemption Date with respect to any shares of Liberty Starz Common Stock will be subsequent to the record date for the payment of a dividend or other distribution thereon or with respect thereto, but prior to the payment of such dividend or distribution, the holders of record of such shares of Liberty Starz Common Stock at the close of business on such record date will be entitled to receive the dividend or other distribution payable on or with respect to such shares on the date set for payment of such dividend or other distribution, notwithstanding the prior conversion

or redemption of such shares.

Before any holder of shares of Liberty Starz Common Stock will be entitled to receive a certificate or certificates representing shares of any kind of capital stock or cash, securities or other assets to be received by such holder with respect to shares of Liberty Starz Common Stock pursuant to paragraph (b) of this Section A.2. or this paragraph (f), such holder will surrender at such place as the Corporation will specify certificates representing such shares of Liberty Starz Common Stock, properly endorsed or assigned for transfer (unless the Corporation will waive such requirement). The Corporation will as soon as practicable after such surrender of a certificate or certificates representing shares of Liberty Starz Common Stock, deliver, or cause to be delivered, at the office of the transfer agent for the shares or other securities to be delivered, to the holder for whose account shares of Liberty Starz Common Stock were so surrendered, or to the nominee or nominees of such holder, a certificate or certificates representing the number of whole shares of the kind of capital stock or cash, securities or other assets to which such Person will be entitled as aforesaid, together with any payment for fractional securities contemplated by paragraph (f)(iv)(I). If less than all of the shares of Liberty Starz Common Stock represented by any one certificate are to be redeemed or converted, the Corporation will issue and deliver a new certificate for the shares of Liberty Starz Common Stock not redeemed or converted. Shares selected for redemption may not thereafter be converted pursuant to paragraph (b)(i)(B) of this Section A.2.

From and after any applicable Starz Group Conversion Date or Starz Group Redemption Date, all rights of a holder of shares of Liberty Starz Common Stock that were converted or redeemed on such Starz Group Conversion Date or Starz Group Redemption Date, as applicable, will cease except for the right, upon surrender of a certificate or certificates representing such shares of Liberty Starz Common Stock, to receive a certificate or certificates representing shares of the kind and amount of capital stock or cash, securities (other than capital stock) or other assets for which such shares were converted or redeemed, as applicable, together with any payment for fractional securities contemplated by paragraph (f)(iv)(I) of this Section A.2. and such holder will have no other or further rights in respect of the shares of Liberty Starz Common Stock so converted or redeemed, including, but not limited to, any rights with respect to any cash, securities or other assets which are reserved or otherwise designated by the Corporation as being held for the satisfaction of the Corporation's obligations to pay or deliver any cash, securities or other assets upon the conversion, exercise or exchange of any Convertible Securities outstanding as of the date of such conversion or redemption. No holder of a certificate which immediately prior to the applicable Starz Group Conversion Date or Starz Group Redemption Date represented shares of Liberty Starz Common Stock will be entitled to receive any dividend or other distribution with respect to shares of any kind of capital

45

stock into or in exchange for which the Liberty Starz Common Stock was converted or redeemed until surrender of such holder's certificate for a certificate or certificates representing shares of such kind of capital stock. Upon such surrender, there will be paid to the holder the amount of any dividends or other distributions (without interest) which theretofore became payable with respect to a record date after the Starz Group Conversion Date or Starz Group Redemption Date, as the case may be, but that were not paid by reason of the foregoing, with respect to the number of whole shares of the kind of capital stock represented by the certificate or certificates issued upon such surrender. From and after a Starz Group Conversion Date or Starz Group Redemption Date, as the case may be, the Corporation will, however, be entitled to treat certificates representing shares of Liberty Starz Common Stock that have not yet been surrendered for conversion or redemption in accordance with clause (G) above as evidencing the ownership of the number of whole shares of the kind or kinds of capital stock for which the shares of Liberty Starz Common Stock represented by such certificates will have been converted or redeemed in accordance with paragraph (b) of this Section A.2 or this paragraph (f), notwithstanding the failure of the holder thereof to surrender such certificates

The Corporation will not be required to issue or deliver fractional shares of any class or series of capital stock or any other securities in a smaller than authorized denomination to any holder of Liberty Starz Common Stock upon any conversion, redemption, dividend or other distribution pursuant to paragraph (b) of this Section A.2. or this paragraph (f). In connection with the determination of the number of shares of any class or series of capital stock that will be issuable or the amount of other securities that will be deliverable to any holder of record of Liberty Starz Common Stock upon any such conversion, redemption, dividend or other distribution (including any fractions of shares or securities), the Corporation may aggregate the shares of Liberty Starz Common Stock held at the relevant time by such holder of record. If the aggregate number of shares of capital stock or other securities to be issued or delivered to any holder of Liberty Starz Common Stock includes a fraction, the Corporation will pay, or will cause to be paid, a cash adjustment in lieu of such fraction in an amount equal to the "value" of such fraction, as the Board of Directors shall in good faith determine to be appropriate (without interest).

Any deadline for effecting a dividend, redemption or conversion prescribed by this paragraph (f) may be extended if deemed necessary or appropriate, in the discretion of the Board of Directors, to enable the Corporation to comply with the U.S. federal securities laws, including the rules and regulations promulgated thereunder.

#### (g) <u>Liquidation and Dissolution</u>.

General. In the event of a liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, after payment or provision for payment of the debts and liabilities of the Corporation and subject to the prior payment in full of the preferential amounts to which any series of Preferred Stock is entitled, the holders of shares of Liberty Capital Common Stock and the holders of shares of Liberty Starz Common Stock will be entitled to receive their proportionate interests in the assets of the Corporation remaining for distribution to holders of Common Stock (regardless of the Group to which such assets are then attributed) in

46

proportion to the respective number of liquidation units per share of Liberty Capital Common Stock and Liberty Starz Common Stock.

- (ii) <u>Liquidation Units</u>. The liquidation units per share of each series of Common Stock will be as follows:
  - (A) each share of Liberty Capital Common Stock will have one liquidation unit; and
- (B) each share of Liberty Starz Common Stock will have a number of liquidation units (including a fraction of one liquidation unit) equal to the amount (calculated to the nearest five decimal places) obtained by dividing (x) the average of the daily volume weighted average prices of the Series A Liberty Starz Common Stock over the 20-Trading Day period commencing on (and including) the first Trading Day on which the Series A Liberty Starz Common Stock trades in the "regular way" market, by (y) the average of the daily volume weighted average prices of the Series A Liberty Capital Common Stock over the 20-Trading Day period commencing on (and including) the first Trading Day on which the Series A Liberty Capital Common Stock trades in the "regular way" market;

provided, that, if, after the Effective Date, the Corporation, at any time or from time to time, subdivides (by stock split, reclassification or otherwise) or combines (by reverse stock split, reclassification or otherwise) the outstanding shares of Liberty Capital Common Stock or Liberty Starz Common Stock, or declares and pays a dividend or distribution in shares of Liberty Capital Common Stock or Liberty Starz Common Stock to holders of Liberty Capital Common Stock or Liberty Starz Common Stock, as applicable, the per share liquidation units of the Liberty Capital Common Stock or Liberty Starz Common Stock, as applicable, will be appropriately adjusted as determined by the Board of Directors, so as to avoid any dilution in the aggregate, relative liquidation rights of the shares of Liberty Capital Common Stock and Liberty Starz Common Stock.

Whenever an adjustment is made to liquidation units under this paragraph (g), the Corporation will promptly thereafter prepare and file a statement of such adjustment with the Secretary of the Corporation. Neither the failure to prepare nor the failure to file any such statement will affect the validity of such adjustment.

(h) <u>Determinations by the Board of Directors.</u> Any determinations made by the Board of Directors under any provision in this Section A.2. will be final and binding on all stockholders of the Corporation, except as may otherwise be required by law. In addition, if different consideration is distributed to different series of Common Stock in a Share Distribution, the determination of the Board of Directors that such Share Distribution was made on an equal

47

per share basis will be final and binding on all stockholders of the Corporation, except as may otherwise be required by law.

- (i) <u>Certain Definitions</u>. Unless the context otherwise requires, the terms defined in this paragraph (i) will have, for all purposes of this Certificate, the meanings herein specified:
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly through one or more intermediaries, Controls, or is Controlled by, or is under common Control with such Person.
- "Approval Date" means the date upon which the Corporation has received each of the Capital Group Redemption Stockholder Approval and/or the Starz Group Redemption Stockholder Approval, to the extent required pursuant to this Section A.2.
- "Average Market Value" of a share of any series of Common Stock or other Publicly Traded capital stock means the average of the daily Market Values of one share of such series of Common Stock or such other capital stock over the applicable period prescribed in this Certificate.
- "Board of Directors" means (i) the Board of Directors of the Corporation and (ii) any duly authorized committee thereof acting at the direction of the Board of Directors (including, without limitation, the Executive Committee).
  - "Capital Group" means, as of any date:
- (i) the direct and indirect interest of the Corporation as of the Effective Date (x) in all of the businesses in which the Corporation is or has been engaged, directly or indirectly (either itself or through direct or indirect Subsidiaries, Affiliates, joint ventures or other investments or any of the predecessors or successors of any of the foregoing), and (y) in the respective assets and liabilities of the Corporation and its Subsidiaries, in each case, other than any businesses, assets or liabilities attributable to the Starz Group as of the Effective Date;
- (ii) all assets, liabilities and businesses acquired or assumed by the Corporation or any of its Subsidiaries for the account of the Capital Group, or contributed, allocated or transferred to the Capital Group (including the net proceeds of any issuances, sales or incurrences for the account of the Capital Group of shares of Liberty Capital Common Stock, Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Capital Common Stock, or indebtedness or Preferred Stock attributed to the Capital Group), in each case, after the Effective Date and as determined by the Board of Directors;
  - (iii) the proceeds of any Disposition of any of the foregoing; and
  - (iv) an Inter-Group Interest in the Starz Group equal to one (1) minus the Starz Group Outstanding Interest Fraction as of such date;

48

provided that the Capital Group will not include (A) any assets, liabilities or businesses disposed of after the Effective Date, including, without limitation, by dividend, to holders of Liberty Capital Common Stock or in redemption of shares of Liberty Capital Common Stock, from and after the date of such Disposition or (B) any assets, liabilities or businesses transferred or allocated after the Effective Date from the Capital Group to the Starz Group (other than through the Capital Group's Inter-Group Interest in the Starz Group, if any, pursuant to clause (iv) above), including, without limitation, any Capital Group Inter-Group Dividend Amount or Capital Group Inter-Group Redemption Amount, from and after the date of such transfer or allocation.

"Capital Group Allocable Net Proceeds" means, with respect to any Capital Group Disposition, (i) if at the time of such Capital Group Disposition, the Capital Group Outstanding Interest Fraction is one (1), the Capital Group Net Proceeds of such Capital Group Disposition, or (ii) if at the time of such Capital Group Disposition the Capital Group Outstanding Interest Fraction is less than one (1), the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Capital Group Net Proceeds of such Capital Group Disposition, by (y) the Capital Group Outstanding Interest Fraction as of such date.

"Capital Group Available Dividend Amount," as of any date, means an amount equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Capital Group Outstanding Interest Fraction, by (y) either: (i) the excess of (A) an amount equal to the total assets of the Capital Group less the total liabilities (not including Preferred Stock attributed to the Capital Group) of the Capital Group as of such date over (B) the aggregate par value of, or any greater amount determined to be capital in respect of, all outstanding shares of Liberty Capital Common Stock and each series of Preferred Stock attributed to the Capital Group or (ii) in case there is no such excess, an amount equal to the Corporation Earnings (Loss) Attributable to the Capital Group (if positive) for the fiscal year in which such date occurs and/or the preceding fiscal year.

"Capital Group Conversion Date" means any date and time fixed by the Board of Directors for a conversion of shares of Liberty Capital Common Stock pursuant to this Section A.2.

"Capital Group Conversion Selection Date" means any date and time fixed by the Board of Directors as the date and time upon which shares to be converted of each series of Liberty Capital Common Stock will be selected for conversion pursuant to this Section A.2. (which, for the avoidance of doubt, may be the same date and time as the Capital Group Conversion Date).

"Capital Group Disposition" means the Disposition, in one transaction or a series of related transactions, by the Corporation and its Subsidiaries of all or substantially all of the assets of the Capital Group to one or more Persons.

49

"Capital Group Net Proceeds" means, as of any date, with respect to any Capital Group Disposition, an amount, if any, equal to the Fair Value of what remains of the gross proceeds of such Disposition to the Corporation after any payment of, or reasonable provision for, (i) any taxes payable by the Corporation or any of its Subsidiaries in respect of such Disposition or in respect of any resulting dividend or redemption pursuant to clause (A), (B) or (D) of paragraph (e)(ii) of this Section A.2. (or that would have been payable but for the utilization of tax benefits attributable to the Starz Group), (ii) any transaction costs, including, without limitation, any legal, investment banking and accounting fees and expenses and (iii) any liabilities and other obligations (contingent or otherwise) of, or attributed to, the Capital Group, including, without limitation, any liabilities for future purchase price adjustments and any preferential amounts plus any accumulated and unpaid dividends and other obligations in respect of Preferred Stock attributed to the Capital Group. For purposes of this definition, any assets of the Capital Group remaining after such Disposition will constitute "reasonable provision" for such amount of taxes, costs, liabilities and other obligations (contingent or otherwise) as can be supported by such assets.

"Capital Group Outstanding Interest Fraction," as of any date, means a fraction the numerator of which is the aggregate number of shares of Liberty Capital Common Stock outstanding on such date and the denominator of which is the amount obtained by adding (i) such aggregate number of shares of Liberty Capital Common Stock outstanding on such date, plus (ii) the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest as of such date, provided that such fraction will in no event be greater than one. If the holders of any Convertible Securities that are convertible into or exercisable or exchangeable for shares of Liberty Capital Common Stock are entitled to participate in any dividend (for purposes of paragraphs (c)(i), (d)(i) or (e)(ii) of this Section A.2.) or redemption (for purposes of paragraph (e) of this Section A.2.) with respect to the Liberty Capital Common Stock (other than by means of an antidilution adjustment), such shares so issuable upon conversion, exercise or exchange will be taken into account in calculating the Capital Group Outstanding Interest Fraction and any related calculations under the applicable provisions of this Section A.2. in such manner as the Board of Directors determines to be appropriate.

"Capital Group Redemption Date" means any date and time fixed by the Board of Directors for a redemption of shares of Liberty Capital Common Stock pursuant to this Section A.2.

"Capital Group Redemption Selection Date" means the date and time fixed by the Board of Directors on which shares of Liberty Capital Common Stock are to be selected for redemption pursuant to this Section A.2. (which, for the avoidance of doubt, may be the same date and time as the Capital Group Redemption Date).

"Capital Group Related Business Transaction" means any Disposition of all or substantially all of the assets of the Capital Group in which the Corporation receives as proceeds of such Disposition primarily equity securities (including, without limitation, capital stock, securities convertible into capital stock or other equity securities, partnership, limited partnership

50

or limited liability company interests and other types of equity securities, without regard to the voting power or contractual or other management or governance rights related to such equity securities) of the purchaser or acquiror of such assets of the Capital Group, any entity which succeeds (by merger, formation of a joint venture enterprise or otherwise) to such assets of the Capital Group, or a third party issuer, if a significant portion of the business or businesses in which such purchaser, acquiror or third party issuer is engaged or proposes to engage consists of one or more businesses similar or complementary to the businesses attributable to the Capital Group prior to such Disposition, as determined in good faith by the Board of Directors.

"Capital Group Share Distribution Ratio" means, as to any Share Distribution consisting of shares of Capital Group Common Stock, the number of shares (including any fraction of a share), of Capital Group Common Stock issuable to a holder for each outstanding share of the applicable series of Common Stock owned by such holder as of the record date for such Share Distribution (rounded, if necessary, to the nearest five decimal places).

"Certificate" means this Restated Certificate of Incorporation, as it may be amended from time to time, including any amendments effected pursuant to the filing of any Preferred Stock Designation.

"Code" means the Internal Revenue Code of 1986, as amended.

"Control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person whether through the ownership of voting securities, by contract or otherwise; provided, however, that for purposes of clause (iii) of the definition of "Exempt Capital Group Disposition" or "Exempt Starz Group Disposition" set forth in this paragraph (i), the Corporation will, without limitation of the foregoing, in any event be deemed to Control any Person in which the Corporation beneficially owns (after giving effect to the applicable Disposition) (i) voting securities having 25% or more of the total voting power of the voting securities of such Person than outstanding, provided that, immediately after giving effect to such Disposition, no other Person that is not Controlled by the Corporation beneficially owns voting securities of such Person having voting power greater than the voting power of the voting securities beneficially owned by the Corporation or (ii) equity securities representing 50% or more of the common equity interest or economic equity interest in such Person.

"Convertible Securities" means (x) any securities of the Corporation (other than any series of Common Stock) or any Subsidiary thereof that are convertible into or exercisable or exchangeable for any shares of any series of Common Stock, whether upon conversion, exercise, exchange, pursuant to antidilution provisions of such securities or otherwise, and (y) any securities of any other Person that are convertible into or exercisable or exchangeable for, securities of such Person or any other Person, whether upon conversion, exercise, exchange, pursuant to antidilution provisions of such securities or otherwise.

"Corporation Earnings (Loss) Attributable to the Capital Group" for any period, means the net earnings or loss of the Capital Group for such period determined on a basis consistent with the determination of the net earnings or loss of the Capital Group for such period

"Corporation Earnings (Loss) Attributable to the Starz Group" for any period, means the net earnings or loss of the Starz Group for such period determined on a basis consistent with the determination of the net earnings or loss of the Starz Group for such period as presented in the reconciling schedules to the consolidated financial statements of the Corporation for such period, including income and expenses of the Corporation attributed to the operations of the Starz Group on a substantially consistent basis, including, without limitation, corporate administrative costs, net interest and income taxes.

"Determination Date" means the date designated by the Board of Directors for determination of any applicable Optional Conversion Ratio.

"Disposition" means the sale, transfer, exchange, assignment or other disposition (whether by merger, consolidation, sale or contribution of assets or stock or otherwise) of assets. The term "Disposition" does not include the consolidation or merger of the Corporation with or into any other Person or Persons or any other business combination involving the Corporation as a whole.

"Effective Date" means the date on which this Restated Certificate of Incorporation is filed with the Secretary of State of Delaware.

"Exempt Capital Group Disposition" means any of the following: (i) the Disposition of all or substantially all of the Corporation's assets in one transaction or a series of related transactions in connection with the liquidation, dissolution or winding up of the Corporation within the meaning of paragraph (g) of this Section A.2., (ii) a dividend, other distribution or redemption in accordance with any provision of paragraph (c), (d) or (e) of this Section A.2., (iii) a Capital Group Disposition to any Person that the Corporation, directly or indirectly, after giving effect to the Disposition, Controls, (iv) a Capital Group Disposition in connection with a Capital Group Related Business Transaction, or (v) a Capital Group Disposition as to which the Board of Directors obtains the requisite approval of the holders of Liberty Capital Voting Securities to classify such Capital Group Disposition as an Exempt Capital Group Disposition in accordance with paragraph (a)(iv).

"Exempt Starz Group Disposition" means any of the following: (i) the Disposition of all or substantially all of the Corporation's assets in one transaction or a series of related transactions in connection with the liquidation, dissolution or winding up of the Corporation within the meaning of paragraph (g) of this Section A.2., (ii) a dividend, other distribution or redemption in accordance with any provision of paragraph (c), (d) or (f) of this Section A.2., (iii) a Starz Group Disposition to any Person that the Corporation, directly or indirectly, after giving effect to the Disposition, Controls, (iv) a Starz Group Disposition in connection with a Starz Group Related Business Transaction, or (v) a Starz Group Disposition as

52

to which the Board of Directors obtains the requisite approval of the holders of Liberty Starz Voting Securities to classify such Starz Group Disposition as an Exempt Starz Group Disposition in accordance with paragraph (a)(iv).

"Fair Value" means, as of any date:

- (i) in the case of any equity security or debt security that is Publicly Traded, the Market Value thereof, as of such date;
- (ii) in the case of any equity security or debt security that is not Publicly Traded, the fair value per share of stock or per other unit of such security, on a fully distributed basis, as determined by an independent investment banking firm experienced in the valuation of securities selected in good faith by the Board of Directors, or, if no such investment banking firm is selected, as determined in the good faith judgment of the Board of Directors;
- (iii) in the case of cash denominated in U.S. dollars, the face amount thereof and in the case of cash denominated in other than U.S. dollars, the face amount thereof converted into U.S. dollars at the rate published in The Wall Street Journal on such date or, if not so published, at such rate as shall be determined in good faith by the Board of Directors based upon such information as the Board of Directors shall in good faith determine to be appropriate; and
- (iv) in the case of assets or property other than securities or cash, the "Fair Value" thereof shall be determined in good faith by the Board of Directors based upon such information (including, if deemed desirable by the Board of Directors, appraisals, valuation reports or opinions of experts) as the Board of Directors shall in good faith determine to be appropriate.

"Group" means the Capital Group or the Starz Group.

"Inter-Group Interest" means, as of any date and with respect to either Group, the proportionate undivided interest, if any, that such Group may be deemed to hold as of such date in the assets, liabilities and businesses of the other Group in accordance with this Certificate. An Inter-Group Interest in the Capital Group held by the Starz Group is expressed in terms of the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest in the Starz Group held by the Capital Group is expressed in terms of the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest.

"Liberty Capital Reference Share" means one share of Series A Liberty Capital Common Stock, unless (i) on any single Trading Day as of which a valuation determination is being made or on the first Trading Day of any Trading Day period with respect to which a valuation determination is being made, in each case, under this Section A.2., the number of shares outstanding of any other Publicly Traded series of Liberty Capital Common Stock exceeds the number of shares outstanding of the Series A Liberty Capital Common Stock, and (ii) the Board of Directors determines to base such valuation determination on such other Publicly Traded series of Liberty Capital Common Stock in lieu of basing it on one share of

53

Series A Liberty Capital Common Stock, in which case the term 'Liberty Capital Reference Share' will mean one share of such other Publicly Traded series of Liberty Capital Common Stock.

"Liberty Capital Voting Securities" means the Series A Liberty Capital Common Stock, the Series B Liberty Capital Common Stock and any series of Preferred Stock which by the terms of its Preferred Stock Designation is designated as a Liberty Capital Voting Security, provided, that each such series of Preferred Stock will be treated as a Liberty Capital Voting Securities only as and to the extent expressly provided for in the applicable Preferred Stock Designation.

"Liberty Starz Reference Share" means one share of Series A Liberty Starz Common Stock, unless (i) on any single Trading Day as of which a valuation determination is being made or on the first Trading Day of any Trading Day period with respect to which a valuation determination is being made, in each case, under this Section A.2., the number of shares outstanding of any other Publicly Traded series of Liberty Starz Common Stock exceeds the number of shares outstanding of the Series A Liberty Starz Common Stock, and (ii) the Board of Directors determines to base such valuation determination on such other Publicly Traded series of Liberty Starz Common Stock in lieu of basing it on one share of Series A Liberty Starz Common Stock, in which case the term "Liberty Starz Reference Share" will mean one share of such other Publicly Traded series of Liberty Starz Common Stock.

"Liberty Starz Voting Securities" means the Series A Liberty Starz Common Stock, the Series B Liberty Starz Common Stock and any series of Preferred Stock which by the terms of its Preferred Stock Designation is designated as a Liberty Starz Voting Security, provided, that each such series of Preferred Stock will

be treated as a Liberty Starz Voting Security and will be entitled to vote together with the other Liberty Starz Voting Securities only as and to the extent expressly provided for in the applicable Preferred Stock Designation.

"Market Value" of a share of any Publicly Traded stock on any Trading Day means the average of the high and low reported sales prices regular way of a share of such stock on such Trading Day, or in case no such reported sale takes place on such Trading Day the average of the reported closing bid and asked prices regular way of a share of such stock on such Trading Day, in either case on the New York Stock Exchange, or if the shares of such stock are not listed on the New York Stock Exchange on such Trading Day, on any tier of the Nasdaq Stock Market, or if the shares of such stock are not listed on any tier of the Nasdaq Stock Market on such Trading Day, the average of the closing bid and asked prices of a share of such stock in the over-the-counter market on such Trading Day as furnished by any New York Stock Exchange member firm selected from time to time by the Corporation, or if such closing bid and asked prices are not made available by any such New York Stock Exchange member firm on such Trading Day, the market value of a share of such stock as determined by the Board of Directors, provided that, for purposes of determining the Average Market Value for any period, (i) the "Market Value" of a share of stock on any day during such period prior to the "ex" date or any similar date for any dividend paid or to be paid with respect to such stock will be reduced by

54

the fair market value of the per share amount of such dividend as determined by the Board of Directors and (ii) the "Market Value" of a share of stock on any day during such period prior to (A) the effective date of any subdivision (by stock split or otherwise) or combination (by reverse stock split or otherwise) of outstanding shares of such stock or (B) the "ex" date or any similar date for any dividend with respect to any such stock in shares of such stock will be appropriately adjusted to reflect such subdivision, combination, dividend or distribution.

"Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest" will initially be zero, and will from time to time thereafter be (without duplication):

- (i) adjusted, if before such adjustment such number is greater than zero, as determined by the Board of Directors to be appropriate to reflect subdivisions (by stock split or otherwise) and combinations (by reverse stock split or otherwise) of the Liberty Capital Common Stock and dividends of shares of Liberty Capital Common Stock to holders of Liberty Capital Common Stock (and, to the extent the Capital Group Outstanding Interest Fraction is less than one (1) as of the record date for such dividend, the applicable treatment of such dividend, as determined by the Board of Directors, with respect to the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest) and other reclassifications of Liberty Capital Common Stock;
- (ii) decreased (but not below zero), if before such adjustment such number is greater than zero, by action of the Board of Directors (without duplication): (A) by a number equal to the aggregate number of shares of Liberty Capital Common Stock issued or sold by the Corporation, the proceeds of which are attributed to the Starz Group; (B) by a number equal to the aggregate number of shares of Liberty Capital Common Stock issued or delivered upon conversion, exercise or exchange of any Convertible Securities that the Board of Directors has determined are attributable to the Starz Group; (C) in accordance with the applicable provisions of paragraph (c) of this Section A.2.; (D) in the event the Board of Directors makes a Capital Group Inter-Group Redemption Election, by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest, as of the Capital Group Redemption Selection Date, by (y) the percentage of the Fair Value of the Capital Group that is represented by the Fair Value of the Corporation's equity interest in the applicable Distributed Capital Group Subsidiary which is attributable to the Capital Group, as determined by the Board of Directors under paragraph (e)(i) for purposes of such redemption; (E) in the event the Board of Directors makes a Capital Group Inter-Group Partial Redemption Election, by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying the Capital Group Inter-Group Redemption Amount by the amount (rounded, if necessary, to the nearest whole number) obtained by dividing the aggregate number of shares of Liberty Capital Common Stock redeemed pursuant to paragraph (e)(ii)(B)(II) of (e)(ii)(D), as applicable, of this Section A.2., by the applicable Capital Group Redemption Amount or the applicable portion of the Capital Group Allocable Net Proceeds applied to such redemption, respectively;

55

the determination to be made pursuant to this clause (F), of assets attributed to the Capital Group that are transferred or allocated from the Capital Group to the Starz Group in consideration of a reduction in the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest, by (y) the Fair Value of the Liberty Capital Reference Share as of the date of such transfer or allocation;

- (iii) increased, by action of the Board of Directors, (A) by a number equal to the aggregate number of shares of Liberty Capital Common Stock that are retired, redeemed or otherwise cease to be outstanding (x) following their purchase or redemption with funds or other assets attributed to the Starz Group, (y) following their retirement or redemption for no consideration if immediately prior thereto they were owned by an asset or business attributed to the Starz Group, or (z) following their conversion into shares of Liberty Starz Common Stock pursuant to clause (C) or (D) of paragraph (e)(ii) of this Section A.2.; (B) in accordance with the applicable provisions of paragraph (c) of this Section A.2.; and (C) by a number equal to, as applicable, the amount (rounded, if necessary, to the nearest whole number) obtained by dividing (I) the Fair Value, as of a date within 90 days of the determination to be made pursuant to this clause (C), of assets theretofore attributed to the Starz Group that are contributed to the Capital Group in consideration of an increase in the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest, by (II) the Fair Value of the Liberty Capital Reference Share as of the date of such contribution; and
- (iv) increased or decreased under such other circumstances as the Board of Directors determines to be appropriate or required by the other terms of this Section A.2. to reflect the economic substance of any other event or circumstance, provided that in each case, the adjustment will be made in a manner that is fair and equitable to holders of all series of Common Stock and intended to reflect the relative economic interest of the Starz Group in the Capital Group.

Whenever a change in the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest occurs, the Corporation will promptly thereafter prepare and file a statement of such change, and the amount to be allocated to the Starz Group with the Secretary of the Corporation. Neither the failure to prepare nor the failure to file any such statement will affect the validity of such change.

"Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest" will initially be zero, and will from time to time thereafter be (without duplication):

(i) adjusted, if before such adjustment such number is greater than zero, as determined by the Board of Directors to be appropriate to reflect subdivisions (by stock split or otherwise) and combinations (by reverse stock split or otherwise) of the Liberty Starz Common Stock and dividends of shares of Liberty Starz Common Stock to holders of Liberty Starz Common Stock (and, to the extent the Starz Group Outstanding Interest Fraction is less than one (1) as of the record date for such dividend, the applicable treatment of such dividend, as determined by the Board of Directors, with respect to the Number of Shares Issuable to the

Capital Group with Respect to the Starz Group Inter-Group Interest) and other reclassifications of Liberty Starz Common Stock;

- decreased (but not below zero), if before such adjustment such number is greater than zero, by action of the Board of Directors (without duplication): (A) by a number equal to the aggregate number of shares of Liberty Starz Common Stock issued or sold by the Corporation, the proceeds of which are attributed to the Capital Group; (B) by a number equal to the aggregate number of shares of Liberty Starz Common Stock issued or delivered upon conversion, exercise or exchange of any Convertible Securities that the Board of Directors has determined are attributable to the Capital Group; (C) in accordance with the applicable provisions of paragraph (c) of this Section A.2.; (D) in the event the Board of Directors makes a Starz Group Inter-Group Redemption Election, by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest, as of the Starz Group Redemption Selection Date, by (y) the percentage of the Fair Value of the Starz Group that is represented by the Fair Value of the Corporation's equity interest in the applicable Distributed Starz Group Subsidiary which is attributable to the Starz Group, as determined by the Board of Directors under paragraph (f)(i) for purposes of such redemption; (E) in the event the Board of Directors makes a Starz Group Inter-Group Partial Redemption Election, by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying the Starz Group Inter-Group Redemption Amount by the amount (rounded, if necessary, to the nearest whole number) obtained by dividing the aggregate number of shares of Liberty Starz Common Stock redeemed pursuant to paragraph (f)(ii)(B)(II) or (f)(ii)(D), as applicable, of this Section A.2., by the applicable Starz Group Redemption Amount or the applicable portion of the Starz Group Allocable Net Proceeds applied to such redemption, respectively; and (F) by a number equal to the amount (rounded, if necessary, to the nearest whole number) obtained by dividing (x) the aggregate Fair Value, as of a date within 90 days of the determination to be made pursuant to this clause (F), of assets attributed to the Starz Group that are transferred or allocated from the Starz Group to the Capital Group in consideration of a reduction in the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest, by (y) the Fair Value of the Liberty Starz Reference Share as of the date of such transfer or allocation;
- (iii) increased, by action of the Board of Directors, (A) by a number equal to the aggregate number of shares of Liberty Starz Common Stock that are retired, redeemed or otherwise cease to be outstanding (x) following their purchase or redemption with funds or other assets attributed to the Capital Group, (y) following their retirement or redemption for no consideration if immediately prior thereto, they were owned by an asset or business attributed to the Capital Group, or (z) following their conversion into shares of Liberty Capital Common Stock pursuant to clause (C) or (D) of paragraph (f)(ii) of this Section A.2.; (B) in accordance with the applicable provisions of paragraph (c) of this Section A.2.; and (C) by a number equal to, as applicable, the amount (rounded, if necessary, to the nearest whole number) obtained by dividing (I) the Fair Value, as of a date within 90 days of the determination to be made pursuant to this clause (C), of assets theretofore attributed to the Capital Group that are contributed to the Starz Group in consideration of an increase in the Number of Shares Issuable to the Capital

57

Group with Respect to the Starz Group Inter-Group Interest, by (II) the Fair Value of the Liberty Starz Reference Share as of the date of such contribution; and

(iv) increased or decreased under such other circumstances as the Board of Directors determines to be appropriate or required by the other terms of this Section A.2. to reflect the economic substance of any other event or circumstance, provided that in each case, the adjustment will be made in a manner that is fair and equitable to holders of all series of Common Stock and intended to reflect the relative economic interest of the Capital Group in the Starz Group.

Whenever a change in the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest occurs, the Corporation will promptly thereafter prepare and file a statement of such change and the amount to be allocated to the Capital Group with the Secretary of the Corporation. Neither the failure to prepare nor the failure to file any such statement will affect the validity of such change.

"Optional Conversion Ratio" means the applicable of the Starz/Capital Group Optional Conversion Ratio and the Capital/Starz Group Optional Conversion Ratio.

"outstanding", when used with respect to the shares of any series of Common Stock, will include, without limitation, the shares of such series, if any, held by any Subsidiary of the Corporation, except as otherwise provided by applicable law with respect to the exercise of voting rights. No shares of any series of Common Stock (or Convertible Securities that are convertible into or exercisable or exchangeable for Common Stock) held by the Corporation in its treasury will be deemed outstanding, nor will any shares be deemed outstanding which are attributable to the Number of Shares Issuable to the Starz Group with Respect to the Capital Group Inter-Group Interest or the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Interest.

"Person" means a natural person, corporation, limited liability company, partnership, joint venture, trust, unincorporated association or other legal entity.

"Publicly Traded" means, with respect to shares of capital stock or other securities, that such shares or other securities are traded on a U.S. securities exchange or quoted on the over-the-counter market.

"Share Distribution" means a dividend payable in shares of any class or series of capital stock, Convertible Securities or other equity securities of the Corporation or any other Person.

"Starz Group" means, as of any date:

(i) the direct and indirect interest of the Corporation, as of the Effective Date, in Starz Entertainment, LLC, Starz Media, LLC, Liberty Sports Interactive, Inc. and each of their Subsidiaries (including any successor to Starz Entertainment, LLC, Liberty Sports Interactive, Inc. or any such Subsidiary by merger, consolidation or sale of all or substantially all of its

58

assets, whether or not in connection with a Starz Group Related Business Transaction) and their respective assets, liabilities and businesses;

- (ii) all other assets, liabilities and businesses of the Corporation or any of its Subsidiaries to the extent attributed to the Starz Group as of the Effective Date;
- (iii) all assets, liabilities and businesses acquired or assumed by the Corporation or any of its Subsidiaries for the account of the Starz Group, or contributed, allocated or transferred to the Starz Group (including the net proceeds of any issuances, sales or incurrences for the account of the Starz Group of shares of Liberty Starz Common Stock, Convertible Securities convertible into or exercisable or exchangeable for shares of Liberty Starz Common Stock, or indebtedness or Preferred Stock attributed to the Starz Group), in each case, after the Effective Date and as determined by the Board of Directors;
  - (iv) the proceeds of any Disposition of any of the foregoing; and
  - (v) an Inter-Group Interest in the Capital Group equal to one (1) minus the Capital Group Outstanding Interest Fraction as of such date;

provided that the Starz Group will not include (A) any assets, liabilities or businesses disposed of after the Effective Date, including, without limitation, by dividend, to holders of Liberty Starz Common Stock or in redemption of shares of Liberty Starz Common Stock, from and after the date of such Disposition or (B) any assets, liabilities or

businesses transferred or allocated after the Effective Date from the Starz Group to the Capital Group (other than through the Starz Group's Inter-Group Interest in the Capital Group, if any, pursuant to clause (v) above), including, without limitation, any Starz Group Inter-Group Dividend Amount or Starz Group Inter-Group Redemption Amount, from and after the date of such transfer or allocation.

"Starz Group Allocable Net Proceeds" means, with respect to any Starz Group Disposition, (i) if at the time of such Starz Group Disposition, the Starz Group Outstanding Interest Fraction is one (1), the Starz Group Net Proceeds of such Starz Group Disposition, or (ii) if at the time of such Starz Group Disposition the Starz Group Outstanding Interest Fraction is less than one (1), the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Starz Group Net Proceeds of such Starz Group Disposition, by (y) the Starz Group Outstanding Interest Fraction as of such date.

"Starz Group Available Dividend Amount," as of any date, means an amount equal to the amount (rounded, if necessary, to the nearest whole number) obtained by multiplying (x) the Starz Group Outstanding Interest Fraction, by (y) either: (i) the excess of (A) an amount equal to the total assets of the Starz Group less the total liabilities (not including Preferred Stock attributed to the Starz Group) of the Starz Group as of such date over (B) the aggregate par value of, or any greater amount determined to be capital in respect of, all outstanding shares of Liberty Starz Common Stock and each series of Preferred Stock attributed to the Starz Group or (ii) in case there is no such excess, an amount equal to the Corporation Earnings (Loss) Attributable to the Starz Group (if positive) for the fiscal year in which such date occurs and/or the preceding fiscal year.

59

"Starz Group Conversion Date" means any date and time fixed by the Board of Directors for a conversion of shares of Liberty Starz Common Stock pursuant to this Section A.2.

"Starz Group Conversion Selection Date" means any date and time fixed by the Board of Directors as the date and time upon which shares to be converted of each series of Liberty Starz Common Stock will be selected for conversion pursuant to this Section A.2. (which, for the avoidance of doubt, may be the same date and time as the Starz Group Conversion Date).

"Starz Group Disposition" means the Disposition, in one transaction or a series of related transactions, by the Corporation and its Subsidiaries of all or substantially all of the assets of the Starz Group to one or more Persons.

"Starz Group Net Proceeds" means, as of any date, with respect to any Starz Group Disposition, an amount, if any, equal to the Fair Value of what remains of the gross proceeds of such Disposition to the Corporation after any payment of, or reasonable provision for, (i) any taxes payable by the Corporation or any of its Subsidiaries in respect of such Disposition or in respect of any resulting dividend or redemption pursuant to clause (A), (B) or (D) of paragraph (f)(ii) of this Section A.2. (or that would have been payable but for the utilization of tax benefits attributable to the Capital Group), (ii) any transaction costs, including, without limitation, any legal, investment banking and accounting fees and expenses and (iii) any liabilities and other obligations (contingent or otherwise) of, or attributed to, the Starz Group, including, without limitation, any liabilities for deferred taxes, any indemnity or guarantee obligations incurred in connection with the Disposition or any liabilities for future purchase price adjustments and any preferential amounts plus any accumulated and unpaid dividends and other obligations in respect of Preferred Stock attributed to the Starz Group. For purposes of this definition, any assets of the Starz Group remaining after such Disposition will constitute "reasonable provision" for such amount of taxes, costs, liabilities and other obligations (contingent or otherwise) as can be supported by such assets.

"Starz Group Outstanding Interest Fraction," as of any date, means a fraction the numerator of which is the aggregate number of shares of Liberty Starz Common Stock outstanding on such date and the denominator of which is the amount obtained by adding (i) such aggregate number of shares of Liberty Starz Common Stock outstanding on such date, plus (ii) the Number of Shares Issuable to the Capital Group with Respect to the Starz Group Inter-Group Interest as of such date, provided that such fraction will in no event be greater than one. If the holders of any Convertible Securities that are convertible into or exercisable or exchangeable for shares of Liberty Starz Common Stock are entitled to participate in any dividend (for purposes of paragraphs (c)(ii), (d)(ii) or (f)(ii) of this Section A.2.) or redemption (for purposes of paragraph (f) of this Section A.2.) with respect to the Liberty Starz Common Stock (other than by means of an antidilution adjustment), such shares so issuable upon conversion, exercise or exchange will be taken into account in calculating the Starz Group Outstanding Interest Fraction and any related calculations under the applicable provisions of this Section A.2. in such manner as the Board of Directors determines to be appropriate.

60

"Starz Group Redemption Date" means any date and time fixed by the Board of Directors for a redemption of shares of Liberty Starz Common Stock pursuant to this Section A.2.

"Starz Group Redemption Selection Date" means the date and time fixed by the Board of Directors on which shares of Liberty Starz Common Stock are to be selected for redemption pursuant to this Section A.2. (which, for the avoidance of doubt, may be the same date and time as the Starz Group Redemption Date).

"Starz Group Related Business Transaction" means any Disposition of all or substantially all of the assets of the Starz Group in which the Corporation receives as proceeds of such Disposition primarily equity securities (including, without limitation, capital stock, securities convertible into capital stock or other equity securities, partnership, limited partnership or limited liability company interests and other types of equity securities, without regard to the voting power or contractual or other management or governance rights related to such equity securities) of the purchaser or acquiror of such assets of the Starz Group, any entity which succeeds (by merger, formation of a joint venture enterprise or otherwise) to such assets of the Starz Group, or a third party issuer, if a significant portion of the businesses in which such purchaser, acquiror or third party issuer is engaged or proposes to engage consists of one or more businesses similar or complementary to the businesses attributable to the Starz Group prior to such Disposition, as determined in good faith by the Board of Directors.

"Starz Group Share Distribution Ratio" means, as to any Share Distribution consisting of shares of Starz Group Common Stock, the number of shares (including any fraction of a share) of Starz Group Common Stock issuable to a holder for each outstanding share of the applicable series of Common Stock owned by such holder as of the record date for such Share Distribution (rounded, if necessary, to the nearest five decimal places).

"Subsidiary," when used with respect to any Person, means (i)(A) a corporation of which a majority in voting power of its share capital or capital stock with voting power, under ordinary circumstances, to elect directors is at the time, directly or indirectly, owned by such Person, by a Subsidiary of such Person, or by such Person and one or more Subsidiaries of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, (B) a partnership or limited liability company in which such Person or a Subsidiary of such Person is, at the date of determination, (1) in the case of a partnership, a general partner of such partnership with the power affirmatively to direct the policies and management of such partnership or (2) in the case of a limited liability company, the managing member or, in the absence of a managing member, a member with the power affirmatively to direct the policies and management of such limited liability company, or (C) any other Person (other than a corporation) in which such Person, a Subsidiary of such Person and one or more Subsidiaries of such Person, directly or indirectly, at the date of determination thereof, has (1) the power to elect or direct the election of a majority of the members of the governing body of such Person, whether or not such power is subject to a voting agreement or similar Encumbrance, or (2) in the absence of such a governing body, at least a majority ownership interest or (ii) any other Person of which an aggregate of more than 50% of the equity interests

are, at the time, directly or indirectly, owned by such Person and/or one or more Subsidiaries of such Person.

"Trading Day" means each day on which the relevant share or security is traded on the New York Stock Exchange or the Nasdaq Stock Market or quoted on the over-the-counter market.

"Voting Securities" means the Liberty Capital Voting Securities, the Liberty Starz Voting Securities and any series of Preferred Stock which by the terms of its Preferred Stock Designation is designated as a Voting Security, provided that each such series of Preferred Stock will be entitled to vote together with the other Voting Securities only as and to the extent expressly provided for in the applicable Preferred Stock Designation.

The following terms have the meanings ascribed thereto in the sections set forth opposite such terms:

Additional Defined Terms	Section
Capital Group Distribution Subsidiary Securities	Article IV, Section A.2(e)(i)
Capital Group Inter-Group Dividend	Article IV, Section A.2(c)(i)(A)
Capital Group Inter-Group Dividend Amount	Article IV, Section A.2(c)(i)(A)
Capital Group Inter-Group Interest Subsidiary Securities	Article IV, Section A.2(e)(i)
Capital Group Inter-Group Partial Redemption Election	Article IV, Section A.2(e)(ii)
Capital Group Inter-Group Redemption Amount	Article IV, Section A.2(e)(ii)
Capital Group Inter-Group Redemption Election	Article IV, Section A.2(e)(i)
Capital Group Redemption Amount	Article IV, Section A.2(e)(ii)(B)(II)
Capital Group Redemption Shares	Article IV, Section A.2(e)(i)
Capital Group Redemption Stockholder Approval	Article IV, Section A.2(a)(v)(A)
Capital/Starz Group Optional Conversion Ratio	Article IV, Section A.2(b)(iii)(B)
Common Stock	Article IV(a)
Corporation	Article I
DGCL	Article III
Distributable Capital Group Subsidiary Securities	Article IV, Section A.2(e)(i)
Distributed Capital Group Subsidiary	Article IV, Section A.2(e)(i)
Distributed Starz Group Subsidiary	Article IV, Section A.2(f)(i)
Distributable Starz Group Subsidiary Securities	Article IV, Section A.2(f)(i)
62	

Additional Defined Terms	Section
Effective Time	Article IV
Liberty Capital Common Stock	Article IV, Section A.1
Liberty Starz Common Stock	Article IV, Section A.1
Preferred Stock	Article IV(b)
Preferred Stock Designation	Article IV, Section B
proceeding	Article V, Section E.2(a)
Series A Liberty Capital Common Stock	Article IV, Section A.1
Series A Liberty Starz Common Stock	Article IV, Section A.1
Series B Liberty Capital Common Stock	Article IV, Section A.1
Series B Liberty Starz Common Stock	Article IV, Section A.1
Series C Liberty Capital Common Stock	Article IV, Section A.1
Series C Liberty Starz Common Stock	Article IV, Section A.1
Starz Group Distribution Subsidiary Securities	Article IV, Section A.2(f)(i)
Starz/Capital Group Optional Conversion Ratio	Article IV, Section A.2(b)(iii)(B)
Starz Group Inter-Group Dividend	Article IV, Section A.2(c)(ii)(A)
Starz Group Inter-Group Dividend Amount	Article IV, Section A.2(c)(ii)(A)
Starz Group Inter-Group Interest Subsidiary Securities	Article IV, Section A.2(f)(i)
Starz Group Inter-Group Partial Redemption Election	Article IV, Section A.2(f)(ii)
Starz Group Inter-Group Redemption Amount	Article IV, Section A.2(f)(ii)
Starz Group Inter-Group Redemption Election	Article IV, Section A.2(f)(i)
Starz Group Redemption Amount	Article IV, Section A.2(f)(ii)(B)(II)
Starz Group Redemption Shares	Article IV, Section A.2(f)(i)
Starz Group Redemption Stockholder Approval	Article IV, Section A.2(a)(v)(B)
substantially all of the assets of the Capital Group	Article IV, Section A.2(e)(ii)
substantially all of the assets of the Starz Group	Article IV, Section A.2(f)(ii)

(j) <u>Reclassification</u>. The Corporation will not reclassify, subdivide or combine one series of Liberty Capital Common Stock without reclassifying, subdividing or combining each other series of Liberty Capital Common Stock on an equal per share basis. The Corporation will

63

not reclassify, subdivide or combine one series of Liberty Starz Common Stock without reclassifying, subdividing or combining each other series of Liberty Starz Common Stock on an equal per share basis.

(k) Transfer Taxes. The Corporation will pay any and all documentary, stamp or similar issue or transfer taxes that may be payable in respect of the issue or delivery of a certificate or certificates representing any shares of capital stock and/or other securities on conversion or redemption of shares of Common Stock pursuant to this Section A.2. The Corporation will not, however, be required to pay any tax that may be payable in respect of any issue or delivery of a certificate or certificates representing any shares of capital stock in a name other than that in which the shares of Common Stock so converted or redeemed were registered and no such issue or delivery will be made unless and until the Person requesting the same has paid to the Corporation or its transfer agent the amount of any such tax, or has established to the satisfaction of the Corporation or its transfer agent that such tax has been paid.

#### PREFERRED STOCK

The Preferred Stock may be divided and issued in one or more series from time to time, with such powers, designations, preferences and relative, participating, optional or other rights and qualifications, limitations or restrictions thereof, as will be stated and expressed in a resolution or resolutions providing for the issue of each such series adopted by the Board of Directors (a "**Preferred Stock Designation**"). The Board of Directors, in the Preferred Stock Designation with respect to a series of Preferred Stock (a copy of which will be filed as required by law), will, without limitation of the foregoing, fix the following with respect to such series of Preferred Stock:

- (i) the distinctive serial designations and the number of authorized shares of such series, which may be increased or decreased, but not below the number of shares thereof then outstanding, by a certificate made, signed and filed as required by law (except where otherwise provided in a Preferred Stock Designation);
- (ii) the dividend rate or amounts, if any, for such series, the date or dates from which dividends on all shares of such series will be cumulative, if dividends on stock of such series will be cumulative, and the relative preferences or rights of priority, if any, or participation, if any, with respect to payment of dividends on shares of such series;
- (iii) the rights of the shares of such series in the event of any voluntary or involuntary liquidation, dissolution or winding up of the Corporation, if any, and the relative preferences or rights of priority, if any, of payment of shares of such series;
- (iv) the right, if any, of the holders of such series to convert or exchange such stock into or for other classes or series of a class of stock or indebtedness of the Corporation or of another Person, and the terms and conditions of such conversion or exchange, including

64

provision for the adjustment of the conversion or exchange rate in such events as the Board of Directors may determine;

- (v) the voting powers, if any, of the holders of such series, including whether such series will be designated as a Capital Group Voting Security, a Starz Group Voting Security and/or a Voting Security and, if so designated, the terms and conditions on which such series may vote together with the holders of any other class or series of capital stock of the Corporation;
  - (vi) the terms and conditions, if any, for the Corporation to purchase or redeem shares of such series; and
  - (vii) any other relative rights, powers, preferences and limitations, if any, of such series.

The Board of Directors is hereby expressly authorized to exercise its authority with respect to fixing and designating various series of the Preferred Stock and determining the relative rights, powers and preferences, if any, thereof to the full extent permitted by applicable law, subject to any stockholder vote that may be required by this Certificate. All shares of any one series of the Preferred Stock will be alike in every particular. Except to the extent otherwise expressly provided in the Preferred Stock Designation for a series of Preferred Stock, the holders of shares of such series will have no voting rights except as may be required by the laws of the State of Delaware. Further, unless otherwise expressly provided in the Preferred Stock Designation for a series of Preferred Stock, no consent or vote of the holders of shares of Preferred Stock or any series thereof will be required for any amendment to this Certificate that would increase the number of authorized shares of Preferred Stock or the number of authorized shares of any series thereof or decrease the number of authorized shares of Preferred Stock or the number of authorized shares of Preferred Stock or such series, as the case may be, then outstanding).

Except as may be provided by the Board of Directors in a Preferred Stock Designation or by law, shares of any series of Preferred Stock that have been redeemed (whether through the operation of a sinking fund or otherwise) or purchased by the Corporation, or which, if convertible or exchangeable, have been converted into or exchanged for shares of stock of any other class or classes will have the status of authorized and unissued shares of Preferred Stock and may be reissued as a part of the series of which they were originally a part or may be reissued as part of a new series of Preferred Stock to be created by a Preferred Stock Designation or as part of any other series of Preferred Stock.

65

### ARTICLE V

# DIRECTORS

# SECTION A

# NUMBER OF DIRECTORS

The governing body of the Corporation will be a Board of Directors. Subject to any rights of the holders of any series of Preferred Stock to elect additional directors, the number of directors will not be less than three (3) and the exact number of directors will be fixed by the Board of Directors by resolution. Election of directors need not be by written ballot.

### SECTION B

# CLASSIFICATION OF THE BOARD

Except as otherwise fixed by or pursuant to the provisions of Article IV hereof relating to the rights of the holders of any series of Preferred Stock to separately elect additional directors, which additional directors are not required to be classified pursuant to the terms of such series of Preferred Stock, the Board of Directors will be divided into three classes: Class I, Class II and Class III. Each class will consist, as nearly as possible, of a number of directors equal to one-third (1/3) of the number of members of the Board of Directors authorized as provided in Section A of this Article V. The term of office of the initial Class I directors will expire at the annual meeting of stockholders in 2012; and the term of office of the initial Class III directors will expire at the annual meeting of stockholders in 2012; and the term of office of the initial Class III directors will expire at the annual meeting of stockholders of the Corporation the successors of that class of directors whose term expires at that meeting will be elected to hold office for a term expiring at the annual meeting of stockholders held in the third year following the year of their election. The directors of each class will hold office until their respective successors are elected and qualified or until such director's earlier death, resignation or removal.

# SECTION C

# REMOVAL OF DIRECTORS

Subject to the rights of the holders of any series of Preferred Stock, directors may be removed from office only for cause upon the affirmative vote of the holders of at least a majority of the total voting power of the then outstanding Voting Securities entitled to vote thereon, voting together as a single class.

### SECTION D

#### NEWLY CREATED DIRECTORSHIPS AND VACANCIES

Subject to the rights of holders of any series of Preferred Stock, vacancies on the Board of Directors resulting from death, resignation, removal, disqualification or other cause, and newly created directorships resulting from any increase in the number of directors on the Board of Directors, will be filled only by the affirmative vote of a majority of the remaining directors then in office (even though less than a quorum) or by the sole remaining director. Any director elected in accordance with the preceding sentence will hold office for the remainder of the full term of the class of directors in which the vacancy occurred or to which the new directorship is apportioned, and until such director's successor will have been elected and qualified or until such director's earlier death, resignation or removal. No decrease in the number of directors constituting the Board of Directors will shorten the term of any incumbent director, except as may be provided with respect to any additional director elected by the holders of the applicable series of Preferred Stock.

#### SECTION E

#### LIMITATION ON LIABILITY AND INDEMNIFICATION

1. **Limitation On Liability.** To the fullest extent permitted by the DGCL as the same exists or may hereafter be amended, a director of the Corporation will not be liable to the Corporation or any of its stockholders for monetary damages for breach of fiduciary duty as a director. Any repeal or modification of this paragraph 1 will be prospective only and will not adversely affect any limitation, right or protection of a director of the Corporation existing at the time of such repeal or modification.

#### 2. Indemnification.

(a) Right to Indemnification. The Corporation will indemnify, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any person who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "proceeding") by reason of the fact that he, or a person for whom he is the legal representative, is or was a director or officer of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees) incurred by such person. Such right of indemnification will inure whether or not the claim asserted is based on matters which antedate the adoption of this Section E. The Corporation will be required to indemnify or make advances to a person in connection with a proceeding (or part thereof) initiated by such person only if the proceeding (or part thereof) was authorized by the Board of Directors.

67

- (b) <u>Prepayment of Expenses</u>. The Corporation will pay the expenses (including attorneys' fees) incurred by a director or officer in defending any proceeding in advance of its final disposition, <u>provided</u>, <u>however</u>, that the payment of expenses incurred by a director or officer in advance of the final disposition of the proceeding will be made only upon receipt of an undertaking by the director or officer to repay all amounts advanced if it should be ultimately determined that the director or officer is not entitled to be indemnified under this paragraph or otherwise.
- (c) <u>Claims</u>. If a claim for indemnification or payment of expenses under this paragraph is not paid in full within 60 days after a written claim therefor has been received by the Corporation, the claimant may file suit to recover the unpaid amount of such claim and, if successful, will be entitled to be paid the expense (including attorney's fees) of prosecuting such claim to the fullest extent permitted by Delaware law. In any such action the Corporation will have the burden of proving that the claimant was not entitled to the requested indemnification or payment of expenses under applicable law.
- (d) Non-Exclusivity of Rights. The rights conferred on any person by this paragraph will not be exclusive of any other rights which such person may have or hereafter acquire under any statute, provision of this Certificate, the Bylaws, agreement, vote of stockholders or resolution of disinterested directors or otherwise.
- (e) Other Indemnification. The Corporation's obligation, if any, to indemnify any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or nonprofit entity will be reduced by any amount such person may collect as indemnification from such other corporation, partnership, joint venture, trust, enterprise or nonprofit entity.
- 3. Amendment or Repeal. Any amendment, modification or repeal of the foregoing provisions of this Section E will not adversely affect any right or protection hereunder of any person in respect of any act or omission occurring prior to the time of such amendment, modification or repeal.

# SECTION F

### AMENDMENT OF BYLAWS

In furtherance and not in limitation of the powers conferred by the DGCL, the Board of Directors, by action taken by the affirmative vote of not less than 75% of the members of the Board of Directors then in office, is hereby expressly authorized and empowered to adopt, amend or repeal any provision of the Bylaws of this Corporation.

### ARTICLE VI

TERM

The term of existence of this Corporation shall be perpetual.

68

## ARTICLE VII

#### STOCK NOT ASSESSABLE

for the debts, obligations or liabilities of this Corporation. This Certificate shall not be subject to amendment in this respect.

#### ARTICLE VIII

### MEETINGS OF STOCKHOLDERS

#### SECTION A

## ANNUAL AND SPECIAL MEETINGS

Subject to the rights of the holders of any series of Preferred Stock, stockholder action may be taken only at an annual or special meeting. Except as otherwise provided in a Preferred Stock Designation with respect to any series of Preferred Stock or unless otherwise prescribed by law or by another provision of this Certificate, special meetings of the stockholders of the Corporation, for any purpose or purposes, will be called by the Secretary of the Corporation (i) upon the written request of the holders of not less than  $66^2/_3\%$  of the total voting power of the then outstanding Voting Securities entitled to vote thereon or (ii) at the request of at least 75% of the members of the Board of Directors then in office.

#### SECTION B

### ACTION WITHOUT A MEETING

No action required to be taken or which may be taken at any annual meeting or special meeting of stockholders may be taken without a meeting, and the power of stockholders to consent in writing, without a meeting, to the taking of any action is specifically denied; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, holders of any series of Preferred Stock may take action by written consent to the extent provided in a Preferred Stock Designation with respect to such series.

#### ARTICLE IX

### ACTIONS REQUIRING SUPERMAJORITY STOCKHOLDER VOTE

Subject to the rights of the holders of any series of Preferred Stock, the affirmative vote of the holders of at least 66/3% of the total voting power of the then outstanding Voting Securities entitled to vote thereon, voting together as a single class at a meeting

69

specifically called for such purpose, will be required in order for the Corporation to take any action to authorize:

- (i) the amendment, alteration or repeal of any provision of this Certificate or the addition or insertion of other provisions herein; provided, however, that this clause (i) will not apply to any such amendment, alteration, repeal, addition or insertion (A) as to which the laws of the State of Delaware, as then in effect, do not require the consent of this Corporation's stockholders, or (B) that at least 75% of the members of the Board of Directors then in office have approved;
- (ii) the adoption, amendment or repeal of any provision of the Bylaws of the Corporation; provided, however, that this clause (ii) will not apply to, and no vote of the stockholders of the Corporation will be required to authorize, the adoption, amendment or repeal of any provision of the Bylaws of the Corporation by the Board of Directors in accordance with the power conferred upon it pursuant to Section F of Article V of this Certificate;
- (iii) the merger or consolidation of this Corporation with or into any other corporation; provided, however, that this clause (iii) will not apply to any such merger or consolidation (A) as to which the laws of the State of Delaware, as then in effect, do not require the consent of this Corporation's stockholders, or (B) that at least 75% of the members of the Board of Directors then in office have approved;
- (iv) the sale, lease or exchange of all, or substantially all, of the property or assets of the Corporation<u>provided, however</u>, that this clause (iv) will not apply to any such sale, lease or exchange that at least 75% of the members of the Board of Directors then in office have approved; or
- (v) the dissolution of the Corporation; <u>provided</u>, <u>however</u>, that this clause (v) will not apply to such dissolution if at least 75% of the members of the Board of Directors then in office have approved such dissolution.

Nothing contained in Section A.2 of this Certificate shall in any way limit, modify or otherwise affect any voting requirement set forth in this Article IX. Any stockholder approval required pursuant to this Article IX or the DGCL will be in addition to, and not in lieu of, any approval of the holders of Liberty Capital Common Stock or Liberty Starz Common Stock required pursuant to Section A.2. of this Certificate.

All rights at any time conferred upon the stockholders of the Corporation, pursuant to this Certificate are granted subject to the provisions of this Article IX.

70

IN WITNESS WHEREOF, the undersigned has executed this Restated Certificate of Incorporation this 23<sup>rd</sup> day of September, 2011.

# LIBERTY MEDIA CORPORATION

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe

Title: Executive Vice President and General Counsel

### CERTIFICATE OF AMENDMENT TO THE CERTIFICATE OF INCORPORATION OF LIBERTY CAPSTARZ, INC.

The undersigned, in accordance with the applicable provisions of Sections 228 and 242 of the Delaware General Corporation Law, does hereby certify as follows:

- 1. The name of the corporation is Liberty CapStarz, Inc. (the "Corporation").
- 2. The Certificate of Incorporation, as amended, of the Corporation is hereby amended by striking out Article FIRST thereof and substituting in lieu of said Article the following new Article:

"FIRST. The name of the corporation is Liberty Media Corporation (the "Corporation")."

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Amendment to be signed by the undersigned this 22nd day of September, 2011.

## LIBERTY CAPSTARZ, INC.

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe Title: Executive Vice President

#### LIBERTY MEDIA CORPORATION

A Delaware Corporation

BYLAWS

### ARTICLE I

#### STOCKHOLDERS

#### Section 1.1 <u>Annual Meeting</u>.

An annual meeting of stockholders for the purpose of electing directors and of transacting any other business properly brought before the meeting pursuant to these Bylaws shall be held each year at such date, time and place, either within or without the State of Delaware or, if so determined by the Board of Directors in its sole discretion, at no place (but rather by means of remote communication), as may be specified by the Board of Directors in the notice of meeting.

#### Section 1.2 Special Meetings.

Except as otherwise provided in the terms of any series of preferred stock or unless otherwise provided by law or by the Certificate of Incorporation, special meetings of stockholders of the Corporation, for the transaction of such business as may properly come before the meeting, may be called by the Secretary of the Corporation only (i) upon written request made in compliance with the advance notice requirements of these Bylaws of the holders of not less than 80% of the total voting power of the outstanding capital stock of the Corporation entitled to vote at such meeting or (ii) at the request of not less than 75% of the members of the Board of Directors then in office. Only such business may be transacted as is specified in the

notice of the special meeting. The Board of Directors shall have the sole power to determine the time, date and place, either within or without the State of Delaware, for any special meeting of stockholders. Following such determination, it shall be the duty of the Secretary to cause notice to be given to the stockholders entitled to vote at such meeting that a meeting will be held at the time, date and place and in accordance with the record date determined by the Board of Directors.

#### Section 1.3 Record Date.

In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date: (i) in the case of determination of stockholders entitled to vote at any meeting of stockholders or adjournment thereof, shall, unless otherwise required by the laws of the State of Delaware, not be more than sixty (60) nor less than ten (10) days before the date of such meeting, and (ii) in the case of any other lawful action, shall not be more than sixty (60) days prior to such other action. If no record date is fixed by the Board of Directors: (i) the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held, and (ii) the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto. A determination of stockholders of record entitled to notice of or to

2

vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

#### Section 1.4 <u>Notice of Meetings.</u>

Notice of all stockholders meetings, stating the place, if any, date and hour thereof; the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting; the place within the city, other municipality or community or electronic network at which the list of stockholders may be examined; and, in the case of a special meeting, the purpose or purposes for which the meeting is called, shall be delivered in accordance with applicable law and applicable stock exchange rules and regulations by the Chairman of the Board, the President, any Vice President, the Secretary or an Assistant Secretary, to each stockholder entitled to vote thereat at least ten (10) days but not more than sixty (60) days before the date of the meeting, unless a different period is prescribed by law, or the lapse of the prescribed period of time shall have been waived.

#### Section 1.5 Notice of Stockholder Business.

## (a) <u>Annual Meetings of Stockholders.</u>

(1) At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly brought before an annual meeting, nominations for persons for election to the Board of Directors and the proposal of business to be considered by the stockholders must be (i) specified in the notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors (or any duly authorized committee thereof), (ii) otherwise properly brought before the meeting by or at the direction of the Board of Directors (or any duly authorized committee thereof), or (iii)

3

otherwise properly be requested to be brought before the meeting by a stockholder in compliance with the procedures set forth in this Section 1.5.

(2) In addition to any other requirements under applicable law and the Corporation's Certificate of Incorporation, for a nomination for election to the Board of Directors or the proposal of business to be properly requested to be brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in proper written form to the Secretary of the Corporation and any such proposed business, other than the nominations of persons for election to the Board of

Directors, must constitute a proper matter for stockholder action. To be timely, a stockholder's notice must be delivered to or mailed and received at the principal executive offices of the Corporation (x) in the case of an annual meeting that is called for a date that is within thirty (30) days before or after the anniversary date of the immediately preceding annual meeting of stockholders, not less than sixty (60) days nor more than ninety (90) days prior to the meeting, and (y) in the case of an annual meeting that is called for a date that is not within thirty (30) days before or after the anniversary date of the immediately preceding annual meeting, not later than the close of business on the tenth day following the day on which notice of the date of the meeting was communicated to stockholders or public disclosure of the date of the meeting was made, whichever occurs first. For purposes of the first annual meeting of stockholders of the Corporation, the first anniversary date shall be May 23, 2012. In no event shall the public announcement of an adjournment or postponement of a meeting of stockholders commence a new time period (or extend any time period) for the giving of a stockholder notice as described herein. To be in proper written form, such stockholder's notice to the Secretary shall set forth in writing (x) as to each person whom the stockholder proposes to nominate for election as a director (i) all information relating to such

4

person that is required to be disclosed in solicitations of proxies for election of directors in an election contest, or is otherwise required, in each case pursuant to and in accordance with Regulation 14A under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and (ii) such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected; (y) as to any other business that the stockholder proposes to bring before the annual meeting, (i) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting, (ii) the text of the proposal or business (including the text of any resolutions proposed for consideration and, in the event that such business includes a proposal to amend the Bylaws of the Corporation, the language of the proposed amendment), and (iii) any material interest of the stockholder and beneficial owner, if any, on whose behalf the proposal is made (i) the name and address, as they appear on the Corporation's books, of such stockholder and of such beneficial owner, (ii) the class or series and number of shares of the capital stock of the Corporation that are owned beneficially and of record by such stockholder and such beneficial owner, (iii) a description of all arrangements or understandings between such stockholder and/or beneficial owner and any other person or persons (including their names) pursuant to which the proposals are to be made by such stockholder, (iv) a representation that such stockholder is a holder of record of stock of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to propose such business or nomination set forth in its notice, (v) a representation (A) whether any such person or such stockholder has received any financial assistance, funding or other consideration from any other person in respect of the nomination (and the details thereof) (a "Stockholder Associated

5

and (B) whether and the extent to which any hedging, derivative or other transaction has been entered into with respect to the Corporation within the past six months by, or is in effect with respect to, such stockholder, any person to be nominated by such stockholder or any Stockholder Associated Person, the effect or intent of which transaction is to mitigate loss to or manage risk or benefit of share price changes for, or to increase or decrease the voting power of, such stockholder, nominee or any such Stockholder Associated Person, and (vi) a representation whether the stockholder or the beneficial owner, if any, intends or is part of a group that intends to (1) deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the proposal or elect the nominee and/or (2) otherwise solicit proxies from stockholders in support of such proposal, and (vii) any other information relating to such stockholder or beneficial owner that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies in support of such proposal pursuant to Section 14 of the Exchange Act, and any rules and regulations promulgated thereunder. The foregoing notice requirements of this Section 1.5 shall not apply to any proposal made pursuant to Rule 14a-8 (or any successor thereof) promulgated under the Exchange Act shall be deemed satisfied if the stockholder making such proposal complies with the provisions of Rule 14a-8 and has notified the Corporation of his or her intention to present a proposal at an annual meeting in compliance with Rule 14a-8 and such stockholder's proposal has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for such annual meeting. The Corporation may require any proposed nominee to furnish such other information as it may reasonably require to determine (x) the eligibility of such proposed nominee to

6

Corporation and (y) whether the nominee would be considered "independent" under the independence requirements set forth in the Corporate Governance Rules of NASDAQ (or the rules and regulations of the principal securities exchange on which the Corporation's equity securities are then listed) in effect from time to time.

- (3) Notwithstanding anything in paragraph (a)(2) of this Section 1.5 to the contrary, in the event that the number of directors to be elected to the Board of Directors of the Corporation at an annual meeting is increased and there is no public announcement by the Corporation naming all of the nominees for director or specifying the size of the increased Board of Directors at least one hundred (100) days prior to the anniversary date of the immediately preceding annual meeting, a stockholder's notice required by this Section 1.5 shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delivered to the Secretary at the principal executive offices of the Corporation not later than the close of business on the tenth (10th) day following the day on which such public announcement is first made by the Corporation.
- (b) Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more directors to the Board of Directors, any such stockholder entitled to vote at such meeting may nominate a person or persons (as the case may be) for election to such position(s) as specified in the Corporation's notice of meeting, if the stockholder's notice meeting the requirements of paragraph (a)(2) of this Section 1.5 (substituting special meeting for annual meeting as applicable) shall be delivered to the Secretary at the principal executive offices of the Corporation not earlier than the close of business on the

7

ninetieth (90th) day prior to such special meeting and not later than the close of business on the later of the sixtieth (60th) day prior to such special meeting or the tenth (10th) day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the Board of Directors to be elected at such meeting. In no event shall the public announcement of an adjournment or postponement of a special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

(c) General. (1) Only such persons who are nominated in accordance with the procedures set forth in this Section 1.5 shall be eligible to be elected at an annual or special meeting of stockholders of the Corporation to serve as directors and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 1.5. Except as otherwise provided by law, the chairman of the meeting shall have the power and duty (i) to determine whether a nomination or any business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 1.5 (including whether the stockholder or beneficial owner, if any, on whose behalf the nomination or proposal is made solicited (or is part of a group which solicited) or did not so solicit, as the case may be, proxies in support of such stockholder's nominee or proposal in compliance with such stockholder's representation as required by clause (a)(2)(z)(vi) of this Section 1.5) and (ii) if any proposed nomination or proposed business was not made or proposed in compliance with this Section 1.5, to declare that such nomination shall be disregarded or that such proposed business shall not be transacted. Notwithstanding the foregoing

special meeting of stockholders of the Corporation to present the nomination to the Board of Directors or to present the proposed business, such nomination shall be disregarded and such proposed business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For purposes of this Section 1.5, to be considered a qualified representative of the stockholder, a person must be authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders.

(2) For purposes of this Section 1.5, "public announcement" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to the Exchange Act.

(3) Notwithstanding the foregoing provisions of this Section 1.5, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 1.5. Nothing in this Section 1.5 shall be deemed to affect any rights (i) of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act or (ii) of the holders of any series of preferred stock to elect directors pursuant to any applicable provisions of the Corporation's Certificate of Incorporation.

#### Section 1.6 Quorum.

Subject to the rights of the holders of any series of preferred stock and except as otherwise provided by law or in the Certificate of Incorporation or these Bylaws, at any meeting of stockholders, the holders of a majority in total voting power of the outstanding shares of stock

9

entitled to vote at the meeting shall be present or represented by proxy in order to constitute a quorum for the transaction of any business. The chairman of the meeting shall have the power and duty to determine whether a quorum is present at any meeting of the stockholders. Shares of its own stock belonging to the Corporation or to another corporation, if a majority of the shares entitled to vote in the election of directors of such other corporation is held, directly or indirectly, by the Corporation, shall neither be entitled to vote nor be counted for quorum purposes; provided, however, that the foregoing shall not limit the right of the Corporation or any subsidiary of the Corporation to vote stock, including, but not limited to, its own stock, held by it in a fiduciary capacity. In the absence of a quorum, the chairman of the meeting may adjourn the meeting from time to time in the manner provided in Section 1.7 hereof until a quorum shall be present.

### Section 1.7 Adjournment.

Any meeting of stockholders, annual or special, may be adjourned from time to time solely by the chairman of the meeting because of the absence of a quorum or for any other reason and to reconvene at the same or some other time, date and place, if any. Notice need not be given of any such adjourned meeting if the time, date and place thereof are announced at the meeting at which the adjournment is taken. The chairman of the meeting shall have full power and authority to adjourn a stockholder meeting in his sole discretion even over stockholder opposition to such adjournment. The stockholders present at a meeting shall not have the authority to adjourn the meeting. If the time, date and place, if any, thereof, and the means of remote communication, if any, by which the stockholders and the proxy holders may be deemed to be present and in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken and the adjournment is for less than thirty (30) days, no notice need be given of any such adjourned meeting. If the adjournment is for more than thirty (30)

10

days and the time, date and place, if any, and the means of remote communication, if any, by which the stockholders and the proxy holders may be deemed to be present and in person are not announced at the meeting at which the adjournment is taken, or if after the adjournment a new record date is fixed for the adjourned meeting, then notice shall be given by the Secretary as required for the original meeting. At the adjourned meeting, the Corporation may transact any business that might have been transacted at the original meeting.

# Section 1.8 Organization.

The Chairman of the Board, or in his absence the President, or in their absence any Vice President, shall call to order meetings of stockholders and preside over and act as chairman of such meetings. The Board of Directors or, if the Board fails to act, the stockholders, may appoint any stockholder, director or officer of the Corporation to act as chairman of any meeting in the absence of the Chairman of the Board, the President and all Vice Presidents. The date and time of the opening and closing of the polls for each matter upon which the stockholders will vote at a meeting shall be determined by the chairman of the meeting and announced at the meeting. The Board of Directors may adopt by resolution such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Unless otherwise determined by the Board of Directors, the chairman of the meeting shall have the exclusive right to determine the order of business and to prescribe other such rules, regulations and procedures and shall have the authority in his discretion to regulate the conduct of any such meeting. Such rules, regulations or procedures, whether adopted by the Board of Directors or prescribed by the chairman of the meeting, may include, without limitation, the following: (i) rules and procedures for maintaining order at the meeting and the safety of those present; (ii) limitations on attendance at or participation in the meeting to stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as the chairman

11

of the meeting shall determine; (iii) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (iv) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board of Directors or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

The Secretary shall act as secretary of all meetings of stockholders, but, in the absence of the Secretary, the chairman of the meeting may appoint any other person to act as secretary of the meeting.

### Section 1.9 <u>Postponement or Cancellation of Meeting.</u>

Any previously scheduled annual or special meeting of the stockholders may be postponed or canceled by resolution of the Board of Directors upon public notice given prior to the time previously scheduled for such meeting of stockholders.

#### Section 1.10 Voting.

Subject to the rights of the holders of any series of preferred stock and except as otherwise provided by law, the Certificate of Incorporation or these Bylaws and except for the election of directors, at any meeting duly called and held at which a quorum is present, the affirmative vote of a majority of the combined voting power of the outstanding shares present in person or represented by proxy at the meeting and entitled to vote on the subject matter shall be the act of the stockholders. Subject to the rights of the holders of any series of preferred stock, at any meeting duly called and held for the election of directors at which a quorum is present, directors shall be elected by a plurality of the combined voting power of the outstanding shares present in person or represented by proxy at the meeting and entitled to vote on the election of directors.

10

#### ARTICLE II

### BOARD OF DIRECTORS

### Section 2.1 Number and Term of Office.

(a) The governing body of this Corporation shall be a Board of Directors. Subject to any rights of the holders of any series of preferred stock to elect additional directors, the Board of Directors shall be comprised of not less than three (3) members, or such other number as may be fixed from time to time by the Board of Directors by resolution adopted by the affirmative vote of 75% of the members of the Board of Directors then in office. Directors need not be stockholders of the Corporation. The Corporation shall nominate the persons holding the offices of Chairman of the Board and President for election as directors at any meeting at which such persons are subject to election as directors.

(b) Except as otherwise fixed by the Certificate of Incorporation relating to the rights of the holders of any series of preferred stock to separately elect additional directors, which additional directors are not required to be classified pursuant to the terms of such series of preferred stock, the Board of Directors shall be divided into three classes: Class I, Class II and Class III. Each class shall consist, as nearly as possible, of a number of directors equal to one-third (33<sup>1</sup>/<sub>3</sub>%) of the then authorized number of members of the Board of Directors. The term of office of the initial Class II directors shall expire at the annual meeting of stockholders in 2014; the term of office of the initial Class II directors shall expire at the annual meeting of stockholders in 2013. At each annual meeting of stockholders of the Corporation the successors of that class of directors whose term expires at that meeting shall be elected to hold office for a term expiring at the annual meeting of stockholders held in the third

13

year following the year of their election. The directors of each class will serve until the earliest to occur of their death, resignation, removal or disqualification or the election and qualification of their respective successors.

#### Section 2.2 Resignations.

Any director of the Corporation, or any member of any committee, may resign at any time by giving written notice to the Board of Directors, the Chairman of the Board or the President or Secretary of the Corporation. Any such resignation shall take effect at the time specified therein or, if the time be not specified therein, then upon receipt thereof. The acceptance of such resignation shall not be necessary to make it effective unless otherwise stated therein.

# Section 2.3 <u>Removal of Directors.</u>

Subject to the rights of the holders of any series of preferred stock, directors may be removed from office only for cause upon the affirmative vote of the holders of not less than a majority of the total voting power of the then outstanding shares entitled to vote at an election of directors voting together as a single class.

#### Section 2.4 Newly Created Directorships and Vacancies.

Subject to the rights of the holders of any series of preferred stock, vacancies on the Board of Directors resulting from death, resignation, removal, disqualification or other cause, and newly created directorships resulting from any increase in the number of directors on the Board of Directors, shall be filled by the affirmative vote of a majority of the remaining directors then in office (even though less than a quorum) or by the sole remaining director at any regular or special meeting of the Board of Directors. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the class of directors in which the vacancy occurred or to which the new directorship is apportioned, and until such

14

director's successor shall have been elected and qualified. No decrease in the number of directors constituting the Board of Directors shall shorten the term of any incumbent director, except as may be provided in the terms of any series of preferred stock with respect to any additional director elected by the holders of such series of preferred stock. Notwithstanding Article 1 of these Bylaws, in case the entire Board of Directors shall die or resign, the President or Secretary of the Corporation, or any ten (10) stockholders may call and cause notice to be given for a special meeting of stockholders in the same manner that the Chairman of the Board may call such a meeting, and directors for the unexpired terms may be elected at such special meeting.

### Section 2.5 Meetings.

The annual meeting of each newly elected Board of Directors may be held on such date and at such time and place as the Board of Directors determines. The annual meeting may be held immediately following the annual meeting of stockholders, and if so held, no notice of such meeting shall be necessary to the newly elected directors in order to hold the meeting legally, provided that a quorum shall be present thereat.

Notice of each regular meeting shall be furnished in writing to each member of the Board of Directors not less than five (5) days in advance of said meeting, unless such notice requirement is waived in writing by each member. No notice need be given of the meeting immediately following an annual meeting of stockholders.

Special meetings of the Board of Directors shall be held at such time and place as shall be designated in the notice of the meeting. Special meetings of the Board of Directors may be called by the Chairman of the Board, and shall be called by the President or Secretary of the Corporation upon the written request of not less than 75% of the members of the Board of Directors then in office

### Section 2.6 Notice of Special Meetings.

The Secretary, or in his absence any other officer of the Corporation, shall give each director notice of the time and place of holding of special meetings of the Board of Directors by mail at least ten (10) days before the meeting, or by facsimile transmission, electronic mail or personal service at least twenty-four (24) hours before the meeting unless such notice requirement is waived in writing by each member. Unless otherwise stated in the notice thereof, any and all business may be transacted at any meeting without specification of such business in the notice.

### Section 2.7 <u>Conference Telephone Meeting.</u>

Members of the Board of Directors, or any committee thereof, may participate in a meeting of the Board of Directors or such committee by means of telephone conference or other similar communications equipment by means of which all persons participating in the meeting can hear each other and communicate with each other, and such participation in a meeting shall constitute presence in person at such meeting.

#### Section 2.8 Quorum and Organization of Meetings.

A majority of the total number of members of the Board of Directors as constituted from time to time shall constitute a quorum for the transaction of business, but, if at any meeting of the Board of Directors (whether or not adjourned from a previous meeting) there shall be less than a quorum present, a majority of those present may adjourn the meeting to another time, date and place, and the meeting may be held as adjourned without further notice or waiver. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, a majority of the directors present at any meeting at which a quorum is present may decide any question brought before such meeting. Meetings shall be presided over by the Chairman of the Board or in his absence by such other person as the directors may select. The Board of Directors

16

shall keep written minutes of its meetings. The Secretary of the Corporation shall act as secretary of the meeting, but in his absence the chairman of the meeting may appoint any person to act as secretary of the meeting.

The Board may designate one or more committees, each committee to consist of one or more of the directors of the Corporation. The Board may designate one or more Directors as alternate members of any committee to replace absent or disqualified members at any meeting of such committee. If a member of a committee shall be absent from any meeting, or disqualified from voting thereat, the remaining member or members present and not disqualified from voting, whether or not such member or members constitute a quorum, may, by a unanimous vote, appoint another member of the Board of Directors to act at the meeting in place of any such absent or disqualified member. Any such committee, to the extent provided in a resolution of the Board of Directors passed as aforesaid, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be impressed on all papers that may require it, but no such committee shall have the power or authority of the Board of Directors in reference to (i) approving or adopting, or recommending to the stockholders, any action or matter expressly required by the laws of the State of Delaware to be submitted to the stockholders for approval or (ii) adopting, amending or repealing any Bylaw of the Corporation. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors. Unless otherwise specified in the resolution of the Board of Directors designating a committee, at all meetings of such committee a majority of the total number of members of the committee shall constitute a quorum for the transaction of business, and the vote of a majority of the members of the committee present at any meeting at

17

which there is a quorum shall be the act of the committee. Each committee shall keep regular minutes of its meetings. Unless the Board of Directors otherwise provides, each committee designated by the Board of Directors may make, alter and repeal rules for the conduct of its business. In the absence of such rules each committee shall conduct its business in the same manner as the Board of Directors conducts its business pursuant to Article 2 of these Bylaws.

## Section 2.9 <u>Indemnification</u>.

The Corporation shall indemnify members of the Board of Directors and officers of the Corporation and their respective heirs, personal representatives and successors in interest for or on account of any action performed on behalf of the Corporation, to the fullest extent permitted by the laws of the State of Delaware and the Corporation's Certificate of Incorporation, as now or hereafter in effect.

# Section 2.10 <u>Indemnity Undertaking</u>.

To the extent not prohibited by law, the Corporation shall indemnify any person who is or was made, or threatened to be made, a party to any threatened, pending or completed action, suit or proceeding (a "Proceeding"), whether civil, criminal, administrative or investigative, including, without limitation, an action by or in the right of the Corporation to procure a judgment in its favor, by reason of the fact that such person, or a person of whom such person is the legal representative, is or was a director or officer of the Corporation, or is or was serving in any capacity at the request of the Corporation for any other corporation, partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprises (an "Other Entity"), against judgments, fines, penalties, excise taxes, amounts paid in settlement and costs, charges and expenses (including attorneys' fees and disbursements). Persons who are not directors or officers of the Corporation may be similarly indemnified in respect of service to the Corporation or to an Other Entity at the request of the Corporation to the extent the Board of

18

Directors at any time specifies that such persons are entitled to the benefits of this Section 2.10.

Except as otherwise provided in Section 2.12 hereof, the Corporation shall be required to indemnify a person in connection with a proceeding (or part thereof) commenced by such person only if the commencement of such proceeding (or part thereof) by the person was authorized by the Board of Directors.

### Section 2.11 Advancement of Expenses.

The Corporation shall, from time to time, reimburse or advance to any director or officer or other person entitled to indemnification hereunder the funds necessary for payment of expenses, including attorneys' fees and disbursements, incurred in connection with any Proceeding in advance of the final disposition of such Proceeding; provided, however, that, if required by the laws of the State of Delaware, such expenses incurred by or on behalf of any director or officer or other person may be paid in advance of the final disposition of a Proceeding only upon receipt by the Corporation of an undertaking, by or on behalf of such director or officer (or other person indemnified hereunder), to repay any such amount so advanced if it shall ultimately be determined by final judicial decision from which there is no further right of appeal that

such director, officer or other person is not entitled to be indemnified for such expenses. Except as otherwise provided in Section 2.12 hereof, the Corporation shall be required to reimburse or advance expenses incurred by a person in connection with a proceeding (or part thereof) commenced by such person only if the commencement of such proceeding (or part thereof) by the person was authorized by the Board of Directors.

Section 2.12 Claims.

If a claim for indemnification or advancement of expenses under this Article 2 is not paid in full within thirty (30) days after a written claim therefor by the person seeking indemnification or reimbursement or advancement of expenses has been received by the

19

Corporation, the person may file suit to recover the unpaid amount of such claim and, if successful, in whole or in part, shall be entitled to be paid the expense of prosecuting such claim. In any such action the Corporation shall have the burden of proving that the person seeking indemnification or reimbursement or advancement of expenses is not entitled to the requested indemnification, reimbursement or advancement of expenses under applicable law.

### Section 2.13 <u>Amendment, Modification or Repeal.</u>

Any amendment, modification or repeal of the foregoing provisions of this Article 2 shall not adversely affect any right or protection hereunder of any person entitled to indemnification under Section 2.9 hereof in respect of any act or omission occurring prior to the time of such repeal or modification.

#### Section 2.14 <u>Executive Committee of the Board of Directors.</u>

The Board of Directors, by the affirmative vote of not less than 75% of the members of the Board of Directors then in office, may designate an executive committee, all of whose members shall be directors, to manage and operate the affairs of the Corporation or particular properties or enterprises of the Corporation. Subject to the limitations of the law of the State of Delaware and the Certificate of Incorporation, such executive committee shall exercise all powers and authority of the Board of Directors in the management of the business and affairs of the Corporation including, but not limited to, the power and authority to authorize the issuance of shares of common or preferred stock. The executive committee shall keep minutes of its meetings and report to the Board of Directors not less often than quarterly on its activities and shall be responsible to the Board of Directors for the conduct of the enterprises and affairs entrusted to it. Regular meetings of the executive committee, of which no notice shall be necessary, shall be held at such time, dates and places as shall be fixed by resolution adopted by the executive committee. Special meetings of the executive committee shall be called at the

20

request of the President or of any member of the executive committee, and shall be held upon such notice as is required by these Bylaws for special meetings of the Board of Directors, provided that oral notice by telephone or otherwise shall be sufficient if received not later than the day immediately preceding the day of the meeting.

### Section 2.15 Other Committees of the Board of Directors.

The Board of Directors may by resolution establish committees other than an executive committee and shall specify with particularity the powers and duties of any such committee. Subject to the limitations of the laws of the State of Delaware and the Certificate of Incorporation, any such committee shall exercise all powers and authority specifically granted to it by the Board of Directors, which powers may include the authority to authorize the issuance of shares of common or preferred stock. Such committees shall serve at the pleasure of the Board of Directors, keep minutes of their meetings and have such names as the Board of Directors by resolution may determine and shall be responsible to the Board of Directors for the conduct of the enterprises and affairs entrusted to them.

### Section 2.16 <u>Directors' Compensation</u>.

Directors shall receive such compensation for attendance at any meetings of the Board and any expenses incidental to the performance of their duties as the Board of Directors shall determine by resolution. Such compensation may be in addition to any compensation received by the members of the Board of Directors in any other capacity.

### Section 2.17 <u>Action Without Meeting.</u>

Nothing contained in these Bylaws shall be deemed to restrict the power of members of the Board of Directors or any committee designated by the Board of Directors to take any action required or permitted to be taken by them without a meeting.

2.1

### ARTICLE III

### OFFICERS

# Section 3.1 <u>Executive Officers.</u>

The Board of Directors shall elect from its own number, at its first meeting after each annual meeting of stockholders, a Chairman of the Board and a President. The Board of Directors may also elect such Vice Presidents as in the opinion of the Board of Directors the business of the Corporation requires, a Treasurer and a Secretary, any of whom may or may not be directors. The Board of Directors may also elect, from time to time, such other or additional officers as in its opinion are desirable for the conduct of business of the Corporation. Each officer shall hold office until the first meeting of the Board of Directors following the next annual meeting of stockholders following their respective election. Any person may hold at one time two or more offices; provided, however, that the President shall not hold any other office except that of Chairman of the Board.

## Section 3.2 <u>Powers and Duties of Officers.</u>

The Chairman of the Board shall have overall responsibility for the management and direction of the business and affairs of the Corporation and shall exercise such duties as customarily pertain to the office of Chairman of the Board and such other duties as may be prescribed from time to time by the Board of Directors. He shall be the senior officer of the Corporation and in case of the inability or failure of the President to perform his duties, he shall perform the duties of the President. He may appoint and terminate the appointment or election of officers, agents or employees other than those appointed or elected by the Board of Directors. He may sign, execute and deliver, in the name of the Corporation, powers of attorney, contracts, bonds and other obligations. The Chairman shall preside at all meetings of stockholders and of

the Board of Directors at which he is present, and shall perform such other duties as may be prescribed from time to time by the Board of Directors or these Bylaws.

The President of the Corporation shall have such powers and perform such duties as customarily pertain to a chief executive officer and the office of a president, including, without limitation, being responsible for the active direction of the daily business of the Corporation, and shall exercise such other duties as may be prescribed from time to time by the Board of Directors. The President may sign, execute and deliver, in the name of the Corporation, powers of attorney, contracts, bonds and other obligations. In the absence or disability of the Chairman of the Board, the President shall perform the duties and exercise the powers of the Chairman of the Board.

Vice Presidents shall have such powers and perform such duties as may be assigned to them by the Chairman of the Board, the President, the executive committee, if any, or the Board of Directors. A Vice President may sign and execute contracts and other obligations pertaining to the regular course of his duties which implement policies established by the Board of Directors.

The Treasurer shall be the chief financial officer of the Corporation. Unless the Board of Directors otherwise declares by resolution, the Treasurer shall have general custody of all the funds and securities of the Corporation and general supervision of the collection and disbursement of funds of the Corporation. He shall endorse for collection on behalf of the Corporation checks, notes and other obligations, and shall deposit the same to the credit of the Corporation in such bank or depository as the Board of Directors may designate. He may sign, with the Chairman of the Board, President or such other person or persons as may be designated for the purpose by the Board of Directors, all bills of exchange or promissory notes of

23

the Corporation. He shall enter or cause to be entered regularly in the books of the Corporation a full and accurate account of all moneys received and paid by him on account of the Corporation, shall at all reasonable times exhibit his books and accounts to any director of the Corporation upon application at the office of the Corporation during business hours and, whenever required by the Board of Directors or the President, shall render a statement of his accounts. He shall perform such other duties as may be prescribed from time to time by the Board of Directors or by these Bylaws. He may be required to give bond for the faithful performance of his duties in such sum and with such surety as shall be approved by the Board of Directors. Any Assistant Treasurer shall, in the absence or disability of the Treasurer, perform the duties and exercise the powers of the Treasurer and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

The Secretary shall keep the minutes of all meetings of the stockholders and of the Board of Directors. The Secretary shall cause notice to be given of meetings of stockholders, of the Board of Directors, and of any committee appointed by the Board of Directors. He shall have custody of the corporate seal, minutes and records relating to the conduct and acts of the stockholders and Board of Directors, which shall, at all reasonable times, be open to the examination of any director. The Secretary or any Assistant Secretary may certify the record of proceedings of the meetings of the stockholders or of the Board of Directors or resolutions adopted at such meetings, may sign or attest certificates, statements or reports required to be filed with governmental bodies or officials, may sign acknowledgments of instruments, may give notices of meetings and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.

24

# Section 3.3 Bank Accounts.

In addition to such bank accounts as may be authorized in the usual manner by resolution of the Board of Directors, the Treasurer, with approval of the Chairman of the Board or the President, may authorize such bank accounts to be opened or maintained in the name and on behalf of the Corporation as he may deem necessary or appropriate, provided payments from such bank accounts are to be made upon and according to the check of the Corporation, which may be signed jointly or singularly by either the manual or facsimile signature or signatures of such officers or bonded employees of the Corporation as shall be specified in the written instructions of the Treasurer or Assistant Treasurer of the Corporation with the approval of the Chairman of the Board or the President of the Corporation.

# Section 3.4 <u>Proxies; Stock Transfers.</u>

Unless otherwise provided in the Certificate of Incorporation or directed by the Board of Directors, the Chairman of the Board or the President or any Vice President or their designees shall have full power and authority on behalf of the Corporation to attend and to vote upon all matters and resolutions at any meeting of stockholders of any corporation in which this Corporation may hold stock, and may exercise on behalf of this Corporation any and all of the rights and powers incident to the ownership of such stock at any such meeting, whether regular or special, and at all adjournments thereof, and shall have power and authority to execute and deliver proxies and consents on behalf of this Corporation in connection with the exercise by this Corporation of the rights and powers incident to the ownership of such stock, with full power of substitution or revocation. Unless otherwise provided in the Certificate of Incorporation or directed by the Board of Directors, the Chairman of the Board or the President or any Vice President or their designees shall have full power and authority on behalf of the Corporation to transfer, sell or dispose of stock of any corporation in which this Corporation may hold stock.

2.5

#### ARTICLE IV

#### CAPITAL STOCK

# Section 4.1 Shares.

The shares of the corporation shall be represented by a certificate or shall be uncertificated. Certificates shall be signed by the Chairman of the Board of Directors or the President and by the Secretary or the Treasurer, and sealed with the seal of the Corporation. Such seal may be a facsimile, engraved or printed. Within a reasonable time after the issuance or transfer of uncertificated shares, the Corporation shall send to the registered owner thereof a written notice containing the information required to be set forth or stated on certificates pursuant to Sections 151, 156, 202(a) or 218(a) of the Delaware General Corporation Law or a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative participating, optional or other special rights of each class of stock or series thereof and the qualification, limitations or restrictions of such preferences and/or rights.

Any of or all the signatures on a certificate may be facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such an officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if such officer, transfer agent or registrar had not ceased to hold such position at the time of its issuance.

#### Section 4.2 Transfer of Shares.

- (a) Upon surrender to the Corporation or the transfer agent of a certificate for shares duly endorsed or accompanied by proper evidence of succession, assignation or authority to transfer, it shall be the duty of the Corporation to issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books. Upon receipt of proper transfer instructions from the registered owner of uncertificated shares such uncertificated shares shall be cancelled, and the issuance of new equivalent uncertificated shares or certificated shares shall be made to the person entitled thereto and the transaction shall be recorded upon the books of the Corporation.
- (b) The person in whose name shares of stock stand on the books of the Corporation shall be deemed by the Corporation to be the owner thereof for all purposes, and the Corporation shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State of Delaware.

#### Section 4.3 Lost Certificates.

The Board of Directors or any transfer agent of the Corporation may direct a new certificate or certificates or uncertificated shares representing stock of the Corporation to be issued in place of any certificate or certificates theretofore issued by the Corporation, alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate to be lost, stolen or destroyed. When authorizing such issue of a new certificate or certificates or uncertificated shares, the Board of Directors (or any transfer agent of the Corporation authorized to do so by a resolution of the Board of Directors) may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost,

2.7

stolen or destroyed certificate or certificates, or his legal representative, to give the Corporation a bond in such sum as the Board of Directors (or any transfer agent so authorized) shall direct to indemnify the Corporation and the transfer agent against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed or the issuance of such new certificates or uncertificated shares, and such requirement may be general or confined to specific instances.

### Section 4.4 <u>Transfer Agent and Registrar.</u>

The Board of Directors may appoint one or more transfer agents and one or more registrars, and may require all certificates for shares to bear the manual or facsimile signature or signatures of any of them.

#### Section 4.5 Regulations.

The Board of Directors shall have power and authority to make all such rules and regulations as it may deem expedient concerning the issue, transfer, registration, cancellation and replacement of certificates representing stock of the Corporation or uncertificated shares, which rules and regulations shall comply in all respects with the rules and regulations of the transfer agent.

### ARTICLE V

### GENERAL PROVISIONS

#### Section 5.1 Offices.

The Corporation shall maintain a registered office in the State of Delaware as required by the laws of the State of Delaware. The Corporation may also have offices in such other places, either within or without the State of Delaware, as the Board of Directors may from time to time designate or as the business of the Corporation may require.

28

# Section 5.2 <u>Corporate Seal.</u>

The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization, and the words "Corporate Seal" and "Delaware."

## Section 5.3 <u>Fiscal Year</u>.

The fiscal year of the Corporation shall be determined by resolution of the Board of Directors.

### Section 5.4 <u>Notices and Waivers Thereof.</u>

Whenever any notice is required by the laws of the State of Delaware, the Certificate of Incorporation or these Bylaws to be given to any stockholder, director or officer, such notice, except as otherwise provided by law, may be given personally, or by mail, or, in the case of directors or officers, by electronic mail or facsimile transmission, addressed to such address as appears on the books of the Corporation. Any notice given by electronic mail or facsimile transmission shall be deemed to have been given when it shall have been transmitted and any notice given by mail shall be deemed to have been given three (3) business days after it shall have been deposited in the United States mail with postage thereon prepaid.

Whenever any notice is required to be given by law, the Certificate of Incorporation, or these Bylaws, a written waiver thereof, signed by the person entitled to such notice, whether before or after the meeting or the time stated therein, shall be deemed equivalent in all respects to such notice to the full extent permitted by law.

### Section 5.5 Saving Clause.

These Bylaws are subject to the provisions of the Certificate of Incorporation and applicable law. In the event any provision of these Bylaws is inconsistent

the extent only of such conflict, and such conflict shall not affect the validity of any other provision of these Bylaws.

## Section 5.6 <u>Amendments</u>.

In furtherance and not in limitation of the powers conferred by the laws of the State of Delaware, the Board of Directors, by action taken by the affirmative vote of not less than 75% of the members of the Board of Directors then in office, is hereby expressly authorized and empowered to adopt, amend or repeal any provision of the Bylaws of this Corporation.

Subject to the rights of the holders of any series of preferred stock, these Bylaws may be adopted, amended or repealed by the affirmative vote of the holders of not less than 80% of the total voting power of the then outstanding capital stock of the Corporation entitled to vote thereon; provided, however, that this paragraph shall not apply to, and no vote of the stockholders of the Corporation shall be required to authorize, the adoption, amendment or repeal of any provision of the Bylaws by the Board of Directors in accordance with the preceding paragraph.

Incorporated Under the Laws of the State of Delaware

Number A- Shares -0-Cusip No.

# LIBERTY MEDIA CORPORATION

# Series A Liberty Capital Common Stock, par value \$.01 per share

This Certifies that	is the owner of	FULLY PAID AND NON-ASSESSABLE
SHARES OF SERIES A LIBERTY CAPI	TAL COMMON STOCK, PAR VALUE \$0.01 PER SHA	ARE, OF LIBERTY MEDIA CORPORATION (hereinafter called the
"Corporation") transferable on the books of	of the Corporation by the holder hereof in person or by du	aly authorized attorney upon surrender of the Certificate properly endorsed.
This Certificate is not valid unless counter	signed by the Transfer Agent and registered by the Regis	strar.
Witness, the seal of the Corporation and the signatures of its duly authorized officers.		
,	<u> </u>	
Dated:		
Liberty Media Corporation		
President		Secretary
		•

Number B-

Shares -0-Cusip No.

# LIBERTY MEDIA CORPORATION

# Series B Liberty Capital Common Stock, par value \$.01 per share

This Certifies that	is the owner of	FULLY PAID AND NON-ASSESSABLE
SHARES OF SERIES B LIBERTY CAPITA	AL COMMON STOCK, PAR VALUE \$0.01 PER SH.	ARE, OF LIBERTY MEDIA CORPORATION (hereinafter called the
"Corporation") transferable on the books of	the Corporation by the holder hereof in person or by de	aly authorized attorney upon surrender of the Certificate properly endorsed.
This Certificate is not valid unless countersignates	gned by the Transfer Agent and registered by the Regis	strar.
Witness, the seal of the Corporation and the signatures of its duly authorized officers.		
•		
Dated:		
Liberty Media Corporation		
President		Secretary

Incorporated Under the Laws of the State of Delaware

Number A- Shares -0-Cusip No.

# LIBERTY MEDIA CORPORATION

# Series A Liberty Starz Common Stock, par value \$.01 per share

This Certifies that	is the owner of	FULLY PAID AND NON-ASSESSABLE
SHARES OF SERIES A LIBERTY STAF	RZ COMMON STOCK, PAR VALUE \$0.01 PER SHAF	RE, OF LIBERTY MEDIA CORPORATION (hereinafter called the
"Corporation") transferable on the books of	of the Corporation by the holder hereof in person or by d	uly authorized attorney upon surrender of the Certificate properly endorsed.
This Certificate is not valid unless counter	signed by the Transfer Agent and registered by the Regis	strar.
Witness, the seal of the Corporation and the signatures of its duly authorized officers.		
•	•	
Dated:		
Liberty Media Corporation		
President		Secretary

Number B-

Shares -0-Cusip No.

# LIBERTY MEDIA CORPORATION

# Series B Liberty Starz Common Stock, par value \$.01 per share

	orporation by the holder hereof in person or by d	FULLY PAID AND NON-ASSESSABLE RE, OF LIBERTY MEDIA CORPORATION (hereinafter called the luly authorized attorney upon surrender of the Certificate properly endorsed. strar.
Witness, the seal of the Corporation and the signatures of its duly authorized officers.		
Dated:		
Liberty Media Corporation		
President		Secretary

# BAKER BOTTS IIP

THE WARNER 1299 PENNSYLVANIA AVE., NW WASHINGTON, D.C. 20004-2400

TEL +1 202.639.7700 FAX +1 202.639.7890 www.bakerbotts.com ABU DHABI AUSTIN BEIJING DALLAS DUBAI HONG KONG HOUSTON LONDON MOSCOW NEW YORK PALO ALTO RIYADH WASHINGTON

September 23, 2011

Liberty Interactive Corporation 12300 Liberty Boulevard Englewood, CO 80112

Ladies and Gentlemen:

As counsel for Liberty Interactive Corporation, a Delaware corporation formerly known as Liberty Media Corporation (**Liberty**"), we have examined and are familiar with the Registration Statement on Form S-4 (File No. 333-171201) (the "**Registration Statement**"), which was filed by Liberty Media Corporation, a Delaware corporation formerly known as Liberty CapStarz, Inc. and as Liberty Splitco, Inc. ("**Splitco**"), on April 18, 2011 with the Securities and Exchange Commission (the "**SEC**") and declared effective by the SEC on April 19, 2011 (which is being amended on the date hereof for the purpose of filing of this opinion) for the purpose of registering, under the Securities Act of 1933, as amended (the "**Securities Act**"), the distribution of:

- (i) all of the outstanding shares of Splitco's Series A Liberty Capital common stock, par value \$.01 per share (the "Series A Splitco Capital Common Stock"), to the holders of Liberty's Series A Liberty Capital common stock, par value \$.01 per share (the "Series A Liberty Capital Common Stock"), in exchange for all of the outstanding shares of Series A Liberty Capital Common Stock;
- (ii) all of the outstanding shares of Splitco's Series B Liberty Capital common stock, par value \$.01 per share (the 'Series B Splitco Capital Common Stock,' and together with the Series A Splitco Capital Common Stock, the "Splitco Capital Common Stock"), to the holders of Liberty's Series B Liberty Capital common stock, par value \$.01 per share (the "Series B Liberty Capital Common Stock,' and together with the Series A Liberty Capital Common Stock, the "Liberty Capital Common Stock"), in exchange for all of the outstanding shares of Series B Liberty Capital Common Stock;
- (iii) all of the outstanding shares of Splitco's Series A Liberty Starz common stock, par value \$.01 per share (the 'Series A Splitco Starz Common Stock'), to the holders of Liberty's Series A Liberty Starz common stock, par value \$.01 per share (the "Series A Liberty Starz Common Stock"), in exchange for all of the outstanding shares of Series A Liberty Starz Common Stock; and
- (iv) all of the outstanding shares of Splitco's Series B Liberty Starz common stock, par value \$.01 per share (the 'Series B Splitco Starz Common Stock," and together with the Series A Splitco Starz Common Stock, the "Splitco Starz Common Stock"), to the holders of Liberty's Series B Liberty Starz common stock, par value \$.01 per share (the "Series B Liberty Starz Common Stock," and together with the Series A Liberty Starz Common Stock, the 'Liberty Starz Common Stock, in exchange for all of the outstanding shares of Series B Liberty Starz Common Stock (together (i), (ii), (iii), and (iv), the 'Split-Off').

The terms of Split-Off, the Splitco Capital Common Stock, and the Splitco Starz Common Stock are described in the proxy statement/prospectus (the 'Proxy Statement') which forms a part of the Registration Statement. We have participated in the preparation of the discussion set forth in the Proxy Statement under the heading "Material U.S. Federal Income Tax Consequences of the Split-Off" (the "Discussion"). References herein to the "Code" shall refer to the Internal Revenue Code of 1986, as amended (the "Code").

In providing this opinion, we have examined and relied upon the facts, information, statements, covenants, representations, and warranties contained in the originals or copies, certified or otherwise identified to our satisfaction, of (i) PLR 131418-10 (published as PLR 201135025) that was issued to Liberty on March 3, 2011 (the "Ruling"), (ii) the request for the Ruling, dated July 28, 2010, filed by Liberty with the Internal Revenue Service (the TRS") in connection with the Split-Off, as modified and supplemented by certain supplemental submissions and correspondence filed by Liberty with the IRS, (iii) the letter from J.P. Morgan Securities Inc. to the Liberty board of directors, dated July 27, 2010, (iv) the Registration Statement, (v) the Proxy Statement, (vi) Liberty's restated certificate of incorporation, (vii) the Reorganization Agreement dated August 30, 2011, by and between Liberty and Splitco, (viii) Splitco's restated certificate of incorporation, a form of which is attached as Exhibit 3.1 to the Registration Statement, (ix) the representation letters, delivered by each of Liberty, Splitco, and Mr. John C. Malone to us, dated the date hereof, and (x) such other documents and records, and information provided to us by Liberty, as we have deemed necessary or appropriate as a basis for this opinion. References to each of the documents above include references to any exhibits, attachments, appendices, and schedules thereto. Unless otherwise specified, capitalized terms used herein shall have the meaning given to them in the Reorganization Agreement.

This opinion assumes and is conditioned on, among other things, the initial and continuing accuracy and completeness, which we have neither investigated nor verified, of the facts, information, statements, covenants, representations, and warranties set forth in each of the documents referred to above. This opinion further assumes that all of the statements and representations set forth in the representation letters referred to above are true, correct, and complete without regard to any qualification for knowledge or belief. This opinion is premised upon the continued validity of the Ruling and the accuracy of all of the facts,

2

information, statements, covenants, representations, and warranties upon which the Ruling and this opinion are based.

In our examination, we have assumed the genuineness of all signatures, the legal capacity of natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as certified, photostatic, electronic, or facsimile copies and the authenticity of the originals of such documents. We also have assumed that the Split-Off will be consummated in the manner described in the Proxy Statement, the Reorganization Agreement, and the Ruling, and that none of the terms and conditions of the Split-Off will have been waived or modified in any respect. Any inaccuracy in any of the aforementioned facts,

information, statements, representations, warranties, or assumptions, or any breach of any of the aforementioned covenants could adversely affect this opinion.

Subject to the foregoing and subject to the conditions, limitations, and qualifications described herein and in the Discussion, it is our opinion that, under presently applicable U.S. federal income tax law:

- (i) the Contribution and the Split-Off will qualify as a tax-free transaction under sections 355 and 368(a)(1)(D) of the Code;
- (ii) no gain or loss will be recognized by Liberty upon the distribution of Splitco Capital Common Stock and Splitco Starz Common Stock in the Split-Off;
- (iii) no gain or loss will be recognized by, and no amount will be included in the income of, holders of Liberty Capital Common Stock and Liberty Starz Common Stock upon the exchange of their shares of Liberty Capital Common Stock and Liberty Starz Common Stock for shares of Splitco Capital Common Stock and Splitco Starz Common Stock, respectively, in the Split-Off;
- (iv) the Splitco Capital Common Stock and the Splitco Starz Common Stock distributed in the Split-Off will be treated as stock of Splitco for U.S. federal income tax purposes; and
- (v) the Splitco Capital Common Stock and the Splitco Starz Common Stock distributed in the Split-Off will not constitute section 306 stock within the meaning of section 306(c) of the Code.

This opinion is based on our interpretation of the Code, applicable Treasury regulations, judicial authority, and administrative rulings and practice, all as of the date hereof. There can be no assurance that future legislative, judicial, or administrative changes or interpretations will not adversely affect the accuracy of the conclusions set forth herein.

3

We do not undertake to advise you as to any such future changes or interpretations unless we are specifically retained to do so. This opinion is not binding upon the IRS or any court and will not preclude the IRS or such court from adopting a contrary position. We express no other opinion as to the U.S. federal tax consequences of the Contribution or the Split-Off, including, without limitation, under those provisions of the Code and applicable Treasury regulations relating to U.S. federal consolidated income tax returns, and we express no opinion as to the state, local, foreign, or other tax consequences, of the Contribution or the Split-Off.

This opinion is delivered to you solely in connection with and for purposes of the transactions contemplated by the Proxy Statement and is not to be relied upon by any other person, quoted in whole or in part, or otherwise referred to (except in a list of closing documents), nor is it to be provided to any other person without our prior written consent. Notwithstanding the foregoing sentence, we consent to the filing of this letter with the SEC as an exhibit to the Registration Statement and to the references to our firm name in the Proxy Statement. In giving such consent, we do not thereby admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act or the rules of the SEC thereunder.

Sincerely, /s/ Baker Botts L.L.P.

#### LIBERTY MEDIA CORPORATION 2011 INCENTIVE PLAN

#### ARTICLE I

#### PURPOSE OF PLAN; EFFECTIVE DATE

- 1.1 Purpose. The purpose of the Plan is to promote the success of the Company by providing a method whereby (i) eligible employees of the Company and its Subsidiaries and (ii) independent contractors providing services to the Company and its Subsidiaries may be awarded additional remuneration for services rendered and may be encouraged to invest in capital stock of the Company, thereby increasing their proprietary interest in the Company's businesses, encouraging them to remain in the employ or service of the Company or its Subsidiaries, and increasing their personal interest in the continued success and progress of the Company and its Subsidiaries. The Plan is also intended to aid in (i) attracting Persons of exceptional ability to become officers and employees of the Company and its Subsidiaries and (ii) inducing independent contractors to agree to provide services to the Company and its Subsidiaries.
  - 1.2 Effective Date. The Plan shall be effective as of the Redemption Date (the "Effective Date").

#### ARTICLE II

#### **DEFINITIONS**

- 2.1 Certain Defined Terms. Capitalized terms not defined elsewhere in the Plan shall have the following meanings (whether used in the singular or plural):
  - "Account" has the meaning ascribed thereto in Section 8.2.
- "Affiliate" of the Company means any corporation, partnership or other business association that, directly or indirectly, through one or more intermediaries, controlls, is controlled by, or is under common control with the Company.
- "Agreement" means a stock option agreement, stock appreciation rights agreement, restricted shares agreement, restricted stock units agreement, cash award agreement or an agreement evidencing more than one type of Award, specified in Section 10.5, as any such Agreement may be supplemented or amended from time to time
- "Approved Transaction" means any transaction in which the Board (or, if approval of the Board is not required as a matter of law, the stockholders of the Company) shall approve (i) any consolidation or merger of the Company, or binding share exchange, pursuant to which shares of Common Stock of the Company would be changed or converted into or exchanged for cash, securities, or other property, other than any such transaction in which the common stockholders of the Company immediately

prior to such transaction have the same proportionate ownership of the Common Stock of, and voting power with respect to, the surviving corporation immediately after such transaction, (ii) any merger, consolidation or binding share exchange to which the Company is a party as a result of which the Persons who are common stockholders of the Company immediately prior thereto have less than a majority of the combined voting power of the outstanding capital stock of the Company ordinarily (and apart from the rights accruing under special circumstances) having the right to vote in the election of directors immediately following such merger, consolidation or binding share exchange, (iii) the adoption of any plan or proposal for the liquidation or dissolution of the Company, or (iv) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, of the assets of the Company.

- "Award" means a grant of Options, SARs, Restricted Shares, Restricted Stock Units, Performance Awards, Cash Awards and/or cash amounts under the Plan.
- "Board" means the Board of Directors of the Company.
- "Board Change" means, during any period of two consecutive years, individuals who at the beginning of such period constituted the entire Board cease for any reason to constitute a majority thereof unless the election, or the nomination for election, of each new director was approved by a vote of at least two-thirds of the directors then still in office who were directors at the beginning of the period.
- "Cash Award" means an Award made pursuant to Section 9.1 of the Plan to a Holder that is paid solely on account of the attainment of one or more Performance Objectives that have been preestablished by the Committee.
- "Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute or statutes thereto. Reference to any specific Code section shall include any successor section.
  - "Committee" means the committee of the Board appointed pursuant to Section 3.1 to administer the Plan.
  - "Common Stock" means each or any (as the context may require) series of the Company's common stock.
  - "Company" means Liberty Media Corporation (f/k/a Liberty CapStarz, Inc.), a Delaware corporation.
- "Control Purchase" means any transaction (or series of related transactions) in which any person (as such term is defined in Sections 13(d)(3) and 14(d)(2) of the Exchange Act), corporation or other entity (other than the Company, any Subsidiary of the Company or any employee benefit plan sponsored by the Company or any Subsidiary of the Company or any Exempt Person (as defined below)) shall become the "beneficial owner" (as such term is defined in Rule 13d-3 under the Exchange Act), directly or

means the spouse, siblings and lineal descendants of such Person.

"Disability" means the inability to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than 12 months.

"Dividend Equivalents" means, with respect to Restricted Stock Units, to the extent specified by the Committee only, an amount equal to all dividends and other distributions (or the economic equivalent thereof) which are payable to stockholders of record during the Restriction Period on a like number and kind of shares of Common Stock. Notwithstanding any provision of the Plan to the contrary, Dividend Equivalents with respect to a Performance Award may only be paid to the extent the Performance Award is actually paid to the Holder.

"Domestic Relations Order" means a domestic relations order as defined by the Code or Title I of the Employee Retirement Income Security Act of 1974, or the rules thereunder.

"Equity Security" shall have the meaning ascribed to such term in Section 3(a)(11) of the Exchange Act, and an equity security of an issuer shall have the meaning ascribed thereto in Rule 16a-1 promulgated under the Exchange Act, or any successor Rule.

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor statute or statutes thereto. Reference to any specific Exchange Act section shall include any successor section.

"Fair Market Value" of a share of any series of Common Stock on any day means (i) for Option and SAR exercise transactions effected on any third-party incentive award administration system provided by the Company, the current high bid price of a share of any series of Common Stock as reported on the consolidated transaction reporting system on the principal national securities exchange on which shares of such series of Common Stock are listed on such day or if such shares are not then listed on a national securities exchange, then as quoted by Pink OTC Markets Inc., or (ii) for all other

3

purposes under this Plan, the last sale price (or, if no last sale price is reported, the average of the high bid and low asked prices) for a share of such series of Common Stock on such day (or, if such day is not a trading day, on the next preceding trading day) as reported on the consolidated transaction reporting system for the principal national securities exchange on which shares of such series of Common Stock are listed on such day or if such shares are not then listed on a national securities exchange, then as quoted by Pink OTC Markets Inc. If for any day the Fair Market Value of a share of the applicable series of Common Stock is not determinable by any of the foregoing means, then the Fair Market Value for such day shall be determined in good faith by the Committee on the basis of such quotations and other considerations as the Committee deems appropriate.

"Free Standing SAR" has the meaning ascribed thereto in Section 7.1.

"Holder" means a Person who has received an Award under the Plan.

"Option" means a stock option granted under Article VI.

"Performance Award" means an Award made pursuant to Article IX of the Plan to a Holder that is subject to the attainment of one or more Performance Objectives.

"Performance Objective" means a standard established by the Committee to determine in whole or in part whether a Performance Award shall be earned.

"Person" means an individual, corporation, limited liability company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Plan" means this Liberty Media Corporation 2011 Incentive Plan.

"Redemption" means the redemption by Liberty Interactive Corporation (f/k/a Liberty Media Corporation) of all of the outstanding shares of each series of its Liberty Capital Common Stock and Liberty Starz Common Stock for shares of the Company's Capital Common Stock and Starz Common Stock, respectively.

"Redemption Date" means 5:00 p.m., New York City time, on the date on which the Redemption occurs.

"Restricted Shares" means shares of any series of Common Stock awarded pursuant to Section 8.1.

"Restricted Stock Unit" means a unit evidencing the right to receive in specified circumstances one share of the specified series of Common Stock or the equivalent value in cash, which right is subject to a Restriction Period or forfeiture provisions.

4

"Restriction Period" means a period of time beginning on the date of each Award of Restricted Shares or Restricted Stock Units and ending on the Vesting Date with respect to such Award.

"Retained Distribution" has the meaning ascribed thereto in Section 8.3.

"SARs" means stock appreciation rights, awarded pursuant to Article VII, with respect to shares of any specified series of Common Stock.

"Subsidiary" of a Person means any present or future subsidiary (as defined in Section 424(f) of the Code) of such Person or any business entity in which such Person owns, directly or indirectly, 50% or more of the voting, capital or profits interests. An entity shall be deemed a subsidiary of a Person for purposes of this definition only for such periods as the requisite ownership or control relationship is maintained.

"Tandem SARs" has the meaning ascribed thereto in Section 7.1.

"Vesting Date," with respect to any Restricted Shares or Restricted Stock Units awarded hereunder, means the date on which such Restricted Shares or Restricted Stock Units cease to be subject to a risk of forfeiture, as designated in or determined in accordance with the Agreement with respect to such Award of Restricted Shares or Restricted Stock Units pursuant to Article VIII. If more than one Vesting Date is designated for an Award of Restricted Shares or Restricted Stock Units, reference in the Plan to a Vesting Date in respect of such Award shall be deemed to refer to each part of such Award and the Vesting Date for such part.

#### ADMINISTRATION

- 3.1 Committee. The Plan shall be administered by the Compensation Committee of the Board unless a different committee is appointed by the Board. The Committee shall be comprised of not less than two Persons. The Board may from time to time appoint members of the Committee in substitution for or in addition to members previously appointed, may fill vacancies in the Committee and may remove members of the Committee. The Committee shall select one of its members as its chairman and shall hold its meetings at such times and places as it shall deem advisable. A majority of its members shall constitute a quorum and all determinations shall be made by a majority of such quorum. Any determination reduced to writing and signed by all of the members shall be as fully effective as if it had been made by a majority vote at a meeting duly called and held.
- 3.2 Powers. The Committee shall have full power and authority to grant to eligible Persons Options under Article VI of the Plan, SARs under Article VII of the Plan, Restricted Shares under Article VIII of the Plan, Cash Awards under Article IX of the Plan and/or Performance Awards under Article IX of the Plan, to determine the terms and conditions (which need not be identical) of all Awards so granted, to interpret the provisions of the Plan and any Agreements relating to Awards granted under the

5

Plan and to supervise the administration of the Plan. The Committee in making an Award may provide for the granting or issuance of additional, replacement or alternative Awards upon the occurrence of specified events, including the exercise of the original Award. The Committee shall have sole authority in the selection of Persons to whom Awards may be granted under the Plan and in the determination of the timing, pricing and amount of any such Award, subject only to the express provisions of the Plan. In making determinations hereunder, the Committee may take into account the nature of the services rendered by the respective employees and independent contractors, their present and potential contributions to the success of the Company and its Subsidiaries, and such other factors as the Committee in its discretion deems relevant.

3.3 Interpretation. The Committee is authorized, subject to the provisions of the Plan, to establish, amend and rescind such rules and regulations as it deems necessary or advisable for the proper administration of the Plan and to take such other action in connection with or in relation to the Plan as it deems necessary or advisable. Each action and determination made or taken pursuant to the Plan by the Committee, including any interpretation or construction of the Plan, shall be final and conclusive for all purposes and upon all Persons. No member of the Committee shall be liable for any action or determination made or taken by such member or the Committee in good faith with respect to the Plan.

#### ARTICLE IV

#### SHARES SUBJECT TO THE PLAN

4.1 Number of Shares. Subject to the provisions of this Article IV, the maximum number of shares of Common Stock with respect to which Awards may be granted during the term of the Plan shall be 25,000,000 shares. Shares of Common Stock will be made available from the authorized but unissued shares of the Company or from shares reacquired by the Company, including shares purchased in the open market. The shares of Common Stock subject to (i) any Award granted under the Plan that shall expire, terminate or be cancelled or annulled for any reason without having been exercised (or considered to have been exercised as provided in Section 7.2), (ii) any Award of any SARs granted under the Plan the terms of which provide for settlement in cash, and (iii) any Award of Restricted Shares or Restricted Stock Units that shall be forfeited prior to becoming vested (provided that the Holder received no benefits of ownership of such Restricted Shares or Restricted Stock Units other than voting rights and the accumulation of Retained Distributions and unpaid Dividend Equivalents that are likewise forfeited) shall again be available for purposes of the Plan. Notwithstanding the foregoing, the following shares of Common Stock may not again be made available for issuance as Awards under the Plan: (a) shares of Common Stock not issued or delivered as a result of the net settlement of an outstanding Option or SAR, (b) shares of Common Stock used to pay the purchase price or withholding taxes related to an outstanding Award, or (c) shares of Common Stock repurchased on the open market with the proceeds of an Option purchase price. Except for Awards described in Section 10.1, no Person may be granted in any calendar year Awards covering more than 8 million shares of Common Stock (as such amount may be adjusted from time to time as provided in Section 4.2). No Person shall receive payment for Cash Awards during any calendar year aggregating in excess of \$10 million.

6

#### 4.2 Adjustments.

- (a) If the Company subdivides its outstanding shares of any series of Common Stock into a greater number of shares of such series of Common Stock (by stock dividend, stock split, reclassification, or otherwise) or combines its outstanding shares of any series of Common Stock into a smaller number of shares of such series of Common Stock (by reverse stock split, reclassification, or otherwise) or if the Committee determines that any stock dividend, extraordinary cash dividend, reclassification, recapitalization, reorganization, stock redemption, split-up, spin-off, combination, exchange of shares, warrants or rights offering to purchase such series of Common Stock or other similar corporate event (including mergers or consolidations other than those which constitute Approved Transactions, adjustments with respect to which shall be governed by Section 10.1(b)) affects any series of Common Stock so that an adjustment is required to preserve the benefits or potential benefits intended to be made available under the Plan, then the Committee, in such manner as the Committee, in its sole discretion, deems equitable and appropriate, shall make such adjustments to any or all of (i) the number and kind of shares of stock which thereafter may be awarded, optioned or otherwise made subject to the benefits contemplated by the Plan, (ii) the number and kind of shares of stock subject to outstanding Awards, and (iii) the purchase or exercise price and the relevant appreciation base with respect to any of the foregoing, provided, however, that the number of shares subject to any Award shall always be a whole number. The Committee may, if deemed appropriate, provide for a cash payment to any Holder of an Award in connection with any adjustment made pursuant to this Section 4.2.
- (b) Notwithstanding any provision of the Plan to the contrary, in the event of a corporate merger, consolidation, acquisition of property or stock, separation, reorganization or liquidation, the Committee shall be authorized, in its discretion, (i) to provide, prior to the transaction, for the acceleration of the vesting and exercisability of, or lapse of restrictions with respect to, the Award and, if the transaction is a cash merger, provide for the termination of any portion of the Award that remains unexercised at the time of such transaction, or (ii) to cancel any such Awards and to deliver to the Holders cash in an amount that the Committee shall determine in its sole discretion is equal to the fair market value of such Awards on the date of such event, which in the case of Options or SARs shall be the excess of the Fair Market Value (as determined in sub-section (ii) of the definition of such term) of Common Stock on such date over the purchase price of the Options or the base price of the SARs, as applicable.
- (c) No adjustment or substitution pursuant to this Section 4.2 shall be made in a manner that results in noncompliance with the requirements of Section 409A of the Code, to the extent applicable.

# ARTICLE V

## ELIGIBILITY

5.1 *General*. The Persons who shall be eligible to participate in the Plan and to receive Awards under the Plan shall, subject to Section 5.2, be such Persons who are employees (including officers and directors) of or independent contractors providing services to the Company or its Subsidiaries as the Committee shall select.

or independent contractors who hold or have held Awards under the Plan or any similar or other awards under any other plan of the Company or any of its Affiliates.

5.2 *Ineligibility*. No member of the Committee, while serving as such, shall be eligible to receive an Award.

#### ARTICLE VI

#### STOCK OPTIONS

- 6.1 Grant of Options. Subject to the limitations of the Plan, the Committee shall designate from time to time those eligible Persons to be granted Options, the time when each Option shall be granted to such eligible Persons, the series and number of shares of Common Stock subject to such Option, and, subject to Section 6.2, the purchase price of the shares of Common Stock subject to such Option.
- 6.2 *Option Price.* The price at which shares may be purchased upon exercise of an Option shall be fixed by the Committee and may be no less than the Fair Market Value of the shares of the applicable series of Common Stock subject to the Option as of the date the Option is granted.
- 6.3 *Term of Options.* Subject to the provisions of the Plan with respect to death, retirement and termination of employment, the term of each Option shall be for such period as the Committee shall determine as set forth in the applicable Agreement; provided that such term may not exceed ten years.
- 6.4 Exercise of Options. An Option granted under the Plan shall become (and remain) exercisable during the term of the Option to the extent provided in the applicable Agreement and the Plan and, unless the Agreement otherwise provides, may be exercised to the extent exercisable, in whole or in part, at any time and from time to time during such term; provided, however, that subsequent to the grant of an Option, the Committee, at any time before complete termination of such Option, may accelerate the time or times at which such Option may be exercised in whole or in part (without reducing the term of such Option).
  - 6.5 Manner of Exercise.
  - (a) Form of Payment. An Option shall be exercised by written notice to the Company upon such terms and conditions as the Agreement may provide and in accordance with such other procedures for the exercise of Options as the Committee may establish from time to time. The method or methods of payment of the purchase price for the shares to be purchased upon exercise of an Option and of any amounts required by Section 10.9 shall be determined by the Committee and may consist of (i) cash, (ii) check, (iii) promissory note (subject to applicable law), (iv) whole shares of any series of Common Stock, (v) the withholding of shares of the applicable series of Common Stock issuable upon such exercise of the Option, (vi) the delivery, together with a properly executed exercise notice, of irrevocable instructions to a broker to deliver promptly to the Company the amount of sale or loan proceeds required to pay the purchase price, or (vii) any combination of the foregoing methods of payment, or such

8

other consideration and method of payment as may be permitted for the issuance of shares under the Delaware General Corporation Law. The permitted method or methods of payment of the amounts payable upon exercise of an Option, if other than in cash, shall be set forth in the applicable Agreement and may be subject to such conditions as the Committee deems appropriate.

- (b) Value of Shares. Unless otherwise determined by the Committee and provided in the applicable Agreement, shares of any series of Common Stock delivered in payment of all or any part of the amounts payable in connection with the exercise of an Option, and shares of any series of Common Stock withheld for such payment, shall be valued for such purpose at their Fair Market Value as of the exercise date.
- (c) Issuance of Shares. The Company shall effect the transfer of the shares of Common Stock purchased under the Option as soon as practicable after the exercise thereof and payment in full of the purchase price therefor and of any amounts required by Section 10.9, and within a reasonable time thereafter, such transfer shall be evidenced on the books of the Company. Unless otherwise determined by the Committee and provided in the applicable Agreement, (i) no Holder or other Person exercising an Option shall have any of the rights of a stockholder of the Company with respect to shares of Common Stock subject to an Option granted under the Plan until due exercise and full payment has been made, and (ii) no adjustment shall be made for cash dividends or other rights for which the record date is prior to the date of such due exercise and full payment.
- 6.6 Nontransferability. Unless otherwise determined by the Committee and provided in the applicable Agreement, Options shall not be transferable other than by will or the laws of descent and distribution or pursuant to a Domestic Relations Order, and, except as otherwise required pursuant to a Domestic Relations Order, Options may be exercised during the lifetime of the Holder thereof only by such Holder (or his or her court-appointed legal representative).

### ARTICLE VII

# SARS

- 7.1 Grant of SARs. Subject to the limitations of the Plan, SARs may be granted by the Committee to such eligible Persons in such numbers, with respect to any specified series of Common Stock, and at such times during the term of the Plan as the Committee shall determine. A SAR may be granted to a Holder of an Option (hereinafter called a "related Option") with respect to all or a portion of the shares of Common Stock subject to the related Option (a "Tandem SAR") or may be granted separately to an eligible employee (a "Free Standing SAR"). Subject to the limitations of the Plan, SARs shall be exercisable in whole or in part upon notice to the Company upon such terms and conditions as are provided in the Agreement.
- 7.2 Tandem SARs. A Tandem SAR may be granted either concurrently with the grant of the related Option or at any time thereafter prior to the complete exercise, termination, expiration or cancellation of such related Option. Tandem SARs shall be exercisable only at the time and to the extent that the related Option is exercisable (and may be subject to such additional limitations on exercisability as the Agreement may provide) and in no event after the

9

entitled to receive from the Company, for each share of the applicable series of Common Stock with respect to which the Tandem SAR is being exercised, consideration (in the form determined as provided in Section 7.4) equal in value to the excess of the Fair Market Value of a share of the applicable series of Common Stock with respect to which the Tandem SAR was granted on the date of exercise over the related Option purchase price per share, and (ii) the related Option with respect thereto shall be canceled automatically to the extent of the number of shares of Common Stock with respect to which the Tandem SAR was so exercised.

- 7.3 Free Standing SARs. Free Standing SARs shall be exercisable at the time, to the extent and upon the terms and conditions set forth in the applicable Agreement. The base price of a Free Standing SAR may be no less than the Fair Market Value of the applicable series of Common Stock with respect to which the Free Standing SAR was granted as of the date the Free Standing SAR is granted. Subject to the limitations of the Plan, upon the exercise of a Free Standing SAR and unless otherwise determined by the Committee and provided in the applicable Agreement, the Holder thereof shall be entitled to receive from the Company, for each share of the applicable series of Common Stock with respect to which the Free Standing SAR is being exercised, consideration (in the form determined as provided in Section 7.4) equal in value to the excess of the Fair Market Value of a share of the applicable series of Common Stock with respect to which the Free Standing SAR was granted on the date of exercise over the base price per share of such Free Standing SAR. The term of a Free Standing SAR may not exceed ten years.
- 7.4 Consideration. The consideration to be received upon the exercise of a SAR by the Holder shall be paid in cash, shares of the applicable series of Common Stock with respect to which the SAR was granted (valued at Fair Market Value on the date of exercise of such SAR), a combination of cash and such shares of the applicable series of Common Stock or such other consideration, in each case, as provided in the Agreement. No fractional shares of Common Stock shall be issuable upon exercise of a SAR, and unless otherwise provided in the applicable Agreement, the Holder will receive cash in lieu of fractional shares. Unless the Committee shall otherwise determine, to the extent a Free Standing SAR is exercisable, it will be exercised automatically for cash on its expiration date.
- 7.5 Limitations. The applicable Agreement may provide for a limit on the amount payable to a Holder upon exercise of SARs at any time or in the aggregate, for a limit on the number of SARs that may be exercised by the Holder in whole or in part for cash during any specified period, for a limit on the time periods during which a Holder may exercise SARs, and for such other limits on the rights of the Holder and such other terms and conditions of the SAR, including a condition that the SAR may be exercised only in accordance with rules and regulations adopted from time to time, as the Committee may determine. Unless otherwise so provided in the applicable Agreement, any such limit relating to a Tandem SAR shall not restrict the exercisability of the related Option. Such rules and regulations may govern the right to

10

exercise SARs granted prior to the adoption or amendment of such rules and regulations as well as SARs granted thereafter.

- 7.6 Exercise. For purposes of this Article VII, the date of exercise of a SAR shall mean the date on which the Company shall have received notice from the Holder of the SAR of the exercise of such SAR (unless otherwise determined by the Committee and provided in the applicable Agreement).
- 7.7 Nontransferability. Unless otherwise determined by the Committee and provided in the applicable Agreement, (i) SARs shall not be transferable other than by will or the laws of descent and distribution or pursuant to a Domestic Relations Order, and (ii) except as otherwise required pursuant to a Domestic Relations Order, SARs may be exercised during the lifetime of the Holder thereof only by such Holder (or his or her court-appointed legal representative).

#### ARTICLE VIII

### RESTRICTED SHARES AND RESTRICTED STOCK UNITS

- 8.1 Grant of Restricted Shares. Subject to the limitations of the Plan, the Committee shall designate those eligible Persons to be granted Awards of Restricted Shares, shall determine the time when each such Award shall be granted, and shall designate (or set forth the basis for determining) the Vesting Date or Vesting Dates for each Award of Restricted Shares, and may prescribe other restrictions, terms and conditions applicable to the vesting of such Restricted Shares in addition to those provided in the Plan. The Committee shall determine the price, if any, to be paid by the Holder for the Restricted Shares; provided, however, that the issuance of Restricted Shares shall be made for at least the minimum consideration necessary to permit such Restricted Shares to be deemed fully paid and nonassessable. All determinations made by the Committee pursuant to this Section 8.1 shall be specified in the Agreement.
- 8.2 Issuance of Restricted Shares. An Award of Restricted Shares shall be registered in a book entry account (the "Account") in the name of the Holder to whom such Restricted Shares shall have been awarded. During the Restriction Period, the Account, any certificates representing the Restricted Shares that may be issued during the Restriction Period and any securities constituting Retained Distributions shall bear a restrictive legend to the effect that ownership of the Restricted Shares (and such Retained Distributions), and the enjoyment of all rights appurtenant thereto, are subject to the restrictions, terms and conditions provided in the Plan and the applicable Agreement. Any such certificates shall remain in the custody of the Company or its designee, and the Holder shall deposit with the custodian stock powers or other instruments of assignment, each endorsed in blank, so as to permit retransfer to the Company of all or any portion of the Restricted Shares and any securities constituting Retained Distributions that shall be forfeited or otherwise not become vested in accordance with the Plan and the applicable Agreement.
- 8.3 Restrictions with Respect to Restricted Shares. During the Restriction Period, Restricted Shares shall constitute issued and outstanding shares of the applicable series of Common Stock for all corporate purposes. The Holder will have the right to vote such Restricted Shares, to receive and retain such dividends and distributions, as the Committee may

11

designate, paid or distributed on such Restricted Shares, and to exercise all other rights, powers and privileges of a Holder of shares of the applicable series of Common Stock with respect to such Restricted Shares; *except, that*, unless otherwise determined by the Committee and provided in the applicable Agreement, (i) the Holder will not be entitled to delivery of the stock certificate or certificates representing such Restricted Shares until the Restriction Period shall have expired and unless all other vesting requirements with respect thereto shall have been fulfilled or waived; (ii) the Company or its designee will retain custody of the stock certificate or certificates representing the Restricted Shares during the Restriction Period as provided in Section 8.2; (iii) other than such dividends and distributions as the Committee may designate, the Company or its designee will retain custody of all distributions ("Retained Distributions") made or declared with respect to the Restricted Shares (and such Retained Distributions will be subject to the same restrictions, terms and vesting, and other conditions as are applicable to the Restricted Shares) until such time, if ever, as the Restricted Shares with respect to which such Retained Distributions shall have been made, paid or declared shall have become vested, and such Retained Distributions shall not bear interest or be segregated in a separate account; (iv) the Holder may not sell, assign, transfer, pledge, exchange, encumber or dispose of the Restricted Shares or any Retained Distributions or such Holder's interest in any of them during the Restriction Period; and (v) a breach of any restrictions, terms or conditions provided in the Plan or established by the Committee with respect to any Restricted Shares or Retained Distributions will cause a forfeiture of such Restricted Shares and any Retained Distributions with respect thereto.

8.4 Grant of Restricted Stock Units. Subject to the limitations of the Plan, the Committee shall designate those eligible Persons to be granted Awards of Restricted Stock Units, the value of which is based, in whole or in part, on the Fair Market Value of the shares of any specified series of Common Stock. Subject to the provisions of the Plan, including any rules established pursuant to Section 8.5, Awards of Restricted Stock Units shall be subject to such terms, restrictions, conditions, vesting requirements and payment rules as the Committee may determine in its discretion, which need not be identical for each Award. Such Awards may provide for the payment of cash consideration by the Person to whom such Award is granted or provide that the Award, and any shares of Common Stock to be issued in connection therewith, if applicable, shall be delivered without the payment of cash consideration; provided, however, that the issuance of any shares of Common Stock in connection with an Award of Restricted Stock Units shall be for at least the minimum consideration necessary to permit such shares to be deemed fully paid and nonassessable. The determinations

made by the Committee pursuant to this Section 8.4 shall be specified in the applicable Agreement.

- 8.5 Restrictions with Respect to Restricted Stock Units. Any Award of Restricted Stock Units, including any shares of Common Stock which are part of an Award of Restricted Stock Units, may not be assigned, sold, transferred, pledged or otherwise encumbered prior to the date on which the shares are issued or, if later, the date provided by the Committee at the time of the Award. A breach of any restrictions, terms or conditions provided in the Plan or established by the Committee with respect to any Award of Restricted Stock Units will cause a forfeiture of such Restricted Stock Units and any Dividend Equivalents with respect thereto.
- 8.6 Issuance of Restricted Stock Units. Restricted Stock Units shall be issued at the beginning of the Restriction Period, shall not constitute issued and outstanding shares of the

12

applicable series of Common Stock, and the Holder shall not have any of the rights of a stockholder with respect to the shares of Common Stock covered by such an Award of Restricted Stock Units, in each case until such shares shall have been issued to the Holder at the end of the Restriction Period. If and to the extent that shares of Common Stock are to be issued at the end of the Restriction Period, the Holder shall be entitled to receive Dividend Equivalents with respect to the shares of Common Stock covered thereby either (i) during the Restriction Period or (ii) in accordance with the rules applicable to Retained Distributions, as the Committee may specify in the Agreement.

- 8.7 Cash Payments. In connection with any Award of Restricted Shares or Restricted Stock Units, an Agreement may provide for the payment of a cash amount to the Holder of such Awards at any time after such Awards shall have become vested. Such cash amounts shall be payable in accordance with such additional restrictions, terms and conditions as shall be prescribed by the Committee in the Agreement and shall be in addition to any other salary, incentive, bonus or other compensation payments which such Holder shall be otherwise entitled or eligible to receive from the Company.
- 8.8 Completion of Restriction Period. On the Vesting Date with respect to each Award of Restricted Shares or Restricted Stock Units and the satisfaction of any other applicable restrictions, terms and conditions, (i) all or the applicable portion of such Restricted Shares or Restricted Stock Units shall become vested, (ii) any Retained Distributions with respect to such Restricted Shares and any unpaid Dividend Equivalents with respect to such Restricted Stock Units shall become vested to the extent that the Award related thereto shall have become vested, and (iii) any cash amount to be received by the Holder with respect to such Restricted Shares or Restricted Stock Units shall become payable, all in accordance with the terms of the applicable Agreement. Any such Restricted Shares, Restricted Stock Units, Retained Distributions and any unpaid Dividend Equivalents that shall not become vested shall be forfeited to the Company, and the Holder shall not thereafter have any rights (including dividend and voting rights) with respect to such Restricted Shares, Restricted Stock Units, Retained Distributions and any unpaid Dividend Equivalents that shall have been so forfeited. The Committee may, in its discretion, provide that the delivery of any Restricted Shares, Restricted Stock Units, Retained Distributions and unpaid Dividend Equivalents that shall have become vested, and payment of any related cash amounts that shall have become payable under this Article VIII, shall be deferred until such date or dates as the recipient may elect. Any election of a recipient pursuant to the preceding sentence shall be filed in writing with the Committee in accordance with such rules and regulations, including any deadline for the making of such an election, as the Committee may provide, and shall be made in compliance with Section 409A of the Code.

# ARTICLE IX

#### CASH AWARDS AND PERFORMANCE AWARDS

9.1 Cash Awards. In addition to granting Options, SARs, Restricted Shares and Restricted Stock Units, the Committee shall, subject to the limitations of the Plan, have authority to grant to eligible Persons Cash Awards. Each Cash Award shall be subject to such terms and conditions, restrictions and contingencies, if any, as the Committee shall determine. Restrictions and contingencies limiting the right to receive a cash payment pursuant to a Cash Award shall be

13

based upon the achievement of single or multiple Performance Objectives over a performance period established by the Committee. The determinations made by the Committee pursuant to this Section 9.1 shall be specified in the applicable Agreement.

- 9.2 Designation as a Performance Award. The Committee shall have the right to designate any Award of Options, SARs, Restricted Shares or Restricted Stock Units as a Performance Award. All Cash Awards shall be designated as Performance Awards.
- Performance Objectives. The grant or vesting of a Performance Award shall be subject to the achievement of Performance Objectives over a performance period established by the Committee based upon one or more of the following business criteria that apply to the Holder, one or more business units, divisions or Subsidiaries of the Company or the applicable sector of the Company, the Company as a whole, or any entity or entities to which the Company or Subsidiaries of the Company are providing services, and if so desired by the Committee, by comparison with a peer group of companies: increased revenue; net income measures (including income after capital costs and income before or after taxes); stock price measures (including growth measures and total stockholder return); price per share of Common Stock; market share; earnings per share (actual or targeted growth); earnings before interest, taxes, depreciation and amortization (EBITDA); operating income before depreciation and amortization (OIBDA); economic value added (or an equivalent metric); market value added; debt to equity ratio; cash flow measures (including cash flow return on tangible capital, net cash flow and net cash flow before financing activities); return measures (including operating income, funds from operations, cash from operations, after-tax operating income, sales volumes, production volumes and production efficiency); expense measures (including overhead cost and general and administrative expense); margins; stockholder value; total stockholder return; proceeds from dispositions; total market value and corporate values measures (including ethics compliance, environmental and safety). Unless otherwise stated, such a Performance Objective need not be based upon an increase or positive result under a particular business criterion and could include, for example, maintaining the status quo or limiting economic losses (measured, in each case, by reference to specific business criteria). The Committee shall have the authority to determine whether the
- 9.4 Section 162(m) of the Code. Notwithstanding the foregoing provisions, if the Committee intends for a Performance Award to be granted and administered in a manner designed to preserve the deductibility of the compensation resulting from such Award in accordance with Section 162(m) of the Code, then the Performance Objectives for such particular Performance Award relative to the particular period of service to which the Performance Objectives relate shall be established by the Committee in writing (i) no later than 90 days after the beginning of such period and (ii) prior to the completion of 25% of such period.
- 9.5 Waiver of Performance Objectives. The Committee shall have no discretion to modify or waive the Performance Objectives or conditions to the grant or vesting of a Performance Award unless such Award is not intended to qualify as qualified performance-based

compensation under Section 162(m) of the Code and the relevant Agreement provides for such discretion.

#### ARTICLE X

# GENERAL PROVISIONS

# 10.1 Acceleration of Awards.

- (a) Death or Disability. If a Holder's employment shall terminate by reason of death or Disability, notwithstanding any contrary waiting period, installment period, vesting schedule or Restriction Period in any Agreement or in the Plan, unless the applicable Agreement provides otherwise: (i) in the case of an Option or SAR, each outstanding Option or SAR granted under the Plan shall immediately become exercisable in full in respect of the aggregate number of shares covered thereby; (ii) in the case of Restricted Shares, the Restriction Period applicable to each such Award of Restricted Shares shall be deemed to have expired and all such Restricted Shares and any related Retained Distributions shall become vested and any related cash amounts payable pursuant to the applicable to each such Award of Restricted Stock Units shall be deemed to have expired and all such Restricted Stock Units and any unpaid Dividend Equivalents shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be adjusted in such manner as may be provided in the Agreement.
- (b) Approved Transactions; Board Change; Control Purchase. In the event of any Approved Transaction, Board Change or Control Purchase, notwithstanding any contrary waiting period, installment period, vesting schedule or Restriction Period in any Agreement or in the Plan, unless the applicable Agreement provides otherwise: (i) in the case of an Option or SAR, each such outstanding Option or SAR granted under the Plan shall become exercisable in full in respect of the aggregate number of shares covered thereby; (ii) in the case of Restricted Shares, the Restriction Period applicable to each such Award of Restricted Shares shall be deemed to have expired and all such Restricted Shares and any related Retained Distributions shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be adjusted in such manner as may be provided in the Agreement; and (iii) in the case of Restricted Stock Units, the Restriction Period applicable to each such Award of Restricted Stock Units shall be deemed to have expired and all such Restricted Stock Units and any unpaid Dividend Equivalents shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be adjusted in such manner as may be provided in the Agreement, in each case effective upon the Board Change or Control Purchase or immediately prior to consummation of the Approved Transaction. The effect, if any, on a Cash Award of an Approved Transaction, Board Change or Control Purchase shall be prescribed in the applicable Agreement. Notwithstanding the foregoing, unless otherwise provided in the applicable Agreement, the Committee may, in its discretion, determine that any or all outstanding Awards of any or all types granted pursuant to the Plan will not vest or become exercisable on an accelerated basis in connection with an Approved

15

Transaction if effective provision has been made for the taking of such action which, in the opinion of the Committee, is equitable and appropriate to substitute a new Award for such Award or to assume such Award and to make such new or assumed Award, as nearly as may be practicable, equivalent to the old Award (before giving effect to any acceleration of the vesting or exercisability thereof), taking into account, to the extent applicable, the kind and amount of securities, cash or other assets into or for which the applicable series of Common Stock may be changed, converted or exchanged in connection with the Approved Transaction.

# 10.2 Termination of Employment.

- (a) General. If a Holder's employment shall terminate prior to an Option or SAR becoming exercisable or being exercised (or deemed exercised, as provided in Section 7.2) in full, or during the Restriction Period with respect to any Restricted Shares or any Restricted Stock Units, then such Option or SAR shall thereafter become or be exercisable, and the Holder's rights to any unvested Restricted Shares, Retained Distributions and related cash amounts and any unvested Restricted Stock Units, unpaid Dividend Equivalents and related cash amounts shall thereafter vest, in each case solely to the extent provided in the applicable Agreement; provided, however, that, unless otherwise determined by the Committee and provided in the applicable Agreement, (i) no Option or SAR may be exercised after the scheduled expiration date thereof; (ii) if the Holder's employment terminates by reason of death or Disability, the Option or SAR shall remain exercisable for a period of at least one year following such termination (but not later than the scheduled expiration of such Option or SAR); and (iii) any termination of the Holder's employment for cause will be treated in accordance with the provisions of Section 10.2(b). The effect on a Cash Award of the termination of a Holder's employment for any reason, other than for cause, shall be prescribed in the applicable Agreement.
- (b) Termination for Cause. If a Holder's employment with the Company or a Subsidiary of the Company shall be terminated by the Company or such Subsidiary for "cause" during the Restriction Period with respect to any Restricted Shares or Restricted Stock Units or prior to any Option or SAR becoming exercisable or being exercised in full or prior to the payment in full of any Cash Award (for these purposes, "cause" shall have the meaning ascribed thereto in any employment agreement to which such Holder is a party or, in the absence thereof, shall include insubordination, dishonesty, incompetence, moral turpitude, other misconduct of any kind and the refusal to perform such Holder's duties and responsibilities for any reason other than illness or incapacity; provided, however, that if such termination occurs within 12 months after an Approved Transaction or Control Purchase or Board Change, termination for "cause" shall mean only a felony conviction for fraud, misappropriation, or embezzlement), then, unless otherwise determined by the Committee and provided in the applicable Agreement, (i) all Options and SARs and all unpaid Cash Awards held by such Holder shall immediately terminate, and (ii) such Holder's rights to all Restricted Shares, Restricted Stock Units, Retained Distributions, any unpaid Dividend Equivalents and any related cash amounts shall be forfeited immediately.

16

- (c) Miscellaneous. The Committee may determine whether any given leave of absence constitutes a termination of employment; provided, however, that for purposes of the Plan, (i) a leave of absence, duly authorized in writing by the Company for military service or sickness, or for any other purpose approved by the Company if the period of such leave does not exceed 90 days, and (ii) a leave of absence in excess of 90 days, duly authorized in writing by the Company provided the employee's right to reemployment is guaranteed either by statute or contract, shall not be deemed a termination of employment. Unless otherwise determined by the Committee and provided in the applicable Agreement, Awards made under the Plan shall not be affected by any change of employment so long as the Holder continues to be an employee of the Company.
- 10.3 Right of Company to Terminate Employment. Nothing contained in the Plan or in any Award, and no action of the Company or the Committee with respect thereto, shall confer or be construed to confer on any Holder any right to continue in the employ of the Company or any of its Subsidiaries or interfere in any way with the right of the Company or any Subsidiary of the Company to terminate the employment of the Holder at any time, with or without cause, subject, however, to the provisions of any employment agreement between the Holder and the Company or any Subsidiary of the Company.
- Nonalienation of Benefits. Except as set forth herein, no right or benefit under the Plan shall be subject to anticipation, alienation, sale, assignment, hypothecation, pledge, exchange, transfer, garnishment, encumbrance or charge, and any attempt to anticipate, alienate, sell, assign, hypothecate, pledge, exchange, transfer, garnish, encumber or charge the same shall be void. No right or benefit hereunder shall in any manner be liable for or subject to the debts, contracts, liabilities or torts of the Person entitled to such benefits.
- 10.5 Written Agreement. Each Award under the Plan shall be evidenced by a written agreement, in such form as the Committee shall approve from time to time in its discretion, specifying the terms and provisions of such Award which may not be inconsistent with the provisions of the Plan; provided, however, that if more than one

type of Award is made to the same Holder, such Awards may be evidenced by a single Agreement with such Holder. Each grantee of an Option, SAR, Restricted Shares, Restricted Stock Units or Performance Award (including a Cash Award) shall be notified promptly of such grant, and a written Agreement shall be promptly delivered by the Company. Any such written Agreement may contain (but shall not be required to contain) such provisions as the Committee deems appropriate (i) to insure that the penalty provisions of Section 4999 of the Code will not apply to any stock or cash received by the Holder from the Company or (ii) to provide cash payments to the Holder to mitigate the impact of such penalty provisions upon the Holder. Any such Agreement may be supplemented or amended from time to time as approved by the Committee as contemplated by Section 10.7(b).

10.6 Designation of Beneficiaries. Each Person who shall be granted an Award under the Plan may designate a beneficiary or beneficiaries and may change such designation from time to time by filing a written designation of beneficiary or beneficiaries with the Committee on a form to be prescribed by it, provided that no such designation shall be effective unless so filed prior to the death of such Person.

17

#### 10.7 Termination and Amendment.

- (a) General. Unless the Plan shall theretofore have been terminated as hereinafter provided, no Awards may be made under the Plan on or after the fifth anniversary of the Effective Date. The Plan may be terminated at any time prior to such date and may, from time to time, be suspended or discontinued or modified or amended if such action is deemed advisable by the Committee.
- (b) Modification. No termination, modification or amendment of the Plan may, without the consent of the Person to whom any Award shall theretofore have been granted, adversely affect the rights of such Person with respect to such Award. No modification, extension, renewal or other change in any Award granted under the Plan shall be made after the grant of such Award, unless the same is consistent with the provisions of the Plan. With the consent of the Holder and subject to the terms and conditions of the Plan (including Section 10.7(a)), the Committee may amend outstanding Agreements with any Holder, including any amendment which would (i) accelerate the time or times at which the Award may be exercised and/or (ii) extend the scheduled expiration date of the Award. Without limiting the generality of the foregoing, the Committee may, but solely with the Holder's consent unless otherwise provided in the Agreement, agree to cancel any Award under the Plan and grant a new Award in substitution therefor, provided that the Award so substituted shall satisfy all of the requirements of the Plan as of the date such new Award is made. Nothing contained in the foregoing provisions of this Section 10.7(b) shall be construed to prevent the Committee from providing in any Agreement that the rights of the Holder with respect to the Award evidenced thereby shall be subject to such rules and regulations as the Committee may, subject to the express provisions of the Plan, adopt from time to time or impair the enforceability of any such provision.
- 10.8 Government and Other Regulations. The obligation of the Company with respect to Awards shall be subject to all applicable laws, rules and regulations and such approvals by any governmental agencies as may be required, including the effectiveness of any registration statement required under the Securities Act of 1933, and the rules and regulations of any securities exchange or association on which the Common Stock may be listed or quoted. For so long as any series of Common Stock are registered under the Exchange Act, the Company shall use its reasonable efforts to comply with any legal requirements (i) to maintain a registration statement in effect under the Securities Act of 1933 with respect to all shares of the applicable series of Common Stock that may be issuable, from time to time, to Holders under the Plan and (ii) to file in a timely manner all reports required to be filed by it under the Exchange Act.
- 10.9 Withholding. The Company's obligation to deliver shares of Common Stock or pay cash in respect of any Award under the Plan shall be subject to applicable federal, state and local tax withholding requirements. Federal, state and local withholding tax due at the time of an Award, upon the exercise of any Option or SAR or upon the vesting of, or expiration of restrictions with respect to, Restricted Shares or Restricted Stock Units or the satisfaction of the Performance Objectives applicable to a Performance Award, as appropriate, may, in the discretion of the Committee, be paid in shares of Common Stock already owned by the Holder or through the withholding of shares otherwise issuable to such Holder, upon such terms and

18

conditions (including the conditions referenced in Section 6.5) as the Committee shall determine. If the Holder shall fail to pay, or make arrangements satisfactory to the Committee for the payment to the Company of, all such federal, state and local taxes required to be withheld by the Company, then the Company shall, to the extent permitted by law, have the right to deduct from any payment of any kind otherwise due to such Holder an amount equal to any federal, state or local taxes of any kind required to be withheld by the Company with respect to such Award.

- 10.10 Nonexclusivity of the Plan. The adoption of the Plan by the Board shall not be construed as creating any limitations on the power of the Board to adopt such other incentive arrangements as it may deem desirable, including the granting of stock options and the awarding of stock and cash otherwise than under the Plan, and such arrangements may be either generally applicable or applicable only in specific cases.
- 10.11 Exclusion from Pension and Profit-Sharing Computation. By acceptance of an Award, unless otherwise provided in the applicable Agreement, each Holder shall be deemed to have agreed that such Award is special incentive compensation that will not be taken into account, in any manner, as salary, compensation or bonus in determining the amount of any payment under any pension, retirement or other employee benefit plan, program or policy of the Company or any Subsidiary of the Company. In addition, each beneficiary of a deceased Holder shall be deemed to have agreed that such Award will not affect the amount of any life insurance coverage, if any, provided by the Company on the life of the Holder which is payable to such beneficiary under any life insurance plan covering employees of the Company or any Subsidiary of the Company.
- 10.12 Unfunded Plan. Neither the Company nor any Subsidiary of the Company shall be required to segregate any cash or any shares of Common Stock which may at any time be represented by Awards, and the Plan shall constitute an "unfunded" plan of the Company. Except as provided in Article VIII with respect to Awards of Restricted Shares and except as expressly set forth in an Agreement, no employee shall have voting or other rights with respect to the shares of Common Stock covered by an Award prior to the delivery of such shares. Neither the Company nor any Subsidiary of the Company shall, by any provisions of the Plan, be deemed to be a trustee of any shares of Common Stock or any other property, and the liabilities of the Company and any Subsidiary of the Company to any employee pursuant to the Plan shall be those of a debtor pursuant to such contract obligations as are created by or pursuant to the Plan, and the rights of any employee, former employee or beneficiary under the Plan shall be limited to those of a general creditor of the Company or the applicable Subsidiary of the Company, as the case may be. In its sole discretion, the Board may authorize the creation of trusts or other arrangements to meet the obligations of the Company under the Plan, provided, however, that the existence of such trusts or other arrangements is consistent with the unfunded status of the Plan.
  - 10.13 Governing Law. The Plan shall be governed by, and construed in accordance with, the laws of the State of Delaware.
- 10.14 Accounts. The delivery of any shares of Common Stock and the payment of any amount in respect of an Award shall be for the account of the Company or the applicable Subsidiary of the Company, as the case may be, and any such delivery or payment shall not be

made until the recipient shall have paid or made satisfactory arrangements for the payment of any applicable withholding taxes as provided in Section 10.9.

- 10.15 Legends. Any certificate evidencing shares of Common Stock subject to an Award shall bear such legends as the Committee deems necessary or appropriate to reflect or refer to any terms, conditions or restrictions of the Award applicable to such shares, including any to the effect that the shares represented thereby may not be disposed of unless the Company has received an opinion of counsel, acceptable to the Company, that such disposition will not violate any federal or state securities laws.
- 10.16 Company's Rights. The grant of Awards pursuant to the Plan shall not affect in any way the right or power of the Company to make reclassifications, reorganizations or other changes of or to its capital or business structure or to merge, consolidate, liquidate, sell or otherwise dispose of all or any part of its business or assets.
- 10.17 Section 409A. It is the intent of the Company that Awards under this Plan comply with the requirements of, or be exempt from the application of, Section 409A of the Code and related regulations and United States Department of the Treasury pronouncements ("Section 409A"), and the provisions of this Plan will be administered, interpreted and construed accordingly. Notwithstanding anything in this Plan to the contrary, if any Plan provision or Award under the Plan would result in the imposition of an additional tax under Section 409A, that Plan provision or Award will be construed or reformed to avoid imposition of the applicable tax and no action taken to comply with Section 409A shall be deemed to adversely affect the Holder's rights to an Award.

# LIBERTY MEDIA CORPORATION 2011 NONEMPLOYEE DIRECTOR INCENTIVE PLAN

#### ARTICLE I

#### PURPOSE OF PLAN

1.1 *Purpose*. The purpose of the Plan is to provide a method whereby eligible Nonemployee Directors of the Company may be awarded additional remuneration for services rendered and encouraged to invest in capital stock of the Company, thereby increasing their proprietary interest in the Company's businesses and increasing their personal interest in the continued success and progress of the Company. The Plan is also intended to aid in attracting Persons of exceptional ability to become Nonemployee Directors of the Company.

#### ARTICLE II

#### DEFINITIONS

- 2.1 Certain Defined Terms. Capitalized terms not defined elsewhere in the Plan shall have the following meanings (whether used in the singular or plural):
  - "Account" has the meaning ascribed thereto in Section 8.2.
- "Affiliate" of the Company means any corporation, partnership, or other business association that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with the Company.
- "Agreement" means a stock option agreement, stock appreciation rights agreement, restricted shares agreement, restricted stock units agreement, or an agreement evidencing more than one type of Award, specified in Section 10.4, as any such Agreement may be supplemented or amended from time to time.
- "Approved Transaction" means any transaction in which the Board (or, if approval of the Board is not required as a matter of law, the stockholders of the Company) shall approve (i) any consolidation or merger of the Company, or binding share exchange, pursuant to which shares of Common Stock of the Company would be changed or converted into or exchanged for cash, securities, or other property, other than any such transaction in which the common stockholders of the Company immediately prior to such transaction have the same proportionate ownership of the Common Stock of, and voting power with respect to, the surviving corporation immediately after such transaction, (ii) any merger, consolidation, or binding share exchange to which the Company is a party as a result of which the Persons who are common stockholders of the

Company immediately prior thereto have less than a majority of the combined voting power of the outstanding capital stock of the Company ordinarily (and apart from the rights accruing under special circumstances) having the right to vote in the election of directors immediately following such merger, consolidation, or binding share exchange, (iii) the adoption of any plan or proposal for the liquidation or dissolution of the Company, or (iv) any sale, lease, exchange, or other transfer (in one transaction or a series of related transactions) of all, or substantially all, of the assets of the Company.

- "Award" means a grant of Options, SARs, Restricted Shares, Restricted Stock Units and/or cash under this Plan.
- "Board" means the Board of Directors of the Company.
- "Board Change" means, during any period of two consecutive years, individuals who at the beginning of such period constituted the entire Board cease for any reason to constitute a majority thereof unless the election, or the nomination for election, of each new director was approved by a vote of at least two-thirds of the directors then still in office who were directors at the beginning of the period.
- "Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute or statutes thereto. Reference to any specific Code section shall include any successor section.
  - "Common Stock" means each or any (as the context may require) series of the Company's common stock.
  - "Company" means Liberty Media Corporation (f/k/a Liberty CapStarz, Inc.), a Delaware corporation.
- "Control Purchase" means any transaction (or series of related transactions) in which any person (as such term is defined in Sections 13(d)(3) and 14(d)(2) of the Exchange Act), corporation, or other entity (other than the Company, any Subsidiary of the Company, or any employee benefit plan sponsored by the Company or any Subsidiary of the Company or any Exempt Person (as defined below)) shall become the "beneficial owner" (as such term is defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing 20% or more of the combined voting power of the then outstanding securities of the Company ordinarily (and apart from the rights accruing under special circumstances) having the right to vote in the election of directors (calculated as provided in Rule 13d-3(d) under the Exchange Act in the case of rights to acquire the Company's securities), other than in a transaction (or series of related transactions) approved by the Board. For purposes of this definition, "Exempt Person" means each of (a) the Chairman of the Board, the President and each of the directors of the Company as of the Redemption Date, and (b) the respective family members, estates, and heirs of each of the Persons referred to in clause (a) above and any trust or other investment vehicle for the primary benefit of any of such Persons or their

2

respective family members or heirs. As used with respect to any Person, the term "family member" means the spouse, siblings and lineal descendants of such Person.

- "Director Compensation" means the annual retainer and meeting fees, and any other regular cash compensation payable by the Company to a Nonemployee Director for service on the Board.
- "Disability" means the inability to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than 12 months.
- "Dividend Equivalents" means, with respect to Restricted Stock Units, to the extent specified by the Board only, an amount equal to all dividends and other distributions (or the economic equivalent thereof) which are payable to stockholders of record during the Restriction Period on a like number and kind of shares of

Common Stock.

"Domestic Relations Order" means a domestic relations order as defined by the Code or Title I of the Employee Retirement Income Security Act of 1974, or the rules thereunder.

"Effective Date" means the date on which the Redemption Date.

"Equity Security" shall have the meaning ascribed to such term in Section 3(a)(11) of the Exchange Act, and an equity security of an issuer shall have the meaning ascribed thereto in Rule 16a-1 promulgated under the Exchange Act, or any successor Rule.

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor statute or statutes thereto. Reference to any specific Exchange Act section shall include any successor section.

"Fair Market Value" of a share of any series of Common Stock on any day means (i) for Option and SAR exercise transactions effected on any third-party incentive award administration system provided by the Company, the current high bid price of a share of any series of Common Stock as reported on the consolidated transaction reporting system on the principal national securities exchange on which shares of such series of Common Stock are listed on such day or if such shares are not then listed on a national securities exchange, then as quoted by Pink OTC Markets Inc., or (ii) for all other purposes under this Plan, the last sale price (or, if no last sale price is reported, the average of the high bid and low asked prices) for a share of such series of Common Stock on such day (or, if such day is not a trading day, on the next preceding trading day) as reported on the consolidated transaction reporting system for the principal national securities exchange on which shares of such series of Common Stock are listed on such day or if such shares are not then listed on a national securities exchange, then as quoted

3

by Pink OTC Markets Inc. If for any day the Fair Market Value of a share of the applicable series of Common Stock is not determinable by any of the foregoing means, then the Fair Market Value for such day shall be determined in good faith by the Board on the basis of such quotations and other considerations as the Board deems appropriate.

"Free Standing SAR" has the meaning ascribed thereto in Section 7.1.

"Holder" means a Person who has received an Award under this Plan.

"Nonemployee Director" means an individual who is a member of the Board and who is not an employee of the Company or any Subsidiary.

"Option" means a stock option granted under Article VI.

"Person" means an individual, corporation, limited liability company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Plan" means this Liberty Media Corporation 2011 Nonemployee Director Incentive Plan.

"Redemption" means the redemption by Liberty Interactive Corporation (f/k/a Liberty Media Corporation) of all of the outstanding shares of each series of Liberty Capital common stock and Liberty Starz common stock for all of the outstanding shares of the Company's Capital common stock and Starz common stock, respectively.

"Redemption Date" means 5:00 p.m., New York City time, on the date on which the Redemption occurs.

"Restricted Shares" means shares of any series of Common Stock awarded pursuant to Section 8.1.

"Restricted Stock Unit" means a unit evidencing the right to receive in specified circumstances one share of the specified series of Common Stock or the equivalent value in cash, which right is subject to a Restriction Period or forfeiture provisions.

"Restriction Period" means a period of time beginning on the date of each Award of Restricted Shares or Restricted Stock Units and ending on the Vesting Date with respect to such Award.

"Retained Distribution" has the meaning ascribed thereto in Section 8.3.

"SARs" means stock appreciation rights, awarded pursuant to Article VII, with respect to shares of any specified series of Common Stock.

"Subsidiary" of a Person means any present or future subsidiary (as defined in Section 424(f) of the Code) of such Person or any business entity in which such Person

4

owns, directly or indirectly, 50% or more of the voting, capital, or profits interests. An entity shall be deemed a subsidiary of a Person for purposes of this definition only for such periods as the requisite ownership or control relationship is maintained.

"Tandem SARs" has the meaning ascribed thereto in Section 7.1.

"Vesting Date," with respect to any Restricted Shares or Restricted Stock Units awarded hereunder, means the date on which such Restricted Shares or Restricted Stock Units cease to be subject to a risk of forfeiture, as designated in or determined in accordance with the Agreement with respect to such Award of Restricted Shares or Restricted Stock Units pursuant to Article VIII. If more than one Vesting Date is designated for an Award of Restricted Stock Units, reference in the Plan to a Vesting Date in respect of such Award shall be deemed to refer to each part of such Award and the Vesting Date for such part.

# ARTICLE III

# ADMINISTRATION

3.1 Administration. The Plan shall be administrated by the Board, provided that it may delegate to employees of the Company certain administrative or ministerial duties in carrying out the purposes of the Plan.

- 3.2 Powers. The Board shall have full power and authority to grant to eligible Persons Options under Article VI of the Plan, SARs under Article VII of the Plan, Restricted Shares under Article VIII of the Plan, and/or Stock Units under Article IX of the Plan, to determine the terms and conditions (which need not be identical) of all Awards so granted, to interpret the provisions of the Plan and any Agreements relating to Awards granted under the Plan, and to supervise the administration of the Plan. The Board in making an Award may provide for the granting or issuance of additional, replacement, or alternative Awards upon the occurrence of specified events, including the exercise of the original Award. The Board shall have sole authority in the selection of Persons to whom Awards may be granted under the Plan and in the determination of the timing, pricing, and amount of any such Award, subject only to the express provisions of the Plan. In making determinations hereunder, the Board may take into account such factors as the Board in its discretion deems relevant.
- 3.3 Interpretation. The Board is authorized, subject to the provisions of the Plan, to establish, amend, and rescind such rules and regulations as it deems necessary or advisable for the proper administration of the Plan and to take such other action in connection with or in relation to the Plan as it deems necessary or advisable. Each action and determination made or taken pursuant to the Plan by the Board, including any interpretation or construction of the Plan, shall be final and conclusive for all purposes and upon all Persons. No member of the Board shall be liable for any action or determination made or taken by such member or the Board in good faith with respect to the Plan.

5

#### ARTICLE IV

# SHARES SUBJECT TO THE PLAN

A.1 Number of Shares. Subject to the provisions of this Article IV, the maximum number of shares of Common Stock (i) which may be issued in lieu of Director Compensation pursuant to Section 9.1 and (ii) with respect to which Awards may be granted during the term of the Plan shall be 1,500,000 shares. Shares of Common Stock will be made available from the authorized but unissued shares of the Company or from shares reacquired by the Company, including shares purchased in the open market. The shares of Common Stock subject to (i) any Award granted under the Plan that shall expire, terminate or be cancelled or annulled for any reason without having been exercised (or considered to have been exercised as provided in Section 7.2), (ii) any Award of any SARs granted under the Plan the terms of which provide for settlement in cash, and (iii) any Award of Restricted Shares or Restricted Stock Units that shall be forfeited prior to becoming vested (provided that the Holder received no benefits of ownership of such Restricted Shares or Restricted Stock Units other than voting rights and the accumulation of Retained Distributions and unpaid Dividend Equivalents that are likewise forfeited) shall again be available for purposes of the Plan. Notwithstanding the foregoing, the following shares of Common Stock may not again be made available for issuance as Awards under the Plan: (a) shares of Common Stock not issued or delivered as a result of the net settlement of an outstanding Option or SAR, (b) shares of Common Stock used to pay the purchase price or withholding taxes related to an outstanding Award, or (c) shares of Common Stock repurchased on the open market with the proceeds of an Option purchase price.

#### 4.2 Adjustments.

(a) If the Company subdivides its outstanding shares of any series of Common Stock into a greater number of shares of such series of Common Stock (by stock dividend, stock split, reclassification, or otherwise) or combines its outstanding shares of any series of Common Stock into a smaller number of shares of such series of Common Stock (by reverse stock split, reclassification, or otherwise) or if the Board determines that any stock dividend, extraordinary cash dividend, reclassification, recapitalization, reorganization, stock redemption, split-up, spin-off, combination, exchange of shares, warrants or rights offering to purchase such series of Common Stock or other similar corporate event (including mergers or consolidations other than those which constitute Approved Transactions, adjustments with respect to which shall be governed by Section 10.1(b)) affects any series of Common Stock so that an adjustment is required to preserve the benefits on potential benefits intended to be made available under the Plan, then the Board, in such manner as the Board, in its sole discretion, deems equitable and appropriate, shall make such adjustments to any or all of (i) the number and kind of shares of stock which thereafter may be awarded, optioned or otherwise made subject to the benefits contemplated by the Plan, (ii) the number and kind of shares of stock subject to outstanding Awards, and (iii) the purchase or exercise price and the relevant appreciation base with respect to any of the foregoing, provided, however, that the number of shares subject to any Award shall always be a whole number. The Board

6

may, if deemed appropriate, provide for a cash payment to any Holder of an Award in connection with any adjustment made pursuant to this Section 4.2.

- (b) Notwithstanding any provision of the Plan to the contrary, in the event of a corporate merger, consolidation, acquisition of property or stock, separation, reorganization or liquidation, the Board shall be authorized, in its discretion, (i) to provide, prior to the transaction, for the acceleration of the vesting and exercisability of, or lapse of restrictions with respect to, the Award and, if the transaction is a cash merger, provide for the termination of any portion of the Award that remains unexercised at the time of such transaction, or (ii) to cancel any such Awards and to deliver to the Holders cash in an amount that the Board shall determine in its sole discretion is equal to the fair market value of such Awards on the date of such event, which in the case of Options or SARs shall be the excess of the Fair Market Value (as determined in sub-section (ii) of the definition of such term) of Common Stock on such date over the purchase price of the Options or the base price of the SARs, as applicable.
- (c) No adjustment or substitution pursuant to this Section 4.2 shall be made in a manner that results in noncompliance with the requirements of Section 409A of the Code, to the extent applicable.

# ARTICLE V

# ELIGIBILITY

- 5.1 General. The Persons who shall be eligible to participate in the Plan and to receive Awards under the Plan shall, subject to Section 5.2, be such Persons who are Nonemployee Directors as the Board shall select. Awards may be made to Nonemployee Directors who hold or have held Awards under this Plan or any similar or other awards under any other plan of the Company or any of its Affiliates.
  - 5.2 Ineligibility. No Person who is not a Nonemployee Director shall be eligible to receive an Award.

# ARTICLE VI

# STOCK OPTIONS

- 6.1 Grant of Options. Subject to the limitations of the Plan, the Board shall designate from time to time those eligible Persons to be granted Options, the time when each Option shall be granted to such eligible Persons, the series and number of shares of Common Stock subject to such Option, and, subject to Section 6.2, the purchase price of the shares of Common Stock subject to such Option.
- 6.2 Option Price. The price at which shares may be purchased upon exercise of an Option shall be fixed by the Board and may be no less than the Fair Market Value of the shares

of the applicable series of Common Stock subject to the Option as of the date the Option is granted.

- 6.3 *Term of Options.* Subject to the provisions of the Plan with respect to death, retirement and termination of service, the term of each Option shall be for such period as the Board shall determine as set forth in the applicable Agreement; provided that such term may not exceed ten years.
- 6.4 Exercise of Options. An Option granted under the Plan shall become (and remain) exercisable during the term of the Option to the extent provided in the applicable Agreement and this Plan and, unless the Agreement otherwise provides, may be exercised to the extent exercisable, in whole or in part, at any time and from time to time during such term; provided, however, that subsequent to the grant of an Option, the Board, at any time before complete termination of such Option, may accelerate the time or times at which such Option may be exercised in whole or in part (without reducing the term of such Option).

# 6.5 Manner of Exercise.

- (a) Form of Payment. An Option shall be exercised by written notice to the Company upon such terms and conditions as the Agreement may provide and in accordance with such other procedures for the exercise of Options as the Board may establish from time to time. The method or methods of payment of the purchase price for the shares to be purchased upon exercise of an Option and of any amounts required by Section 10.8 shall be determined by the Board and may consist of (i) cash, (ii) check, (iii) promissory note (subject to applicable law), (iv) whole shares of any series of Common Stock, (v) the withholding of shares of the applicable series of Common Stock issuable upon such exercise of the Option, (vi) the delivery, together with a properly executed exercise notice, of irrevocable instructions to a broker to deliver promptly to the Company the amount of sale or loan proceeds required to pay the purchase price, or (vii) any combination of the foregoing methods of payment, or such other consideration and method of payment as may be permitted for the issuance of shares under the Delaware General Corporation Law. The permitted method or methods of payment of the amounts payable upon exercise of an Option, if other than in cash, shall be set forth in the applicable Agreement and may be subject to such conditions as the Board deems appropriate.
- (b) Value of Shares. Unless otherwise determined by the Board and provided in the applicable Agreement, shares of any series of Common Stock delivered in payment of all or any part of the amounts payable in connection with the exercise of an Option, and shares of any series of Common Stock withheld for such payment, shall be valued for such purpose at their Fair Market Value as of the exercise date.
- (c) Issuance of Shares. The Company shall effect the transfer of the shares of Common Stock purchased under the Option as soon as practicable after the exercise thereof and payment in full of the purchase price therefor and of any amounts required by Section 10.8, and within a reasonable time thereafter, such transfer shall be evidenced on

8

the books of the Company. Unless otherwise determined by the Board and provided in the applicable Agreement, (i) no Holder or other Person exercising an Option shall have any of the rights of a stockholder of the Company with respect to shares of Common Stock subject to an Option granted under the Plan until due exercise and full payment has been made, and (ii) no adjustment shall be made for cash dividends or other rights for which the record date is prior to the date of such due exercise and full payment.

6.6 Nontransferability. Unless otherwise determined by the Board and provided in the applicable Agreement, Options shall not be transferable other than by will or the laws of descent and distribution or pursuant to a Domestic Relations Order, and, except as otherwise required pursuant to a Domestic Relations Order, Options may be exercised during the lifetime of the Holder thereof only by such Holder (or his or her court-appointed legal representative).

# ARTICLE VII

#### SARS

- 7.1 Grant of SARs. Subject to the limitations of the Plan, SARs may be granted by the Board to such eligible Persons in such numbers, with respect to any specified series of Common Stock, and at such times during the term of the Plan as the Board shall determine. A SAR may be granted to a Holder of an Option (hereinafter called a "related Option") with respect to all or a portion of the shares of Common Stock subject to the related Option (a "Tandem SAR") or may be granted separately to an eligible Nonemployee Director (a "Free Standing SAR"). Subject to the limitations of the Plan, SARs shall be exercisable in whole or in part upon notice to the Company upon such terms and conditions as are provided in the Agreement.
- 7.2 Tandem SARs. A Tandem SAR may be granted either concurrently with the grant of the related Option or at any time thereafter prior to the complete exercise, termination, expiration, or cancellation of such related Option. Tandem SARs shall be exercisable only at the time and to the extent that the related Option is exercisable (and may be subject to such additional limitations on exercisability as the Agreement may provide) and in no event after the complete termination or full exercise of the related Option. Upon the exercise or termination of the related Option, the Tandem SARs with respect thereto shall be canceled automatically to the extent of the number of shares of Common Stock with respect to which the related Option was so exercised or terminated. Subject to the limitations of the Plan, upon the exercise of a Tandem SAR and unless otherwise determined by the Board and provided in the applicable Agreement, (i) the Holder thereof shall be entitled to receive from the Company, for each share of the applicable series of Common Stock with respect to which the Tandem SAR is being exercised, consideration (in the form determined as provided in Section 7.4) equal in value to the excess of the Fair Market Value of a share of the applicable series of Common Stock with respect to which the Tandem SAR was granted on the date of exercise over the related Option purchase price per share, and (ii) the related Option with respect thereto shall be canceled automatically to the extent of the number of shares of Common Stock with respect to which the Tandem SAR was so exercised.

9

- 7.3 Free Standing SARs. Free Standing SARs shall be exercisable at the time, to the extent and upon the terms and conditions set forth in the applicable Agreement. The base price of a Free Standing SAR may be no less than the Fair Market Value of the applicable series of Common Stock with respect to which the Free Standing SAR was granted as of the date the Free Standing SAR is granted. Subject to the limitations of the Plan, upon the exercise of a Free Standing SAR and unless otherwise determined by the Board and provided in the applicable Agreement, the Holder thereof shall be entitled to receive from the Company, for each share of the applicable series of Common Stock with respect to which the Free Standing SAR is being exercised, consideration (in the form determined as provided in Section 7.4) equal in value to the excess of the Fair Market Value of a share of the applicable series of Common Stock with respect to which the Free Standing SAR was granted on the date of exercise over the base price per share of such Free Standing SAR. The term of a Free Standing SAR may not exceed ten years.
- 7.4 Consideration. The consideration to be received upon the exercise of a SAR by the Holder shall be paid in cash, shares of the applicable series of Common Stock with respect to which the SAR was granted (valued at Fair Market Value on the date of exercise of such SAR), a combination of cash and such shares of the

applicable series of Common Stock or such other consideration, in each case, as provided in the Agreement. No fractional shares of Common Stock shall be issuable upon exercise of a SAR, and unless otherwise provided in the applicable Agreement, the Holder will receive cash in lieu of fractional shares. Unless the Board shall otherwise determine, to the extent a Free Standing SAR is exercisable, it will be exercised automatically for cash on its expiration date.

- 7.5 Limitations. The applicable Agreement may provide for a limit on the amount payable to a Holder upon exercise of SARs at any time or in the aggregate, for a limit on the number of SARs that may be exercised by the Holder in whole or in part for cash during any specified period, for a limit on the time periods during which a Holder may exercise SARs, and for such other limits on the rights of the Holder and such other terms and conditions of the SAR, including a condition that the SAR may be exercised only in accordance with rules and regulations adopted from time to time, as the Board may determine. Unless otherwise so provided in the applicable Agreement, any such limit relating to a Tandem SAR shall not restrict the exercisability of the related Option. Such rules and regulations may govern the right to exercise SARs granted prior to the adoption or amendment of such rules and regulations as well as SARs granted thereafter.
- 7.6 Exercise. For purposes of this Article VII, the date of exercise of a SAR shall mean the date on which the Company shall have received notice from the Holder of the SAR of the exercise of such SAR (unless otherwise determined by the Board and provided in the applicable Agreement).
- 7.7 Nontransferability. Unless otherwise determined by the Board and provided in the applicable Agreement, (i) SARs shall not be transferable other than by will or the laws of descent and distribution or pursuant to a Domestic Relations Order, and (ii) except as otherwise required pursuant to a Domestic Relations Order, SARs may be exercised during the lifetime of the Holder thereof only by such Holder (or his or her court-appointed legal representative).

10

# ARTICLE VIII

#### RESTRICTED SHARES AND RESTRICTED STOCK UNITS

- 8.1 Grant of Restricted Shares. Subject to the limitations of the Plan, the Board shall designate those eligible Persons to be granted Awards of Restricted Shares, shall determine the time when each such Award shall be granted, and shall designate (or set forth the basis for determining) the Vesting Date or Vesting Dates for each Award of Restricted Shares, and may prescribe other restrictions, terms and conditions applicable to the vesting of such Restricted Shares in addition to those provided in the Plan. The Board shall determine the price, if any, to be paid by the Holder for the Restricted Shares; provided, however, that the issuance of Restricted Shares shall be made for at least the minimum consideration necessary to permit such Restricted Shares to be deemed fully paid and nonassessable. All determinations made by the Board pursuant to this Section 8.1 shall be specified in the Agreement.
- 8.2 Issuance of Restricted Shares. An Award of Restricted Shares shall be registered in a book entry account (the "Account") in the name of the Holder to whom such Restricted Shares shall have been awarded. During the Restriction Period, the Account, any certificates representing the Restricted Shares that may be issued during the Restriction Period and any securities constituting Retained Distributions shall bear a restrictive legend to the effect that ownership of the Restricted Shares (and such Retained Distributions), and the enjoyment of all rights appurtenant thereto, are subject to the restrictions, terms, and conditions provided in the Plan and the applicable Agreement. Any such certificates shall remain in the custody of the Company or its designee, and the Holder shall deposit with the custodian stock powers or other instruments of assignment, each endorsed in blank, so as to permit retransfer to the Company of all or any portion of the Restricted Shares and any securities constituting Retained Distributions that shall be forfeited or otherwise not become vested in accordance with the Plan and the applicable Agreement.
- 8.3 Restrictions with respect to Restricted Shares. During the Restriction Period, Restricted Shares shall constitute issued and outstanding shares of the applicable series of Common Stock for all corporate purposes. The Holder will have the right to vote such Restricted Shares, to receive and retain such dividends and distributions, as the Board may designate, paid or distributed on such Restricted Shares, and to exercise all other rights, powers, and privileges of a Holder of shares of the applicable series of Common Stock with respect to such Restricted Shares; except, that, unless otherwise determined by the Board and provided in the applicable Agreement, (a) the Holder will not be entitled to delivery of the stock certificate or certificates representing such Restricted Shares until the Restriction Period shall have expired and unless all other vesting requirements with respect thereto shall have been fulfilled or waived; (b) the Company or its designee will retain custody of the stock certificate or certificates representing the Restricted Shares during the Restriction Period as provided in Section 8.2; (c) other than such dividends and distributions as the Board may designate, the Company or its designee will retain custody of all distributions ("Retained Distributions") made or declared with respect to the Restricted Shares (and such Retained Distributions will be subject to the same restrictions, terms and vesting, and other conditions as are applicable to the Restricted Shares)

11

until such time, if ever, as the Restricted Shares with respect to which such Retained Distributions shall have been made, paid, or declared shall have become vested, and such Retained Distributions shall not bear interest or be segregated in a separate account; (d) the Holder may not sell, assign, transfer, pledge, exchange, encumber, or dispose of the Restricted Shares or any Retained Distributions or such Holder's interest in any of them during the Restriction Period; and (e) a breach of any restrictions, terms, or conditions provided in the Plan or established by the Board with respect to any Restricted Shares or Retained Distributions will cause a forfeiture of such Restricted Shares and any Retained Distributions with respect thereto.

- 8.4 Grant of Restricted Stock Units. Subject to the limitations of the Plan, the Board shall designate those eligible Persons to be granted Awards of Restricted Stock Units, the value of which is based, in whole or in part, on the Fair Market Value of the shares of any specified series of Common Stock. Subject to the provisions of the Plan, including any rules established pursuant to Section 8.5, Awards of Restricted Stock Units shall be subject to such terms, restrictions, conditions, vesting requirements and payment rules as the Board may determine in its discretion, which need not be identical for each Award. Such Awards may provide for the payment of cash consideration by the Person to whom such Award is granted or provide that the Award, and any shares of Common Stock to be issued in connection therewith, if applicable, shall be delivered without the payment of cash consideration; provided, however, that the issuance of any shares of Common Stock in connection with an Award of Restricted Stock Units shall be for at least the minimum consideration necessary to permit such shares to be deemed fully paid and nonassessable. The determinations made by the Board pursuant to this Section 8.4 shall be specified in the applicable Agreement.
- 8.5 Restrictions with Respect to Restricted Stock Units. Any Award of Restricted Stock Units, including any shares of Common Stock which are part of an Award of Restricted Stock Units, may not be assigned, sold, transferred, pledged or otherwise encumbered prior to the date on which the shares are issued or, if later, the date provided by the Board at the time of the Award. A breach of any restrictions, terms or conditions provided in the Plan or established by the Board with respect to any Award of Restricted Stock Units will cause a forfeiture of such Restricted Stock Units and any Dividend Equivalents with respect thereto.
- 8.6 Issuance of Restricted Stock Units. Restricted Stock Units shall be issued at the end of the Restriction Period, shall not constitute issued and outstanding shares of the applicable series of Common Stock, and the Holder shall not have any of the rights of a stockholder with respect to the shares of Common Stock covered by such an Award of Restricted Stock Units, in each case until such shares shall have been issued to the Holder at the end of the Restriction Period. If and to the extent that shares of Common Stock are to be issued at the end of the Restriction Period, the Holder shall be entitled to receive Dividend Equivalents with respect to the shares of Common Stock covered thereby either (i) during the Restriction Period or (ii) in accordance with the rules applicable to Retained Distributions, as the Board may specify in the Agreement.
- 8.7 Cash Payments. In connection with any Award of Restricted Shares or Restricted Stock Units, an Agreement may provide for the payment of a cash amount to the Holder of such Awards at any time after such Awards shall have become vested. Such cash amounts shall be payable in accordance with such additional restrictions, terms, and conditions as shall be

prescribed by the Board in the Agreement and shall be in addition to any other compensation payments which such Holder shall be otherwise entitled or eligible to receive from the Company.

8.8 Completion of Restriction Period. On the Vesting Date with respect to each Award of Restricted Shares or Restricted Stock Units and the satisfaction of any other applicable restrictions, terms, and conditions, (a) all or the applicable portion of such Restricted Shares or Restricted Stock Units shall become vested, (b) any Retained Distributions with respect to such Restricted Stock Units shall become vested to the extent that the Awards related thereto shall have become vested, and (c) any cash amount to be received by the Holder with respect to such Restricted Shares or Restricted Stock Units shall become payable, all in accordance with the terms of the applicable Agreement. Any such Restricted Shares, Restricted Stock Units, Retained Distributions, and any unpaid Dividend Equivalents that shall not become vested shall be forfeited to the Company, and the Holder shall not thereafter have any rights (including dividend and voting rights) with respect to such Restricted Stock Units, Retained Distributions, and any unpaid Dividend Equivalents that shall have been so forfeited. The Board may, in its discretion, provide that the delivery of any Restricted Shares, Restricted Stock Units, Retained Distributions, and unpaid Dividend Equivalents that shall have become vested, and payment of any related cash amounts that shall have become payable under this Article VIII, shall be deferred until such date or dates as the recipient may elect. Any election of a recipient pursuant to the preceding sentence shall be filed in writing with the Board in accordance with such rules and regulations, including any deadline for the making of such an election, as the Board may provide, and shall be made in compliance with Section 409A of the Code.

#### ARTICLE IX

#### STOCK AWARDS IN LIEU OF CASH DIRECTOR FEES

- 9.1 General. Each Nonemployee Director shall have the option to elect to receive shares of one or more series of Common Stock, as prescribed by the Board, in lieu of all or part of the Director Compensation otherwise payable by the Company during each calendar quarter. Subject to any applicable Purchase Restriction as described in Section 9.3, to the extent a Nonemployee Director has elected in writing to receive stock in lieu of Director Compensation, such Nonemployee Director will receive shares of Common Stock on the last day of the calendar quarter for which the Director Compensation was earned. The Director Compensation shall be converted to a number of shares of Common Stock equal in value to such Director Compensation based on the Fair Market Value of such shares on the last day of the calendar quarter for which the Director Compensation would otherwise be payable to the Nonemployee Director, with any fractional shares paid in cash. For this purpose, if the last day of the calendar quarter is not a trading day, then Fair Market Value shall be determined as of the next succeeding trading day. Any shares issued in lieu of Director Compensation shall be issued free of all restrictions except as required by law.
- 9.2 Timing of Election. A Nonemployee Director's election pursuant to Section 9.1 must be made no later than the 30th calendar day (or such other day as the Board may prescribe) prior to the end of the calendar quarter to which the election applies in accordance with the

13

procedures established by the Board. Once an election is made with respect to a particular calendar quarter, it may not be withdrawn or substituted unless the Board determines, in its sole discretion, that the withdrawal or substitution is occasioned by an extraordinary or unanticipated event.

- 9.3 Election Void During Restricted Period. If, on the date shares would be purchased pursuant to an election under Section 9.1, there is in place any restriction under applicable law (including a blackout period under the Sarbanes-Oxley Act of 2002) or the rules of the principal national securities exchange on which shares of the applicable series of Common Stock are traded (a "Purchase Restriction") which would prohibit the Nonemployee Director from making such a purchase, then such shares shall be purchased on the first trading day following the lapse or removal of the Purchase Restriction based on the Fair Market Value of the shares on such trading day.
- 9.4 *Conditions*. Nothing contained herein shall preclude the Board, in its sole discretion, from imposing conditions on any election made under Section 9.1, including the conditions described in Section 9.3.

#### ARTICLE X

# GENERAL PROVISIONS

### 10.1 Acceleration of Awards.

- (a) Death or Disability. If a Holder's service shall terminate by reason of death or Disability, notwithstanding any contrary waiting period, installment period, vesting schedule, or Restriction Period in any Agreement or in the Plan, unless the applicable Agreement provides otherwise: (i) in the case of an Option or SAR, each outstanding Option or SAR granted under the Plan shall immediately become exercisable in full in respect of the aggregate number of shares covered thereby; (ii) in the case of Restricted Shares, the Restriction Period applicable to each such Award of Restricted Shares shall be deemed to have expired and all such Restricted Shares and any related Retained Distributions shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be Restricted Stock Units, the Restriction Period applicable to each such Award of Restricted Stock Units shall be deemed to have expired and all such Restricted Stock Units and any unpaid Dividend Equivalents shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be adjusted in such manner as may be provided in the Agreement.
- (b) Approved Transactions; Board Change; Control Purchase. In the event of any Approved Transaction, Board Change or Control Purchase, notwithstanding any contrary waiting period, installment period, vesting schedule, or Restriction Period in any Agreement or in the Plan, unless the applicable Agreement provides otherwise: (i) in the case of an Option or SAR, each such outstanding Option or SAR granted under the Plan

14

shall become exercisable in full in respect of the aggregate number of shares covered thereby; (ii) in the case of Restricted Shares, the Restriction Period applicable to each such Award of Restricted Shares shall be deemed to have expired and all such Restricted Shares and any related Retained Distributions shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be adjusted in such manner as may be provided in the Agreement; and (iii) in the case of Restricted Stock Units, the Restriction Period applicable to each such Award of Restricted Stock Units shall be deemed expired and all such Restricted Stock Units and any unpaid Dividend Equivalents shall become vested and any related cash amounts payable pursuant to the applicable Agreement shall be adjusted in such manner as may be provided in the Agreement, in each case effective upon the Board Change or Control Purchase or immediately prior to consummation of the Approved Transaction. Notwithstanding the foregoing, unless otherwise provided in the applicable Agreement, the Board may, in its discretion, determine that any or all outstanding Awards of any or all types granted pursuant to the Plan will not vest or become exercisable on an accelerated basis in connection with an Approved Transaction if effective provision has been made for the taking of such action which, in the opinion of the Board, is equitable and appropriate to substitute a new Award for such Award or to assume such Award and to make such new or assumed Award, as nearly as may be practicable, equivalent to the old Award (before giving

effect to any acceleration of the vesting or exercisability thereof), taking into account, to the extent applicable, the kind and amount of securities, cash, or other assets into or for which the applicable series of Common Stock may be changed, converted, or exchanged in connection with the Approved Transaction.

# 10.2 Termination of Service.

- (a) General. If a Holder's service shall terminate prior to an Option or SAR becoming exercisable or being exercised (or deemed exercised, as provided in Section 7.2), in full, or during the Restriction Period with respect to any Restricted Shares or any Restricted Stock Units, then such Option or SAR shall thereafter become or be exercisable, and the Holder's rights to any unvested Restricted Shares, Retained Distributions and related cash amounts and any unvested Restricted Stock Units, unpaid Dividend Equivalents and related cash amounts shall thereafter vest, in each case solely to the extent provided in the applicable Agreement; provided, however, that, unless otherwise determined by the Board and provided in the applicable Agreement, (i) no Option or SAR may be exercised after the scheduled expiration date thereof; (ii) if the Holder's service terminates by reason of death or Disability, the Option or SAR shall remain exercisable for a period of at least one year following such termination (but not later than the scheduled expiration of such Option or SAR); and (iii) any termination of the Holder's service for cause will be treated in accordance with the provisions of Section 10.2(b).
- (b) Termination for Cause. If a Holder's service on the Board shall be terminated by the Company during the Restriction Period with respect to any Restricted Shares or Restricted Stock Units, or prior to any Option or SAR becoming exercisable or being exercised in full, for "cause" (for these purposes, cause shall include, but not be

15

limited to, insubordination, dishonesty, incompetence, moral turpitude, other misconduct of any kind, and the refusal to perform such Holder's duties and responsibilities for any reason other than illness or incapacity; provided, however, that if such termination occurs within 12 months after an Approved Transaction or Control Purchase or Board Change, termination for cause shall mean only a felony conviction for fraud, misappropriation, or embezzlement), then, unless otherwise determined by the Board and provided in the applicable Agreement, (i) all Options and SARs held by such Holder shall immediately terminate and (ii) such Holder's rights to all Restricted Shares, Restricted Stock Units, Retained Distributions, any unpaid Dividend Equivalents and any related cash amounts shall be forfeited immediately.

- 10.3 Nonalienation of Benefits. Except as set forth herein, no right or benefit under the Plan shall be subject to anticipation, alienation, sale, assignment, hypothecation, pledge, exchange, transfer, encumbrance, or charge, and any attempt to anticipate, alienate, sell, assign, hypothecate, pledge, exchange, transfer, encumber or charge the same shall be void. No right or benefit hereunder shall in any manner be liable for or subject to the debts, contracts, liabilities, or torts of the Person entitled to such benefits.
- Written Agreement. Each Award under the Plan shall be evidenced by a written agreement, in such form as the Board shall approve from time to time in its discretion, specifying the terms and provisions of such Award which may not be inconsistent with the provisions of the Plan; provided, however, that if more than one type of Award is made to the same Holder, such Awards may be evidenced by a single Agreement with such Holder. Each grantee of an Option, SAR, Restricted Shares or Restricted Stock Units shall be notified promptly of such grant, and a written Agreement shall be promptly delivered by the Company. Any such written Agreement may contain (but shall not be required to contain) such provisions as the Board deems appropriate (i) to insure that the penalty provisions of Section 4999 of the Code will not apply to any stock or cash received by the Holder from the Company or (ii) to provide cash payments to the Holder to mitigate the impact of such penalty provisions upon the Holder. Any such Agreement may be supplemented or amended from time to time as approved by the Board as contemplated by Section 10.6(b).
- 10.5 Designation of Beneficiaries. Each Person who shall be granted an Award under the Plan may designate a beneficiary or beneficiaries and may change such designation from time to time by filing a written designation of beneficiary or beneficiaries with the Board on a form to be prescribed by it, provided that no such designation shall be effective unless so filed prior to the death of such Person.

#### 10.6 Termination and Amendment.

(a) General. Unless the Plan shall theretofore have been terminated as hereinafter provided, no Awards may be made under the Plan on or after the fifth anniversary of the Effective Date. The Plan may be terminated at any time prior to such date and may, from time to time, be suspended or discontinued or modified or amended if such action is deemed advisable by the Board.

16

- (b) Modification. No termination, modification or amendment of the Plan may, without the consent of the Person to whom any Award shall theretofore have been granted, adversely affect the rights of such Person with respect to such Award. No modification, extension, renewal, or other change in any Award granted under the Plan shall be made after the grant of such Award, unless the same is consistent with the provisions of the Plan. With the consent of the Holder and subject to the terms and conditions of the Plan (including Section 10.6(a)), the Board may amend outstanding Agreements with any Holder, including any amendment which would (i) accelerate the time or times at which the Award may be exercised and/or (ii) extend the scheduled expiration date of the Award. Without limiting the generality of the foregoing, the Board may, but solely with the Holder's consent unless otherwise provided in the Agreement, agree to cancel any Award under the Plan and grant a new Award in substitution therefor, provided that the Award so substituted shall satisfy all of the requirements of the Plan as of the date such new Award is made. Nothing contained in the foregoing provisions of this Section 10.6(b) shall be construed to prevent the Board from providing in any Agreement that the rights of the Holder with respect to the Award evidenced thereby shall be subject to such rules and regulations as the Board may, subject to the express provisions of the Plan, adopt from time to time or impair the enforceability of any such provision.
- 10.7 Government and Other Regulations. The obligation of the Company with respect to Awards shall be subject to all applicable laws, rules, and regulations and such approvals by any governmental agencies as may be required, including the effectiveness of any registration statement required under the Securities Act of 1933, and the rules and regulations of any securities exchange or association on which the Common Stock may be listed or quoted. For so long as any series of Common Stock are registered under the Exchange Act, the Company shall use its reasonable efforts to comply with any legal requirements (i) to maintain a registration statement in effect under the Securities Act of 1933 with respect to all shares of the applicable series of Common Stock that may be issuable, from time to time, to Holders under the Plan and (ii) to file in a timely manner all reports required to be filed by it under the Exchange Act.
- 10.8 Withholding. The Company's obligation to deliver shares of Common Stock or pay cash in respect of any Award under the Plan shall be subject to applicable federal, state, and local tax withholding requirements. Federal, state, and local withholding tax due at the time of an Award, upon the exercise of any Option or SAR or upon the vesting of, or expiration of restrictions with respect to, Restricted Shares or Restricted Stock Units, as appropriate, may, in the discretion of the Board, be paid in shares of Common Stock already owned by the Holder or through the withholding of shares otherwise issuable to such Holder, upon such terms and conditions (including the conditions referenced in Section 6.5) as the Board shall determine. If the Holder shall fail to pay, or make arrangements satisfactory to the Board for the payment to the Company of, all such federal, state and local taxes required to be withheld by the Company, then the Company shall, to the extent permitted by law, have the right to deduct from any payment of any kind otherwise due to such Holder an amount equal to any federal, state, or local taxes of any kind required to be withheld by the Company with respect to such Award.

- 10.9 Nonexclusivity of the Plan. The adoption of the Plan by the Board shall not be construed as creating any limitations on the power of the Board to adopt such other incentive arrangements as it may deem desirable, including the granting of stock options and the awarding of stock and cash otherwise than under the Plan, and such arrangements may be either generally applicable or applicable only in specific cases.
- 10.10 Exclusion from Other Plans. By acceptance of an Award, unless otherwise provided in the applicable Agreement, each Holder shall be deemed to have agreed that such Award is special incentive compensation that will not be taken into account, in any manner, as compensation or bonus in determining the amount of any payment under any pension, retirement or other benefit plan, program, or policy of the Company or any Subsidiary of the Company. In addition, each beneficiary of a deceased Holder shall be deemed to have agreed that such Award will not affect the amount of any life insurance coverage, if any, provided by the Company on the life of the Holder which is payable to such beneficiary under any life insurance plan of the Company or any Subsidiary of the Company. Director Compensation elected to be received in the form of stock in lieu of cash shall be treated as regular compensation for purposes of any Director retirement or life insurance plan.
- 10.11 Unfunded Plan. Neither the Company nor any Subsidiary of the Company shall be required to segregate any cash or any shares of Common Stock which may at any time be represented by Awards, and the Plan shall constitute an "unfunded" plan of the Company. Except as provided in Article VIII with respect to Awards of Restricted Shares and except as expressly set forth in an Agreement, no Holder shall have voting or other rights with respect to the shares of Common Stock covered by an Award prior to the delivery of such shares. Neither the Company nor any Subsidiary of the Company shall, by any provisions of the Plan, be deemed to be a trustee of any shares of Common Stock or any other property, and the liabilities of the Company and any Subsidiary of the Company to any Holder pursuant to the Plan shall be those of a debtor pursuant to such contract obligations as are created by or pursuant to the Plan, and shall be limited to those of a general creditor of the Company or the applicable Subsidiary of the Company, as the case may be. In its sole discretion, the Board may authorize the creation of trusts or other arrangements to meet the obligations of the Company under the Plan, provided, however, that the existence of such trusts or other arrangements is consistent with the unfunded status of the Plan.
  - 10.12 Governing Law. The Plan shall be governed by, and construed in accordance with, the laws of the State of Delaware.
- 10.13 Accounts. The delivery of any shares of Common Stock and the payment of any amount in respect of an Award shall be for the account of the Company or the applicable Subsidiary of the Company, as the case may be, and any such delivery or payment shall not be made until the recipient shall have paid or made satisfactory arrangements for the payment of any applicable withholding taxes as provided in Section 10.8.

18

- 10.14 Legends. Any certificate evidencing shares of Common Stock subject to an Award shall bear such legends as the Board deems necessary or appropriate to reflect or refer to any terms, conditions, or restrictions of the Award applicable to such shares, including any to the effect that the shares represented thereby may not be disposed of unless the Company has received an opinion of counsel, acceptable to the Company, that such disposition will not violate any federal or state securities laws.
- 10.15 Company's Rights. The grant of Awards pursuant to the Plan shall not affect in any way the right or power of the Company to make reclassifications, reorganizations, or other changes of or to its capital or business structure or to merge, consolidate, liquidate, sell, or otherwise dispose of all or any part of its business or assets.
- 10.16 Section 409A. It is the intent of the Company that Awards under this Plan comply with the requirements of, or be exempt from the application of, Section 409A of the Code and related regulations and United States Department of the Treasury pronouncements ("Section 409A"), and the provisions of this Plan will be administered, interpreted and construed accordingly. Notwithstanding anything in this Plan to the contrary, if any Plan provision or Award under the Plan would result in the imposition of an additional tax under Section 409A, that Plan provision or Award will be construed or reformed to avoid imposition of the applicable tax and no action taken to comply with Section 409A shall be deemed to adversely affect the Holder's rights to an Award.

#### LIBERTY MEDIA CORPORATION

#### TRANSITIONAL STOCK ADJUSTMENT PLAN

#### ARTICLE I

#### PURPOSE AND AMENDMENT OF PLAN

1.1 *Purpose.* The purpose of the Plan is to provide for the supplemental grant of cash awards, stock options to purchase the common stock of Liberty Media Corporation (f/k/a Liberty CapStarz, Inc.) (the "Company"), stock appreciation rights related to the Company's common stock, and restricted shares of the Company's common stock to holders of certain outstanding cash awards, options, stock appreciation rights and restricted shares issued under certain stock-based plans administered by Liberty Interactive Corporation (f/k/a Liberty Media Corporation) ("LMC") in connection with adjustments made to outstanding cash awards, stock incentive awards and restricted shares of LMC Capital Common Stock (as defined below) and LMC Starz Common Stock (as defined below) as a result of the split-off of the Company from LMC.

#### ARTICLE II

#### DEFINITIONS

2.1 Certain Defined Terms. For purposes of the Plan, the following terms shall have the meanings below stated.

"Approved Transaction" means any transaction in which the Board (or, if approval of the Board is not required as a matter of law, the stockholders of the Company) shall approve (i) any consolidation or merger of the Company, or binding share exchange, pursuant to which shares of Common Stock of the Company would be changed or converted into or exchanged for cash, securities, or other property, other than any such transaction in which the common stockholders of the Company immediately prior to such transaction have the same proportionate ownership of the Common Stock of, and voting power with respect to, the surviving corporation immediately after such transaction, (ii) any merger, consolidation or binding share exchange to which the Company is a party as a result of which the Persons who are common stockholders of the Company immediately prior thereto have less than a majority of the combined voting power of the outstanding capital stock of the Company ordinarily (and apart from the rights accruing under special circumstances) having the right to vote in the election of directors immediately following such merger, consolidation or binding share exchange, (iii) the adoption of any plan or proposal for the liquidation or dissolution of the Company, or (iv) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, of the assets of the Company.

"Awards" means collectively Cash Awards and Stock Incentives.

"Board" means the Board of Directors of the Company.

1

"Board Change" means, during any period of two consecutive years, individuals who at the beginning of such period constituted the entire Board cease for any reason to constitute a majority thereof unless the election, or the nomination for election, of each new director was approved by a vote of at least two-thirds of the directors then still in office who were directors at the beginning of the period.

"Cash Award" means an award denominated in cash and granted by the Company pursuant to Section 8.1.

"Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute or statutes thereto. Reference to any specific Code Section shall include any successor section.

"Committee" means the committee of the Board appointed to administer this Plan pursuant to Article IX.

"Common Stock" means each or any (as the context may require) series of the Company's common stock.

"Company" means Liberty Media Corporation (f/k/a Liberty CapStarz, Inc.), a Delaware corporation, and any successor thereto.

"Control Purchase" means any transaction (or series of related transactions) in which (1) any person (as such term is defined in Sections 13(d)(3) and 14(d) (2) of the Exchange Act), corporation or other entity (other than the Company, any Subsidiary of the Company or any employee benefit plan sponsored by the Company or any Subsidiary of the Company) shall purchase any Common Stock of the Company (or securities convertible into Common Stock of the Company) for cash, securities or any other consideration pursuant to a tender offer or exchange offer, without the prior consent of the Board, or (2) any person (as such term is so defined), corporation or other entity (other than the Company, any Subsidiary of the Company, any employee benefit plan sponsored by the Company or any Subsidiary of the Company or any Exempt Person (as defined below)) shall become the "beneficial owner" (as such term is defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing 20% or more of the combined voting power of the then outstanding securities of the Company ordinarily (and apart from the rights accruing under special circumstances) having the right to vote in the election of directors (calculated as provided in Rule 13d-3(d) under the Exchange Act in the case of rights to acquire the Company's securities), other than in a transaction (or series of related transactions) approved by the Board. For purposes of this definition, "Exempt Person" means each of (a) the Chairman of the Board, the President and each of the directors of the Company as of the Redemption Date, and (b) the respective family members of each of the persons referred to in clause (a) above and any trust or other investment vehicle for the primary benefit of any of such persons or their respective family members or heirs. As used with respect to any person, the term "family member" means the spouse, siblings and lineal descendants of such person.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

2

"Fair Market Value" of a share of any series of Common Stock on any day means (i) for Option and SAR exercise transactions effected on any third-party incentive award administration system provided by the Company, the current high bid price of a share of any series of Common Stock as reported on the consolidated transaction reporting system on the principal national securities exchange on which shares of such series of Common Stock are listed on such day or if such shares are not then listed on a national securities exchange, then as quoted by Pink OTC Markets Inc., or (ii) for all other purposes under this Plan, the last sale price (or, if no last sale price is reported, the average of the high bid and low asked prices) for a share of such series of Common Stock on such day (or, if such day is not a trading day, on the next preceding trading day) as reported on the consolidated transaction reporting system for the principal national securities exchange on which shares of such series of Common Stock are listed on such day or if such shares are not then listed on a national securities exchange, then as quoted by Pink OTC Markets Inc. If for any day the Fair Market Value of a share of the applicable series of Common Stock is not determinable by any of the foregoing means, then the Fair Market Value for such day shall be determined in good faith

by the Committee on the basis of such quotations and other considerations as the Committee deems appropriate.

"Incentive Plan" means the Liberty Media Corporation 2010 Incentive Plan, the Liberty Media Corporation 2007 Incentive Plan, Liberty Media Corporation 2000 Incentive Plan (As Amended and Restated Effective February 22, 2007), Liberty Media Corporation 2002 Nonemployee Director Incentive Plan (As Amended and Restated Effective August 15, 2007), in each case as amended on or prior to the date hereof, and any other stock option or incentive plan assumed by LMC pursuant to which any Participant holds an outstanding LMC Award as of the Redemption Date. Depending on the context, "Incentive Plan" shall mean all of such plans or a particular one of such plans.

"LMC Award" means (1) an LMC Cash Award, (2) an unexercised and unexpired option to purchase LMC Capital Common Stock, (3) an unexercised and unexpired option to purchase LMC Starz Common Stock, (4) an LMC Capital SAR, (5) an LMC Starz SAR, (6) an unvested award of restricted shares of LMC Capital Common Stock, or (7) an unvested award of restricted shares of LMC Starz Common Stock.

"LMC Corporate Holder" means an individual who, as of the Redemption Date, is (1) an employee or consultant of LMC or a Qualifying Subsidiary or (2) a member of the board of directors of LMC or a Qualifying Subsidiary. The Committee may, in its discretion, determine that (i) an individual who does not meet any of the foregoing criteria should be classified as an LMC Corporate Holder or (ii) an individual who otherwise would qualify as an LMC Corporate Holder, should not be classified as such.

"LMC Capital Common Stock" means each or any (as the context may require) series of Liberty Capital common stock, par value \$.01 per share.

"LMC Cash Award" means a cash award granted pursuant to an Incentive Plan.

"LMC Starz Common Stock" means each or any (as the context may require) series of Liberty Starz common stock, par value \$.01 per share.

3

"LMC Capital SAR" means a stock appreciation right with respect to any series of LMC Capital Common Stock.

"LMC Starz SAR" means a stock appreciation right with respect to any series of LMC Starz Common Stock

"Option" means an option to purchase Common Stock, granted by the Company to a Participant pursuant to Section 6.1 of the Plan.

"Participant" means a person who is an LMC Corporate Holder and who, as of the Redemption Date, holds an outstanding LMC Award.

"Person" means an individual, corporation, limited liability company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Plan" means the Liberty Media Corporation Transitional Stock Adjustment Plan, as set forth herein and as from time to time amended.

"Qualifying Subsidiary" means a former direct or indirect subsidiary of LMC (or its predecessor Liberty Interactive LLC (f/k/a Liberty Media LLC)), any successor of any such former subsidiary, and the parent company (directly or indirectly) of any such former subsidiary or successor, including without limitation the Company, Ascent Media Corporation, DIRECTV, Discovery Communications, Inc. and Liberty Global, Inc.

"Redemption" means the redemption by LMC of (i) all of the outstanding shares of each series of LMC Capital Common Stock for all of the outstanding shares of Splitco Capital Common Stock and (ii) all of the outstanding shares of each series of LMC Starz Common Stock for all of the outstanding shares of Splitco Starz Common Stock.

"Redemption Date" means 5:00 p.m., New York City time, on the date on which the Redemption occurs.

per share.

Section 5.1.

"Restricted Stock Award" means an award of restricted shares of Common Stock, granted by the Company to a Participant pursuant to Section 5.1.

"SARs" means stock appreciation rights, awarded pursuant to Section 7.1, with respect to shares of any specified series of Common Stock.

"Splitco Capital Common Stock" means each or any (as the context may require) series of the Company's Liberty Capital common stock, par value \$.01

"Splitco Starz Common Stock" means each or any (as the context may require) series of the Company's Liberty Starz common stock, par value \$.01 per share.

"Splitco Capital Option" means an option to purchase Splitco Capital Common Stock, granted by the Company pursuant to Section 6.1.

4

"Splitco Starz Option" means an option to purchase Splitco Starz Common Stock, granted by the Company pursuant to Section 6.1.

"Splitco Capital Restricted Stock Award" means an award of restricted shares of Splitco Capital Common Stock, granted by the Company pursuant to

"Splitco Starz Restricted Stock Award" means an award of restricted shares of Splitco Starz Common Stock, granted by the Company pursuant to Section 5.1.

"Splitco Capital SARs" means stock appreciation rights, awarded pursuant to Section 7.1, with respect to shares of any specified series of Splitco Capital Common Stock.

"Splitco Starz SARs" means stock appreciation rights, awarded pursuant to Section 7.1, with respect to shares of any specified series of Splitco Starz Common Stock.

"Stock Incentives" means collectively the Restricted Stock Awards, SARs and Options.

"Subsidiary" of a Person means any present or future subsidiary (as defined in Section 424(f) of the Code) of such Person or any business entity in which such Person owns, directly or indirectly, 50% or more of the voting, capital or profits interests. An entity shall be deemed a subsidiary of a Person for purposes of this

definition only for such periods as the requisite ownership or control relationship is maintained.

#### ARTICLE III

# RESERVATION OF SHARES

The aggregate number of shares of Splitco Capital Common Stock which may be issued under this Plan shall not exceed 4,520,000 shares, subject to adjustment as hereinafter provided. The aggregate number of shares of Splitco Starz Common Stock which may be issued under this Plan shall not exceed 3,782,000 shares, subject to adjustment as hereinafter provided. Any part of such 4,520,000 shares of Splitco Capital Common Stock and such 3,782,000 shares of Splitco Starz Common Stock may be issued pursuant to Splitco Capital Restricted Stock Awards and Splitco Starz Restricted Stock Awards, respectively. The shares of Common Stock which may be granted pursuant to Stock Incentives will consist of either authorized but unissued shares of Common Stock or shares of Common Stock which have been issued and reacquired by the Company, including shares purchased in the open market. The total number of shares authorized under this Plan shall be subject to increase or decrease in order to give effect to the adjustment provision of Section 11.3 and to give effect to any amendment adopted as provided in Section 11.1.

#### ARTICLE IV

#### PARTICIPATION IN PLAN

4.1 Eligibility to Receive Awards. Awards under this Plan may be granted only to persons who are Participants.

5

4.2 Participation Not Guarantee of Employment. Nothing in this Plan or in the instrument evidencing the grant of an Award shall in any manner be construed to limit in any way the right of the Company, LMC or any of their respective Subsidiaries to terminate a Participant's employment at any time, without regard to the effect of such termination on any rights such Participant would otherwise have under the Plan or any Incentive Plan, or give any right to such a Participant to remain employed by the Company, LMC or any of their respective Subsidiaries in any particular position or at any particular rate of compensation.

#### ARTICLE V

#### STOCK AWARDS

- 5.1 Grant of Restricted Stock Awards.
- (a) Grant. Splitco Capital Restricted Stock Award(s) or Splitco Starz Restricted Stock Award(s) shall be granted to each Participant who, as of the Redemption Date, holds an outstanding LMC Award(s) consisting of unvested restricted shares of LMC Capital Common Stock or LMC Starz Common Stock, respectively.
- (b) Award of Shares. Each Splitco Capital Restricted Stock Award or Splitco Starz Restricted Stock Award shall be for the number and series of shares of Splitco Capital Common Stock or Splitco Starz Common Stock that such holder will receive in exchange for such holder's existing restricted shares of LMC Capital Common Stock or LMC Starz Common Stock, respectively, as a result of the application of the redemption ratio in the Redemption. Each Splitco Capital Restricted Stock Award or Splitco Starz Restricted Stock Award and the restricted shares of Splitco Capital Common Stock or Splitco Starz Common Stock issued thereunder shall continue to be subject to all the terms and conditions of the applicable Incentive Plan and associated instrument under which the corresponding award of restricted shares of LMC Capital Common Stock or LMC Starz Common Stock, respectively, was made and any such terms, conditions and restrictions as may be determined to be appropriate by the Committee.
- (c) Lapse of Restrictions. The restrictions on each Splitco Capital Restricted Stock Award or Splitco Starz Restricted Stock Award shall lapse in accordance with the terms and conditions of the applicable Incentive Plan and associated instrument under which the corresponding award of restricted shares of LMC Capital Common Stock or LMC Starz Common Stock, respectively, was made; provided, however, that a Participant's employment or service with the Company, LMC, a Qualifying Subsidiary or any of their respective Subsidiaries shall be deemed to be employment or service with the Company and LMC for all purposes under a Restricted Stock Award.
- (d) Award Documentation. Splitco Capital Restricted Stock Awards and Splitco Starz Restricted Stock Awards shall be evidenced in such form as the Committee shall approve and contain such terms and conditions as shall be contained therein or incorporated by way of reference to the Incentive Plan or any associated instrument governing the corresponding award of restricted shares of LMC Capital Common Stock or LMC Starz Common Stock, respectively, which need not be the same for all Restricted Stock Awards.

6

(e) Rights with Respect to Shares. No Participant who is granted a Restricted Stock Award shall have any rights as a stockholder by virtue of such grant until shares are actually issued or delivered to the Participant.

# ARTICLE VI

#### **OPTIONS**

- 6.1 Grant of Options.
- (a) Grant. Splitco Capital Option(s) or Splitco Starz Option(s) shall be granted to each Participant who, as of the Redemption Date, holds an outstanding LMC Award(s) consisting of an option to purchase shares of LMC Capital Common Stock or LMC Starz Common Stock, respectively. Except as otherwise provided in this Plan, each Splitco Capital Option or Splitco Starz Option shall continue to be subject to all the terms and conditions of the applicable Incentive Plan and associated instrument under which the corresponding option to purchase LMC Capital Common Stock or LMC Starz Common Stock, respectively, was made and any such terms, conditions and restrictions as may be determined to be appropriate by the Committee.
- (b) Option Shares. Each Splitco Capital Option or Splitco Starz Option shall be for the number and series of shares of Splitco Capital Common Stock or Splitco Starz Common Stock that a Participant would have received in the Redemption if the applicable option for LMC Capital Common Stock or LMC Starz Common Stock, respectively, had been exercised immediately prior to the Redemption Date.
- (c) Option Price. The purchase price per share of Common Stock under each Option shall be established by the Committee. The Option price shall be subject to adjustment in accordance with the provisions of Section 11.3 hereof.
- (d) Option Documentation. Splitco Capital Options and Splitco Starz Options shall be evidenced in such form as the Committee shall approve and contain such terms and conditions as shall be contained therein or incorporated by way of reference to the Incentive Plan or any associated instrument governing the corresponding option to purchase LMC Capital Common Stock or LMC Starz Common Stock, respectively, which need not be the same for all Options.

# 6.2 Exercise and/or Termination of Options.

(a) Terms of Option. Splitco Capital Options or Splitco Starz Options granted under this Plan may be exercised at the same time and in the same manner as the corresponding option to purchase LMC Capital Common Stock or LMC Starz Common Stock, respectively. Splitco Capital Options or Splitco Starz Options granted under this Plan shall expire at the same time and in the same manner as the corresponding option to purchase LMC Capital Common Stock or LMC Starz Common Stock, as provided in the applicable Incentive Plan and any associated instrument governing such option to purchase LMC Capital Common Stock or LMC Starz Common Stock, respectively; provided, however, that a Participant's employment or service with the Company, LMC, a Qualifying Subsidiary or any of their respective Subsidiaries shall be

7

deemed to be employment or service with the Company and LMC for all purposes under an Option.

- (b) Payment on Exercise. No shares of Common Stock shall be issued on the exercise of an Option unless paid for in full at the time of purchase. Payment for shares of Common Stock purchased upon the exercise of an Option and any amounts required under Section 11.4 shall be determined by the Committee and may consist of (i) cash, (ii) check, (iii) promissory note (subject to applicable law), (iv) whole shares of any series of Common Stock, (v) the withholding of shares of the applicable series of Common Stock issuable upon such exercise of the Option, (vi) the delivery, together with a properly executed exercise notice, of irrevocable instructions to a broker to deliver promptly to the Company the amount of sale or loan proceeds required to pay the purchase price, or (vii) any combination of the foregoing methods of payment, or such other consideration and method of payment as may be permitted for the issuance of shares under the Delaware General Corporation Law. The permitted method or methods of payment of the amounts payable upon exercise of an Option, if other than in cash, shall be set forth in the applicable Option agreement and may be subject to such conditions as the Committee deems appropriate.
- (c) Value of Shares. Unless otherwise determined by the Committee and provided in the applicable Option agreement, shares of any series of Common Stock delivered in payment of all or any part of the amounts payable in connection with the exercise of an Option, and shares of any series of Common Stock withheld for such payment, shall be valued for such purpose at their Fair Market Value as of the exercise date.
- (d) Issuance of Shares. The Company shall effect the transfer of the shares of Common Stock purchased under the Option as soon as practicable after the exercise thereof and payment in full of the purchase price therefor and of any amounts required by Section 11.4, and within a reasonable time thereafter, such transfer shall be evidenced on the books of the Company. Unless otherwise determined by the Committee and provided in the applicable Option agreement, (i) no Participant or other person exercising an Option shall have any of the rights of a stockholder of the Company with respect to shares of Common Stock subject to an Option granted under the Plan until due exercise and full payment has been made, and (ii) no adjustment shall be made for cash dividends or other rights for which the record date is prior to the date of such due exercise and full payment.
- (e) Exercise. For purposes of this Article VI, the date of exercise of an Option shall mean the date on which the Company shall have received notice from the holder of the Option of the exercise of such Option (unless otherwise determined by the Committee and provided in the applicable Option agreement).

8

#### ARTICLE VII

# SARS

# 7.1 Grant of SARs.

- (a) Grant. Splitco Capital SARs or Splitco Starz SARs shall be granted to each Participant who, as of the Redemption Date, holds an outstanding LMC Award(s) consisting of an LMC Capital SAR or LMC Starz SAR, respectively. Except as otherwise provided in this Plan, each Splitco Capital SAR or Splitco Starz SAR shall continue to be subject to all the terms and conditions of the applicable Incentive Plan and associated instrument under which the LMC Capital SAR or LMC Starz SAR, respectively, was made and any such terms, conditions and restrictions as may be determined to be appropriate by the Committee.
- (b) SAR Shares. Each Splitco Capital SAR or Splitco Starz SAR shall be for the number and series of shares of Splitco Capital Common Stock or Splitco Starz Common Stock that a Participant would have received in the Redemption if the shares of LMC Capital Common Stock or LMC Starz Common Stock subject to such LMC Capital SAR or LMC Starz SAR, respectively, had been outstanding on the Redemption Date.
- (c) Base Price. The base price per share of Common Stock under each SAR shall be established by the Committee. The base price shall be subject to adjustment in accordance with the provisions of Section 11.3 hereof.
- (d) SAR Documentation. Splitco Capital SARs and Splitco Starz SARs shall be evidenced in such form as the Committee shall approve and contain such terms and conditions as shall be contained therein or incorporated by way of reference to the Incentive Plan or any associated instrument governing the corresponding LMC Capital SAR or LMC Starz SAR, respectively, which need not be the same for all SARs.

# 7.2 Exercise and/or Termination of SARs.

- (a) Terms of SARs. Splitco Capital SARs or Splitco Starz SARs granted under this Plan may be exercised at the same time and in the same manner as the corresponding LMC Capital SAR or LMC Starz SAR, respectively. Splitco Capital SARs or Splitco Starz SARs granted under this Plan shall expire at the same time and in the same manner as the applicable LMC Capital SAR or LMC Starz SAR, as provided in the applicable Incentive Plan and any associated instrument governing the LMC Capital SAR or LMC Starz SAR, respectively; provided, however, that a Participant's employment or service with the Company, LMC, a Qualifying Subsidiary or any of their respective Subsidiaries shall be deemed to be employment or service with the Company and LMC for all purposes under a SAR.
- (b) Consideration. The consideration to be received upon the exercise of a SAR shall be paid in cash, shares of the applicable series of Common Stock with respect to which the SAR was granted (valued at Fair Market Value on the date of exercise of such SAR), a combination of cash and such shares of the applicable series of Common Stock or such other consideration, in each case, as provided in the SAR agreement. No fractional shares of Common Stock shall be issuable upon exercise of a SAR, and unless otherwise provided in the applicable SAR agreement, the holder will receive cash in lieu of fractional shares.
- (c) Exercise. For purposes of this Article VII, the date of exercise of a SAR shall mean the date on which the Company shall have received notice from the holder of the SAR of

the exercise of such SAR (unless otherwise determined by the Committee and provided in the applicable SAR agreement).

#### ARTICLE VIII

#### CASH AWARDS

- 8.1 Grant. Cash Award(s) shall be granted to each Participant who, as of the Redemption Date, holds an outstanding LMC Award(s) consisting of an unvested LMC Cash Award(s).
- 8.2 Value. Each Cash Award shall provide the Participant with the opportunity to earn a cash payment in an amount equal to the amount the Participant would have received pursuant to the unvested LMC Cash Award, in accordance with the terms of such LMC Cash Award.
- 8.3 Lapse of Restrictions. The restrictions on each Cash Award shall lapse in accordance with the terms and conditions of the applicable Incentive Plan and associated instrument under which the corresponding LMC Cash Award was made; provided, however, that a Participant's employment or service the Company, LMC, a Qualifying Subsidiary or any of their respective Subsidiaries shall be deemed to be employment or service with the Company or LMC for all purposes under a Cash Award. A Cash Award shall only be paid following the written certification of the Compensation Committee of the Board that the terms and conditions of the Cash Award are satisfied.
- 8.4 Award Documentation. Cash Awards shall be evidenced in such form as the Committee shall approve and contain such terms and conditions as shall be contained therein or incorporated by way of reference to the Incentive Plan or any associated instrument governing the corresponding award of LMC Cash Awards, which need not be the same for all Cash Awards.

# ARTICLE IX

# ADMINISTRATION OF PLAN

9.1 The Committee. This Plan shall be administered solely by the Compensation Committee of the Board or such other committee of the Board as the Board shall designate to administer the Plan. A majority of the Committee shall constitute a quorum thereof and the actions of a majority of the Committee at a meeting at which a quorum is present, or actions unanimously approved in writing by all members of the Committee, shall be the actions of the Committee. Vacancies occurring on the Committee shall be filled by the Board. The Committee shall have full and final authority to interpret this Plan and any instruments evidencing Awards granted hereunder, to prescribe, amend and rescind rules and regulations, if any, relating to this Plan and to make all determinations necessary or advisable for the administration of this Plan. The Committee's determination in all matters referred to herein shall be conclusive and binding for all purposes and upon all persons including, but without limitation, the Company, LMC, the shareholders of the Company, the shareholders of LMC, the Committee and each of the members

10

thereof, and the Participants, and their respective successors in interest. The Committee may delegate any of its rights, powers and duties to any one or more of its members, or to any other person, by written action as provided herein, acknowledged in writing by the delegate or delegates, except that the Committee may not delegate to any person the authority to grant Stock Incentives to, or take other action with respect to, Participants who are subject to Section 16 of the Exchange Act. Such delegation may include, without limitation, the power to execute any documents on behalf of the Committee.

- 9.2 Liability of Committee. No member of the Committee shall be liable for any action or determination made or taken by him or the Committee in good faith with respect to the Plan. The Committee shall have the power to engage outside consultants, auditors or other professionals to assist in the fulfillment of the Committee's duties under this Plan at the Company's expense.
- 9.3 Determinations of the Committee. The Committee may, in its sole discretion, waive any provisions of any Award, provided such waiver is not inconsistent with the terms of the applicable Incentive Plan, any associated instrument or this Plan as then in effect.

# ARTICLE X

# AMENDMENT AND TERMINATION OF PLAN

- 10.1 Amendment, Modification, Suspension or Termination. The Board may from time to time amend, modify, suspend or terminate the Plan for the purpose of meeting or addressing any changes in legal requirements or for any other purpose permitted by law except that (i) subject to Section 11.6, no amendment or alteration that would impair the rights of any Participant under any Award awarded to such Participant shall be made without such Participant's consent and (ii) no amendment or alteration shall be effective prior to approval by the Company's shareholders to the extent such approval is then required pursuant to applicable legal requirements or the applicable requirements of the securities exchange on which the Company's Common Stock is listed. With the consent of the Participant, or as otherwise permitted under Section 11.6, and subject to the terms and conditions of the Plan, the Committee may amend outstanding Award agreements with any Participant, including any amendment which would (i) accelerate the time or times at which the Award may be exercised and/or (ii) extend the scheduled expiration date of the Award.
- 10.2 Termination. The Board may at any time terminate this Plan as of any date specified in a resolution adopted by the Board. If not earlier terminated, this Plan shall terminate on the last date that any Option or SAR granted hereunder may be exercised or any restriction applicable to a Restricted Stock Award granted hereunder has lapsed, whichever occurs later.

# ARTICLE XI

# MISCELLANEOUS PROVISIONS

11.1 Exclusion from Pension and Profit-Sharing Computation. By acceptance of an Award, unless otherwise provided in the applicable Award agreement, each Participant shall be

11

deemed to have agreed that such Award is special incentive compensation that will not be taken into account, in any manner, as salary, compensation or bonus in determining the amount of any payment under any pension, retirement or other employee benefit plan, program or policy of the Company or any Subsidiary of the Company. In addition, each beneficiary of a deceased Participant shall be deemed to have agreed that such Award will not affect the amount of any life insurance coverage, if any, provided by the Company on the life of the Participant which is payable to such beneficiary under any life insurance plan covering employees of the Company or any Subsidiary of the Company.

11.2 Government and Other Regulations. The obligation of the Company with respect to Awards shall be subject to all applicable laws, rules and regulations and such approvals by any governmental agencies as may be required, including the effectiveness of any registration statement required under the Securities Act of 1933, and the rules and regulations of any securities exchange or association on which the Common Stock may be listed or quoted. For so long as any series of Common Stock is registered under the Exchange Act, the Company shall use its reasonable efforts to comply with any legal requirements (i) to maintain a registration statement in effect under the Securities Act of 1933 with respect to all shares of the applicable series of Common Stock that may be issued to Participants under the Plan and (ii) to file in a timely manner all reports required to be filed by it under the Exchange Act.

# 11.3 Adjustments.

- (a) (i) If the Company subdivides its outstanding shares of any series of Common Stock into a greater number of shares of such series of Common Stock (by stock dividend, stock split, reclassification, or otherwise) or combines its outstanding shares of any series of Common Stock into a smaller number of shares of such series of Common Stock (by reverse stock split, reclassification, or otherwise) or if the Committee determines that any stock dividend, extraordinary cash dividend, reclassification, recapitalization, reorganization, stock redemption, split-up, spin-off, combination, exchange of shares, warrants or rights offering to purchase such series of Common Stock or other similar corporate event (including mergers or consolidations other than those which constitute Approved Transactions, adjustments with respect to which shall be under the Plan, then the Committee, in such manner as the Committee, in its sole discretion, deems equitable and appropriate, shall make such adjustments to any or all of (i) the number and kind of shares of stock subject to outstanding Stock Incentives, (ii) the purchase or exercise price and the relevant appreciation base with respect to any of the foregoing and (iii) the terms of the Cash Awards, provided, however, that the number of shares subject to any Stock Incentive shall always be a whole number. The Committee may, if deemed appropriate, provide for a cash payment to a Participant in connection with any adjustment made pursuant to this Section 11.3(a).
- (ii) Notwithstanding any provision of the Plan to the contrary, in the event of a corporate merger, consolidation, acquisition of property or stock, separation, reorganization or liquidation, the Committee shall be authorized, in its discretion, (i) to provide, prior to the transaction, for the acceleration of the vesting and exercisability of, or lapse of restrictions with respect to, the Award and, if the transaction is a cash merger, provide for the termination of any

12

portion of the Award that remains unexercised at the time of such transaction, or (ii) to cancel any such Awards and to deliver to the Holders cash in an amount that the Committee shall determine in its sole discretion is equal to the fair market value of such Awards on the date of such event, which in the case of Options or SARs shall be the excess of the Fair Market Value (as determined in sub-section (ii) of the definition of such term) of Common Stock on such date over the purchase price of the Options or the base price of the SARs, as applicable.

- (b) Approved Transactions; Board Change; Control Purchase. In the event of any Approved Transaction, Board Change or Control Purchase, notwithstanding any contrary waiting period, installment period, vesting schedule or restriction period in any Award agreement or in the Plan, unless the applicable Award agreement provides otherwise: (i) in the case of an Option or SAR, each such outstanding Option or SAR granted under the Plan shall become exercisable in full in respect of the aggregate number of shares covered thereby; (ii) in the case of Common Stock awarded under a Restricted Stock Award, any restriction period applicable to each such Common Stock shall be deemed to have expired and all such Common Stock shall become vested; and (iii) in the case of a Cash Award, the effect of an Approved Transaction, Board Change or Control Purchase shall be the effect prescribed for the corresponding LMC Cash Award in the event of an Approved Transaction, Board Change or Control Purchase with respect to LMC in the applicable Award agreement. Notwithstanding the foregoing, unless otherwise provided in the applicable Award agreement, the Committee may, in its discretion, determine that any or all outstanding Awards of any or all types granted pursuant to the Plan will not vest or become exercisable on an accelerated basis in connection with an Approved Transaction if effective provision has been made for the taking of such action which, in the opinion of the Committee, is equitable and appropriate to substitute a new Award or to assume such Award and to make such new or assumed Award, as nearly as may be practicable, equivalent to the old Award (before giving effect to any acceleration of the vesting or exercisability thereof), taking into account, to the extent applicable, the kind and amount of securities, cash or other assets into or for which the applicable series of Common Stock may be changed, converted or exchanged in connection with the Approved Transaction.
- (c) Compliance with Section 409A. No adjustment or substitution pursuant to this Section 11.3 shall be made in a manner that results in noncompliance with the requirements of Section 409A, to the extent applicable.
- 11.4 Withholding of Taxes. The Company's obligation to deliver shares of Common Stock or pay cash in respect of any Awards under the Plan shall be subject to applicable federal, state and local tax withholding requirements. Federal, state and local withholding tax due upon the exercise of any Option or SAR, upon the vesting of any Cash Awards or upon the vesting of, or expiration of restrictions with respect to, Common Stock granted under Restricted Stock Awards, may, in the discretion of the Committee, be paid in shares of the applicable series of Common Stock already owned by the Participant or through the withholding of shares otherwise issuable to such Participant, upon such terms and conditions (including the conditions referenced in Section 6.2) as the Committee shall determine. If the Participant shall fail to pay, or make arrangements satisfactory to the Committee for the payment of, all such federal, state and local taxes required to be withheld with respect to an Award, then the Company shall, to the extent permitted by law, have the right to deduct from any payment of any kind otherwise due to such

13

Participant an amount equal to any federal, state or local taxes of any kind required to be withheld with respect to such Award.

- 11.5 Restrictions on Benefit. Notwithstanding any provision of this Plan to the contrary, the provisions of any Incentive Plan concerning restrictions on benefits (in order to avoid excise taxes on the Participant under Section 4999 of the Code or the disallowance of a deduction to the Company pursuant to Section 280G of the Code) are specifically incorporated by this reference.
- 11.6 Section 409A. It is the intent of the Company that Awards under this Plan comply with the requirements of, or be exempt from the application of, Section 409A of the Code and related regulations and United States Department of the Treasury pronouncements ("Section 409A"), and the provisions of this Plan will be administered, interpreted and construed accordingly. Notwithstanding any provision in this Plan or any Incentive Plan to the contrary, if any Plan or Incentive Plan provision or any Award thereunder would result in the imposition of an additional tax under Section 409A, that Plan or Incentive Plan provision and/or that Award will be reformed to avoid imposition of the applicable tax and no action taken to comply with Section 409A shall be deemed to adversely affect the Participant's right to a Awards or require the consent of the Participant.

# TAX SHARING AGREEMENT

# BETWEEN

# LIBERTY INTERACTIVE CORPORATION,

# LIBERTY INTERACTIVE LLC

# AND

# LIBERTY MEDIA CORPORATION

# TABLE OF CONTENTS

		Page
SECTION 1.	Definition of Terms	1
SECTION 2.	Allocation of Tax Liabilities, Tax Benefits and Certain Losses	15
2.1	Liability for and the Payment of Taxes	15
2.2	Allocation Rules	15
SECTION 3.	Preparation and Filing of Tax Returns	19
3.1	Combined Returns	19
3.2 3.3	Separate Returns Provision of Information	19 20
3.4	Special Rules Relating to the Preparation of Tax Returns	20 20
SECTION 4.	Tax Payments	21
4.1	Payment of Taxes to Tax Authority	21
4.2	Indemnification Payments	21
4.3	Payments for Tax Refunds and Tax Benefits	22
4.4	Interest on Late Payments	22
4.5	Initial Determinations and Subsequent Adjustments	22
4.6	Tax Consequences of Payments	23
SECTION 5.	Assistance and Cooperation	24
5.1	Cooperation	24
5.2	Supplemental Rulings	24
SECTION 6.	Tax Records	25
6.1	Retention of Tax Records	25
6.2	Access to Tax Records	25
6.3 6.4	Confidentiality Delivery of Tax Records	26
0.4	Delivery of Tax Records	26
SECTION 7.	Restrictions on Certain Actions of Distributing and Controlled; Indemnity	26
7.1	Restrictive Covenants	26
7.2	Distributing Indemnity	28
7.3	Controlled Indemnity	28
7.4 7.5	Scope Notices of Tax Contests	29 29
7.6	Control of Tax Contests Generally	29
7.7	Cooperation	30
7.8	Joint Claims	30
	i	
7.9	Controlled Claims	31
7.10	Other Claims	31
SECTION 8.	General Provisions	31
8.1	Termination	31
8.2	Predecessors or Successors	31
8.3 8.4	Expenses Governing Law; Jurisdiction	31
		32
8.5 8.6	Waiver of Jury Trial Notices	32 33

8.7	Counterparts	33
8.8	Binding Effect; Assignment	33
8.9	Severability	34
8.10	Amendments; Waivers	34
8.11	Effective Date	34
8.12	Change in Law	34
8.13	Authorization, Etc.	34
8.14	No Third Party Beneficiaries	34
8.15	Entire Agreement	35
8.16	No Strict Construction; Interpretation	35
8.17	Headings	35
8.18	Assignment of Rights under the LEI Tax Sharing Agreement	35
8.19	Assignment of Rights under the Tax Matters Agreement	36
8.20	Assignment of Rights under other Tax Agreements	36

ii

# TAX SHARING AGREEMENT

THIS TAX SHARING AGREEMENT (this "Agreement") is entered into as of September 23, 2011, between Liberty Interactive Corporation, a Delaware corporation formerly known as Liberty Media Corporation ("Distributing"), Liberty Interactive LLC, a Delaware limited liability company formerly known as Liberty Media LLC ("Liberty LLC"), and Liberty Media Corporation, a Delaware corporation formerly known as Liberty CapStarz, Inc. ("Controlled"). Unless otherwise indicated, all "Section" references in this Agreement are to sections of this Agreement.

#### RECITALS

WHEREAS, Controlled is a wholly owned subsidiary of Distributing;

WHEREAS, the Board of Directors of Distributing has determined that it would be appropriate and desirable for Distributing to separate the Controlled Business from the Distributing Business;

WHEREAS, the Board of Directors of Controlled has approved such transaction;

WHEREAS, following the Contribution, Distributing intends to distribute its entire interest in the stock of Controlled to the holders of Liberty Capital Common Stock and Liberty Starz Common Stock in exchange for their shares of Liberty Capital Common Stock and Liberty Starz Common Stock in what is intended to qualify as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code (the "Distribution");

WHEREAS, the parties set forth in the Reorganization Agreement the principal arrangements between them regarding the separation of the Controlled Business from the Distributing Business; and

WHEREAS, the parties desire to provide for and agree upon the allocation between the parties of liabilities for Taxes and credits for Tax Benefits arising prior to, as a result of, and subsequent to the Distribution, and to provide for and agree upon other matters relating to Taxes.

NOW, THEREFORE, in consideration of the foregoing and the covenants and agreements set forth below, and intending to be legally bound hereby, Distributing, Liberty LLC and Controlled hereby agree as follows:

**SECTION 1. Definition of Terms.** For purposes of this Agreement (including the recitals hereof), the following terms have the following meanings:

"2029 Exchangeables" means the 4% Senior Exchangeable Debentures due 2029 issued by Liberty LLC.

1

"2030 Exchangeables" means the 33/4% Senior Exchangeable Debentures due 2030 issued by Liberty LLC.

"2031 Exchangeables" means the 31/4% Senior Exchangeable Debentures due 2031 issued by Liberty LLC.

"Affiliate" means with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such first Person. For the avoidance of doubt, (x) none of Expedia, Inc., any of its present or former Affiliates, TripAdvisor, Inc. or any of its present or former Affiliates will be treated as an Affiliate of any member of the Distributing Group during any period (or portion thereof) in which the voting stock of such corporation owned by any member of the Distributing Group is subject to a proxy in favor of Mr. Barry Diller to vote those shares during such period; (y) no member of the Controlled Group will be treated as an Affiliate of any member of the Distributing Group; and (z) no member of the Distributing Group will be treated as an Affiliate of any member of the Controlled Group.

"Agreement" has the meaning set forth in the preamble hereof.

"Ascent" means Ascent Capital Group, Inc., a Delaware corporation.

"AT&T Tax Sharing Agreement" means the Tax Sharing Agreement dated as of March 9, 1999, as amended, by and among AT&T Corp., Liberty LLC, for itself and each member of the Liberty Group (as defined therein), Tele-Communications, Inc., Liberty Ventures Group LLC, Liberty Media Group LLC, TCI Starz, Inc., TCI CT Holdings, Inc., each Covered Entity (as defined therein) listed on the signature pages thereof, and any entities which became parties thereto pursuant to Section 23 thereof.

"AT&T TSA Benefits" means any right to receive any payment (including any indemnification payment) from AT&T Corp. or any member of the Common Stock Group (as defined in the AT&T Tax Sharing Agreement).

"AT&T TSA Liabilities" means any obligation or liability to make any payment (including any indemnification payment) to AT&T Corp., any member of the Common Stock Group (as defined in the AT&T Tax Sharing Agreement), or any Common Stock Indemnitees (as defined in the AT&T Tax Sharing Agreement).

"business day" means any day other than a Saturday, Sunday or a day on which banking institutions in New York City, New York or London, England are authorized or required by law or executive order to close.

"Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time, or any successor law.

"Combined Return" means (i) with respect to any Tax Return for a Tax Year beginning on or before the Distribution Date, any Tax Return that includes Tax Items of both the Distributing Business and the Controlled Business, determined in

2

accordance with the allocation rules of Section 2.2, and (ii) with respect to any Tax Return for a Tax Year beginning after the Distribution Date, any Tax Return that includes one or more members of the Distributing Group and one or more members of the Controlled Group.

"Company" means Distributing or Controlled, as the context requires.

"Compensatory Equity Interests" means stock, equity interests, options, stock appreciation rights, or other similar rights with respect to the equity of any entity granted prior to the Distribution in connection with employee, independent contractor or director compensation (including, for the avoidance of doubt, stock, equity interests, options, stock appreciation rights, or other similar rights issued in substitution for any of the foregoing by reason of the Distribution or any subsequent transaction).

"Consolidated Group" means a group filing (or required to file) consolidated federal income Tax Returns for any Tax Year (or portion thereof), within the meaning of Treasury Regulations Section 1.1502-1(h).

"Contribution" has the meaning given to such term in the Reorganization Agreement.

"Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of securities or partnership, membership, limited liability company, or other ownership interests, by contract or otherwise and the terms "Controlling" and "Controlled" have meanings correlative to the foregoing.

"Controlled" has the meaning set forth in the preamble hereof.

"Controlled Acquired Subsidiary" has the meaning set forth in Section 2.2(1).

"Controlled Business" means: (i) with respect to any Tax Year (or portion thereof) ending prior to the Issue Date, the assets, liabilities, and businesses of Distributing and its Subsidiaries other than any assets, liabilities, and businesses of, and any equity or debt interests in, QVC, Inc., Provide Commerce, Inc., and each of their respective Subsidiaries, (ii) with respect to any Tax Year (or portion thereof) beginning on the Issue Date and ending at or before the Effective Time, the assets, liabilities and businesses of Distributing and its Subsidiaries, other than the assets, liabilities and businesses that were tracked during such Tax Year (or portion thereof), and only for so long as so tracked, by the Liberty Interactive Common Stock (including any equity or debt interests in any entities so tracked); and (iii) with respect to any Tax Year (or portion thereof) beginning after the Effective Time, the assets, liabilities, and businesses of the Controlled Group during such Tax Year (or portion thereof).

"Controlled Claim" means any pending or threatened Tax Contest, claim, action, suit, investigation or proceeding brought by a third party relating to any

AT&T

3

TSA Liabilities, DHC TSA Liabilities, or LMI TSA Liabilities, in each case for which Distributing is or may be indemnified by Controlled under Section 7.

"Controlled Consolidated Group" means a Consolidated Group of which Controlled is the common parent, within the meaning of Section 1504(a)(1) of the Code.

"Controlled Group" means, with respect to any Tax Year (or portion thereof) beginning after the Effective Time, Controlled and each Subsidiary of Controlled (but only while such Subsidiary is a Subsidiary of Controlled).

"Controlled Indemnitees" has the meaning set forth in Section 7.2.

"Controlled Section 355(e) Event" means the application of Section 355(e) of the Code to the Distribution as a result of the Distribution being "part of a plan (or series of related transactions) pursuant to which 1 or more persons acquire directly or indirectly stock representing a 50-percent or greater interest" in Controlled or any successor corporation (within the meaning of Section 355(e) of the Code).

"Controlled Stock" means Controlled's Series A Liberty Capital common stock, par value \$.01 per share, Series B Liberty Capital common stock, par value \$.01 per share, Series A Liberty Starz common stock, par value \$.01 per share, and if and when issued, Series C Liberty Capital common stock, par value \$.01 per share, and series C Liberty Starz common stock, par value \$.01 per share, and any series or class of stock into which Controlled's Series A, Series B, or Series C Liberty Capital common stock or Series A, Series B, or Series C Liberty Starz common stock is redesignated, reclassified, converted or exchanged following the Effective Time.

"DHC" means Discovery Holding Company, a Delaware corporation.

"DHC Tax Sharing Agreement" means the Tax Sharing Agreement dated July 20, 2005, between Liberty LLC and DHC.

"DHC TSA Benefits" means any right to receive any payment (including any indemnification payment) from DHC or any member of the DHC Group (as defined in the DHC Tax Sharing Agreement) pursuant to the terms of the DHC Tax Sharing Agreement.

"DHC TSA Liabilities" means any obligation or liability to make any payment (including any indemnification payment) to DHC, any member of the DHC Group (as defined in the DHC Tax Sharing Agreement), or any DHC Indemnitee (as defined in the DHC Tax Sharing Agreement) pursuant to the terms of the DHC Tax Sharing Agreement.

"Disclosing Party" has the meaning set forth in Section 6.3.

"Discovery" means Discovery Communications, Inc., a Delaware corporation.

"Distributing" has the meaning set forth in the preamble hereof: <u>provided, however</u>, that with respect to any period (or portion thereof) ending prior to the Issue Effective Time, "Distributing" shall mean Liberty LLC (formerly known as Liberty Media Corporation).

"Distributing Acquired Subsidiary" has the meaning set forth in Section 2.2(1).

"Distributing Business" means, (i) with respect to any Tax Year (or portion thereof) ending prior to the Issue Date, the assets, liabilities, and businesses of, and any equity or debt interests in, QVC, Inc., Provide Commerce, Inc., and each of their respective Subsidiaries; (ii) with respect to any Tax Year (or portion thereof) beginning on the Issue Date and ending at or before the Effective Time, the assets, liabilities and businesses that were tracked during such Tax Year (or portion thereof), and only for so long as so tracked, by the Liberty Interactive Common Stock (including any equity or debt interests in any entities so tracked); and (iii) with respect to any Tax Year (or portion thereof) beginning after the Effective Time, the assets, liabilities, and businesses of the Distributing Group during such Tax Year (or portion thereof).

"Distributing Consolidated Group" means a Consolidated Group of which Distributing is the common parent, within the meaning of Section 1504(a)(1) of the Code.

"Distributing Group" means, with respect to any Tax Year (or portion thereof) beginning after the Effective Time, Distributing and each Subsidiary of Distributing (but only while such Subsidiary is a Subsidiary of Distributing).

"Distributing Indemnitees" has the meaning set forth in Section 7.3.

"Distribution" has the meaning set forth in the recitals hereof.

"Distribution Date" means the date on which the Distribution occurs.

"DTV" means DIRECTV, a Delaware corporation.

"Due Date" has the meaning set forth in Section 4.4.

"Effective Time" means the time at which the Distribution is effected on the Distribution Date.

"Employing Party" means the Company whose Group includes any entity that is required under applicable Tax law to satisfy any Tax reporting obligations with respect to any employee, independent contractor or director compensation attributable to any Compensatory Equity Interests; provided, however, that if an entity in both Groups is permitted under applicable Tax Law to satisfy such Tax reporting obligations, then the Employing Party shall mean the Company to which the Taxes and Tax Items arising with respect to such employee, independent contractor or director compensation are allocated pursuant to Section 2.2(i).

5

"Final Determination" means a determination within the meaning of Section 1313 of the Code or any similar provision of state or local Tax Law.

"Group" means the Distributing Group or the Controlled Group, as the context requires.

"Interactive Entities" means any Person owned directly or indirectly by Distributing prior to the Distribution (other than Live Nation Entertainment, Inc.) that is or was tracked by the Liberty Interactive Common Stock at any time prior to the Distribution and any Person acquired directly or indirectly by Distributing following the Distribution.

"Interactive Pre-Distribution Acquired Subsidiaries" means BUYSEASONS, Inc., Backcountry.com, Inc., Bodybuilding.com, LLC, Celebrate Express, Inc., Celebrate Interactive, Inc., Lockerz, LLC, and each of their respective Subsidiaries.

"Interest Rate" means the Rate determined below, as adjusted as of each Interest Rate Determination Date. The "Rate," means, with respect to each period between two consecutive Interest Rate Determination Dates, a rate determined at approximately 11:00 a.m., London time, two London business days before the earlier Interest Rate Determination Date equal to the greater of: (x) the sum of (i) the six month dollar LIBOR rate as displayed on page "LR" of Bloomberg (or such other appropriate page as may replace such page), plus (ii) 2%, and (y) the interest rate that would be applicable at such time to a "large corporate underpayment" (within the meaning of Section 6621(c) of the Code) under Sections 6601 and 6621 of the Code. Interest will be calculated on the basis of a year of 365 days and the actual number of days for which due.

"Interest Rate Determination Date" means the Due Date and each March 31, June 30, September 30 and December 31 thereafter.

"IRS" means the Internal Revenue Service.

"IRS Submissions" means the Ruling Request, each supplemental submission and any other correspondence or supplemental materials submitted to the IRS in connection with obtaining the Ruling.

"Issue Date" means May 9, 2006.

"Issue Effective Time" means the time at which the Liberty Restructuring was effected on the Issue Date.

"issuing corporation" has the meaning set forth in Section 3.4(e).

"Joint Claim" means any pending or threatened Tax Contest, claim, action, suit, investigation or proceeding brought by a third party relating to (w) any Transaction Taxes or any Transaction Tax-Related Losses, (x) any LEI Transaction

6

Taxes or any LEI Tax-Related Losses, (y) any News Transaction Taxes or any News Tax-Related Losses or (z) any Tracking Stock Taxes and Losses, in each case for which one Company is or may be indemnified by the other Company under Section 7.

"LEI" means Liberty Entertainment, Inc., a Delaware corporation.

"LEI IRS Submissions" means the LEI Ruling Request, each supplemental submission and any other correspondence or supplemental materials submitted to the IRS in connection with obtaining the LEI Ruling and any request, supplemental submission and any other correspondence or supplemental materials submitted to the

IRS in connection with obtaining any private letter ruling which supplements or otherwise modifies the LEI Ruling.

"LEI Opinion" means the opinion delivered by Skadden, Arps, Slate, Meagher & Flom LLP to Distributing in connection with the LEI Transaction to the effect that the LEI Transaction will qualify as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code to Distributing and the holders of Liberty Starz Common Stock (except, in the case of the holders of Liberty Starz Common Stock, with respect to cash received in lieu of fractional shares).

"LEI Ruling" means PLR 201004001 that was issued to Distributing on October 22, 2009.

"LEI Ruling Request" means the request for rulings, dated January 16, 2009, as amended on May 21, 2009, filed by Distributing with the IRS in connection with the LEI Transaction, as the same shall have been amended or supplemented.

"LEI Tax Materials" means (i) the LEI Ruling and any supplemental private letter ruling which supplements or otherwise modifies the LEI Ruling, (ii) each LEI IRS Submission, (iii) the representation letters delivered to Skadden, Arps, Slate, Meagher & Flom LLP in connection with the delivery of the LEI Opinion, and (iv) any other materials delivered by Distributing in connection with the rendering by Skadden, Arps, Slate, Meagher & Flom LLP of the LEI Opinion or the issuance by the IRS of the LEI Ruling.

"LEI Tax-Related Losses" mean any Losses resulting from the failure of the LEI Transaction to qualify (i) as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code (except with respect to cash received in lieu of fractional shares), or (ii) in whole for nonrecognition of income, gain and loss for U.S. federal income tax purposes to Distributing, each of its Subsidiaries immediately prior to the LEI Transaction, and the holders of Liberty Starz Common Stock (except with respect to cash received in lieu of fractional shares).

"LEI Tax Sharing Agreement" means the Tax Sharing Agreement dated as of November 19, 2009, by and among Distributing and LEI, taken together with the Assumption and Joinder Agreement dated as of November 19, 2009, by and among Distributing, LEI and DTV.

7

- "LEI Transaction" means the "Contribution" and the "Split-off," in each case as such terms are defined in the LEI Ruling.
- "LEI Transaction Taxes" means any Taxes resulting from the LEI Transaction.
- "LGI" means Liberty Global, Inc., a Delaware corporation.

"Liberty Capital Common Stock" means Distributing's Series A Liberty Capital common stock, par value \$.01 per share, and Series B Liberty Capital common stock, par value \$.01 per share.

"Liberty Interactive Common Stock" means Distributing's Series A Liberty Interactive common stock, par value \$.01 per share, Series B Liberty Interactive common stock, par value \$.01 per share, and if and when issued, Series C Liberty Interactive common stock, par value \$.01 per share, and any series or class of stock into which Distributing's Series A, Series B, or Series C Liberty Interactive common stock is redesignated, reclassified, converted or exchanged following the Effective Time

"Liberty LLC" has the meaning set forth in the preamble hereof.

"Liberty Restructuring" means the transactions effected on the Issue Date pursuant to which Distributing became the publicly traded parent of Liberty LLC in a reorganization qualifying under Section 368(a)(l)(F) of the Code.

"Liberty Restructuring Opinion" means the tax opinion delivered by Baker Botts L.L.P. to Liberty LLC and Distributing in connection with the Liberty Restructuring.

"Liberty Starz Common Stock" means Distributing's Series A Liberty Starz common stock, par value \$.01 per share, and Series B Liberty Starz common stock, par value \$.01 per share, and for any taxable periods (or portions thereof) prior to the Redesignation, Distributing's Series A Liberty Entertainment common stock, par value \$.01 per share, and Series B Liberty Entertainment common stock, par value \$.01 per share.

"LMI" means Liberty Media International, Inc., a Delaware corporation.

"LMI Tax Sharing Agreement" means the Tax Sharing Agreement dated June 1, 2004, between Liberty LLC and LMI.

"LMI TSA Benefits" means any right to receive any payment (including any indemnification payment) from LMI or any member of the LMI Group (as defined in the LMI Tax Sharing Agreement) pursuant to the terms of the LMI Tax Sharing Agreement.

"LMI TSA Liabilities" means any obligation or liability to make any payment to LMI, any member of the LMI Group (as defined in the LMI Tax Sharing

8

Agreement) or any LMI Indemnitee (as defined in the LMI Tax Sharing Agreement) pursuant to the terms of the LMI Tax Sharing Agreement.

"Losses" means any and all damages, losses, deficiencies, liabilities, obligations, penalties, judgments, settlements, claims, payments, fines, interest, costs and expenses (including, without limitation, the fees and expenses of any and all actions and demands, assessments, judgments, settlements and compromises relating thereto and the costs and expenses of attorneys', accountants', consultants' and other professionals' fees and expenses incurred in the investigation or defense thereof or the enforcement of rights hereunder); provided, however, that "Losses" shall exclude any special or punitive damages; provided, further, that the foregoing proviso will not be interpreted to limit indemnification for Losses incurred as a result of the assertion by a claimant (other than the parties hereto and their successors and assigns) in a third-party claim for special or punitive damages.

"News" means News Corporation, a Delaware corporation.

"News Distributions" means "Distribution 1," "Distribution 2," "Distribution 3," "Distribution 4," and "Distribution 5," in each case as defined in the News Rulings.

"News IRS Submissions" means the News Ruling Requests, each supplemental submission and any other correspondence or supplemental materials submitted to the IRS in connection with obtaining the News Rulings.

"News Opinions" means the tax opinion delivered by Baker Botts L.L.P. to Distributing, and the tax opinion delivered by Skadden, Arps, Slate, Meagher & Flom LLP to News, in connection with the News Transactions.

"News Ruling Requests" means the requests for rulings, dated February 26, 2007 and March 20, 2007, filed by News and Distributing with the IRS in connection with the News Transactions.

"News Rulings" means PLR 200812003 that was issued to News on September 25, 2007 and PLR 200812004 that was issued to Distributing on September 25, 2007.

"News Tax Materials" means (i) the News Rulings, (ii) each News IRS Submission, (iii) the representation letters delivered by Distributing to Baker Botts L.L.P. and Skadden, Arps, Slate, Meagher & Flom LLP in connection with the delivery of the News Opinions, and (iv) any other materials delivered by Distributing in connection with the rendering by Baker Botts L.L.P. and Skadden, Arps, Slate, Meagher & Flom LLP of the News Opinions or the issuance by the IRS of the News Rulings.

"News Tax-Related Losses" mean any Losses resulting from the failure of any of the News Transactions to qualify (i) as a tax-free transaction described under Sections 368(a), 355 and/or 361 of the Code, as applicable, or (ii) in whole for

9

nonrecognition of income, gain and loss for U.S. federal income tax purposes to News and each of its Subsidiaries immediately prior to the News Transactions and to Distributing and each of its Subsidiaries on February 27, 2008.

"News Transactions" means "Contribution 1," "Contribution 2," and "Contribution 3," in each case as defined in the News Rulings, and each of the News Distributions.

"News Transaction Taxes" means any Taxes resulting from the News Transactions.

"Non-Preparer" means the Company that is not responsible for the preparation and filing of the Combined Return or Separate Return, as applicable, pursuant to Section 3.

"Payment Date" means (x) with respect to any U.S. federal income tax return, the due date for any required installment of estimated taxes determined under Section 6655 of the Code, the due date (determined without regard to extensions) for filing the return determined under Section 6072 of the Code, and the date the return is filed, and (y) with respect to any other Tax Return, the corresponding dates determined under the applicable Tax Law.

"Person" means any individual, corporation, company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Post-Distribution Period" means all taxable periods (or portions thereof) beginning after the Effective Time.

"Pre-Distribution Period" means all taxable periods (or portions thereof) ending before or at the Effective Time.

"Preparer" means the Company that is responsible for the preparation and filing of the Combined Return or Separate Return, as applicable, pursuant to Section 3.

"Receiving Party" has the meaning set forth in Section 6.3.

"Reclassification" means the reclassification of Distributing's outstanding stock that was effected on March 3, 2008.

"Reclassification Opinion" means the tax opinion delivered by Baker Botts L.L.P. to Distributing in connection with the Reclassification

"Redesignation" means the filing of Distributing's amended and restated certificate of incorporation on November 19, 2009 to, among other things, rename its "Entertainment Group" as the "Starz Group" and rename its "Series A Liberty Entertainment Common Stock" and "Series B Liberty Entertainment Common Stock" as its "Series A Liberty Starz Common Stock" and "Series B Liberty Starz Common Stock" respectively.

10

"Reorganization Agreement" means the Reorganization Agreement between Distributing and Controlled dated August 30, 2011.

"Requesting Party" has the meaning set forth in Section 5.2.

"Restructuring" has the meaning assigned to such term in the Reorganization Agreement.

"Ruling" means PLR 201135025 that was issued to Distributing on March 3, 2011.

"Ruling Request" means the request for rulings, dated July 28, 2010, filed by Distributing with the IRS in connection with the Distribution, as the same shall have been amended or supplemented.

"Separate Return" means any Tax Return that is not a Combined Return.

"Share Distribution" means the distribution by LTWX V, Inc. of certain shares of stock of Time Warner Inc., Time Warner Cable Inc. and AOL, Inc. which was effected in connection with the reattribution of those shares of stock from Distributing's Capital Group to Distributing's Interactive Group on February 9, 2011.

"Straddle Losses" means approximately \$163 million in straddle losses, which arose in the Pre-Distribution Period, that are related to the stock of Time Warner Inc., Time Warner Cable Inc. and/or AOL, Inc.

"Subsidiary" when used with respect to any Person, means (i)(A) a corporation a majority in voting power of whose share capital or capital stock with voting power, under ordinary circumstances, to elect directors is at the time, directly or indirectly, owned by such Person, by one or more Subsidiaries of such Person, or by such Person and one or more Subsidiaries of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, (B) a partnership or limited liability company in which such Person or a Subsidiary of such Person is, at the date of determination, (1) in the case of a partnership, a general partner of such partnership with the power affirmatively to direct the policies and management of such partnership or (2) in the case of a limited liability company, the managing member or, in the absence of a managing member, a member with the power affirmatively to direct the policies and management of such limited liability company, or (C) any other Person

(other than a corporation) in which such Person, one or more Subsidiaries of such Person or such Person and one or more Subsidiaries of such Person, directly or indirectly, at the date of determination thereof, has or have (1) the power to elect or direct the election of a majority of the members of the governing body of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, or (2) in the absence of such a governing body, at least a majority voting interest or (ii) any other Person of which an aggregate of 50% or more of the equity interests are, at the time, directly or indirectly, owned by such Person and/or one or more Subsidiaries of such Person. For purposes of the foregoing, none of IAC/InteractiveCorp, Expedia, Inc., TripAdvisor, Inc.,

11

or any of their respective present or former Subsidiaries will be treated as Subsidiaries of Distributing or any member of the Distributing Group during any period (or portion thereof) if (i) such corporation would otherwise be a Subsidiary of Distributing solely as a result of Distributing and/or its Subsidiaries owning stock constituting a majority in voting power of the capital stock of such corporation, and (ii) the voting stock owned by Distributing and/or its Subsidiaries in such corporation is subject to a proxy in favor of Mr. Barry Diller to vote those shares during such period.

"Supplemental IRS Submissions" means any request for a Supplemental Ruling, each supplemental submission and any other correspondence or supplemental materials submitted to the IRS in connection with obtaining any Supplemental Ruling.

"Supplemental Ruling" means any private letter ruling obtained by Distributing or Controlled from the IRS which supplements or otherwise modifies the Ruling.

"Tax" or "Taxes" means any income, gross income, gross receipts, profits, capital stock, franchise, withholding, payroll, social security, workers compensation, employment, unemployment, Medicare, disability, property, ad valorem, stamp, excise, severance, occupation, service, sales, use, license, lease, transfer, import, export, value added, alternative minimum, estimated or other similar tax (including any fee, assessment, or other charge in the nature of or in lieu of any tax) imposed by any Tax Authority and any interest, penalties, additions to tax, or additional amounts in respect of the foregoing.

"Tax Authority" means, with respect to any Tax, the governmental entity or political subdivision, agency, commission or authority thereof that imposes such Tax, and the agency, commission or authority (if any) charged with the assessment, determination or collection of such Tax for such entity or subdivision.

"Tax Benefit" means a Tax Item which decreases the Tax liability of a taxpayer, including a Tax Refund.

"Tax Contest" means an audit, review, examination, or any other administrative or judicial proceeding with the purpose, potential or effect of redetermining Taxes of any member of either Group (including any administrative or judicial review of any claim for refund).

"Tax Counsel" means Baker Botts L.L.P.

"Tax Item" means, with respect to any Tax, any item of income, gain, loss, deduction, credit or other attribute that may have the effect of increasing or decreasing any Tax.

"Tax Law" means the law of any governmental entity or political subdivision thereof, and any controlling judicial or administrative interpretations of such law, relating to any Tax.

12

"Tax Materials" means (i) the Ruling and each Supplemental Ruling issued by the IRS in connection with the Distribution, (ii) each IRS Submission and Supplemental IRS Submission, (iii) the representation letters delivered to Tax Counsel in connection with the delivery of the Tax Opinion, and (iv) any other materials delivered or deliverable by Distributing, Controlled and others in connection with the rendering by Tax Counsel of the Tax Opinion or the issuance by the IRS of the Ruling and any Supplemental Ruling.

"Tax Matters Agreement" means the Tax Matters Agreement entered into as of December 22, 2006, by and among News and Distributing.

"Tax Opinion" means the opinion to be delivered by Tax Counsel to Distributing in connection with the Distribution, which will rely on the continued validity of the Ruling as to the matters covered by the Ruling, to the effect that, under applicable U.S. federal income tax law, (i) the Contribution and the Distribution will qualify as a tax-free transaction described under Sections 355 and 368(a)(1)(D) of the Code, (ii) no gain or loss will be recognized by Distributing upon the distribution of Controlled Stock in the Distribution, (iii) no gain or loss will be recognized by, and no amount will be included in the income of, holders of Liberty Capital Common Stock and Liberty Starz Common Stock upon the exchange of their shares of Liberty Capital Common Stock and Liberty Starz Common Stock for shares of Controlled Stock in the Distribution, (iv) the Controlled Stock issued in the Distribution will be treated as stock of Controlled for U.S. federal income tax purposes, and (v) the Controlled stock issued in the Distribution will not constitute Section 306 stock within the meaning of Section 306(c) of the Code.

"Tax Records" means Tax Returns, Tax Return work papers, documentation relating to any Tax Contests, and any other books of account or records required to be maintained under applicable Tax Laws (including but not limited to Section 6001 of the Code) or under any record retention agreement with any Tax Authority.

"Tax Refund" means a refund of Taxes previously paid and any overpayment interest within the meaning of Section 6611 of the Code or any similar provision under applicable Tax Law (whether paid by way of a refund or credited against any liability for related Taxes).

"Tax Return" means any report of Taxes due, any claims for refund of Taxes paid, any information return with respect to Taxes, or any other similar report, statement, declaration, or document filed or required to be filed (by paper, electronically or otherwise) under any applicable Tax Law, including any attachments, exhibits, or other materials submitted with any of the foregoing, and including any amendments or supplements to any of the foregoing.

"Tax Year" means, with respect to any Tax, the year, or shorter period, if applicable, for which the Tax is reported as provided under applicable Tax Law.

13

"Tracking Stock Tax Materials" means (i) the representation letter delivered by Liberty LLC and Distributing to Baker Botts L.L.P. in connection with the Liberty Restructuring Opinion, (ii) the representation letter delivered by Distributing to Baker Botts L.L.P. in connection with the Reclassification Opinion, and (iii) any other materials delivered by Distributing or Liberty LLC in connection with the rendering by Baker Botts L.L.P. of the Liberty Restructuring Opinion or the Reclassification Opinion.

Capital Common Stock, or the Liberty Starz Common Stock as other than stock of Distributing, or as Section 306 stock within the meaning of Section 306(c) of the Code, for any Pre-Distribution Period, (ii) the actual or deemed disposition of any assets caused by the issuance of the Liberty Interactive Common Stock, the Liberty Capital Common Stock, or the Liberty Starz Common Stock for any Pre-Distribution Period, (iii) the treatment of the Controlled Stock as other than stock of Controlled, or as Section 306 stock within the meaning of Section 306(c) of the Code, or (iv) the actual or deemed disposition of any assets caused by the issuance of the Controlled Stock.

"Transaction Taxes" means any Taxes resulting from the Contribution and the Distribution, other than (i) Transfer Taxes, and (ii) any Taxes attributable to "deferred intercompany transactions" or "excess loss accounts" (as those terms are defined by Treasury Regulations) that are triggered as a result of the Contribution and the Distribution.

"Transaction Tax-Related Losses" means any Losses resulting from the Contribution and the Distribution as a result of (i) the failure of the Contribution and Distribution to qualify as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code; (ii) the failure of the Contribution and Distribution to qualify in whole for nonrecognition of income, gain and loss for U.S. federal income tax purposes to Distributing, Controlled, each of their respective Subsidiaries at the Effective Time, and the holders of Liberty Capital Common Stock and Liberty Starz Common Stock that receive stock of Controlled in the Distribution; (iii) the Controlled Stock not being treated as stock of Controlled, or being treated as Section 306 stock within the meaning of Section 306(c) of the Code, for U.S. federal income tax purposes, or (iv) the Liberty Interactive Common Stock, the Liberty Capital Common Stock or the Liberty Starz Common Stock not being treated as stock of Distributing, or being treated as Section 306 stock within the meaning of Section 306(c) of the Code, for U.S. federal income tax purposes. For the avoidance of doubt, "Transaction Tax-Related Losses" shall not include any (i) Transfer Taxes, or (ii) any Taxes attributable to "deferred intercompany transactions" or "excess loss accounts" (as those terms are defined by Treasury Regulations) that are triggered as a result of the Contribution and the Distribution.

"Transfer Taxes" means all U.S. federal, state, local or foreign sales, use, privilege, transfer, documentary, gains, stamp, duties, recording, and similar Taxes and fees (including any penalties, interest or additions thereto) imposed upon any party hereto or any of its Affiliates in connection with the Restructuring or the Distribution.

14

"Treasury Regulations" means the regulations promulgated from time to time under the Code as in effect for the relevant Tax Year.

#### SECTION 2. Allocation of Tax Liabilities, Tax Benefits and Certain Losses.

- 2.1 <u>Liability for and the Payment of Taxes.</u> Except as provided in Section 3.4(e) (Withholding and Reporting) and Section 7.5 (Notices) and in accordance with Section 4:
- (a) Distributing Liabilities and Payments. For any Tax Year (or portion thereof), Distributing shall (i) be liable for the Taxes (determined without regard to Tax Benefits) allocated to it by this Section 2, reduced by any Tax Benefits allocated to Distributing or Controlled that are allowable under applicable Tax Law to reduce such Taxes, (ii) pay such Taxes, as so reduced, either to the applicable Tax Authority or to Controlled as required by Section 4, and (iii) pay Controlled for any Tax Benefits allocated to Controlled by this Section 2 that Distributing uses to reduce Taxes payable by it pursuant to clause (ii) of this Section 2.1(a).
- (b) Controlled Liabilities and Payments. For any Tax Year (or portion thereof), Controlled shall (i) be liable for the Taxes (determined without regard to Tax Benefits) allocated to it by this Section 2, reduced by any Tax Benefits allocated to Distributing or Controlled that are allowable under applicable Tax Law to reduce such Taxes, (ii) pay such Taxes, as so reduced, either to the applicable Tax Authority or to Distributing as required by Section 4, and (iii) pay Distributing for any Tax Benefits allocated to Distributing by this Section 2 that Controlled uses to reduce Taxes payable by it pursuant to clause (ii) of this Section 2.1(b).
- (c) Use of Tax Benefits. For purposes of Section 2.1(a)(i), (x) Distributing shall reduce Taxes allocated to it with any Tax Benefits allocated to Distributing that are allowable under applicable Tax Law in the same Tax Year prior to reducing such Taxes with any Tax Benefits allocated to Controlled, and (y) Distributing shall reduce Taxes allocated to it by Tax Benefits allocated to Controlled only to the extent such Tax Benefits are not taken into account by Controlled pursuant to Section 2.1(b)(i) in the same Tax Year. For purposes of Section 2.1(b)(i), (x) Controlled shall reduce Taxes allocated to it with any Tax Benefits allocated to Controlled that are allowable under applicable Tax Law in the same Tax Year prior to reducing such Taxes with any Tax Benefits allocated to Distributing, and (y) Controlled shall reduce Taxes allocated to it by Tax Benefits allocated to Distributing only to the extent such Tax Benefits are not taken into account by Distributing pursuant to Section 2.1(a)(i) in the same Tax Year.
  - 2.2 <u>Allocation Rules.</u> For purposes of Section 2.1:
- (a) General Rule. Except as otherwise provided in this Section 2.2, Taxes (determined without regard to Tax Benefits) for any Tax Year (or portion thereof) shall be allocated between Controlled and Distributing in proportion to

15

the taxable income or other applicable items attributable to or arising from the respective Controlled Business and Distributing Business (as so defined for such Tax Year or portion thereof) that contribute to such Taxes, and Tax Benefits for any Tax Year (or portion thereof) shall be allocated between Controlled and Distributing in proportion to the losses, credits, or other applicable items attributable to or arising from the respective Controlled Business and Distributing Business (as so defined for such Tax Year or portion thereof) that contribute to such Tax Benefits.

- (b) Transaction Taxes and Transaction Tax-Related Losses.
- (i) Distributing shall be allocated all Transaction Taxes and Transaction Tax-Related Losses other than any Transaction Taxes and Transaction Tax-Related Losses allocated to Controlled pursuant to clause (ii) of this Section 2.2(b).
- (ii) Controlled shall be allocated any Transaction Taxes and Transaction Tax-Related Losses that (w) result primarily from, individually or in the aggregate, any breach by Controlled of any of its covenants set forth in Section 7.1 hereof, (x) result from the Controlled Stock not being treated as stock of Controlled, or being treated as Section 306 stock within the meaning of Section 306(c) of the Code, for U.S. federal income tax purposes, (y) result from the Liberty Interactive Common Stock, the Liberty Capital Common Stock or the Liberty Starz Common Stock not being treated as stock of Distributing, or being treated as Section 306 stock within the meaning of Section 306(c) of the Code, for U.S. federal income tax purposes, or (z) result from a Controlled Section 355(e) Event; provided, however, that Controlled shall not be allocated any Transaction Taxes and Transaction Tax-Related Losses pursuant to clause (x) or (y) of this Section 2.2(b)(ii) to the extent that such Transaction Tax-Related Losses result primarily from, individually or in the aggregate, any breach by Distributing of any of its covenants set forth in Section 7.1 hereof.
  - (c) LEI Transaction Taxes and LEI Tax-Related Losses.
- (i) Controlled shall be allocated all LEI Transaction Taxes and LEI Tax-Related Losses other than any LEI Transaction Taxes and LEI Tax-Related Losses allocated to Distributing pursuant to clause (ii) of this Section 2.2(c).

or in the aggregate, any breach by	(ii) Distributing shall be allocated any LEI Transaction Taxes and LEI Tax-Related Losses that result primarily from, individually Distributing of any of its covenants set forth in Section 7.1 hereof.
(d)	Taxes Resulting from the News Transactions.
Taxes and News Tax-Related Los	(i) Controlled shall be allocated all News Transaction Taxes and News Tax-Related Losses other than any News Transaction sees allocated to Distributing pursuant to clause (ii) of this Section 2.2(d).
individually	(ii) Distributing shall be allocated any News Transaction Taxes and News Tax-Related Losses that result primarily from,
	16
	<del>-</del>
or in the aggregate, any breach by	Distributing of any of its covenants set forth in Section 7.1 hereof.
(e)	Taxes and Losses with Respect to Tracking Stock.
(0)	(i) Controlled shall be allocated all Tracking Stock Taxes and Losses other than any Tracking Stock Taxes and Losses allocated to
Distributing pursuant to clause (ii)	
(as those terms are defined by Tre	(ii) Distributing shall be allocated any Tracking Stock Taxes and Losses that (x) result primarily from, individually or in the ting of any of its covenants set forth in Section 7.1 hereof, or (y) result from "deferred intercompany transactions" or "excess loss accounts" assury Regulations) that are triggered by the actual or deemed disposition of any assets referred to in clause (ii) or (iv) of the definition of and would otherwise be allocated to Distributing, but for clause (i) of this Section 2.2(e).
	Carryovers or Carrybacks of Tax Benefits. If any Tax Item allocable to the Controlled Business in a Tax Year is carried forward or back other Tax Year, then, except as provided in Section 2.2(g), the resulting Tax Benefit shall be allocated to Controlled. If any Tax Item less in a Tax Year is carried forward or back and utilized as a Tax Benefit in another Tax Year, the resulting Tax Benefit shall be allocated to
Period, then, notwithstanding Sect	Controlled Carrybacks from Post-Distribution Period. If, pursuant to Section 3.4(d), any Tax Item allocable to Controlled in a Tax Year Period is carried back and generates a Tax Benefit on a Combined Return filed with respect to a Tax Year beginning in the Pre-Distribution ion 2.2(f), any resulting Tax Benefit shall be allocated to Distributing to the extent, if any, that the carryback of such Tax Item increases the ibuting or reduces the amount of Tax Benefits allocable to Distributing that otherwise could be used with respect to such Tax Year.
	COD Income. Distributing shall be allocated all Taxes and Tax Items resulting from an aggregate of approximately \$846.2 million in net atably in Tax Years 2014 through 2018 for U.S. federal income tax purposes as a result of the cancellation in April 2009 of \$400 million in nangeables and \$350 million in principal amount of the 2030 Exchangeables and the cancellation in March 2009 of \$10 million in principal s.
(i)	Employee Compensation and Employee Benefits.
or settlement of any Compensator Distributing;	(i) <i>Pre-Distribution Period.</i> For any Pre-Distribution Period: (x) Taxes and Tax Items arising from the issuance, vesting, exercise y Equity Interests with respect to any series of Liberty Interactive Common Stock or in any Interactive Entity shall be allocated to
C,	17
	<u> </u>
stock (other than any series of Lib Entity shall be allocated to Control	om the issuance, vesting, exercise or settlement of any Compensatory Equity Interests with respect to any class or series of Distributing erty Interactive Common Stock) or in any Person (including DTV, Discovery, LGI, and Ascent) other than Distributing or any Interactive lled, and (z) any other Taxes or Tax Items related to employee, independent contractor or director compensation or employee benefits shall extent that the Distributing Business is or was responsible for the underlying obligation and to Controlled to the extent that the Controlled the underlying obligation.
Distributing; (y) Taxes and Tax It Controlled stock or in any Person other Taxes or Tax Items related t	(ii) Post-Distribution Period. For any Post-Distribution Period: (x) Taxes and Tax Items arising from the issuance, vesting, pensatory Equity Interests with respect to any class or series of Distributing stock or in any Interactive Entity shall be allocated to ems arising from the issuance, vesting, exercise or settlement of any Compensatory Equity Interests with respect to any class or series of (including DTV, Discovery, LGI, and Ascent) other than Distributing or any Interactive Entity shall be allocated to Controlled, and (z) any of employee, independent contractor or director compensation or employee benefits shall be allocated to Distributing to the extent that the sponsible for the underlying obligation and to Controlled to the extent that the Controlled Business is or was responsible for the underlying
payments previously made by each by such Company pursuant to this	Alternative Minimum Tax Credit. Any credit arising in any Tax Year (or portion thereof) from the payment of any alternative minimum n any Combined Return shall be allocated between Distributing and Controlled in a manner that offsets the excess of the net payment or h Company pursuant to this Agreement in respect of such Combined Return over the net payment or payments that would have been made Agreement in respect of such Combined Return if no alternative minimum consolidated federal tax liability had been owed with respect to sees of this Section 2.2(j), net payments received shall be treated as a negative amount of net payments made.
$\begin{array}{c} \text{(k)} \\ \text{any Tax Year (or portion thereof)} \end{array}$	Interactive Pre-Distribution Acquired Subsidiaries. Taxes and Tax Items of each Interactive Pre-Distribution Acquired Subsidiary for ending on or prior to the date such entity was acquired, directly or indirectly, by Distributing shall be allocated to Distributing.
Taxes and Tax Items of such Cont	Acquired Subsidiaries. If any Person becomes a Subsidiary of any member of the Controlled Group in any transaction after the not a member of the Controlled Group or the Distributing Group prior to such transaction) (a "Controlled Acquired Subsidiary"), then any rolled Acquired Subsidiary for any Tax Year (or portion thereof) ending on or prior to the date of such transaction shall be allocated to s a Subsidiary of any member of the Distributing Group in any transaction after the Distribution (and such Person was not a member of the

Distributing Group prior to such transaction) (a "Distributing Acquired Subsidiary"), then any Taxes and Tax Items of such Distributing Acquired Subsidiary for any Tax Year (or portion thereof) ending on or prior to the date of such transaction shall be allocated to Distributing.

- (m) Straddle Losses. Distributing shall be allocated any Tax Benefit resulting from the Straddle Losses.
- (n) Share Distribution. Controlled shall be allocated all Taxes and Tax Items resulting from the Share Distribution.
- (o) Transfer Taxes. All Transfer Taxes shall be allocated 50% to Distributing and 50% to Controlled.
- (p) Certain Pre-Existing Tax Sharing Agreements. Controlled shall be allocated all Taxes, Tax Items, Losses and payments attributable to any AT&T TSA Liabilities, DHC TSA Liabilities and LMI TSA Liabilities, and Controlled shall be allocated all AT&T TSA Benefits, DHC TSA Benefits and LMI TSA Benefits and any Taxes and Tax Items arising therefrom.

#### **SECTION 3.** Preparation and Filing of Tax Returns.

- 3.1 <u>Combined Returns.</u> Except as otherwise provided in this Section 3:
- (a) Preparation of Combined Returns. Distributing shall be responsible for preparing and filing (or causing to be prepared and filed) all Combined Returns for any Tax Year.
  - 3.2 <u>Separate Returns.</u> Except as otherwise provided in this Section 3:
- (a) Tax Returns to be Prepared by Distributing. Distributing shall be responsible for preparing and filing (or causing to be prepared and filed) (i) all Separate Returns for a Tax Year beginning on or before the Distribution Date that include Tax Items of the Distributing Business, determined in accordance with the allocation rules of Section 2.2, and (ii) all Separate Returns for a Tax Year beginning after the Distribution Date that include one or more members of the Distributing Group.
- (b) Tax Returns to be Prepared by Controlled. Controlled shall be responsible for preparing and filing (or causing to be prepared and filed) (i) all Separate Returns for a Tax Year beginning on or before the Distribution Date that include Tax Items of the Controlled Business, determined in accordance with the allocation rules of Section 2.2, and (ii) all Separate Returns for a Tax Year beginning after the Distribution Date that include one or more members of the Controlled Group.

19

# 3.3 <u>Provision of Information.</u>

- (a) Distributing shall provide to Controlled, and Controlled shall provide to Distributing, any information about members of the Distributing Group or the Controlled Group, respectively, that the Preparer needs to determine the amount of Taxes due on any Payment Date with respect to a Tax Return for which the Preparer is responsible pursuant to Section 3.1 or 3.2 and to properly and timely file all such Tax Returns.
- (b) If a member of the Controlled Group supplies information to a member of the Distributing Group, or a member of the Distributing Group supplies information to a member of the Controlled Group, and an officer of the requesting member intends to sign a statement or other document under penalties of perjury in reliance upon the accuracy of such information, then a duly authorized officer of the member supplying such information shall certify, to the best of such officer's knowledge, the accuracy of the information so supplied.
  - 3.4 Special Rules Relating to the Preparation of Tax Returns.
- (a) General Rule. Except as otherwise provided in this Agreement, the Company responsible for filing (or causing to be filed) a Tax Return pursuant to Sections 3.1 or 3.2 shall have the exclusive right, in its sole discretion, with respect to such Tax Return to determine (i) the manner in which such Tax Return shall be prepared and filed, including the elections, methods of accounting, positions, conventions and principles of taxation to be used and the manner in which any Tax Item shall be reported, (ii) whether any extensions may be requested, (iii) whether an amended Tax Return shall be filed, (iv) whether any claims for refund shall be made, (v) whether any refunds shall be paid by way of refund or credited against any liability for the related Tax and (vi) whether to retain outside firms to prepare or review such Tax Return.
  - (b) Controlled Tax Returns. With respect to any Separate Return for which Controlled is responsible pursuant to Section 3.2(b):
- (i) Controlled may not take (and shall cause the members of the Controlled Group not to take) any positions that it knows, or reasonably should know, would adversely affect any member of the Distributing Group, except to the extent that the failure to take such position would be contrary to applicable Tax Law; and Controlled and the other members of the Controlled Group must (x) allocate Tax Items between such Separate Return for which Controlled is responsible and any related Combined Return for which Distributing is responsible that is filed with respect to the same Tax Year in a manner that is consistent with the reporting of such Tax Items on such related Combined Return and (y) make any applicable elections required under applicable Tax Law (including, without limitation, under Treasury Regulations Section 1.1502-76(b)(2)), necessary to effect such allocation.
  - (c) Election to File Consolidated, Combined or Unitary Tax Returns Distributing shall have the sole discretion of filing any Tax Return on

20

consolidated, combined or unitary basis, if such Tax Return would include at least one member of each Group and the filing of such Tax Return is elective under applicable Tax Law.

- (d) Filing Claims for Carrybacks. If a Tax Item allocable to Controlled is carried back from a Tax Year beginning in the Post-Distribution Period and generates a Tax Benefit on a Combined Return filed with respect to a Tax Year beginning in the Pre-Distribution Period, then, upon the request of Controlled, Distributing may, in its sole discretion, file a claim for refund arising from such Tax Benefit. Any resulting Tax Benefit shall be allocated to Controlled pursuant to Section 2.2(f), except as otherwise provided by Section 2.2(g).
- (e) Withholding and Reporting. Following the Effective Time, in the event any Compensatory Equity Interests are settled (whether by issuance, exercise, vesting or otherwise) by the corporation that is the issuer or obligor under the Compensatory Equity Interest (the "issuing corporation") or by another member of the Group to which the issuing corporation belongs, and the issuing corporation is not a member of the same Group as the Employing Party, the Company whose

Group includes the issuing corporation shall promptly remit to the Employing Party an amount of cash equal to the amount required to be withheld in respect of any withholding Taxes, and in the application of this Agreement, the Employing Party shall not be liable for such withholding Taxes or for failure to remit to the applicable Tax Authority any amount required to have been withheld from the recipient of the Compensatory Equity Interest in connection with such issuance, exercise, vesting or settlement, except to the extent that the Company whose Group includes the issuing corporation shall have remitted such amount to the Employing Party or to the applicable Tax Authority. Distributing shall promptly notify Controlled, and Controlled shall promptly notify Distributing, regarding the exercise of any option or the issuance, vesting, exercise or settlement of any other Compensatory Equity Interest to the extent that, as a result of such issuance, exercise, vesting or settlement, any other party may be entitled to a Tax Benefit or required to pay any Tax, or such information otherwise may be relevant to the preparation of any Tax Return or payment of any Tax by such other party or parties.

# SECTION 4. Tax Payments.

4.1 <u>Payment of Taxes to Tax Authority.</u> Distributing shall be responsible for remitting to the proper Tax Authority the Tax shown on any Tax Return for which it is responsible for the preparation and filing pursuant to Section 3.1(a) or Section 3.2(a), and Controlled shall be responsible for remitting to the proper Tax Authority the Tax shown on any Tax Return for which it is responsible for the preparation and filing pursuant to Section 3.2(b).

# 4.2 <u>Indemnification Payments.</u>

(a) Tax Payments Made by the Distributing Group. If any member of the Distributing Group is required to make a payment to a Tax Authority for Taxes allocated to Controlled under this Agreement, Controlled shall pay the amount of

21

Taxes allocated to it to Distributing not later than the later of (i) five business days after receiving notification requesting such amount, and (ii) one business day prior to the date such payment is required to be made to such Tax Authority.

(b) Tax Payments Made by the Controlled Group. If any member of the Controlled Group is required to make a payment to a Tax Authority for Taxes allocated to Distributing under this Agreement, Distributing shall pay the amount of Taxes allocated to it to Controlled not later than the later of (i) five business days after receiving notification requesting such amount, and (ii) one business day prior to the date such payment is required to be made to such Tax Authority.

# 4.3 Payments for Tax Refunds and Tax Benefits.

- (a) Tax Refund or Tax Benefit Received by Distributing Group. If a member of the Distributing Group receives a Tax Refund with respect to Taxes for which Controlled is liable hereunder or uses a Tax Benefit for which Controlled is entitled to reimbursement pursuant to clause (ii) of Section 2.1(a), Distributing shall pay to Controlled, within five business days following the receipt of the Tax Refund or the use of such Tax Benefit, an amount equal to such Tax Refund or Tax Benefit.
- (b) Tax Refund or Tax Benefit Received by Controlled Group. If a member of the Controlled Group receives a Tax Refund with respect to Taxes for which Distributing is liable hereunder or uses a Tax Benefit for which Distributing is entitled to reimbursement pursuant to clause (ii) of Section 2.1(b), Controlled shall pay to Distributing, within five business days following the receipt of the Tax Refund or the use of such Tax Benefit, an amount equal to such Tax Refund or Tax Benefit
- (c) Rules Regarding Tax Benefits. For purposes of this Agreement, a Tax Benefit (other than a Tax Refund) shall be considered used or received (i) at the time the Tax Return is filed with respect to such Tax Benefit, or (ii) if no Tax Return is filed, (x) at the time a Tax Refund generated by use of such Tax Benefit is received or (y) if no Tax Refund is received, at the time the Tax would have been due in the absence of such Tax Benefit. The amount of such Tax Benefit shall be the amount by which Taxes are actually reduced by such Tax Benefit.
- 4.4 <u>Interest on Late Payments.</u> Payments pursuant to this Agreement that are not made by the date prescribed in this Agreement or, if no such date is prescribed, not later than five business days after demand for payment is made (the "Due Date") shall bear interest for the period from and including the date immediately following the Due Date through and including the date of payment at the Interest Rate. Such interest will be payable at the same time as the payment to which it relates.
- 4.5 <u>Initial Determinations and Subsequent Adjustments.</u> The initial determination of the amount of any payment that one Company is required to make to another under this Agreement shall be made on the basis of the Tax Return as filed, or, if the Tax to which the payment relates is not reported in a Tax Return, on the basis of the amount of Tax initially paid to the Tax Authority. The amounts paid under this

22

Agreement shall be redetermined, and additional payments relating to such redetermination shall be made, as appropriate, if as a result of an audit by a Tax Authority or for any other reason (w) additional Taxes to which such determination relates are subsequently paid, (x) a Tax Refund or a Tax Benefit relating to such Taxes is received or used, (y) the amount or character of any Tax Item is adjusted or redetermined, or (z) a Tax Benefit allocable to Distributing that is reduced in one Tax Year by reason of the carryback of a Tax Item allocable to the Controlled Business, resulting in an allocation to Distributing of a Tax Benefit pursuant to Section 2.2(g), is used by Distributing in a subsequent Tax Year. Each payment required by the immediately preceding sentence (i) as a result of a payment of additional Taxes will be due five business days after the date on which the additional Taxes were paid or, if later, five business days after the date of a request from the other Company for the payment, (ii) as a result of the receipt or use of a Tax Refund or Tax Benefit will be due five business days after the Tax Refund or Tax Benefit was received or used, or (iii) as a result of an adjustment or redetermination of the amount or character of a Tax Item will be due five business days after the date on which the final action resulting in such adjustment or redetermination is taken by a Tax Authority or either Company or any of their Subsidiaries. If a payment is made as a result of an audit by a Tax Authority which does not conclude the matter, further adjusting payments will be made, as appropriate, to reflect the outcome of subsequent administrative or judicial proceedings.

4.6 Tax Consequences of Payments. For all Tax purposes and to the extent permitted by applicable Tax Law, the parties hereto shall treat any payment made pursuant to this Agreement as a capital contribution or a distribution between Distributing and Controlled, as the case may be, immediately prior to the Distribution. If the receipt or accrual of any payment under this Agreement causes, directly or indirectly, an increase in the taxable income of the recipient under one or more applicable Tax Laws, such payment shall be increased so that, after the payment of any Taxes with respect to the payment, the recipient thereof shall have realized the same net amount it would have realized had the payment not resulted in taxable income. To the extent that Taxes for which any party hereto (the indemnifying party) is required to pay another party (the indemnified party) pursuant to this Agreement may be deducted or credited in determining the amount of any other Taxes required to be paid by the indemnified party (for example, state Taxes which are permitted to be deducted in determining federal Taxes), the amount of any payment made to the indemnified party by the indemnified party occurs following the payment made to the indemnified Party with respect to the relevant indemnified Taxes, the indemnified party shall promptly repay the indemnifying party the amount of such reduction when actually realized. If the Tax Benefit arising from the foregoing reduction of Taxes described in this Section 4.6 is subsequently decreased or eliminated, then the indemnifying party shall promptly pay the indemnified party the amount of the decrease in such Tax Benefit.

# **SECTION 5.** Assistance and Cooperation.

5.1 <u>Cooperation.</u> In addition to the obligations enumerated in Sections 3.3 and 7.7, Distributing and Controlled shall cooperate (and shall cause their respective Subsidiaries and Affiliates to cooperate) with each other and with each other's agents, including accounting firms and legal counsel, in connection with Tax matters, including provision of relevant documents and information in their possession and making available to each other, as reasonably requested and available, personnel (including officers, directors, employees and agents of the parties or their respective Subsidiaries or Affiliates) responsible for preparing, maintaining, and interpreting information and documents relevant to Taxes, and personnel reasonably required as witnesses or for purposes of providing information or documents in connection with any administrative or judicial proceedings relating to Taxes.

# 5.2 Supplemental Rulings.

- (a) Each of the Companies agrees that, at the reasonable request of the other Company (the "Requesting Party"), each Company shall (and shall cause their respective Subsidiaries and Affiliates to) cooperate and use reasonable best efforts to obtain, as expeditiously as reasonably practicable, a Supplemental Ruling from the IRS. The Requesting Party shall reimburse the other parties for all reasonable out-of-pocket costs and expenses incurred by such parties or their Subsidiaries or Affiliates in connection with obtaining or requesting such Supplemental Ruling within five business days after receiving an invoice from such party therefor.
- (b) Distributing shall provide Controlled with a reasonable opportunity to review and comment on each Supplemental IRS Submission to be filed by Distributing prior to the filing of such Supplemental IRS Submission with the IRS, and Controlled shall provide Distributing with a reasonable opportunity to review and comment on each Supplemental IRS Submission to be filed by Controlled prior to the filing of such Supplemental IRS Submission with the IRS. No Supplemental IRS Submission with the IRS unless, prior to such filing Distributing shall have agreed as to the contents of such Supplemental IRS Submission to the extent that the Supplemental IRS Submission (i) includes statements or representations relating to facts that are or will be under the control of any member of the Distributing Group or any of its Affiliates or (ii) is relevant to, or creates, any actual or potential obligations of, or limitations on, any member of the Distributing Group or any of their Affiliates; provided, however, that if the IRS requests same-day filing of a Supplemental IRS Submission that does not include any material issue or statement, then Controlled is required only to make a good faith effort to notify Distributing's representatives and to give such representatives an opportunity to review and comment on such Supplemental IRS Submission prior to filing it with the IRS. No Supplemental IRS Submission shall be filed by Distributing with the IRS unless, prior to the filing, Controlled shall have agreed as to the contents of such Supplemental IRS Submission to the extent that the Supplemental IRS Submission (i) includes statements or representations relating to facts that are or will be under the control of any member of the Controlled Group or any of its Affiliates or (ii) is relevant to, or creates, any actual or

24

potential obligations of, or limitations on, any member of the Controlled Group or any of their Affiliates; provided, however, that if the IRS requests same-day filing of a Supplemental IRS Submission that does not include any material issue or statement, then Distributing is required only to make a good faith effort to notify Controlled's representatives and to give such representatives an opportunity to review and comment on such Supplemental IRS Submission prior to filing it with the IRS. Prior to filing any Supplemental IRS Submission that includes any material issue or statement, each Company shall represent to the other Company that (i) it has reviewed the Supplemental IRS Submission, and (ii) all information and representations, if any, relating to such Company, each member of its Group, and their respective Affiliates that are contained in the Supplemental IRS Submission are true, correct and complete in all material respects. Each Company (or its representatives) shall provide the other Company (or its representatives) with copies of each Supplemental IRS Submission filed with the IRS promptly following the filing thereof.

(c) Neither Company nor its representatives shall conduct any substantive communications with the IRS regarding any material issue arising with respect to any Supplemental Ruling, including meetings or conferences with IRS personnel, whether telephonically, in person or otherwise, without first notifying the other Company (or its representatives) and giving the other Company (or its representatives) a reasonable opportunity to participate, and a reasonable number of such Company's representatives shall have an opportunity to participate in all conferences or meetings with IRS personnel that take place in person, regardless of the nature of the issues expected to be discussed; provided, however, that in the case of communications concerning a Supplemental Ruling that occur during an unscheduled conference initiated by the IRS or a conference initiated by a Company or its representatives for a purpose unrelated to a Supplemental Ruling in connection with which it is not reasonably practicable to provide to the other Company or its representatives advance notice and an opportunity to participate, such Company (or its representatives) shall promptly update the other Company and its representatives as to the content of such communications. Each Company shall promptly provide the other Company (or its representatives) with copies of any correspondence received by such Company (or its representatives) from the IRS relating to any Supplemental Ruling.

# SECTION 6. Tax Records.

- Records that are in their possession, and that could affect the liability of any member of the other Group for Taxes, for as long as the contents thereof may become material in the administration of any matter under applicable Tax Law, but in any event until the later of (x) the expiration of any applicable statutes of limitation, as extended, and (y) seven years after the Distribution Date.
- 6.2 Access to Tax Records. Controlled shall make available, and cause its Subsidiaries to make available, to members of the Distributing Group for inspection and copying (x) all Tax Records in their possession that relate to a Pre-

25

Distribution Period, and (y) the portion of any Tax Record in their possession that relates to a Post-Distribution Period and which is reasonably necessary for the preparation of a Tax Return by a member of the Distributing Group or any of their Affiliates or with respect to any audit, litigation or other proceeding by a Tax Authority relating to such return. Distributing shall make available, and cause its Subsidiaries to make available, to members of the Controlled Group for inspection and copying the portion of any Tax Record in their possession that relates to a Pre-Distribution Period and which is reasonably necessary for the preparation of a Tax Return by a member of the Controlled Group or any of their Affiliates or with respect to any audit, litigation or other proceeding by a Tax Authority relating to such return.

- 6.3 Confidentiality. Each party hereby agrees that it will hold, and shall use its reasonable best efforts to cause its officers, directors, employees, accountants, counsel, consultants, advisors and agents to hold, in confidence all records and information prepared and shared by and among the parties in carrying out the intent of this Agreement, except as may otherwise be necessary in connection with the filing of Tax Returns or any administrative or judicial proceedings relating to Taxes or unless disclosure is compelled by a governmental authority. Information and documents of one party (the "Disclosing Party") shall not be deemed to be confidential for purposes of this Section 6.3 to the extent such information or document (i) is previously known to or in the possession of the other party or parties (the "Receiving Party") and is not otherwise subject to a requirement to be kept confidential, (ii) becomes publicly available by means other than unauthorized disclosure under this Agreement by the Receiving Party or (iii) is received from a third party without, to the knowledge of the Receiving Party after reasonable diligence, a duty of confidentiality owed to the Disclosing Party.
- 6.4 <u>Delivery of Tax Records.</u> On or before the Distribution Date, Distributing shall provide to Controlled (to the extent not previously provided or held by any member of the Controlled Group on the Distribution Date) copies of (i) the Separate Returns of any member of the Controlled Group, (ii) the relevant portions of any other Tax Returns with respect to any member of the Controlled Group, and (iii) other existing Tax Records (or the relevant portions thereof) reasonably necessary to

prepare and file any Tax Returns of, or with respect to, the members of the Controlled Group, or to defend or contest Tax matters relevant to the members of the Controlled Group, including in each case, all Tax Records related to Tax attributes of the members of the Controlled Group and any and all communications or agreements with, or rulings by, any Tax Authority with respect to any member of the Controlled Group.

# SECTION 7. Restrictions on Certain Actions of Distributing and Controlled; Indemnity.

#### 7.1 Restrictive Covenants.

(a) General Restrictions. Following the Effective Time, Controlled shall not, and shall cause the members of the Controlled Group and their Affiliates not to, and Distributing shall not, and shall cause the members of the Distributing Group and their Affiliates not to, take any action that, or fail to take any

26

action the failure of which, (i) would cause Distributing or any Subsidiary of Distributing immediately prior to the Distribution to recognize gain or loss, or otherwise include any amount in income, as a result of the Restructuring for U.S. federal income tax purposes (except with respect to the Share Distribution), (ii) would be inconsistent with the Contribution and Distribution qualifying, or would preclude the Contribution and Distribution from qualifying, as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code, (iii) would cause Distributing, Controlled, any of their respective Subsidiaries at the Effective Time, or the holders of Liberty Capital Common Stock or Liberty Starz Common Stock that receive stock of Controlled in the Distribution, to recognize gain or loss, or otherwise include any amount in income, as a result of the Contribution and/or the Distribution for U.S. federal income tax purposes, (iv) would be inconsistent with the LEI Transaction qualifying, or would preclude the LEI Transaction from qualifying, as a tax-free transaction described under Sections 368(a), 355 and 361 of the Code (except with respect to cash received in lieu of fractional shares), (v) would cause Distributing, any Subsidiary of Distributing immediately prior to the LEI Transaction, or the holders of Liberty Starz Common Stock that received stock of LEI in the LEI Transaction to recognize gain or loss, or otherwise include any amount in income, as a result of the LEI Transaction for U.S. federal income tax purposes (except with respect to cash received in lieu of fractional shares), (vi) would be inconsistent with the News Transactions qualifying, or would preclude the News Transactions from qualifying, as tax-free transactions described under Sections 368(a), 355 and/or 361 of the Code, as applicable, (vii) would cause News or any Subsidiary of News immediately prior to any of the News Transactions, or Distributing or any of its Subsidiaries on February 27, 2008, to recognize gain or loss, or otherwise include any amount in income, as a result of the News Transactions for U.S. federal income tax purposes, (viii) would be inconsistent with the Liberty Restructuring qualifying, or would preclude the Liberty Restructuring from qualifying, as a tax-free reorganization described under Section 368(a)(1)(F) of the Code (except with respect to cash received in lieu of fractional shares), (ix) would cause Distributing, Liberty LLC, any of their respective Subsidiaries immediately prior to the Liberty Restructuring or any of the former holders of stock of Liberty LLC that received stock of Distributing in the Liberty Restructuring to recognize gain or loss, or otherwise include any amount in income, as a result of the Liberty Restructuring for U.S. federal income tax purposes (except with respect to cash received in lieu of fractional shares), (ix) would be inconsistent with the Reclassification qualifying, or would preclude the Reclassification from qualifying, as a tax-free reorganization described under Section 368(a)(1)(E) of the Code, or (x) would cause Distributing, any of its Subsidiaries immediately prior to the Reclassification or any holders of Distributing stock that received stock of Distributing in the Reclassification to recognize gain or loss, or otherwise include any amount in income, as a result of the Reclassification.

(b) Restricted Actions. Without limiting the provisions of Section 7.1(a) hereof, following the Effective Time, Controlled shall not, and shall cause the members of the Controlled Group and their Affiliates not to, and Distributing shall not, and shall cause the members of the Distributing Group and their Affiliates not to, take any action that, or fail to take any action the failure of which, (i) would be inconsistent with, or would cause any Person to be in breach of, any representation or

27

covenant, or any material statement, made in the Tax Materials, the LEI Tax Materials, the News Tax Materials or the Tracking Stock Tax Materials, or (ii) would be inconsistent with, or would cause Distributing to be in breach of, any representation or covenant made in the LEI Tax Sharing Agreement or the Tax Matters Agreement.

- (c) Reporting. Unless and until there has been a Final Determination to the contrary, each party agrees not to take any position on any Tax Return, in connection with any Tax Contest, or otherwise for Tax purposes that is inconsistent with the Ruling or the Tax Opinion.
- 7.2 <u>Distributing Indemnity.</u> Distributing agrees to indemnify and hold harmless each member of the Controlled Group and their respective directors, officers, employees, agents, successors and assigns (the "Controlled Indemnitees") from and against any and all (without duplication) (a) Taxes, Tax Items, Losses and payments allocated to Distributing pursuant to Section 2.2(b), (c) LEI Transaction Taxes and LEI Tax-Related Losses allocated to Distributing pursuant to Section 2.2(d), (e) Tracking Stock Taxes and Losses allocated to Distributing pursuant to Section 2.2(d), (e) Tracking Stock Taxes and Losses allocated to Distributing pursuant to Section 2.2(e), (f) Taxes and Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Distributing contained in this Agreement, (g) Transfer Taxes allocated to Distributing pursuant to Section 2.2(o), and (h) Losses, including reasonable out-of-pocket legal, accounting and other advisory and court fees and expenses, incurred in connection with the items described in clauses (a) through (g); <u>provided, however</u>, that notwithstanding clauses (a), (f) and (h) of this Section 7.2, Distributing shall not be responsible for, and shall have no obligation to indemnify or hold harmless any Controlled Indemnitee for, (x) any Transaction Taxes, Transaction Tax-Related Losses, News Transaction Taxes, News Tax-Related Losses or Tracking Stock Taxes and Losses that are allocated to Controlled pursuant to Sections 2.2(b)(ii), (c)(ii), (d)(i) or (e)(i), or (y) any Taxes or Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Controlled contained in this Agreement.
- 7.3 Controlled Indemnity. Controlled agrees to indemnify and hold harmless each member of the Distributing Group and their respective directors, officers, employees, agents, successors and assigns (the "Distributing Indemnitees") from and against any and all (without duplication) (a) Taxes, Tax Items, Losses and payments allocated to Controlled pursuant to Section 2.2, (b) Transaction Taxes and Transaction Tax-Related Losses allocated to Controlled pursuant to Section 2.2(b), (c) LEI Transaction Taxes and LEI Tax-Related Losses allocated to Controlled pursuant to Section 2.2(c), (d) News Transaction Taxes and News Tax-Related Losses allocated to Controlled pursuant to Section 2.2(e), (f) Taxes and Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Controlled contained in this Agreement, (g) Transfer Taxes allocated to Controlled pursuant to Section 2.2(o), and (h) Losses, including reasonable out-of-pocket legal,

28

accounting and other advisory and court fees, incurred in connection with the items described in clauses (a) through (g); provided, however, that notwithstanding clauses (a), (f) and (h) of this Section 7.3, Controlled shall not be responsible for, and shall have no obligation to indemnify or hold harmless any Distributing Indemnitee for, (x) any Transaction Taxes, Transaction Taxe-Related Losses, LEI Transaction Taxes, LEI Tax-Related Losses, News Transaction Taxes, News Tax-Related Losses or Tracking Stock Taxes and Losses that are allocated to Distributing pursuant to Sections 2.2(b)(i), (c)(ii), (d)(ii) or (e)(ii), or (y) any Taxes or Losses arising out of or based upon any breach or nonperformance of any covenant or agreement made or to be performed by Distributing contained in this Agreement.

7.4 Scope. The provisions of this Section 7 are intended to be for the benefit of, and shall be enforceable by, each Distributing Indemnitee and its successors in interest and each Controlled Indemnitee and its successors in interest.

7.5 Notices of Tax Contests. Each Company shall provide prompt notice to the other Company of any pending or threatened Tax audit, assessment, proceeding or other Tax Contest, Joint Claim or Controlled Claim of which it becomes aware relating to Taxes, Losses or any other liabilities or amounts for which it is or may be indemnified by such other Company hereunder. Such notice shall contain (i) factual information (to the extent known) describing any asserted Tax liability or other claim in reasonable detail and shall be accompanied by copies of any notice and other documents received from any Tax Authority or third party in respect of any such matters, and (ii) the amount of such asserted Tax liability or other claim. Such notice shall be given within a reasonable period of time after notice thereof was received by such Company, but any failure to give timely notice shall not affect the indemnities given hereunder except, and only to the extent that, the indemnifying Company shall have been actually prejudiced as a result of such failure. Thereafter, the indemnified Company shall deliver to the indemnifying Company such additional information with respect to such Tax Contest, Joint Claim or Controlled Claim in its possession that the indemnifying Company may reasonably request.

#### 7.6 Control of Tax Contests Generally.

- (a) General Rule. Except as provided in Sections 7.6(b), 7.8, and 7.9, each Company (or the appropriate member of its Group) shall have full responsibility, control and discretion in handling, defending, settling or contesting any Tax Contest involving a Tax reported (or that, it is asserted, should have been reported) on a Tax Return for which such Company is responsible for preparing and filing (or causing to be prepared and filed) pursuant to Section 3 of this Agreement.
- (b) Non-Preparer Participation Rights. With respect to a Tax Contest (other than with respect to a Joint Claim or Controlled Claim) of any Tax Return which could result in a Tax liability for which the Non-Preparer may be liable under this Agreement or the reduction in any Tax Benefit to which the Non-Preparer may be entitled to under this Agreement, (i) the Non-Preparer shall, at its own cost and expense, be entitled to participate in such Tax Contest, (ii) the Preparer shall keep the Non-

29

Preparer updated and informed, and shall consult with the Non-Preparer, (iii) the Preparer shall act in good faith with a view to the merits in connection with the Tax Contest, and (iv) the Preparer shall not settle or compromise such Tax Contest without the prior written consent of the Non-Preparer (which consent shall not be unreasonably withheld) if the settlement or compromise could have a more than de minimis impact on the Non-Preparer and the other members of its Group, taken as a whole.

- 7.7 Cooperation. The parties shall provide each other with all information relating to a Tax Contest, Joint Claim or Controlled Claim which is needed by the other party or parties to handle, participate in, defend, settle or contest the Tax Contest, Joint Claim or Controlled Claim. At the request of any party, the other parties shall take any reasonable action (e.g., executing a power of attorney) that is necessary to enable the requesting party to exercise its rights under this Agreement in respect of a Tax Contest, Joint Claim or Controlled Claim. Controlled shall assist Distributing, and Distributing shall assist Controlled, in taking any remedial actions that are necessary or desirable to minimize the effects of any adjustment made by a Tax Authority. The indemnifying party or parties shall reimburse the indemnified party or parties for any reasonable out-of-pocket costs and expenses incurred in complying with this Section 7.7.
- 7.8 Joint Claims. Distributing and Controlled will have the right to jointly control the defense, compromise or settlement of any Joint Claim; provided, however, that with respect to any Joint Claim arising under the LEI Tax Sharing Agreement or the Tax Matters Agreement (or otherwise subject to the indemnification provisions of the LEI Tax Sharing Agreement or the Tax Matters Agreement), Controlled's rights to jointly control, or otherwise participate in the defense, compromise or settlement of, any such Joint Claim will be subject to the terms of the LEI Tax Sharing Agreement or Tax Matters Agreement, as applicable. Distributing shall use reasonable efforts to provide Controlled with the right to jointly control with Distributing and otherwise participate in the defense, compromise or settlement of, any Joint Claim arising under the LEI Tax Sharing Agreement (or otherwise subject to the indemnification provisions of the LEI Tax Sharing Agreement or the Tax Matters Agreement), including taking action on behalf of Controlled (at the request of Controlled) to the extent any other party to the LEI Tax Sharing Agreement or the Tax Matters Agreement does not recognize Controlled's ability to act thereunder; provided, however, that Distributing shall not be required to relinquish any rights that it has to control the defense, compromise or settlement of any such Joint Claim (other than to Controlled pursuant to the foregoing). No indemnified Company shall settle or compromise or consent to entry of any judgment with respect to any such Joint Claim without the prior written consent of the indemnifying Company, which consent may be withheld in the indemnifying Company's sole discretion. No indemnifying Company shall settle or compromise or consent to entry of any judgment with respect to any such Joint Claim unless such settlement, compromise or consent (x) includes an unconditional release of the indemnified Company and (y) does not enjoin or restrict in any way the future actions or conduct of the indemnified Compan

30

- Claim, and Distributing and Liberty LLC shall use reasonable efforts to provide Controlled with the right to control, and otherwise participate in the defense, compromise or settlement of, any Controlled Claim arising under the AT&T Tax Sharing Agreement, the DHC Tax Sharing Agreement or the LMI Tax Sharing Agreement, as applicable (or otherwise subject to the indemnification provisions of any of those agreements), including taking action at the direction of Controlled to the extent any other party to the AT&T Tax Sharing Agreement, the DHC Tax Sharing Agreement, as the case may be, does not recognize Controlled's ability to act thereunder. Distributing and Liberty LLC shall not settle or compromise or consent to entry of any judgment with respect to any Controlled Claim without the prior written consent of Controlled, which consent may be withheld in Controlled's sole discretion. Controlled shall not settle or compromise or consent to entry of any judgment with respect to any such Controlled Claim unless such settlement, compromise or consent (x) includes an unconditional release of Distributing and Liberty LLC and (y) does not enjoin or restrict in any way the future actions or conduct of Distributing or Liberty LLC (other than with respect to their performance hereunder).
- 7.10 Other Claims. In the event any Distributing Indemnitee should have a claim against Controlled, or any Controlled Indemnitee should have a claim against Distributing, under this Section 7 that does not involve a third party action, such indemnified Company (or Distributing on behalf of all Distributing Indemnitees or Controlled on behalf of all Controlled Indemnitees, as applicable) shall as promptly as practicable notify the indemnifying Company of such claim, describing such claim and the factual basis thereof, the amount of such claim (if known) and the method of computation of such amount, all with reasonable particularity.

# **SECTION 8.** General Provisions.

- 8.1 <u>Termination</u>. This Agreement shall terminate at such time as all obligations and liabilities of the parties hereto have been satisfied. The obligations and liabilities of the parties arising under this Agreement shall continue in full force and effect until all such obligations have been satisfied and such liabilities have been paid in full, whether by expiration of time, operation of law, or otherwise.
- 8.2 <u>Predecessors or Successors.</u> Any reference to Distributing, Controlled, their respective Subsidiaries, or any other Person in this Agreement shall include any predecessors or successors (*e.g.*, by merger or other reorganization, liquidation, conversion, or election under Treasury Regulations Section 301.7701-3) of Distributing, Controlled, such Subsidiary, or such Person, respectively.
- 8.3 Expenses. Except as otherwise expressly provided for herein, each Company and its Subsidiaries shall bear their own expenses incurred in connection with preparation of Tax Returns and other matters related to Taxes under the provisions of this Agreement for which they are liable; provided, however, that any fees or expenses incurred in connection with the preparation of a Combined Return shall be allocated

between Distributing and Controlled in a manner resulting in Distributing and Controlled, respectively, bearing a reasonable approximation of the actual amount of such fees or expenses hereunder reasonably related to, and for the benefit of, their respective Groups.

- Governing Law; Jurisdiction. This Agreement and the legal relations among the parties hereto will be governed in all respects, including validity, interpretation and effect, by the laws of the State of Delaware applicable to contracts made and performed wholly therein, without giving effect to any choice or conflict of laws provisions or rules that would cause the application of the laws of any other jurisdiction. Each of the parties hereto irrevocably agrees that any legal action or proceeding with respect to this Agreement, and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement, and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware). Each of the parties hereto hereby irrevocably submits with regard to any such action or proceeding for itself and in respect of its property, generally and unconditionally, to the personal jurisdiction of the aforesaid courts and agrees that it will not bring any action relating to this Agreement or the transactions contemplated hereby in any court other than the aforesaid courts. Each of the parties hereto hereby irrevocably waives, and agrees not to assert as a defense, counterclaim or otherwise, in any action or proceeding with respect to this Agreement (a) any claim that it is not personally subject to the jurisdiction of the above named courts for any reason other than the failure to serve in accordance with Section 8.6 and this Section 8.4, (b) any claim that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) to the fullest extent permitted by applicable law, any claim that (i) the suit, action or proceeding in such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper or (iii) this Agreement or the subject matter hereof may not be enforced in or by such courts. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 8.6 shall be deemed effective service of process on such party.
- 8.5 <u>Waiver of Jury Trial.</u> EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT. EACH PARTY HERETO CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY

32

OF ANY OTHER PARTY HERETO HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF SUCH ACTION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.5.

- 8.6 Notices. All notices, requests, and other communications hereunder shall be in writing and shall be delivered in person, by facsimile (with confirming copy sent by one of the other delivery methods specified herein), by overnight courier or sent by certified, registered or express air mail, postage prepaid, and shall be deemed given when so delivered in person, or when so received by facsimile or courier, or, if mailed, three (3) calendar days after the date of mailing, as follows:
  - (a) If to Distributing or Liberty LLC, to:

Liberty Interactive Corporation 12300 Liberty Boulevard Englewood, Colorado 80112

Attn: Albert E. Rosenthaler Facsimile: (720) 875-5447

(b) If to Controlled, to:

Liberty Media Corporation 12300 Liberty Boulevard Englewood, Colorado 80112

Attn: Charles Y. Tanabe Facsimile: (720) 875-5382

or to such other address as the party to whom notice is given may have previously furnished to the other parties in writing in the manner set forth above.

- 8.7 <u>Counterparts.</u> This Agreement may be executed in two or more identical counterparts, each of which shall be deemed to be an original, and all of which together shall constitute one and the same agreement.
- 8.8 <u>Binding Effect; Assignment.</u> This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Except with respect to a merger of a party, neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto without the prior written consent of the other parties; provided, however, that each of Distributing and Controlled may assign its respective rights, interests, liabilities and obligations under this Agreement to any other

33

member of its Group, but such assignment shall not relieve Distributing or Controlled, as the assignor, of its liabilities or obligations hereunder.

- 8.9 Severability. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof. Any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Upon a determination that any provision of this Agreement is prohibited or unenforceable in any jurisdiction, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the provisions contemplated hereby are consummated as originally contemplated to the fullest extent possible.
  - 8.10 Amendments; Waivers. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and

is signed, in the case of an amendment, by each party to this Agreement, or in the case of a waiver, by the party against whom the waiver is to be effective. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. Except as otherwise provided herein, the rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by applicable law. Any consent provided under this Agreement must be in writing, signed by the party against whom enforcement of such consent is sought.

- 8.11 <u>Effective Date.</u> This Agreement shall become effective on the date recited above on which the parties entered into this Agreement.
- 8.12 <u>Change in Law.</u> Any reference to a provision of the Code or any other Tax Law shall include a reference to any applicable successor provision or
- 8.13 Authorization, Etc. Each of the parties hereto hereby represents and warrants that it has the power and authority to execute, deliver and perform this Agreement, that this Agreement has been duly authorized by all necessary corporate action on the part of such party, that this Agreement constitutes a legal, valid and binding obligation of such party and that the execution, delivery and performance of this Agreement by such party does not contravene or conflict with any provision of law or of its charter or bylaws or any agreement, instrument or order binding such party.
- 8.14 No Third Party Beneficiaries. Except as provided in Sections 7.2, 7.3, and 8.8 of this Agreement, this Agreement is solely for the benefit of the parties and their respective Subsidiaries and is not intended to confer upon any other Person any rights or remedies hereunder. Notwithstanding anything in this Agreement to the contrary, this Agreement is not intended to confer upon any Controlled Indemnitees any rights or remedies against Controlled hereunder, and this Agreement is not intended to

34

confer upon any Distributing Indemnitees any rights or remedies against Distributing hereunder.

8.15 Entire Agreement. This Agreement embodies the entire understanding among the parties relating to its subject matter and supersedes and terminates any prior agreements and understandings among the parties with respect to such subject matter, and no party to this Agreement shall have any right, responsibility, obligation or liability under any such prior agreement or understanding. Any and all prior correspondence, conversations and memoranda are merged herein and shall be without effect hereon. No promises, covenants or representations of any kind, other than those expressly stated herein, have been made to induce any party to enter into this Agreement.

### 8.16 No Strict Construction; Interpretation.

law

- (a) Distributing and Controlled each acknowledge that this Agreement has been prepared jointly by the parties hereto and shall not be strictly construed against any party hereto.
- (b) When a reference is made in this Agreement to an Article, Section, Exhibit or Schedule, such reference shall be to an Article of, a Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation". The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. All terms defined in this Agreement shall have the defined meanings when used in any certificate or other document made or delivered pursuant hereto unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to herein or in any agreement that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein. References to a Person are also to its permitted successors and assigns.
- 8.17 <u>Headings.</u> The headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement.
- 8.18 Assignment of Rights under the LEI Tax Sharing Agreement Distributing hereby assigns to Controlled all of its rights to indemnification payments and related rights under the LEI Tax Sharing Agreement with respect to any liability for LEI

35

Transaction Taxes, LEI Tax-Related Losses, News Transaction Taxes, News Tax-Related Losses, or any other Taxes or Losses that, in each case, is allocated to Controlled hereunder and with respect to which Controlled has paid in whole. If any Joint Claim is made against any member of the Distributing Group or the Controlled Group with respect to any LEI Transaction Taxes, LEI Tax-Related Losses, News Transaction Taxes or News Tax-Related Losses, or any other claim is made against any member of the Distributing Group or the Controlled Group would be entitled to indemnification under the LEI Tax Sharing Agreement, then at Controlled's request, Distributing shall assert a claim for indemnification against LEI and DTV under the LEI Tax Sharing Agreement in respect of such LEI Transaction Taxes, LEI Tax-Related Losses, News Transaction Taxes, News Tax-Related Losses, or other Taxes or Losses, as applicable, to the extent such a claim would not be frivolous. Controlled and Distributing shall jointly control the prosecution of any such claim related to LEI Transaction Taxes, LEI Tax-Related Losses, News Transaction Taxes, LEI Tax-Related Losses, News Transaction Taxes, and the principles of Section 7.6 shall govern any claim that is not a Joint Claim. Distributing shall not amend, modify or terminate the LEI Tax Sharing Agreement, or waive any rights thereunder, without the prior written consent of Controlled, which consent shall not be unreasonably withheld.

- 8.19 Assignment of Rights under the Tax Matters Agreement. Distributing hereby assigns to Controlled all its rights to indemnification payments and related rights under the Tax Matters Agreement with respect to any liability for News Transaction Taxes, News Tax-Related Losses, or any other Taxes or Losses that, in each case, is allocated to Controlled hereunder and with respect to which Controlled has paid in whole. If any Joint Claim is made against any member of the Distributing Group or the Controlled Group with respect to News Transaction Taxes or News Tax-Related Losses, or any other claim is made against any member of the Distributing Group or the Controlled Group with respect to any other Taxes or Losses for which any member of the Distributing Group or the Controlled Group would be entitled to indemnification under the Tax Matters Agreement, then at Controlled's request, Distributing shall assert a claim for indemnification against News under the Tax Matters Agreement in respect of such News Transaction Taxes, News Tax-Related Losses, or other Taxes or Losses, as applicable, to the extent such a claim would not be frivolous. Controlled and Distributing shall jointly control the prosecution of any such claim related to News Transaction Taxes or News Tax-Related Losses under the principles contained in Section 7.8, and the principles of Section 7.6 shall govern any claim that is not a Joint Claim. Distributing shall not amend, modify or terminate the Tax Matters Agreement, or waive any rights thereunder, without the prior written consent of Controlled, which consent shall not be unreasonably withheld.
- 8.20 <u>Assignment of Rights under other Tax Agreements.</u> Liberty LLC hereby assigns to Controlled all of its rights to indemnification payments and related rights under the AT&T Tax Sharing Agreement, the LMI Tax Sharing Agreement and the DHC Tax Sharing Agreement with respect to any liability for Taxes, Tax Items, Losses or payments that is allocated to Controlled hereunder and with respect to which Controlled has paid in whole. If any Controlled Claim is made against any

the Distributing Group or the Controlled Group with respect to any Taxes or Losses for which any member of the Distributing Group or the Controlled Group would be entitled to indemnification under the AT&T Tax Sharing Agreement, the LMI Tax Sharing Agreement or the DHC Tax Sharing Agreement, then at Controlled's request, Distributing shall assert a claim for indemnification against the appropriate party pursuant to such agreement to the extent such a claim would not be frivolous, and Controlled shall control, directly or indirectly, the prosecution of such claim under the principles contained in Section 7.9. Distributing and Liberty LLC shall not amend, modify or terminate the AT&T Tax Sharing Agreement, the LMI Tax Sharing Agreement or the DHC Tax Sharing Agreement, or waive any rights thereunder, without the prior written consent of Controlled, which consent shall not be unreasonably withheld.

37

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by the respective officers as of the date set forth above.

# LIBERTY INTERACTIVE CORPORATION

By: /s/ Albert E. Rosenthaler

Name: Albert E. Rosenthaler Title: Senior Vice President

#### LIBERTY INTERACTIVE LLC

By: /s/ Albert E. Rosenthaler

Name: Albert E. Rosenthaler Title: Senior Vice President

#### LIBERTY MEDIA CORPORATION

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe

Title: Executive Vice President and General Counsel

#### SERVICES AGREEMENT

SERVICES AGREEMENT (this "Agreement"), dated as of September 23, 2011, by and between Liberty Media Corporation (f/k/a Liberty CapStarz, Inc.), a Delaware corporation (the "Provider"), and Liberty Interactive Corporation (f/k/a Liberty Media Corporation), a Delaware corporation ('Liberty'').

#### RECITALS

WHEREAS, on the date hereof the Provider is a wholly owned subsidiary of Liberty and holds the assets and liabilities attributed to Liberty's Liberty Capital group and Liberty Starz group following the consummation of the transactions described in the plan of restructuring set forth in Schedule 1.1 to the Reorganization Agreement, dated as of August 30, 2011 (the "Reorganization Agreement"), to which the Provider and Liberty are each parties;

WHEREAS, in accordance with the Reorganization Agreement and the Restated Certificate of Incorporation of Liberty, Liberty will effect the redemptions of (i) 100% of the issued and outstanding shares of its Liberty Capital common stock for shares of a new Capital Group tracking stock of the Provider, and (ii) 100% of the issued and outstanding shares of its Liberty Starz common stock for shares of a new Starz Group tracking stock of the Provider, with the effect that Provider will be split-off (the "Split-Off") from Liberty and cease to be a wholly owned subsidiary of Liberty;

WHEREAS, immediately following the Split-Off, Liberty's assets and liabilities will consist solely of those assets and liabilities attributed to Liberty's Interactive group and Provider's assets and liabilities will consist solely of those assets and liabilities that, immediately prior to the time of the Split-Off, were attributed to Liberty's Liberty Capital group and Liberty Starz group;

WHEREAS, immediately following the Split-Off, the Provider and Liberty will be separate and independent publicly-traded companies;

WHEREAS, Liberty and the Provider desire that, following the Split-Off, Liberty obtain from the Provider the services described herein, and that Liberty compensate the Provider for the performance of such services on the basis set forth in this Agreement; and

WHEREAS, on the date hereof a subsidiary of the Provider is also entering into a facilities sharing agreement with Liberty with respect to 12300 Liberty Boulevard, Englewood, Colorado (the "Facilities Sharing Agreement").

#### **AGREEMENT**

NOW THEREFORE, in consideration of the foregoing recitals, the mutual agreements contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be bound legally, agree as follows:

#### ARTICLE I

### ENGAGEMENT AND SERVICES

Section 1.1 <u>Engagement</u>. Liberty engages the Provider to provide to Liberty, commencing on the date of the Split-Off (the 'Split-Off Effective Date''), the services set forth in Section 1.2 (collectively, the "Services"), and the Provider accepts such engagement, subject to and upon the terms and conditions of this Agreement. The parties acknowledge that certain of the Services will be performed by officers, employees or consultants of the Provider, who also serve as officers, employees or consultants of Liberty.

# Section 1.2 <u>Services</u>.

- (a) The Services will include the following, if and to the extent required by Liberty during the Term of this Agreement:
  - (i) insurance administration and risk management services;
- (ii) technical and information technology assistance (including management information systems, computer, data storage network and telecommunications services), computers, office supplies, postage, courier service and other office services;
- (iii) services performed by the Provider's finance, accounting, payroll, treasury, cash management, legal, disclosure compliance, human resources, employee benefits, investor relations, tax and real estate management departments; and
- (iv) such other services as the Provider may obtain from its officers, employees and consultants in the management of its own operations that Liberty may from time to time require.
- (b) The Services are intended to be those services and functions that are appropriate for the operation and management of Liberty as a publicly-traded company, and are not intended to be duplicative of services and functions for the operating subsidiaries of Liberty that are to be performed by officers, employees and consultants of those companies.
- Section 1.3 <u>Services Not to Interfere with the Provider's Business</u> Liberty acknowledges and agrees that in providing Services hereunder the Provider will not be required to take any action that would disrupt, in any material respect, the orderly operation of the Provider's business activities.
- Section 1.4 <u>Books and Records.</u> The Provider will maintain books and records, in reasonable detail in accordance with the Provider's standard business practices, with respect to its provision of Services to Liberty pursuant to this Agreement, including records supporting the allocation of Employee Compensation (as defined below) and other costs and expenses to Liberty pursuant to Article II (collectively, "Supporting Records"). The Provider will give Liberty and its duly authorized representatives, agents, and attorneys reasonable access to all

#### COMPENSATION

#### Section 2.1 <u>Allocated Expenses</u>.

- (a) For each officer, employee or consultant of Provider that provides Services to Liberty (each, an "Employee"), Liberty shall be allocated an amount (the "Employee Compensation") equal to his or her aggregate salary, bonus and health, retirement and other compensation and benefits paid by Provider (other than, in all cases, equity-based compensation paid by Provider) multiplied by his or her Liberty Percentage (as defined below); provided, however, that the Employee Compensation allocable to any Employee who holds the office of Vice President or higher of Liberty (each, a "Liberty Officer") will not take into account any bonus amounts paid to such Employee by Provider. The "Liberty Percentage" with respect to any Employee will equal the estimate of the relative amount of time that such Employee will spend providing Services to Liberty over a defined period of time. Liberty will pay the Provider the aggregate Employee Compensation so allocated to it under this Section 2.1(a), together with such other costs and expenses as may be incurred by the Provider in connection with the provision of Services to Liberty hereunder (collectively with the aggregate Employee Compensation but exclusive of any Out-of-Pocket Costs (as defined below), the "Allocated Expenses"). The Allocated Expenses will be more fully set forth in, or determined from time to time in the manner set forth in, Schedule 2.1 attached hereto, as such Schedule may be periodically amended and revised by the parties. It is intended that the payments by Liberty to Provider under this Agreement in respect of Allocated Expenses are equivalent to those which Liberty would pay to a third party on an arm's length basis for the same services. If the cost of any Service to be provided to Liberty under this Agreement is included in the Annual Allocation Expense (as such term is defined in the Facilities Sharing Agreement) payable by Liberty under this Agreement.
- (b) The Liberty Percentage applicable to each Employee and the Allocated Expenses will be determined by the Provider, in consultation with Liberty, on or about each December 15th during the Term based on the anticipated Services to be provided by Employees to Liberty during the upcoming fiscal year, among other things deemed relevant by the parties. The Provider and Liberty will review and evaluate the Liberty Percentages and Allocated Expenses for reasonableness semiannually during the Term, and will negotiate in good faith to reach agreement on any appropriate adjustments to the Liberty Percentages and Allocated Expenses based on such review and evaluation, including: (i) adjustments that reflect changes to the Employee Compensation of each Employee; (ii) adjustments that reflect changes to any other costs or expenses included in the Allocated Expenses; (iii) adjustments that reflect changes in the allocable percentages of time spent by particular Employees providing Services to Liberty; and (iv) agreeing on the appropriate effective date (which may be retroactive) of any such adjustment to the Liberty Percentages and Allocated Expenses.

3

Section 2.2 <u>Cost Reimbursement.</u> In addition to (and without duplication of) the Allocated Expenses payable pursuant to Section 2.1, Liberty also will reimburse the Provider for all direct out-of-pocket costs, with no markup ("<u>Out-of-Pocket Costs</u>"), incurred by the Provider in performing the Services (e.g., postage and courier charges, software license fees attributable to desktop or laptop computers utilized by Employees, travel, meals and entertainment expenses, and other miscellaneous expenses that are incurred by the Provider or the Employees in the conduct of the Services).

#### Section 2.3 Payment Procedures.

- (a) Liberty will pay the Provider, by wire or intrabank transfer of funds or in such other manner specified by the Provider to Liberty, in arrears on or before the last day of each calendar month beginning with October 2011, the Allocated Expenses then in effect, in monthly installments.
- (b) Any reimbursement to be made by Liberty to the Provider pursuant to Section 2.2 will be paid by Liberty to the Provider within 15 days after receipt by Liberty of an invoice therefor, by wire or intrabank transfer of funds or in such other manner specified by the Provider to Liberty. The Provider will invoice Liberty monthly for reimbursable expenses incurred by the Provider on behalf of Liberty during the preceding calendar month as contemplated in Section 2.2; provided, however, that the Provider may separately invoice Liberty at any time for any single reimbursable expense incurred by the Provider on behalf of Liberty in an amount equal to or greater than \$25,000. Any invoice or statement pursuant to this Section 2.3(b) will be accompanied by supporting documentation in reasonable detail consistent with Provider's own expense reimbursement policy with respect to the costs and expenses incurred by the Provider for which the Provider is seeking reimbursement hereunder.
- (c) Any payments not made when due under this Section 2.3 will bear interest at the rate of 1.5% per month on the outstanding amount from and including the due date to but excluding the date paid.
  - Section 2.4 Survival. The terms and conditions of this Article II will survive the expiration or earlier termination of this Agreement.

# ARTICLE III

#### TERM

- Section 3.1 <u>Term Generally.</u> The term of this Agreement will commence on the Split-Off Effective Date and will continue until the third anniversary of the Split-Off Effective Date (the "<u>Term</u>"). This Agreement is subject to termination prior to the end of the Term in accordance with Section 3.3.
- Section 3.2 <u>Discontinuance of Select Services</u>. At any time during the Term, on not less than 30 days' prior notice by Liberty to the Provider, Liberty may elect to discontinue obtaining any of the Services from the Provider. In such event, the Provider's obligation to provide Services that have been discontinued pursuant to this Section 3.2, and Liberty's obligation to compensate the Provider for such Services, will cease as of the end of such 30-day

4

period (or such later date as may be specified in the notice), and this Agreement will remain in effect for the remainder of the Term with respect to those Services that have not been so discontinued. The Provider and Liberty will promptly evaluate the Allocated Expenses for reasonableness following the discontinuance of any Services and will negotiate in good faith to reach agreement on any appropriate adjustment to the Allocated Expenses. Each party will remain liable to the other for any required payment or performance accrued prior to the effective date of discontinuance of any Service.

- Section 3.3 Termination. This Agreement will be terminated prior to the expiration of the Term in the following events:
  - (a) at any time upon at least 30 days' prior written notice by Liberty to the Provider;
- (b) immediately upon written notice (or at any time specified in such notice) by the Provider to Liberty if a Change in Control or Bankruptcy Event occurs with respect to Liberty; or
- (c) immediately upon notice (or at any time specified in such notice) by Liberty to the Provider if a Change in Control or Bankruptcy Event occurs with respect to the Provider.

For purposes of this Section 3.3, a "Change in Control" will be deemed to have occurred with respect to a party if a merger, consolidation, binding share exchange, acquisition, or similar transaction (each, a "Transaction"), or series of related Transactions, involving such party occurs as a result of which the voting power of all voting

securities of such party outstanding immediately prior thereto represent (either by remaining outstanding or being converted into voting securities of the surviving entity) less than 75% of the voting power of such party or the surviving entity of the Transaction outstanding immediately after such Transaction (or if such party or the surviving entity after giving effect to such Transaction is a subsidiary of the issuer of securities in such Transaction, then the voting power of all voting securities of such party outstanding immediately prior to such Transaction represent (by being converted into voting securities of such issuer) less than 75% of the voting power of the issuer outstanding immediately after such Transaction).

For purposes of this Section 3.3, a "Bankruptcy Event" will be deemed to have occurred with respect to a party upon such party's insolvency, general assignment for the benefit of creditors, such party's voluntary commencement of any case, proceeding, or other action seeking reorganization, arrangement, adjustment, liquidation, dissolution, or consolidation of such party's debts under any law relating to bankruptcy, insolvency, or reorganization, or relief of debtors, or seeking appointment of a receiver, trustee, custodian, or other similar official for such party or for all or any substantial part of such party's assets (each, a "Bankruptcy Proceeding"), or the involuntary filing against Liberty or the Provider, as applicable, of any Bankruptcy Proceeding that is not stayed within 60 days after such filing.

Each party will remain liable to the other for any required payment accrued prior to the termination of this Agreement.

5

#### ARTICLE IV

## PERSONNEL AND EMPLOYEES

## Section 4.1 <u>Personnel to Provide Services.</u>

- (a) The Provider will make available to Liberty, on a non-exclusive basis, the appropriate personnel to perform the Services. The personnel made available to perform selected Services are expected to be substantially the same personnel who provide similar services in connection with the management and administration of the business and operations of the Provider.
  - (b) Liberty acknowledges that:
  - (i) certain of the Employees also will be performing services for the Provider and may be performing services for certain Subsidiaries and Affiliates of the Provider, in each case, while also potentially performing services directly for Liberty and certain of its Subsidiaries and Affiliates under a direct employment, consultancy or other service relationship between such Employee and Liberty and irrespective of this Agreement; and
  - (ii) the Provider may elect, in its discretion, to utilize independent contractors rather than employees of the Provider to perform Services from time to time, and such independent contractors will be deemed included within the definition of "Employees" for all purposes of this Agreement.
- Section 4.2 Provider as Payor. The parties acknowledge and agree that the Provider, and not Liberty, will be solely responsible for the payment of salaries, wages, benefits (including health insurance, retirement, and other similar benefits, if any) and other compensation applicable to all Employees; provided, however, that (a) Liberty is responsible for the payment of the Allocated Expenses in accordance with Section 2.1 and for the portion of any severance payments allocated to Liberty under Section 4.3, and (b) Liberty is responsible for the payment of (i) all compensation based on, comprised of or related to the equity securities of Liberty and (ii) any bonus amounts payable to any Liberty Officer with respect to services performed for the benefit of Liberty (together with (i), "Excluded Compensation"). The parties acknowledge that Employees may provide services directly to Liberty in consideration for the receipt of Excluded Compensation pursuant to such Employees' separate employment, consultancy or other service relationship with Liberty. All Employees will be subject to the personnel policies of the Provider and will be eligible to participate in the Provider's employee benefit plans to the same extent as similarly situated employees of the Provider performing services in connection with the Provider's business. Except as otherwise required by the terms of the Tax Sharing Agreement, the Provider will be responsible for the payment of all federal, state, and local withholding taxes on the compensation of all Employees (other than Excluded Compensation) and other such employment related taxes as are required by law, and Liberty will be responsible for the payment of all federal, state, and local withholding taxes on Excluded Compensation paid to any Employee by Liberty and other such employment related taxes as are required by law. Each of Liberty and Provider will cooperate with the other to facilitate the other's compliance with

6

applicable federal, state, and local laws, rules, regulations, and ordinances applicable to the employment of all Employees by either party.

Section 4.3 <u>Additional Employee Provisions</u>. The Provider will have the right to terminate its employment of any Employee at any time. A portion of any severance payments payable to any Employee who separates from employment with the Provider during the Term will be allocated to Liberty based on the monthly average Liberty Percentage applicable to such Employee since the Split-Off Effective Date; *provided*, *however*, that such severance payments will not take into account (i) any bonus-related compensation arrangements payable to any Liberty Officer and (ii) any equity-based compensation arrangements paid or provided by Provider.

#### ARTICLE V

## REPRESENTATIONS AND WARRANTIES

- Section 5.1 Representations and Warranties of the Provider. The Provider represents and warrants to Liberty as follows:
  - (a) The Provider is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware.
  - (b) The Provider has the power and authority to enter into this Agreement and to perform its obligations under this Agreement, including the Services.
- (c) The Provider is not subject to any contractual or other legal obligation that materially interferes with its full, prompt, and complete performance under this Agreement.
  - (d) The individual executing this Agreement on behalf of the Provider has the authority to do so.
  - Section 5.2 Representations and Warranties of Liberty. Liberty represents and warrants to the Provider as follows:
    - (a) Liberty is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware.
    - (b) Liberty has the power and authority to enter into this Agreement and to perform its obligations under this Agreement.
    - (c) Liberty is not subject to any contractual or other legal obligation that materially interferes with its full, prompt, and complete performance under

this Agreement.

#### ARTICLE VI

#### INDEMNIFICATION

Section 6.1 <u>Indemnification by the Provider.</u> The Provider will indemnify, defend, and hold harmless Liberty and each of its Subsidiaries, Affiliates, officers, directors, employees and agents, successors and assigns (collectively, the "<u>Liberty Indemnitees</u>"), from and against any and all Actions, judgments, Liabilities, losses, costs, damages, or expenses, including reasonable counsel fees, disbursements, and court costs (collectively, "<u>Losses</u>"), that any Liberty Indemnitee may suffer arising from or out of, or relating to, (a) any material breach by the Provider of its obligations under this Agreement, or (b) the gross negligence, willful misconduct, fraud, or bad faith of the Provider

Section 6.2 <u>Indemnification by Liberty.</u> Liberty will indemnify, defend, and hold harmless the Provider and its Subsidiaries, Affiliates, officers, directors, employees and agents, successors and assigns (collectively, the "<u>Provider Indemnitees</u>"), from and against any and all Losses that any Provider Indemnitee may suffer arising from or out of, or relating to (a) any material breach by Liberty of its obligations under this Agreement, or (b) any acts or omissions of the Provider in providing the Services pursuant to this Agreement (except to the extent such Losses (i) arise from or relate to any material breach by the Provider of its obligations under this Agreement, (ii) are attributable to the gross negligence, willful misconduct, fraud, or bad faith of the Provider or any other Provider Indemnitee seeking indemnification under this Section 6.2, (iii) are fully covered by insurance maintained by the Provider or such other Provider Indemnitee, or (iv) are payable by the Provider pursuant to Section 7.11).

#### Section 6.3 <u>Indemnification Procedures</u>.

(a) (i) In connection with any indemnification provided for in Section 6.1 or 6.2, the party seeking indemnification (the 'Indemnitee') will give the party from which indemnification is sought (the "Indemniter") prompt notice whenever it comes to the attention of the Indemnitee that the Indemnitee has suffered or incurred, or may suffer or incur, any Losses for which it is entitled to indemnification under Section 6.1 or 6.2, and, if and when known, the facts constituting the basis for such claim and the projected amount of such Losses (which shall not be conclusive as to the amount of such Losses), in each case in reasonable detail. Without limiting the generality of the foregoing, in the case of any Action commenced by a third party for which indemnification is being sought (a "Third-Party Claim"), such notice will be given no later than ten business days following receipt by the Indemnitee of written notice of such Third-Party Claim. Failure by any Indemnitee to so notify the Indemnitor will not affect the rights of such Indemnitee hereunder except to the extent that such failure has a material prejudicial effect on the defenses or other rights available to the Indemnitor with respect to such Third Party Claim. The Indemnitee will deliver to the Indemnitor as promptly as practicable, and in any event within five business days after Indemnitee's receipt, copies of all notices, court papers and other documents received by the Indemnitee relating to any Third-Party Claim.

8

After receipt of a notice pursuant to Section 6.3(a)(i) with respect to any Third-Party Claim, the Indemnitor will be entitled, if it so elects, to take control of the defense and investigation with respect to such Third-Party Claim and to employ and engage attorneys reasonably satisfactory to the Indemnitee to handle and defend such claim, at the Indemnitor's cost, risk and expense, upon written notice to the Indemnitee of such election, which notice acknowledges the Indemnitor's obligation to provide indemnification under this Agreement with respect to any Losses arising out of or relating to such Third-Party Claim. The Indemnitor will not settle any Third-Party Claim that is the subject of indemnification without the written consent of the Indemnitee, which consent will not be unreasonably withheld, conditioned or delayed; provided, however, that, after reasonable notice, the Indemnitor may settle a claim without the Indemnitee's consent if such settlement (A) makes no admission or acknowledgment of Liability or culpability with respect to the Indemnitee, (B) includes a complete release of the Indemnitee and (C) does not seek any relief against the Indemnitee other than the payment of money damages to be borne by the Indemnitor. The Indemnitee will cooperate in all reasonable respects with the Indemnitor and its attorneys in the investigation, trial and defense of any lawsuit or action with respect to such claim and any appeal arising therefrom (including the filing in the Indemnitee's name of appropriate cross-claims and counterclaims). The Indemnitee may, at its own cost, participate in any investigation, trial and defense of any Third-Party Claim controlled by the Indemnitor and any appeal arising therefrom, including participating in the process with respect to the potential settlement or compromise thereof. If the Indemnitee has been advised by its counsel that there may be one or more legal defenses available to the Indemnitee that conflict with those available to, or that are not available to, the Indemnitor ("Separate Legal Defenses"), or that there may be actual or potential differing or conflicting interests between the Indemnitor and the Indemnitee in the conduct of the defense of such Third-Party Claim, the Indemnitee will have the right, at the expense of the Indemnitor, to engage separate counsel reasonably acceptable to the Indemnitor to handle and defend such Third-Party Claim, provided, that, if such Third-Party Claim can be reasonably separated between those portion(s) for which Separate Legal Defenses are available ("Separable Claims") and those for which no Separate Legal Defenses are available, the Indemnitee will instead have the right, at the expense of the Indemnitor, to engage separate counsel reasonably acceptable to the Indemnitor to handle and defend the Separable Claims, and the Indemnitor will not have the right to control the defense or investigation of such Third-Party Claim or such Separable Claims, as the case may be (and, in which latter case, the Indemnitor will have the right to control the defense or investigation of the remaining portion(s) of such Third-Party Claim).

(iii) If, after receipt of a notice pursuant to Section 6.3(a)(i) with respect to any Third-Party Claim as to which indemnification is available hereunder, the Indemnitor does not undertake to defend the Indemnitee against such Third-Party Claim, whether by not giving the Indemnitee timely notice of its election to so defend or otherwise, the Indemnitee may, but will have no obligation to, assume its own defense, at the expense of the Indemnitor (including attorneys fees and costs), it being understood that the Indemnitee's right to indemnification for such Third Party Claim shall not be adversely affected by its assuming the defense of such Third Party Claim. The Indemnitor will be bound by the result obtained with respect thereto by the Indemnitee; provided, that the Indemnitee may not settle any lawsuit or action with respect to which the Indemnitee is entitled to indemnification hereunder without the consent of the Indemnitor, which consent will not be unreasonably withheld, conditioned or delayed; provided further, that such consent shall not be required if (i) the Indemnitor had the right under this

Ç

Section 6.3 to undertake control of the defense of such Third-Party Claim and, after notice, failed to do so within thirty days of receipt of such notice (or such lesser period as may be required by court proceedings in the event of a litigated matter), or (ii) (x) the Indemnitor does not have the right to control the defense of the entirety of such Third-Party Claim pursuant to Section 6.3(a)(ii) or (y) the Indemnitor does not have the right to control the defense of any Separable Claim pursuant to Section 6.3(a)(ii) (in which case such settlement may only apply to such Separable Claims), the Indemnitee provides reasonable notice to Indemnitor of the settlement, and such settlement (A) makes no admission or acknowledgment of Liability or culpability with respect to the Indemnitor, (B) does not seek any relief against the Indemnitor and (C) does not seek any relief against the Indemnitor is responsible other than the payment of money damages.

(b) In no event will the Indemnitor be liable to any Indemnitee for any special, consequential, indirect, collateral, incidental or punitive damages, however caused and on any theory of liability arising in any way out of this Agreement, whether or not such Indemnitor was advised of the possibility of any such damages; *provided*, that the foregoing limitations shall not limit a party's indemnification obligations for any Losses incurred by an Indemnitee as a result of the assertion of a Third Party Claim.

- (c) The Indemnitor and the Indemnitee shall use commercially reasonable efforts to avoid production of confidential information, and to cause all communications among employees, counsel and others representing any party with respect to a Third Party Claim to be made so as to preserve any applicable attorney-client or work-product privilege.
- (d) The Indemnitor shall pay all amounts payable pursuant to this Section 6.3 by wire transfer of immediately available funds, promptly following receipt from an Indemnitee of a bill, together with all accompanying reasonably detailed backup documentation, for any Losses that are the subject of indemnification hereunder, unless the Indemnitor in good faith disputes the amount of such Losses or whether such Losses are covered by the Indemnitor's indemnification obligation in which event the Indemnitor shall promptly so notify the Indemnitee. In any event, the Indemnitor shall pay to the Indemnitee, by wire transfer of immediately available funds, the amount of any Losses for which it is liable hereunder no later than three (3) days following any final determination of the amount of such Losses and the Indemnitor's liability therefor. A "final determination" shall exist when (a) the parties to the dispute have reached an agreement in writing or (b) a court of competent jurisdiction shall have entered a final and non-appealable order or judgment.
- (e) If the indemnification provided for in this Section 6.3 shall, for any reason, be unavailable or insufficient to hold harmless an Indemnitee in respect of any Losses for which it is entitled to indemnification hereunder, then the Indemnitor shall contribute to the amount paid or payable by such Indemnitee as a result of such Losses, in such proportion as shall be appropriate to reflect the relative benefits received by and the relative fault of the Indemnitor on the one hand and the Indemnitee on the other hand with respect to the matter giving rise to such Losses.
- (f) The remedies provided in this Section 6.3 shall be cumulative and shall not preclude assertion by any Indemnitee of any other rights or the seeking of any and all other remedies against an Indemnitor, subject to Section 6.3(b).

10

- (g) For the avoidance of doubt, the provisions of this Article VI are not intended to, and shall not, apply to any Loss, claim or Liability to which the provisions of the Tax Sharing Agreement are applicable.
- (h) To the fullest extent permitted by applicable law, the Indemnitor will indemnify the Indemnitee against any and all reasonable fees, costs and expenses (including attorneys' fees), incurred in connection with the enforcement of his, her or its rights under this Article VI.
  - Section 6.4 Survival. The terms and conditions of this Article VI will survive the expiration or termination of this Agreement.

## ARTICLE VII

#### MISCELLANEOUS

## Section 7.1 <u>Defined Terms</u>.

(a) The following terms will have the following meanings for all purposes of this Agreement:

"Action" means any demand, action, claim, suit, countersuit, litigation, arbitration, prosecution, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any court, grand jury or other governmental authority or any arbitrator or arbitration panel.

"Affiliate" means, with respect to any Person, any other Person controlled by such first Person, with "control" for such purpose meaning the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities or voting interests, by contract, or otherwise. Notwithstanding the foregoing, for purposes of this Agreement, (i) none of Expedia, Inc., any of its present or former Affiliates, TripAdvisor, Inc. or any of its present or former Affiliates will be deemed to be Affiliates of the Provider, Liberty or any of their respective Subsidiaries for any purpose during any period in which the voting stock of such corporation owned by any of the Provider, por any of their respective Subsidiaries is subject to a proxy in favor of Mr. Barry Diller to vote those shares during such period, and (ii) none of the Provider or any of its Subsidiaries will be deemed to be Affiliates of Liberty or any of its Subsidiaries, nor will Liberty or any of its Subsidiaries be deemed to be Subsidiaries of Provider or any of its Subsidiaries.

"Confidential Information" means any information marked, noticed, or treated as confidential by a party which such party holds in confidence, including all trade secrets, technical, business, or other information, including customer or client information, however communicated or disclosed, relating to past, present and future research, development and business activities.

"Liabilities" means any and all debts, liabilities, commitments and obligations, whether or not fixed, contingent or absolute, matured or unmatured, direct or indirect, liquidated or

11

unliquidated, accrued or unaccrued, known or unknown, and whether or not required by GAAP to be reflected in financial statements or disclosed in the notes thereto (other than taxes).

"Person" means any natural person, corporation, limited liability corporation, partnership, trust, unincorporated organization, association, governmental authority, or other entity.

"Subsidiary" when used with respect to any Person, means (i)(A) a corporation a majority in voting power of whose share capital or capital stock with voting power, under ordinary circumstances, to elect directors is at the time, directly or indirectly, owned by such Person, by one or more Subsidiaries of such Person, or by such Person and one or more Subsidiaries of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, (B) a partnership or limited liability company in which such Person or a Subsidiary of such Person is, at the date of determination, (1) in the case of a partnership, a general partner of such partnership with the power affirmatively to direct the policies and management of such partnership or (2) in the case of a limited liability company, the managing member or, in the absence of a managing member, a member with the power affirmatively to direct the policies and management of such limited liability company, or (C) any other Person (other than a corporation) in which such Person, one or more Subsidiaries of such Person and one or more Subsidiaries of such Person, directly or indirectly, at the date of determination thereof, has or have (1) the power to elect or direct the election of a majority of the members of the governing body of such Person, whether or not such power is subject to a voting agreement or similar encumbrance, or (2) in the absence of such a governing body, at least a majority ownership interest or (ii) any other Person of which an aggregate of 50% or more of the equity interests are, at the time, directly or indirectly, owned by such Person and/or one or more Subsidiaries of such Person.

Notwithstanding the foregoing, for purposes of this Agreement, (A) none of Expedia, Inc., TripAdvisor, Inc., or any of their present or former Subsidiaries will be deemed to be Subsidiaries is subject to a proxy in favor of Mr. Barry Diller to vote those shares during such period, and (B) none of the Subsidiaries of the Provider will be deemed to be S

"Tax Sharing Agreement" means the Tax Sharing Agreement among Liberty, Liberty Interactive LLC (f/k/a Liberty Media LLC) and Provider.

(b) The following terms will have the meanings for all purposes of this Agreement set forth in the Section reference provided next to such term:

Section Reference		
Preamble		
Section 2.1(a)		
Section 3.3		
Section 3.3		
Section 3.3		
Section 6.1		
Section 2.1(a)		

12

Definition	Section Reference
Employee Compensation	Section 2.1(a)
Excluded Compensation	Section 4.2
Facilities Sharing Agreement	Recitals
Indemnitee	Section 6.3(a)(i)
Indemnitor	Section 6.3(a)(i)
Liberty	Preamble
Liberty Indemnitees	Section 6.1
Liberty Officer	Section 2.1(a)
Liberty Percentage	Section 2.1(a)
Losses	Section 6.1
Out-of-Pocket Costs	Section 2.2
Provider	Preamble
Provider Indemnitees	Section 6.2
Reorganization Agreement	Recitals
Separable Claim	Section 6.3(a)(ii)
Separate Legal Defenses	Section 6.3(a)(ii)
Services	Section 1.1
Split-Off	Recitals
Split-Off Effective Date	Section 1.1
Supporting Records	Section 1.4
Term	Section 3.1
Third- Party Claim	Section 6.3(a)(i)
Transaction	Section 3.3

Section 7.2 Entire Agreement; Severability. This Agreement (including the Schedules attached hereto), the Tax Sharing Agreement, the Facilities Sharing Agreement and the Reorganization Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and thereof, and supersedes all prior agreements and understandings, oral and written, among the parties hereto with respect to such subject matter. It is the intention of the parties hereto that the provisions of this Agreement will be enforced to the fullest extent permissible under all applicable laws and public policies, but that the unenforceability of any provision hereof (or the modification of any provision hereof to conform with such laws or public policies, as provided in the next sentence) will not render unenforceable or impair the remainder of this Agreement. Accordingly, if any provision is determined to be invalid or unenforceable either in whole or in part, this Agreement will be deemed amended to delete or modify, as necessary, the invalid or unenforceable provisions and to alter the balance of this Agreement in order to render the same valid and enforceable, consistent (to the fullest extent possible) with the intent and purposes hereof. If the provisions of this Agreement conflict with any provisions of the Facilities Sharing Agreement, the provisions of the Tax Sharing Agreement, the

13

Section 7.3 Notices. All notices and communications hereunder will be in writing and will be deemed to have been duly given if delivered personally or mailed, certified or registered mail with postage prepaid, or sent by confirmed facsimile, addressed as follows:

If to the Provider: Liberty Media Corporation

12300 Liberty Boulevard Englewood, Colorado 80112 Attention: General Counsel Facsimile: (720) 875-5401

If to Liberty: Liberty Interactive Corporation

12300 Liberty Boulevard Englewood, Colorado 80112 Attention: General Counsel Facsimile: (720) 875-5401

or to such other address (or to the attention of such other person) as the parties may hereafter designate in writing. All such notices and communications will be deemed to have been given on the date of delivery if sent by facsimile or personal delivery, or the third day after the mailing thereof, except that any notice of a change of address will be deemed to have been given only when actually received.

Section 7.4 Governing Law. This Agreement and the legal relations among the parties hereto will be governed in all respects, including validity, interpretation and effect, by the laws of the State of Colorado applicable to contracts made and performed wholly therein, without giving effect to any choice or conflict of laws provisions or rules that would cause the application of the laws of any other jurisdiction.

Section 7.5 Rules of Construction. The descriptive headings in this Agreement are inserted for convenience of reference only and are not intended to be part of

or to affect the meaning or interpretation of this Agreement. Words used in this Agreement, regardless of the gender and number specifically used, will be deemed and construed to include any other gender, masculine, feminine, or neuter, and any other number, singular or plural, as the context requires. As used in this Agreement, the word "including" or any variation thereof is not limiting, and the word "or" is not exclusive. The word day means a calendar day. If the last day for giving any notice or taking any other action is a Saturday, Sunday, or a day on which banks in New York, New York or Denver, Colorado are closed, the time for giving such notice or taking such action will be extended to the next day that is not such a day.

Section 7.6 No Third-Party Rights. Nothing expressed or referred to in this Agreement is intended or will be construed to give any Person other than the parties hereto and their respective successors and permitted assigns any legal or equitable right, remedy or claim under or with respect to this Agreement, or any provision hereof, it being the intention of the parties hereto that this Agreement and all of its provisions and conditions are for the sole and exclusive benefit of the parties to this Agreement and their respective successors and assigns.

14

- Section 7.7 <u>Counterparts.</u> This Agreement may be executed in one or more counterparts, each of which will be an original and all of which together will constitute one and the same instrument.
- Section 7.8 <u>Payment of Expenses.</u> Liberty will be responsible for all costs and expenses incurred with respect to the preparation of this Agreement. From and after the Split-Off Effective Date, and except as otherwise expressly provided in this Agreement, each of the parties to this Agreement will bear its own expenses, including the fees of any attorneys and accountants engaged by such party, in connection with this Agreement.
  - Section 7.9 <u>Binding Effect; Assignment.</u>
- (a) This Agreement will inure to the benefit of and be binding on the parties to this Agreement and their respective legal representatives, successors and permitted assigns.
- (b) Except as expressly contemplated hereby (including by Section 4.1), this Agreement, and the obligations arising hereunder, may not be assigned by either party to this Agreement, *provided*, *however*, that Liberty and Provider may assign their respective rights, interests, duties, liabilities and obligations under this Agreement to any of their respective wholly-owned Subsidiaries, but such assignment shall not relieve Liberty or the Provider, as the assignor, of its obligations hereunder.
- Section 7.10 Amendment, Modification, Extension or Waiver. Any amendment, modification or supplement of or to any term or condition of this Agreement will be effective only if in writing and signed by both parties hereto. Either party to this Agreement may (a) extend the time for the performance of any of the obligations or other acts of the other party to this Agreement, or (b) waive compliance by the other party with any of the agreements or conditions contained herein or any breach thereof. Any agreement on the part of either party to any such extension or waiver will be valid only if set forth in an instrument in writing signed on behalf of such party. No waiver of any term, provision or condition of this Agreement, whether by conduct or otherwise, in any one or more instance, will be deemed or construed as a further or continuing waiver of any such term, provision or condition or of any other term, provision or condition, but any party hereto may waive its rights in any particular instance by written instrument of waiver.
- Section 7.11 <u>Legal Fees; Costs.</u> If either party to this Agreement institutes any action or proceeding to enforce any provision of this Agreement, the prevailing party will be entitled to receive from the other party reasonable attorneys' fees, disbursements and costs incurred in such action or proceeding, whether or not such action or proceeding is prosecuted to judgment.
- Section 7.12 Force Majeure. Neither party will be liable to the other party with respect to any nonperformance or delay in performance of its obligations under this Agreement to the extent such failure or delay is due to any action or claims by any third party, labor dispute, labor strike, weather conditions or any cause beyond a party's reasonable control. Each party agrees that it will use all commercially reasonable efforts to continue to perform its obligations under this Agreement, to resume performance of its obligations under this Agreement, and to minimize

15

any delay in performance of its obligations under this Agreement notwithstanding the occurrence of any such event beyond such party's reasonable control.

- Section 7.13 Specific Performance. Each party agrees that irreparable damage would occur and that the parties would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that each of the parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, this being in addition to any other remedy to which they are entitled at law or in equity.
- Section 7.14 <u>Further Actions</u>. The parties will execute and deliver all documents, provide all information, and take or forbear from all actions that may be necessary or appropriate to achieve the purposes of this Agreement.
  - Section 7.15 Confidentiality.
    - (a) Except with the prior consent of the disclosing party, each party will:
  - (i) limit access to the Confidential Information of the other party disclosed to such party hereunder to its employees, agents, representatives, and consultants on a need-to-know basis;
  - (ii) advise its employees, agents, representatives, and consultants having access to such Confidential Information of the proprietary nature thereof and of the obligations set forth in this Agreement; and
  - (iii) safeguard such Confidential Information by using a reasonable degree of care to prevent disclosure of the Confidential Information to third parties, but not less than that degree of care used by that party in safeguarding its own similar information or material.
- (b) A party's obligations respecting confidentiality under Section 7.15(a) will not apply to any of the Confidential Information of the other party that a party can demonstrate: (i) was, at the time of disclosure to it, in the public domain; (ii) after disclosure to it, is published or otherwise becomes part of the public domain through no fault of the receiving party; (iii) was in the possession of the receiving party at the time of disclosure to it without being subject to any obligation of confidentiality; (iv) was received after disclosure to it from a third party who, to its knowledge, had a lawful right to disclose such information to it; (v) was independently developed by the receiving party without reference to the Confidential Information; (vi) was required to be disclosed to any regulatory body having jurisdiction over a party or any of their respective clients; or (vii) was required to be disclosed by reason of legal, accounting, or regulatory requirements beyond the reasonable control of the receiving party. In the case of any disclosure pursuant to clauses (vi) or (vii) of this paragraph (b), to the extent practical, the receiving party will give prior notice to the disclosing party of the required disclosure and will use commercially reasonable efforts to obtain a protective order covering such disclosure.

(c) The provisions of this Section 7.15 will survive the expiration or termination of this Agreement.

[Signature Page Follows]

17

IN WITNESS WHEREOF, each of the parties has signed this Agreement, or has caused this Agreement to be signed by its duly authorized officer, as of the date first above written.

# PROVIDER:

LIBERTY MEDIA CORPORATION (f/k/a LIBERTY CAPSTARZ, INC.)

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe

Title: Executive Vice President & General Counsel

LIBERTY:

LIBERTY INTERACTIVE CORPORATION (f/k/a LIBERTY MEDIA CORPORATION)

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe

Title: Executive Vice President & General Counsel

[Signature Page to Services Agreement]

#### Schedule 2.1

Allocated Expenses in Effect for Calendar Year 2011

# **List of Omitted Schedules**

The following schedule to the Services Agreement, dated as of September 23, 2011, by and between Liberty Media Corporation and Liberty Interactive Corporation, has not been provided herein:

Schedule 2.1 - Allocated Expenses in Effect for Calendar Year 2011

The undersigned registrant hereby undertakes to furnish supplementally a copy of the omitted schedule to the Securities and Exchange Commission upon request.

## LIBERTY PROPERTY HOLDINGS, INC. 12300 LIBERTY BOULEVARD ENGLEWOOD, CO 80112

September 23, 2011

Liberty Interactive Corporation 12300 Liberty Boulevard Englewood, CO 80112 Attention: Legal Department

Re: Facilities Sharing Agreement.

Ladies and Gentlemen:

As you are aware, Liberty Interactive Corporation (f/k/a Liberty Media Corporation), a Delaware corporation (<u>\*Liberty</u>), and Liberty Media Corporation (f/k/a Liberty CapStarz, Inc.), a Delaware corporation (<u>\*Splitco</u>), have entered into a Reorganization Agreement, dated as of August 30, 2011 (the <u>\*Reorganization Agreement</u>), pursuant to which various assets and businesses of Liberty and its subsidiaries have been, or will be, transferred to Splitco and its subsidiaries. Among the assets to be transferred to Splitco is all of the capital stock of Liberty Property Holdings, Inc., a Delaware corporation (<u>\*LPH</u>), which is the owner of 12300 Liberty Boulevard, Englewood, Colorado (the <u>\*Premises</u>).

The Reorganization Agreement contemplates that Liberty and Splitco will be separated into two independent public companies (the <u>Split-Off</u>") by means of the redemption of all of the issued and outstanding shares of Liberty's Liberty Capital common stock and Liberty Starz common stock, in exchange for shares of a corresponding series of Splitco's Liberty Capital common stock and Liberty Starz common stock, respectively, on the terms and subject to the conditions set forth in the Reorganization Agreement and the Restated Certificate of Incorporation of Liberty. After the Split-Off, Splitco, through LPH, will own the Premises and will occupy the office facilities in the Premises that are currently occupied by Liberty.

In connection with the Split-Off, Liberty and Splitco have entered into a Services Agreement, dated the date hereof (the <u>Services Agreement</u>"), pursuant to which Splitco will provide to Liberty certain services performed by Splitco's insurance, risk management, IT, clerical, finance, accounting, payroll, treasury, cash management, legal, human resources, employee benefits, disclosure compliance, investor relations, tax and real estate management departments, as well as such other services as Liberty may from time to time request. The persons who perform these services will be officers, employees or consultants of Splitco (each, an "<u>Employee</u>"). Each Employee will be paid his or her entire salary and bonus by Splitco and will receive health, welfare and other benefits directly from Splitco (in each case, other than in the limited circumstances described in the Services Agreement) (collectively, "<u>Employee</u>"

1

Compensation"). In accordance with the Services Agreement, the Employee Compensation for each Employee will be allocated between Splitco and Liberty based on an estimate of the relative amount of time such Employee spends on matters for Splitco and Liberty, respectively (the percentage of time spent by an Employee on Liberty matters being referred to herein as his or her "Individual Liberty Percentage," and the average percentage of time spent by all Employees on Liberty matters being referred to herein as the "Liberty Percentage"). The Individual Liberty Percentage for each Employee will be determined as set forth in Section 2.1 of the Services Agreement.

Liberty desires to continue to occupy and use the office and parking facilities that are currently utilized by it within the Premises (the <u>Shared Facilities</u>") following the Split-Off through a sharing arrangement, and Splitco and LPH are amenable to such a sharing arrangement, on the terms and subject to the conditions set forth herein

Based on the premises and the mutual agreements of the parties, and for other good and valuable consideration the receipt of which is hereby acknowledged, Liberty and LPH hereby agree as follows:

Section 1. <u>Use of Facilities</u>. The Shared Facilities consist of 40,115 square feet, in the aggregate, of fully-furnished executive offices, working stations for secretarial and other support staff and common areas, including the main reception area, conference facilities, hallways, stairways, restrooms, kitchenettes, the employee cafeteria, the fitness area and parking facilities (collectively, the "<u>Shared Facilities Space</u>"), located within the Premises known as 12300 Liberty Boulevard, Englewood, CO 80112

Section 2. Sharing Fee. Liberty will pay to LPH a monthly fee (the 'Sharing Fee"), by wire or intrabank transfer of funds or in such other manner as may be agreed upon by the parties, in arrears on or before the last day of each calendar month beginning with the first full calendar month following the date of the Split-Off, equal to one-twelfth of the sum of (A) the product of (i) the then-current Liberty Percentage multiplied by (ii) the product of the total square footage of space within the Shared Facilities Space and the Square Foot Rate (as defined below), plus (B) the Annual Allocation Expense (as defined below). For this purpose, Liberty and LPH agree that, until December 31, 2011, the fair market "fully loaded" rental rate per square foot, including parking facilities, for space comparable to the Shared Facilities in Englewood, Colorado will be \$30.00 per square foot (the "Square Foot Rate"). The Square Foot Rate will be automatically increased on the first day of the first month of each calendar year thereafter in an amount equal to the percentage increase in the U.S. Department of Labor Consumer Price Index All Items, All Urban Consumers Denver-Boulder-Greeley ("CPI") for the same period. The Square Foot Rate does not include charges for expenses related to the use of the Shared Facilities, including, but not limited to, utilities, security and janitorial services, office equipment rent, office supplies, use of the cafeteria facilities onsite at the Shared Facilities, maintenance and repairs, telephone, satellite, video and information technology (including network maintenance and data storage, computer and telephone support and maintenance, and management and information systems (servers, hardware and related software)) ("Allocations"). With respect to each calendar year during the term of this Agreement, Liberty shall reimburse Splitco in an amount (the "Annual Allocation Expense") equal to the

2

product of (x) the aggregate amount of the estimated Allocations for such year, as determined in good faith by Splitco and notified to Liberty prior to the commencement of such calendar year, and (y) the Liberty Percentage applicable to such calendar year; provided that, if the Liberty Percentage changes during any calendar year, the Annual Allocation Expense applicable to such calendar year shall be appropriately adjusted.

The terms and conditions of this Section 2 will survive the expiration or earlier termination of this Agreement.

- (i) The term of this Agreement will commence on the Split-Off Effective Date (such term as defined in the Services Agreement) and will continue until the third anniversary of the Split-Off Effective Date (the "Term"). This Agreement is subject to termination prior to the end of the Term in accordance with Section 3(ii).
  - (ii) This Agreement will be terminated prior to the expiration of the Term in the following events:
  - at any time upon at least 30 days' prior written notice by Liberty to LPH;
  - · immediately upon written notice (or any time specified in such notice) by LPH to Liberty if Liberty shall default in the performance of any of its material obligations hereunder and such default shall remain unremedied for a period of 30 days after written notice thereof is given by LPH to Liberty;
  - · immediately upon written notice (or at any time specified in such notice) by LPH to Liberty if a Change in Control or Bankruptcy Event occurs with respect to Liberty; or
  - · immediately upon written notice (or at any time specified in such notice) by Liberty to LPH if a Change in Control or Bankruptcy Event occurs with respect to Splitco.

For purposes of this Section 3(ii), a "Change in Control" will be deemed to have occurred with respect to a party if a merger, consolidation, binding share exchange, acquisition, or similar transaction (each, a "Transaction"), or series of related Transactions, involving such party occurs as a result of which the voting power of all voting securities of such party outstanding immediately prior thereto represent (either by remaining outstanding or being converted into voting securities of the surviving entity) less than 75% of the voting power of such party or the surviving entity of the Transaction outstanding immediately after such Transaction (or if such party or the surviving entity after giving effect to such Transaction is a subsidiary of the issuer of securities in such Transaction, then the voting power of all voting securities of such party outstanding immediately prior to such Transaction represent (by being converted into voting securities of such issuer) less than 75% of the voting power of the issuer outstanding immediately after such Transaction).

For purposes of this Section 3(ii), a "Bankruptcy Event" will be deemed to have occurred with respect to a party upon such party's insolvency, general assignment for the benefit of creditors, such party's voluntary commencement of any case, proceeding, or other action

3

seeking reorganization, arrangement, adjustment, liquidation, dissolution, or consolidation of such party's debts under any law relating to bankruptcy, insolvency, or reorganization, or relief of debtors, or seeking appointment of a receiver, trustee, custodian, or other similar official for such party or for all or any substantial part of such party's assets (each, a "Bankruptcy Proceeding"), or the involuntary filing against Liberty or Splitco, as applicable, of any Bankruptcy Proceeding that is not stayed within 60 days after such filing.

## Section 4. Miscellaneous.

(i) Entire Agreement; Severability. This Agreement (including the schedules hereto), the Reorganization Agreement, the Services Agreement and the Tax Sharing Agreement constitutes the entire agreement among the parties hereto with respect to the subject matter hereof and thereof, and supersedes all prior agreements and understandings, oral and written, among the parties hereto with respect to such subject matter. It is the intention of the parties hereto that the provisions of this Agreement will be enforced to the fullest extent permissible under all applicable laws and public policies, but that the unenforceability of any provision hereof (or the modification of any provision hereof to conform with such laws or public policies, as provided in the next sentence) will not render unenforceable or impair the remainder of this Agreement. Accordingly, if any provision is determined to be invalid or unenforceable either in whole or in part, this Agreement will be deemed amended to delete or modify, as necessary, the invalid or unenforceable provisions and to alter the balance of this Agreement in order to render the same valid and enforceable, consistent (to the fullest extent possible) with the intent and purposes hereof. If the cost of any service to be provided to Liberty under the Services Agreement is included in the Annual Allocation Expense payable hereunder, then the cost of such service shall not also be payable by Liberty under the Services Agreement.

(ii) Notices. All notices and communications hereunder will be in writing and will be deemed to have been duly given if delivered personally or mailed, certified or registered mail with postage prepaid, or sent by confirmed facsimile, addressed as follows:

# If to LPH:

Liberty Property Holdings, Inc. c/o Liberty Media Corporation 12300 Liberty Boulevard Englewood, Colorado 80112 Attention: General Counsel Facsimile: (720) 875-5401

### If to Liberty:

Liberty Interactive Corporation 12300 Liberty Boulevard Englewood, Colorado 80112 Attention: General Counsel Facsimile: (720) 875-5401

4

or to such other address (or to the attention of such other person) as the parties may hereafter designate in writing. All such notices and communications will be deemed to have been given on the date of delivery if sent by facsimile or personal delivery, or the third day after the mailing thereof, except that any notice of a change of address will be deemed to have been given only when actually received.

- (iii) Governing Law. This Agreement and the legal relations among the parties hereto will be governed in all respects, including validity, interpretation and effect, by the laws of the State of Colorado applicable to contracts made and performed wholly therein, without giving effect to any choice or conflict of laws provisions or rules that would cause the application of the laws of any other jurisdiction.
- (iv) No Third-Party Rights. Nothing expressed or referred to in this Agreement is intended or will be construed to give any person other than the parties hereto and their respective successors and permitted assigns any legal or equitable right, remedy or claim under or with respect to this Agreement, or any provision hereof, it being the intention of the parties hereto that this Agreement and all of its provisions and conditions are for the sole and exclusive benefit of the parties to this Agreement and their respective successors and assigns.
- (v) <u>Assignment</u>. This Agreement will inure to the benefit of and be binding on the parties to this Agreement and their respective legal representatives, successors and permitted assigns. Except as expressly contemplated hereby, this Agreement, and the obligations arising hereunder, may not be assigned by either party to this

Agreement, provided, however, that LPH and Liberty may assign their respective rights, interests, duties, liabilities and obligations under this Agreement to any of their respective wholly-owned Affiliates (as defined in the Services Agreement), but such assignment shall not relieve LPH or Liberty, as the assignor, of its obligations hereunder.

- (vi) Amendment. Any amendment, modification or supplement of or to any term or condition of this Agreement will be effective only if in writing and signed by both parties hereto.
- (vii) <u>Further Actions</u>. The parties will execute and deliver all documents, provide all information, and take or forbear from all actions that may be necessary or appropriate to achieve the purposes of this Agreement.

(viii) Force Majeure. Neither party will be liable to the other party with respect to any nonperformance or delay in performance of its obligations under this Agreement to the extent such failure or delay is due to any action or claims by any third party, labor dispute, labor strike, weather conditions or any cause beyond a party's reasonable control. Each party agrees that it will use all commercially reasonable efforts to continue to perform its obligations under this Agreement, to resume performance of its obligations under this Agreement notwithstanding the occurrence of any such event beyond such party's reasonable control.

5

If the foregoing meets with your approval, kindly execute below and return a copy to the undersigned.

Very truly yours,

LIBERTY PROPERTY HOLDINGS, INC.

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe

Title: Executive Vice President and General Counsel

Accepted and agreed this 23rd day of September, 2011:

LIBERTY INTERACTIVE CORPORATION

By: /s/ Charles Y. Tanabe

Name: Charles Y. Tanabe

Title: Executive Vice President and General Counsel

#### AIRCRAFT TIME SHARING AGREEMENT

This Aircraft Time Sharing Agreement ("Agreement") is effective as of the 23<sup>d</sup> day of September, 2011 ("Effective Date"), by and between Liberty Media Corporation (formerly known as Liberty CapStarz, Inc.), with an address of 12300 Liberty Boulevard, Englewood, Colorado 80112 ("Lessor"), and Liberty Interactive Corporation (formerly known as Liberty Media Corporation), with an address of 12300 Liberty Boulevard, Englewood, Colorado 80112 ("Lessee").

#### RECITALS

WHEREAS, Lessor is the owner of that certain Dassault Falcon 900EX aircraft, bearing manufacturer's serial number 101 (the "Aircraft"), currently registered with the Federal Aviation Administration ("FAA") as N730LM;

WHEREAS, Lessor employs a fully qualified flight crew to operate the Aircraft;

WHEREAS, Lessor desires to lease the Aircraft to Lessee and to provide a fully qualified flight crew for all operations on a periodic, non-exclusive time sharing basis, as defined in Section 91.501(c)(1) of the Federal Aviation Regulations ("FAR"); and

WHEREAS, the use of the Aircraft by Lessee shall at all times be pursuant to and in full compliance with the requirements of FAR Sections 91.501(b)(6), 91.501(c) (1) and 91.501(d).

NOW, THEREFORE, in consideration of the mutual promises and considerations contained in this Agreement, the parties agree as follows:

- 1. Lessor agrees to lease the Aircraft to Lessee on a periodic, non-exclusive basis, and to provide a fully qualified flight crew for all operations, pursuant and subject to the provisions of FAR Section 91.501(c)(1) and the terms of this Agreement. The parties expressly acknowledge and agree that, regardless of any employment, contractual or other relationship of any kind or nature, at all times that the Aircraft is operated under this Agreement, Lessor, as the party furnishing the Aircraft and flight crew and exercising complete control over all phases of aircraft operation, shall be deemed to have operational control of the Aircraft as such term is defined in 14 C.F.R. Section 1.1. This Agreement will commence on the Effective Date and continue until the first anniversary of the Effective Date. Thereafter, this Agreement shall be automatically renewed on a month-to-month basis, unless sooner terminated by either party as hereinafter provided. Either party may at any time terminate this Agreement (including during the initial term) upon 30 days' prior written notice to the other party.
- 2. Lessee shall pay Lessor an amount equal to 200% of the actual expenses for fuel for each flight conducted under this Agreement, as permitted by FAR Section 91.501(d) (the "Time Sharing Charge"). For clarification, in no event shall the Time Sharing Charge paid by Lessee exceed those charges specifically permitted in FAR Section 91.501(d), which are as follows:
  - (a) Fuel, oil, lubricants, and other additives;
  - (b) Travel expenses of the crew, including food, lodging and ground transportation;
  - (c) Hangar and tie down costs away from the Aircraft's base of operation;
  - (d) Insurance obtained for the specific flight;
  - (e) Landing fees, airport taxes and similar assessments;
  - (f) Customs, foreign permit, and similar fees directly related to the flight;
  - (g) In-flight food and beverages;
  - (h) Passenger ground transportation;
  - (i) Flight planning and weather contract services; and
  - (j) An additional charge equal to 100% of the expenses listed in subparagraph (a) of this paragraph.
- 3. Lessor will pay all expenses related to the operation of the Aircraft when incurred, and will bill Lessee on a monthly basis as soon as practicable after the last day of each calendar month for the Time Sharing Charge for any and all flights for the account of Lessee pursuant to this Agreement during the preceding month. Lessee shall pay Lessor for all flights for the account of Lessee pursuant to this Agreement within 30 days of receipt of the invoice therefor. If requested by Lessee, Lessor will provide Lessee with a detailed accounting of the expenses composing the Time Sharing Charge for each flight for the account of Lessee pursuant to this Agreement. Without limiting the foregoing, amounts payable by Lessee to Lessor under this Agreement may include any federal excise tax that may be imposed under Internal Revenue Code Section 4261 or any similar excise taxes, if any.
- 4. Lessee will provide Lessor with requests for flight time and proposed flight schedules as far in advance of any given flight as possible, and in any case, at least 24 hours in advance of Lessee's planned departure unless Lessor otherwise agrees. Requests for flight time shall be in a form, whether written or oral, mutually convenient to, and agreed upon by the parties. In addition to the proposed schedules and flight times, Lessee shall provide at least the following information for each proposed flight at some time prior to scheduled departure as required by Lessor or Lessor's flight crew:
  - (a) proposed departure point;
  - (b) destinations;
  - (c) date and time of flight;
  - (d) the number of anticipated passengers;
  - (e) the identity of each anticipated passenger;
  - (f) the nature and extent of luggage and/or cargo to be carried;
  - (g) the date and time of return flight, if any; and
  - (h) any other information concerning the proposed flight that may be pertinent or required by Lessor or Lessor's flight crew.
- 5. Lessor shall have sole and exclusive authority over the scheduling of the Aircraft, including any limitations on the number of passengers on any flight; provided, however, that Lessor will use commercially reasonable efforts to accommodate Lessee's needs and to avoid conflicts in scheduling.

2

6. As between Lessor and Lessee, Lessor shall be solely responsible for securing maintenance, preventive maintenance and required or otherwise necessary inspections on the Aircraft, and shall take such requirements into account in scheduling the Aircraft. No period of maintenance, preventative maintenance or inspection shall be delayed or postponed for the purpose of scheduling the Aircraft, unless said maintenance or inspection can be safely conducted at a later time in compliance with all applicable laws and regulations, and within the sound discretion of the pilot in command. The pilot in command shall have final and complete authority to cancel any flight for any reason or condition that in his judgment would compromise the safety of the flight.

- 7. Lessor shall employ, pay for and provide to Lessee a qualified flight crew for each flight undertaken under this Agreement.
- 8. In accordance with applicable FARs, the qualified flight crew provided by Lessor will exercise all of its duties and responsibilities in regard to the safety of each flight conducted hereunder. Lessee specifically agrees that the flight crew, in its sole discretion, may terminate any flight, refuse to commence any flight or take other action which in the considered judgment of the pilot in command is necessitated by considerations of safety. No such action of the pilot in command shall create or support any liability for loss, injury, damage or delay to Lessee or any other person. The parties further agree that Lessor shall not be liable for delay or failure to furnish the Aircraft and crew pursuant to this Agreement when such failure is caused by government regulation or authority, mechanical difficulty, war, civil commotion, strikes or labor disputes, weather conditions, or acts of God or any other event or circumstance beyond the reasonable control of Lessor.
- 9. (a) At all times during the term of this Agreement, Lessor shall cause to be carried and maintained, at Lessor's cost and expense, physical damage insurance with respect to the Aircraft, third party aircraft liability insurance, passenger legal liability insurance, property damage liability insurance, and medical expense insurance in such amounts and on such terms and conditions as Lessor shall determine in its sole discretion. Lessor shall also bear the cost of paying any deductible amount on any policy of insurance in the event of a claim or loss.
- (b) Any policies of insurance carried in accordance with this Agreement: (i) shall name Lessee as an additional insured; (ii) shall contain a waiver by the underwriter thereof of any right of subrogation against Lessee; and (iii) shall require the insurers to provide at least 30 days' prior written notice (or at least seven days' in the case of any war-risk insurance) to Lessee if the insurers cancel insurance for any reason whatsoever; provided, however, that the insurers shall provide at least ten days' prior written notice if the same is allowed to lapse for non-payment of premium. Each liability policy shall be primary without right of contribution from any other insurance that is carried by Lessee or Lessor and shall expressly provide that all of the provisions thereof, except the limits of liability, shall operate in the same manner as if there were a separate policy covering each insured.
- (c) Lessor shall obtain the approval of this Agreement by the insurance carrier for each policy of insurance on the Aircraft. If requested by Lessee, Lessor shall arrange for a Certificate of Insurance evidencing the insurance coverage with respect to the Aircraft carried and maintained by Lessor to be given by its insurance carriers to Lessee or will provide Lessee

3

with a copy of such insurance policies. Lessor will give Lessee reasonable advance notice of any material modifications to insurance coverage relating to the Aircraft.

10. (a) Lessee agrees that the proceeds of insurance will be Lessee's sole recourse against Lessor with respect to any claims that Lessee may have under this Agreement, except in the event of gross negligence or willful misconduct by Lessor.

# (b) THE PROVISIONS OF THIS SECTION 10 SHALL SURVIVE INDEFINITELY THE TERMINATION OR EXPIRATION OF THE AGREEMENT.

- 11. Lessee warrants that:
- (a) It will not use the Aircraft for the purpose of providing transportation of passengers or cargo in air commerce for compensation or hire, for any illegal purpose, or in violation of any insurance policies with respect to the Aircraft;
- (b) It will refrain from incurring any mechanics, international interest, prospective international interest or other lien and shall not attempt to convey, mortgage, assign, lease or grant or obtain an international interest or prospective international interest or in any way alienate the Aircraft or create any kind of lien or security interest involving the Aircraft or do anything or take any action that might mature into such a lien; and
- (c) It will comply with all applicable laws, governmental and airport orders, rules and regulations, as shall from time to time be in effect relating in any way to the operation and use of the Aircraft under this Agreement.
  - 12. For purposes of this Agreement, the permanent base of operation of the Aircraft shall be Centennial Airport, Englewood, Colorado.
- 13. A copy of this Agreement shall be carried in the Aircraft and available for review upon the request of the Federal Aviation Administration on all flights conducted pursuant to this Agreement.
- 14. Lessee shall not assign this Agreement or its interest herein to any other person or entity without the prior written consent of Lessor, which may be granted or denied in Lessor's sole discretion. Subject to the preceding sentence, this Agreement shall inure to the benefit of and be binding upon the parties hereto, and their respective heirs, representatives, successors and assigns, and does not confer any rights on any other person.
- 15. This Agreement constitutes the entire understanding between the parties with respect to the subject matter hereof and supersedes any prior understandings and agreements between the parties respecting such subject matter. This Agreement may be amended or supplemented and any provision hereof waived only by a written instrument signed by all parties. The failure or delay on the part of any party to insist on strict performance of any of the terms and conditions of this Agreement or to exercise any rights or remedies hereunder shall not constitute a waiver of any such provisions, rights or remedies. This Agreement may be executed in counterparts, which shall, singly or in the aggregate, constitute a fully executed and binding Agreement. Words of gender used in this Agreement may be read as masculine, feminine or

4

neuter as required by the context. Words of number may be read as singular or plural, as required by the context. The word "include" and derivatives of that word are used in this Agreement in an illustrative sense rather than a limiting sense. The word "or" is not exclusive and shall be interpreted as meaning "and/or." The words "shall" and "will" are used interchangeably and are intended to have the same meaning. Where applicable, this Agreement may be referred to as "this Lease."

- 16. Except as otherwise set forth in Section 4, all communications and notices provided for herein shall be in writing and shall become effective when delivered by facsimile transmission or by personal delivery, Federal Express or other overnight courier or four days following deposit in the United States mail, with correct postage for first-class mail prepaid, addressed to Lessor or Lessee at their respective addresses set forth above, or else as otherwise directed by the other party from time to time in writing.
- 17. If any one or more provisions of this Agreement shall be held invalid, illegal or unenforceable by a court of competent jurisdiction, the remaining provisions of this Agreement shall be unimpaired, and the invalid, illegal or unenforceable provisions shall be replaced by a mutually acceptable provision, which, being valid, legal and enforceable, comes closest to the intention of the parties underlying the invalid, illegal or unenforceable provision. To the extent permitted by applicable law, the parties hereby waive any provision of law, that renders any provision of this Agreement prohibited or unenforceable in any respect.
  - 18. This Agreement is entered into under, and is to be construed in accordance with, the laws of the State of Colorado, without reference to conflicts of laws.

#### 19 TRUTH IN LEASING STATEMENT UNDER FAR SECTION 91.23

THE AIRCRAFT, A DASSAULT FALCON 900EX, MANUFACTURER'S SERIAL NO. 101, CURRENTLY REGISTERED WITH THE FEDERAL AVIATION ADMINISTRATION AS N730LM, HAS BEEN MAINTAINED AND INSPECTED UNDER FAR PART 91 DURING THE 12 MONTH PERIOD PRECEDING THE DATE OF THIS LEASE.

THE AIRCRAFT WILL BE MAINTAINED AND INSPECTED UNDER FAR PART 91 FOR OPERATIONS TO BE CONDUCTED UNDER THIS LEASE. DURING THE DURATION OF THIS LEASE, LIBERTY MEDIA CORPORATION, 12300 LIBERTY BOULEVARD, ENGLEWOOD, COLORADO 80112 IS CONSIDERED RESPONSIBLE FOR OPERATIONAL CONTROL OF THE AIRCRAFT UNDER THIS LEASE.

AN EXPLANATION OF THE FACTORS BEARING ON OPERATIONAL CONTROL AND PERTINENT FEDERAL AVIATION REGULATIONS CAN BE OBTAINED FROM THE NEAREST FAA FLIGHT STANDARDS DISTRICT OFFICE.

THE "INSTRUCTIONS FOR COMPLIANCE WITH TRUTH IN LEASING REQUIREMENTS" ATTACHED HERETO ARE INCORPORATED HEREIN BY REFERENCE.

LIBERTY MEDIA CORPORATION, LOCATED AT 12300 LIBERTY BOULEVARD, ENGLEWOOD, COLORADO 80112, THROUGH ITS UNDERSIGNED AUTHORIZED SIGNATORY BELOW, CERTIFIES THAT LESSOR IS RESPONSIBLE FOR OPERATIONAL CONTROL OF THE AIRCRAFT AND THAT IT UNDERSTANDS ITS RESPONSIBILITIES FOR COMPLIANCE WITH APPLICABLE FEDERAL AVIATION REGULATIONS.

IN WITNESS WHEREOF, the parties have executed this Agreement to be effective as of the date first above written.

LESSOR		LESSEE	
LIBERTY MEDIA CORPORATION		LIBERTY INTERACTIVE CORPORATION	
Ву:	/s/ Charles Y. Tanabe	By:	/s/ Charles Y. Tanabe
Name:	Charles Y. Tanabe	Name:	Charles Y. Tanabe
Title:	Executive Vice President	Title:	Executive Vice President

### INSTRUCTIONS FOR COMPLIANCE WITH "TRUTH IN LEASING" REQUIREMENTS

1. Mail a copy of the lease to the following address via certified mail, return receipt requested, immediately upon execution of the lease (14 C.F.R. 91.23 requires that the copy be sent within 24 hours after it is signed):

> Federal Aviation Administration Aircraft Registration Branch ATTN: Technical Section P.O. Box 25724 Oklahoma City, Oklahoma 73125

- 2. Telephone the nearest Flight Standards District Office at least 48 hours prior to the first flight under this lease.
- 3. Carry a copy of the lease in the aircraft at all times.

#### AIRCRAFT TIME SHARING AGREEMENT

This Aircraft Time Sharing Agreement ("Agreement") is effective as of the 23d day of September, 2011 ("Effective Date"), by and between Liberty Media Corporation (formerly known as Liberty CapStarz, Inc.), with an address of 12300 Liberty Boulevard, Englewood, Colorado 80112 ("Lessor"), and Liberty Interactive Corporation (formerly known as Liberty Media Corporation), with an address of 12300 Liberty Boulevard, Englewood, Colorado 80112 ("Lessee").

## RECITALS

WHEREAS, Lessor is the owner of that certain Dassault Falcon 900EX aircraft, bearing manufacturer's serial number 74 (the "Aircraft"), currently registered with the Federal Aviation Administration ("FAA") as N740LM;

WHEREAS, Lessor employs a fully qualified flight crew to operate the Aircraft;

WHEREAS, Lessor desires to lease the Aircraft to Lessee and to provide a fully qualified flight crew for all operations on a periodic, non-exclusive time sharing basis, as defined in Section 91.501(c)(1) of the Federal Aviation Regulations ("FAR"); and

WHEREAS, the use of the Aircraft by Lessee shall at all times be pursuant to and in full compliance with the requirements of FAR Sections 91.501(b)(6), 91.501(c) (1) and 91.501(d).

NOW, THEREFORE, in consideration of the mutual promises and considerations contained in this Agreement, the parties agree as follows:

- 1. Lessor agrees to lease the Aircraft to Lessee on a periodic, non-exclusive basis, and to provide a fully qualified flight crew for all operations, pursuant and subject to the provisions of FAR Section 91.501(c)(1) and the terms of this Agreement. The parties expressly acknowledge and agree that, regardless of any employment, contractual or other relationship of any kind or nature, at all times that the Aircraft is operated under this Agreement, Lessor, as the party furnishing the Aircraft and flight crew and exercising complete control over all phases of aircraft operation, shall be deemed to have operational control of the Aircraft as such term is defined in 14 C.F.R. Section 1.1. This Agreement will commence on the Effective Date and continue until the first anniversary of the Effective Date. Thereafter, this Agreement shall be automatically renewed on a month-to-month basis, unless sooner terminated by either party as hereinafter provided. Either party may at any time terminate this Agreement (including during the initial term) upon 30 days' prior written notice to the other party.
- 2. Lessee shall pay Lessor an amount equal to 200% of the actual expenses for fuel for each flight conducted under this Agreement, as permitted by FAR Section 91.501(d) (the "Time Sharing Charge"). For clarification, in no event shall the Time Sharing Charge paid by Lessee exceed those charges specifically permitted in FAR Section 91.501(d), which are as follows:
  - (a) Fuel, oil, lubricants, and other additives;
  - (b) Travel expenses of the crew, including food, lodging and ground transportation;
  - (c) Hangar and tie down costs away from the Aircraft's base of operation;
  - (d) Insurance obtained for the specific flight;
  - (e) Landing fees, airport taxes and similar assessments;
  - (f) Customs, foreign permit, and similar fees directly related to the flight;
  - (g) In-flight food and beverages;
  - (h) Passenger ground transportation;
  - (i) Flight planning and weather contract services; and
  - (j) An additional charge equal to 100% of the expenses listed in subparagraph (a) of this paragraph.
- 3. Lessor will pay all expenses related to the operation of the Aircraft when incurred, and will bill Lessee on a monthly basis as soon as practicable after the last day of each calendar month for the Time Sharing Charge for any and all flights for the account of Lessee pursuant to this Agreement during the preceding month. Lessee shall pay Lessor for all flights for the account of Lessee pursuant to this Agreement within 30 days of receipt of the invoice therefor. If requested by Lessee, Lessor will provide Lessee with a detailed accounting of the expenses composing the Time Sharing Charge for each flight for the account of Lessee pursuant to this Agreement. Without limiting the foregoing, amounts payable by Lessee to Lessor under this Agreement may include any federal excise tax that may be imposed under Internal Revenue Code Section 4261 or any similar excise taxes, if any.
- 4. Lessee will provide Lessor with requests for flight time and proposed flight schedules as far in advance of any given flight as possible, and in any case, at least 24 hours in advance of Lessee's planned departure unless Lessor otherwise agrees. Requests for flight time shall be in a form, whether written or oral, mutually convenient to, and agreed upon by the parties. In addition to the proposed schedules and flight times, Lessee shall provide at least the following information for each proposed flight at some time prior to scheduled departure as required by Lessor or Lessor's flight crew:
  - (a) proposed departure point;
  - (b) destinations;
  - (c) date and time of flight;
  - (d) the number of anticipated passengers;
  - (e) the identity of each anticipated passenger;
  - (f) the nature and extent of luggage and/or cargo to be carried;
  - (g) the date and time of return flight, if any; and
  - (h) any other information concerning the proposed flight that may be pertinent or required by Lessor or Lessor's flight crew.
- 5. Lessor shall have sole and exclusive authority over the scheduling of the Aircraft, including any limitations on the number of passengers on any flight; provided, however, that Lessor will use commercially reasonable efforts to accommodate Lessee's needs and to avoid conflicts in scheduling.

2

- 6. As between Lessor and Lessee, Lessor shall be solely responsible for securing maintenance, preventive maintenance and required or otherwise necessary inspections on the Aircraft, and shall take such requirements into account in scheduling the Aircraft. No period of maintenance, preventative maintenance or inspection shall be delayed or postponed for the purpose of scheduling the Aircraft, unless said maintenance or inspection can be safely conducted at a later time in compliance with all applicable laws and regulations, and within the sound discretion of the pilot in command. The pilot in command shall have final and complete authority to cancel any flight for any reason or condition that in his judgment would compromise the safety of the flight.
  - 7. Lessor shall employ, pay for and provide to Lessee a qualified flight crew for each flight undertaken under this Agreement.
- 8. In accordance with applicable FARs, the qualified flight crew provided by Lessor will exercise all of its duties and responsibilities in regard to the safety of each flight conducted hereunder. Lessee specifically agrees that the flight crew, in its sole discretion, may terminate any flight, refuse to commence any flight or take other action which in the considered judgment of the pilot in command is necessitated by considerations of safety. No such action of the pilot in command shall create or support any liability for loss, injury, damage or delay to Lessee or any other person. The parties further agree that Lessor shall not be liable for delay or failure to furnish the Aircraft and crew pursuant to this Agreement when such failure is caused by government regulation or authority, mechanical difficulty, war, civil commotion, strikes or labor disputes, weather conditions, or acts of God or any other event or circumstance beyond the reasonable control of Lessor.
- 9. (a) At all times during the term of this Agreement, Lessor shall cause to be carried and maintained, at Lessor's cost and expense, physical damage insurance with respect to the Aircraft, third party aircraft liability insurance, passenger legal liability insurance, property damage liability insurance, and medical expense insurance in such amounts and on such terms and conditions as Lessor shall determine in its sole discretion. Lessor shall also bear the cost of paying any deductible amount on any policy of insurance in the event of a claim or loss.
- (b) Any policies of insurance carried in accordance with this Agreement: (i) shall name Lessee as an additional insured; (ii) shall contain a waiver by the underwriter thereof of any right of subrogation against Lessee; and (iii) shall require the insurers to provide at least 30 days' prior written notice (or at least seven days' in the case of any war-risk insurance) to Lessee if the insurers cancel insurance for any reason whatsoever; provided, however, that the insurers shall provide at least ten days' prior written notice if the same is allowed to lapse for non-payment of premium. Each liability policy shall be primary without right of contribution from any other insurance that is carried by Lessee or Lessor and shall expressly provide that all of the provisions thereof, except the limits of liability, shall operate in the same manner as if there were a separate policy covering each insured.
  - (c) Lessor shall obtain the approval of this Agreement by the insurance carrier for each policy of insurance on the Aircraft. If requested by Lessee,

3

with a copy of such insurance policies. Lessor will give Lessee reasonable advance notice of any material modifications to insurance coverage relating to the Aircraft.

10. (a) Lessee agrees that the proceeds of insurance will be Lessee's sole recourse against Lessor with respect to any claims that Lessee may have under this Agreement, except in the event of gross negligence or willful misconduct by Lessor.

# (b) THE PROVISIONS OF THIS SECTION 10 SHALL SURVIVE INDEFINITELY THE TERMINATION OR EXPIRATION OF THE AGREEMENT.

#### 11. Lessee warrants that:

- (a) It will not use the Aircraft for the purpose of providing transportation of passengers or cargo in air commerce for compensation or hire, for any illegal purpose, or in violation of any insurance policies with respect to the Aircraft;
- (b) It will refrain from incurring any mechanics, international interest, prospective international interest or other lien and shall not attempt to convey, mortgage, assign, lease or grant or obtain an international interest or prospective international interest or in any way alienate the Aircraft or create any kind of lien or security interest involving the Aircraft or do anything or take any action that might mature into such a lien; and
- (c) It will comply with all applicable laws, governmental and airport orders, rules and regulations, as shall from time to time be in effect relating in any way to the operation and use of the Aircraft under this Agreement.
  - 12. For purposes of this Agreement, the permanent base of operation of the Aircraft shall be Centennial Airport, Englewood, Colorado.
- 13. A copy of this Agreement shall be carried in the Aircraft and available for review upon the request of the Federal Aviation Administration on all flights conducted pursuant to this Agreement.
- 14. Lessee shall not assign this Agreement or its interest herein to any other person or entity without the prior written consent of Lessor, which may be granted or denied in Lessor's sole discretion. Subject to the preceding sentence, this Agreement shall inure to the benefit of and be binding upon the parties hereto, and their respective heirs, representatives, successors and assigns, and does not confer any rights on any other person.
- 15. This Agreement constitutes the entire understanding between the parties with respect to the subject matter hereof and supersedes any prior understandings and agreements between the parties respecting such subject matter. This Agreement may be amended or supplemented and any provision hereof waived only by a written instrument signed by all parties. The failure or delay on the part of any party to insist on strict performance of any of the terms and conditions of this Agreement or to exercise any rights or remedies hereunder shall not constitute a waiver of any such provisions, rights or remedies. This Agreement may be executed in counterparts, which shall, singly or in the aggregate, constitute a fully executed and binding Agreement. Words of gender used in this Agreement may be read as masculine, feminine or

4

neuter as required by the context. Words of number may be read as singular or plural, as required by the context. The word "include" and derivatives of that word are used in this Agreement in an illustrative sense rather than a limiting sense. The word "or" is not exclusive and shall be interpreted as meaning "and/or." The words "shall" and "will" are used interchangeably and are intended to have the same meaning. Where applicable, this Agreement may be referred to as "this Lease."

- 16. Except as otherwise set forth in Section 4, all communications and notices provided for herein shall be in writing and shall become effective when delivered by facsimile transmission or by personal delivery, Federal Express or other overnight courier or four days following deposit in the United States mail, with correct postage for first-class mail prepaid, addressed to Lessor or Lessee at their respective addresses set forth above, or else as otherwise directed by the other party from time to time in writing.
- 17. If any one or more provisions of this Agreement shall be held invalid, illegal or unenforceable by a court of competent jurisdiction, the remaining provisions of this Agreement shall be unimpaired, and the invalid, illegal or unenforceable provisions shall be replaced by a mutually acceptable provision, which, being valid, legal and enforceable, comes closest to the intention of the parties underlying the invalid, illegal or unenforceable provision. To the extent permitted by applicable law, the parties hereby waive any provision of law, that renders any provision of this Agreement prohibited or unenforceable in any respect.
  - 18. This Agreement is entered into under, and is to be construed in accordance with, the laws of the State of Colorado, without reference to conflicts of laws.

## REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

5

# 19. TRUTH IN LEASING STATEMENT UNDER FAR SECTION 91.23

THE AIRCRAFT, A DASSAULT FALCON 900EX, MANUFACTURER'S SERIAL NO. 74, CURRENTLY REGISTERED WITH THE FEDERAL AVIATION ADMINISTRATION AS N740LM, HAS BEEN MAINTAINED AND INSPECTED UNDER FAR PART 91 DURING THE 12 MONTH PERIOD PRECEDING THE DATE OF THIS LEASE.

THE AIRCRAFT WILL BE MAINTAINED AND INSPECTED UNDER FAR PART 91 FOR OPERATIONS TO BE CONDUCTED UNDER THIS LEASE. DURING THE DURATION OF THIS LEASE, LIBERTY MEDIA CORPORATION, 12300 LIBERTY BOULEVARD, ENGLEWOOD, COLORADO 80112 IS CONSIDERED RESPONSIBLE FOR OPERATIONAL CONTROL OF THE AIRCRAFT UNDER THIS LEASE.

AN EXPLANATION OF THE FACTORS BEARING ON OPERATIONAL CONTROL AND PERTINENT FEDERAL AVIATION REGULATIONS CAN BE OBTAINED FROM THE NEAREST FAA FLIGHT STANDARDS DISTRICT OFFICE.

#### REFERENCE.

LIBERTY MEDIA CORPORATION, LOCATED AT 12300 LIBERTY BOULEVARD, ENGLEWOOD, COLORADO 80112, THROUGH ITS UNDERSIGNED AUTHORIZED SIGNATORY BELOW, CERTIFIES THAT LESSOR IS RESPONSIBLE FOR OPERATIONAL CONTROL OF THE AIRCRAFT AND THAT IT UNDERSTANDS ITS RESPONSIBILITIES FOR COMPLIANCE WITH APPLICABLE FEDERAL AVIATION REGULATIONS.

IN WITNESS WHEREOF, the parties have executed this Agreement to be effective as of the date first above written.

LESSOR	l .	LESSEE	
LIBERT	Y MEDIA CORPORATION	LIBERTY CORPOR	INTERACTIVE ATION
By:	/s/ Charles Y. Tanabe	By:	/s/ Charles Y. Tanabe
Name:	Charles Y. Tanabe	Name:	Charles Y. Tanabe
Title:	Executive Vice President	Title:	Executive Vice President

# INSTRUCTIONS FOR COMPLIANCE WITH "TRUTH IN LEASING" REQUIREMENTS

1. Mail a copy of the lease to the following address via certified mail, return receipt requested, immediately upon execution of the lease (14 C.F.R. 91.23 requires that the copy be sent within 24 hours after it is signed):

Federal Aviation Administration Aircraft Registration Branch ATTN: Technical Section P.O. Box 25724 Oklahoma City, Oklahoma 73125

- Telephone the nearest Flight Standards District Office at least 48 hours prior to the first flight under this lease.
- 3. Carry a copy of the lease in the aircraft at all times.